# TESI DI DOTTORATO

#### Alberto Roncoroni

Symmetry and quantitative stability results for problems in geometric analysis and functional inequalities

Dottorato in Matematica, Pavia (2019).

<http://www.bdim.eu/item?id=tesi\_2019\_RoncoroniAlberto\_1>

L'utilizzo e la stampa di questo documento digitale è consentito liberamente per motivi di ricerca e studio. Non è consentito l'utilizzo dello stesso per motivi commerciali. Tutte le copie di questo documento devono riportare questo avvertimento.

bdim (Biblioteca Digitale Italiana di Matematica) SIMAI & UMI http://www.bdim.eu/



Università degli Studi di Pavia Dipartimento di Matematica Felice Casorati Joint PhD Program in Mathematics Pavia - Milano Bicocca - INdAM

#### Symmetry and quantitative stability results for problems in geometric analysis and functional inequalities

**Doctoral Dissertation of:** Alberto Roncoroni

**PhD Coordinator:** Prof. Thomas Weigel

Supervisors: Prof. Giulio Ciraolo Prof. Luigi Vezzoni

**Tutor:** Prof. Giulio Schimperna

December 13, 2019 - XXXII Cycle

In direzione ostinata e contraria (Fabrizio De André)

### Abstract

This thesis addresses the characterization of geometric properties for problems in Partial Differential Equations (PDEs), geometric analysis and functional inequalities, with particular interest in the study of symmetry and quantitative stability issues. The thesis consists in four chapters.

Chapter 1 is about symmetry and quantitative studies for hypersurfaces embedded in space forms with some curvature close to a constant. The starting point is the well-known Alexandrov Soap Bubble Theorem which asserts that the distance spheres are the only embedded closed connected hypersurfaces in space forms (i.e. the Euclidean space, the hyperbolic space and the hemisphere) having constant mean curvature. Actually the theorem can be extended to more general functions of the principal curvatures  $f(k_1, \ldots, k_{n-1})$  satisfying suitable conditions. The main result in Chapter 1 are sharp quantitative estimates of proximity to a single sphere for Alexandrov Soap Bubble Theorem in space forms when the curvature operator f is close to a constant. Under an assumption that prevents bubbling (the uniform touching ball condition), the proximity to a single sphere is optimally quantified in terms of the oscillation of the curvature function f. Our approach provides a unified picture of quantitative studies of the method of moving planes, i.e. the original method introduced by Alexandrov to prove its theorem, in space forms.

Chapter 2 is about symmetry results for Serrin-type overdetermined problems. Serrin's symmetry results asserts that if there exists a solution to an overdetermined boundary value problem associated to the equation  $\Delta u = -1$ in an open domain of the Euclidean space, then the domain must be a ball.

The first result in Chapter 2 is a Serrin's symmetry result for an overdetermined boundary value problem in a particular class of Riemannian manifolds, the so-called model manifolds. We prove an Euclidean symmetry result under a suitable compatibility assumption between the solution and the geometry of the manifolds.

The second result in Chapter 2 is a Serrin's symmetry result for overdetermined boundary value problems in convex cones for (possibly) degenerate operators, such as the p-Laplace operator, in the Euclidean space as well as for a suitable generalization of the problem for convex cones in space forms. We prove symmetry results by showing that the existence of a solution implies that the domain is a spherical sector.

Chapter 3 is about symmetry results for critical anisotropic *p*-Laplace equations in convex cones. Given 1 , we consider the*critical p-Laplacian*  $equation <math>\Delta_p u + u^{p^*-1} = 0$ , which is related to critical points of the Sobolev inequality and to the Yamabe problem. Exploiting the moving planes method, it has been recently shown that positive and entire solutions to the ciritical *p*-Laplacian equation are classified and are given by the so-called Aubin-Talenti bubbles. Since the moving planes method strongly relies on the symmetries of the equation and the domain, in Chapter 3 we provide a new approach to this Liouville-type problem that allows us to give a complete classification of solutions in an anisotropic setting and in convex cones. More precisely, we characterize solutions to the critical *p*-Laplacian equation induced by a smooth norm inside any convex cone of the Euclidean space. One can show that the critical *p*-Laplace equation that we consider is related to the critical points of the anisotropic Sobolev inequality in convex cones. Since a sharp Sobolev inequality was missing in this setting, in Appendix B, we prove a general class of (weighted) anisotropic Sobolev inequalities inside arbitrary convex cones by using the optimal transport approach.

Chapter 4 is about functional inequalities on a particular class of Riemannian manifolds. In particular we consider the so-called Cartan-Hadamard manifolds, i.e. complete, simply connected, non-compact Riemannian manifolds with negative sectional curvatures everywhere. It is well-known that on every Cartan-Hadamard manifold the Sobolev inequality holds true, moreover if the sectional curvatures are bounded above by a negative constant then also the Poincaré inequality holds true. In Chapter 4 we investigate the validity, as well as the failure, of Sobolev-type inequalities on Cartan-Hadamard manifolds under suitable bounds on the sectional and the Ricci curvatures. More specifically, we prove that if the sectional curvatures are bounded from above by a negative power of the distance from a fixed pole (times a negative constant), then all the  $L^{p}$ -inequalities that interpolate between Poincaré and Sobolev hold in the radial setting; provided such power lies in the interval (-2, 0), except the Poincaré inequality. If the power is equal to -2 then p must necessarily be strictly larger (in a quantitative way) than 2. Upon assuming similar bounds from below on the Ricci curvature, we show that the non-radial version of such inequalities fails, except for the Sobolev one. Finally, we prove optimal smoothing effects for a porous medium equation set up on the Cartan-Hadamard manifolds we are considering which follows from the Sobolev-type inequalities that we prove.

## Contents

#### Introduction

#### I Symmetry and quantitative stability results for problems in geometric analysis

v

1

1	Qua	antitative stability for almost constant mean curvature hy-			
	persurfaces				
	1.1	The method of the moving planes	16		
	1.2	Local quantitative estimates in space forms	19		
		1.2.1 Quantitative stability of the parallel transport	24		
	1.3	Curvatures of projected surfaces in conformally Euclidean spaces	33		
	1.4	Approximate symmetry in one direction	42		
	1.5	Global approximate symmetry	54		
	1.6	Proof of Theorem 1.1 and Corollary 1.1	58		
<b>2</b>	Syn	nmetry results for Serrin-type overdetermined problems	60		
2	<b>Syn</b> 2.1	nmetry results for Serrin-type overdetermined problemsA symmetry result in model manifolds	<b>60</b> 60		
2	<b>Syn</b> 2.1	nmetry results for Serrin-type overdetermined problemsA symmetry result in model manifolds2.1.1Explicit computations towards the proof of Theorem 2.1	<b>60</b> 60 62		
2	<b>Syn</b> 2.1	A symmetry results for Serrin-type overdetermined problemsA symmetry result in model manifolds2.1.1Explicit computations towards the proof of Theorem 2.12.1.2Proof of Theorem 2.1	60 60 62 64		
2	<b>Syn</b> 2.1 2.2	A symmetry results for Serrin-type overdetermined problemsA symmetry result in model manifolds2.1.1Explicit computations towards the proof of Theorem 2.12.1.2Proof of Theorem 2.1Symmetry results in convex cones	60 60 62 64 69		
2	<b>Syn</b> 2.1 2.2	nmetry results for Serrin-type overdetermined problemsA symmetry result in model manifolds2.1.1Explicit computations towards the proof of Theorem 2.12.1.2Proof of Theorem 2.1Symmetry results in convex cones2.2.1Preliminary results for Theorem 2.2	60 62 64 69 73		
2	<b>Syn</b> 2.1 2.2	A symmetry results for Serrin-type overdetermined problemsA symmetry result in model manifolds2.1.1Explicit computations towards the proof of Theorem 2.12.1.2Proof of Theorem 2.1Symmetry results in convex cones2.2.1Preliminary results for Theorem 2.22.2.2Proof of Theorem 2.2	60 60 62 64 69 73 83		
2	<b>Syn</b> 2.1 2.2	A symmetry results for Serrin-type overdetermined problemsA symmetry result in model manifolds2.1.1Explicit computations towards the proof of Theorem 2.12.1.2Proof of Theorem 2.1Symmetry results in convex cones2.2.1Preliminary results for Theorem 2.22.2.2Proof of Theorem 2.22.2.3Cones in space forms: proof of Theorem 2.3	60 60 62 64 69 73 83 86		

# II Symmetry and validity results related to functional inequalities 90

3	Syn	results for critical anisotropic p-Laplacian equations	quations			
in convex cones						
	3.1	ninary results	99			
		3.1.1	Boundeness of solutions	99		
		3.1.2	Asymptotic bounds on $u$ and $\nabla u$	103		
		3.1.3	Asymptotic estimates on higher order derivatives	109		
	3.2	$\mathbf{Proof}$	of Theorem 3.1	113		
		3.2.1	An integral inequality	114		
		3.2.2	Conclusion	119		

4	Sobolev-type inequalities on Cartan-Hadamard manifolds		
	4.1	Statements of the main results	121
		4.1.1 Basic notations	123
		4.1.2 The sub-hyperbolic radial case	124
		4.1.3 The quasi-Euclidean radial case	125
		4.1.4 Failure of the nonradial inequalities	125
	4.2	Geometric and functional preliminaries	126
		4.2.1 Notations from Riemannian geometry	126
		4.2.2 Laplacian comparison	127
		4.2.3 Geometric interpretation of the scalar curvature	129
		4.2.4 One-dimensional weighted inequalities	129
	4.3	Proofs of the radial results	129
		4.3.1 The sub-hyperbolic case	130
		4.3.2 The quasi-Euclidean case	136
	4.4	The Poincaré inequality: McKean's Theorem and related issues .	140
		4.4.1 Negative curvatures outside a ball	143
	4.5	Failure of the inequalities in the nonradial framework	146
	4.6	The porous medium equation on Cartan-Hadamard manifolds	148
II	Ι.	Appendixes	152
		••	
Α	Ag	general result on Riemannian manifolds with boundary	153
в	Sha	orn anisotropic Sobolov inequalities with weight in conve	v
Б	con	arp anisotropic bobolev mequanties with weight in conve	^ 156
	con		100
$\mathbf{C}$	Syr	nmetry results for critical $p$ -Laplace equation in $\mathbb{R}^n$	160
F	a	, .	
D	Ser	rin vs Alexandrov	165
	D.I	Serrin implies Alexandrov	165
	D.2	Alexandrov implies Serrin	167
т	7 1	Bibliography	160
T,	v 1	Dibilography	103

### Introduction

This thesis addresses the characterization of geometric properties for solutions to problems in Partial Differential Equations (PDEs), geometric analysis and functional inequalities, with particular interest in the study of symmetry and quantitative stability issues. The thesis is divided in four parts.

Part I is about symmetry and quantitative studies for hypersurfaces embedded in space forms with some curvature close to a constant, for instance the mean curvature, and overdetermined problems for PDEs. This part collects the results obtained in [63, 65, 196].

Part II is about symmetry results for critical anisotropic p-Laplace equations in convex cones and functional inequalities on a particular class of Riemannian manifolds. Part II collects the results obtained in [57] and [174].

Part III contains the four appendixes of the thesis and Part IV collects the bibliographic references of the thesis.

This is a first introduction to the topics and results contained in this thesis. We refer to the introductions to Part I and to Part II for a more detailed description of the results contained in this thesis. We will start by describing classical results about the isoperimetric inequality and the Sobolev inequality, which motivate most of the results contained in this thesis. Roughly speaking, we can say that most of the problems that we will consider in the thesis are related to the Euler-Lagrange equations associated to these two inequalities. Moreover we will emphasize several similarities that arise when considering different proofs of these problems.

The isoperimetric problem goes back to the ancient Greece (the well-known Dido's Problem) and it has been object of studies for generations of mathematicians. The isoperimetric problem consists in minimizing the surface area among all domains having fixed volume, or equivalently maximizing the volume among all domains whose boundary surface has fixed (n-1)-dimensional area. It is well known that is that the unique extremal of these problems is the ball. An equivalent and more analytic formulation of the isoperimetric problem is the so-called *isoperimetric inequality*: if  $\Omega \subset \mathbb{R}^n$  is a bounded domain, and  $\partial\Omega$  denotes its boundary, then

$$|\partial \Omega| \ge n \omega_n^{\frac{1}{n}} |\Omega|^{1 - \frac{1}{n}} \,, \tag{1}$$

where  $\omega_n$  denotes the volume of the unit sphere in  $\mathbb{R}^n$  and  $|\cdot|$  denotes either the *n*-dimensional (Lebesgue) measure or the (n-1)-dimensional surface measure of a subset of  $\mathbb{R}^n$ . Moreover, the equality in (1) is attained if and only if the domain is a ball (see [78] and also [179]).

The natural functional associated to (1) is the *isoperimetric functional* given by:

$$\mathcal{P}(\Omega) := \frac{|\partial \Omega|^n}{|\Omega|^{n-1}}.$$
(2)

We are interested in studying *critical points* of the functional  $\mathcal{P}$ . The most direct approach is to perform the methods of the calculus of variations in order to write the Euler-Lagrange equation associated to (2). Let  $\Omega \subset \mathbb{R}^n$  be a critical domain for the functional (2) and assume that  $\Omega$  is bounded by a smooth hypersurface S, i.e.  $S = \partial \Omega$ . The idea is to compute the first variation of the functional  $\mathcal{P}$ : we take a smooth function  $h: S \to \mathbb{R}$  and we consider the normal variation of S defined by  $\psi_t: S \to \mathbb{R}^n$  such that

$$\psi_t(p) = p + th(p)\nu(p) \,,$$

where  $\nu$  is the unitary exterior normal field to S. We denote by  $S_t$  the hypersurface given by  $\psi_t(S_t)$  and by  $\Omega_t$  the domain enclosed by  $S_t$ ; observe that  $S_t$  is nothing but the displacement of each point of S by the vector  $th\nu$ . For simplicity of notation, we set

$$\mathcal{A}(t) := |S_t| = \int_{S_t} d\sigma$$
 and  $\mathcal{V}(t) := |\Omega_t| = \int_{\Omega_t} dx$ .

Then the first variation of the functional (2) above is given by (see e.g. [161] and [208])

$$\mathcal{A}'(0) = -(n-1) \int_{S} hH \, d\sigma \quad \text{and} \quad \mathcal{V}'(0) = \int_{S} h \, d\sigma \,, \tag{3}$$

where H is the mean curvature of S. Since  $\Omega$  is a critical domain, we have that

$$\left. \frac{d}{dt} \right|_{t=0} \mathcal{P}(\Omega_t) = 0 \,,$$

or equivalently, from (2) and (3),

$$\int_{S} h(n|\Omega|H - |S|) \, d\sigma = 0, \quad \text{for all } h,$$

and this implies that S has constant mean curvature. Hence we prove that the Euler-Lagrange equation associated to the functional (2) (and so to the inequality (1)) is

$$H \equiv \frac{|\partial\Omega|}{n|\Omega|} \,. \tag{4}$$

Now the following characterization proved by Alexandrov in the 50's comes into play:

**Theorem A** ([6]): let S be a  $C^2$ -regular, connected, closed (i.e. compact and without boundary) hypersurface embedded in the Euclidean space  $\mathbb{R}^n$ . Then S has constant mean curvature if and only if is a sphere.

Putting the previous computation and Theorem A together, we prove that critical points of the functional (2) must be spheres, and in particular the equality in (1) is attained if and only if the domain is a ball.

We mention that the previous argument (in particular Alexandrov's Theorem) works if we assume to work with smooth boundaries and this is not the general situation where one wants to prove the isoperimetric inequality (we refer to the *theory of sets of finite perimeter* developed by Caccioppoli and De Giorgi in the 30-50's, see also [163] for a recent reference). So we emphasize that in this first part of the Introduction we only wanted to stress the link between the isoperimetric inequality and Alexandrov's Theorem because this is, probably, the most important motivation in order to study Alexandrov's Theorem. We mention that Theorem A is stated in its simplest formulation, indeed Alexandrov's Theorem has been widely studied and extended in several directions. Moreover, in the last years, also the quantitative version of the isoperimetric inequality has been the object of several studies. In [104] the sharp quantitative version of the isoperimetric inequality in  $\mathbb{R}^n$  has been proved. Once one has a quantitative result for the minimizers of (2) it is of strong interests to understand the shape of critical points of (2) (see [58, 67]). In Chapter 1 we study a quantitative version of Alexandrov Theorem for hypersurfaces embedded in space forms with some curvature (not necessary the mean curvature) close to a constant.

Alexandrov Theorem is also related to an important and well-known result in the theory of elliptic PDEs, which is the following characterization of domains which support a solution to an overdetermined problem.

**Theorem B** ([204]): let  $\Omega \subset \mathbb{R}^n$  be a bounded domain with boundary of class  $C^2$  and let  $\nu$  be the outward normal to  $\partial\Omega$ . Then there exists a solution  $u \in C^2(\Omega) \cap C^1(\overline{\Omega})$  to

$$\begin{cases} \Delta u = -1 & \text{in } \Omega\\ u = 0 & \text{on } \partial \Omega \end{cases},$$
(5)

such that, for some constant c,

$$\partial_{\nu} u = c \quad on \; \partial\Omega \,, \tag{6}$$

if and only if  $\Omega$  is a ball and u is a radial function.

Theorem B was proved by Serrin in 1971 by using the so-called *method of moving planes*, which takes inspiration from *the reflection principle* introduced by Alexandrov to prove Theorem A. As we will show in Appendix D, a further link between these two theorems is that they are "equivalent", in the sense that we can prove Theorem A assuming Theorem B holds and viceversa (we mention that the technique used to prove this equivalence takes inspiration from the proof of Theorem B proposed by Weinberger in [215]).

Before giving an idea of the proofs of the two theorems, we mention that Serrin's problem (5)-(6) is called an overdetermined boundary value problem since the Dirichlet problem (5) already admits a unique solution; hence condition (6) is an additional requirement and in general the whole problem (5)-(6) may not admit a solution. Thus, the remaining data of the problem, the domain  $\Omega$ , cannot be given arbitrarly, i.e. there is a requirement also on the domain: this phenomenon is called rigidity and it implies that the domain and the solution itself satisfy some symmetry.

We emphasize that overdetermined boundary value problems like (5)-(6) arise in many context. The following is one of the physical interpretations given

by Serrin in [204]: consider a viscous incompressible fluid moving in straight parallel streamlines through a straight pipe of given cross sectional form  $\Omega$ , then it is standard that the velocity flow u satisfies (5). In this context, Serrin's result can be formulated as follows: the tangential stress on the pipe wall is the same at all points of the wall if and only if the pipe has a circular cross section.

The same differential equation and boundary condition arise in the linear theory of torsion of a solid straight bar of cross section  $\Omega$  (see [204] for more details), In this context, Serrin's result states that, when a solid straight bar is subject to torsion, the magnitude of the resulting traction which occurs at the surface of the bar is independent of position if and only if the bar has a circular cross section. Moreover, overdetermined problems like (5)-(6) arise when one study critical domains for the torsional rigidity.

Now we describe the proofs given by Alexandrov and Serrin for Theorems A and B, respectively. The idea behind Alexandrov's proof is the following: under the assumptions of Theorem A, S is a sphere if and only if for every  $\omega \in \mathbb{R}^n$ there exists a hyperplane  $\pi_{\omega}$  orthogonal to  $\omega$  such that the hypersurface S is symmetric about  $\pi_{\omega}$ . In order to achieve this symmetry property, let  $\omega \in \mathbb{R}^n$  be a fixed direction and let  $\pi_{\lambda}$  be a 1-parameter family of hyperplanes orthogonal to  $\omega$ . We move these hyperplanes towards S until we touch S and we continue to move them until we find a critical hyperplane  $\pi^*$ . In order to define the critical hyperplane  $\pi^*$  we define:  $\mathcal{T}^*$  is the halfspace such that  $\partial \mathcal{T}^* = \pi^*$ ,  $S' = S \cap \mathcal{T}^*$ is the part of S contained in  $\mathcal{T}^*$  and  $S^*$  is the reflection with respect to  $\pi^*$  of S'. With these notations the critical hyperplane  $\pi^*$  is defined in the following way:

- (i) either  $S^*$  is tangent to S at a point  $p \in S \setminus \pi^*$ ;
- (*ii*) or  $\pi^*$  is orthogonal to S at a point  $q \in S \cap \pi^*$ .

Since S and S<sup>\*</sup> are tangent to p (or q), we can locally write the two hypersurfaces as graphs of function u and  $u^*$ , respectively, on the tangent space at p (or q). Moreover, by construction  $u - u^*$  has a sign, and u and  $u^*$  satisfy the following elliptic equation:

$$H = \frac{1}{n-1} \operatorname{div} \left( \frac{\nabla v}{\sqrt{1+|\nabla v|^2}} \right), \tag{7}$$

with H constant (equal to the value of the mean curvature) and where the righthand side is the expression of the mean curvature of the graph of a function v. At this point the PDE's tools come into our help: by using the strong maximum principle in case (i) and the Hopf's boundary point Lemma in case (ii) we conclude that  $u = u^*$ , which implies that the hypersurface S and its reflection  $S^*$  with respect to the plane  $\pi^*$  coincide. Since the direction  $\omega$  is arbitrary we conclude that S must be spherically symmetric and hence S is a sphere.

The proof of Theorem B given by Serrin generalizes Alexandrov's reflection principle. Also in this case the goal is to prove that  $\Omega$  is symmetric with respect to the hyperplane  $\pi_{\omega}$  orthogonal to  $\omega \in \mathbb{R}^n$ . As before, let  $\mathcal{T}^*$  be the halfspace such that  $\partial \mathcal{T}^* = \pi^*$ ,  $\Omega' = \Omega \cap \mathcal{T}^*$  is the part of  $\Omega$  contained in  $\mathcal{T}^*$  and  $\Omega^*$ is the reflection with respect to  $\pi^*$  of  $\Omega'$ . We consider the critical hyperplane  $\pi^*$  at the two critical positions (i) and (ii) with  $S = \partial \Omega$ . We compare the solution u to (5)-(6) to its reflection v, defined in  $\Omega^*$ . Since u - v is harmonic and non negative either u - v > 0 in  $\Omega^*$  or  $u - v \equiv 0$  in  $\Omega^*$ . If u - v > 0 occurs, the Hopf's Lemma gives a contradiction in case (i). Case (ii) is more delicate and one has to use Serrin's corner Lemma (see [204, Lemma 1]) to find a contradiction. Hence  $u - v \equiv 0$  in  $\Omega^*$  and  $\Omega$  is symmetric about  $\pi^*$ .

Both Alexandrov's and Serrin's results and proofs originate a great interest in geometric analysis and PDE's communities. Indeed the method of moving planes is a powerful tool which has been used to prove several results in geometric analysis, for elliptic and parabolic PDEs, Harnack's inequalities and many others results (see e.g. [9, 23, 24, 36, 37, 38, 146, 155, 168, 192, 193, 202]). Moreover, one of the most influencing application of the method of moving planes in the theory of PDEs is the approach of Gidas-Ni-Nirenberg in [109] and [110] where the authors prove symmetry results for positive solutions of elliptic PDEs.

As for Alexandrov Theorem, we mention that Theorem B is stated in its simplest version and many generalizations have been studied. In Chapter 2 we will study symmetry results in the spirit of Theorem B in more general Riemannian manifolds (see Section 2.1) and for quasilinear operators in convex cones (see Section 2.2).

As already mentioned, the inequality that motivates the study in Part II is the *Sobolev inequality* (see [210] and also [105, 177]) in  $\mathbb{R}^n$ ,  $n \geq 2$ : given  $1 there exists a constant <math>C_{n,p} > 0$  such that the following inequality

$$\left(\int_{\mathbb{R}^n} |u|^{p^*} dx\right)^{p/p^*} \le C_{n,p} \int_{\mathbb{R}^n} |\nabla u|^p dx \tag{8}$$

holds for every function u in the homogeneous space

$$\mathcal{D}^{1,p}(\mathbb{R}^n) := \left\{ u \in L^{p^*}(\mathbb{R}^n) : \nabla u \in L^p(\mathbb{R}^n) \right\},\tag{9}$$

where  $p^*$  is the *critical Sobolev exponent*, i.e.

$$p^* = \frac{np}{n-p} \,. \tag{10}$$

Sobolev inequality appears when one considers the embedding of the Sobolev space in some  $L^p$ -space. Moreover it naturally appears in regularity theory for PDEs and it is related to the isoperimetric inequality as well as to many other important topics in mathematical analysis.

The natural functional associated to (8) is the Sobolev functional given by:

$$\mathcal{J}(u) := \frac{\int_{\mathbb{R}^n} |\nabla u|^p \, dx}{\left(\int_{\mathbb{R}^n} |u|^{p^*} \, dx\right)^{p/p^*}} \,. \tag{11}$$

We are interested in *critical points* of this functional  $\mathcal{J}$  and we want to write and study the Euler-Lagrange equation associated to (11). Let u be a (positive) critical point of (11) and we compute the first variation of the functional. Let  $\varepsilon > 0$  and let  $\varphi \in C_c^{\infty}(\mathbb{R}^n)$  be a test function. Since u is a critical point we get

$$\frac{d}{d\varepsilon}\Big|_{\varepsilon=0}\mathcal{J}(u+\varepsilon\varphi)=0\,,$$

or equivalently, from a direct computation,

$$\int_{\mathbb{R}^n} |\nabla u|^{p-2} \nabla u \cdot \nabla \varphi \, dx - \frac{\int_{\mathbb{R}^n} |\nabla u|^p \, dx}{\int_{\mathbb{R}^n} |u|^{p^*} \, dx} \int_{\mathbb{R}^n} u^{p^*-1} \varphi \, dx = 0,$$

for all  $\varphi \in C_c^{\infty}(\mathbb{R}^n)$ . By multiplying u by an appropriate constant we may assume that u satisfies

$$\int_{\mathbb{R}^n} |\nabla u|^{p-2} \nabla u \cdot \nabla \varphi \, dx - \int_{\mathbb{R}^n} u^{p^*-1} \varphi \, dx = 0 \,, \quad \text{for all } \varphi \in C_c^\infty(\mathbb{R}^n) \,. \tag{12}$$

Equation (12) is the weak formulation of the following quasilinear PDE:

$$\Delta_p u + u^{p^* - 1} = 0 \quad \text{in } \mathbb{R}^n,$$

where  $\Delta_p u$  is the usual *p*-Laplace operator, i.e.

$$\Delta_p u = \operatorname{div}\left(|\nabla u|^{p-2} \nabla u\right),$$

and  $p^*$  is the Sobolev critical exponent defined in (10). Hence we prove that the Euler-Lagrange equation (for positive critical point) associated to the functional (11) (and so to the inequality (8)) is the following *critical p-Laplace equation in*  $\mathbb{R}^n$ :

$$\begin{cases} \Delta_p u + u^{p^* - 1} = 0 & \text{ in } \mathbb{R}^n \\ u > 0 & \text{ in } \mathbb{R}^n. \end{cases}$$
(13)

We observe some properties of (13) which will be useful in the following: if u is a solution to (13) then also  $u(x+\alpha)$ , for  $\alpha \in \mathbb{R}^n$ , and  $\lambda^{\frac{n-p}{p}}u(\lambda x)$ , for  $\lambda \in \mathbb{R}$ , are solutions to (13). These two properties are called *invariance under translations* and *invariance under rescaling* of (13).

As mentioned in [224], it is well known that the profile of solutions of the equation

$$\Delta_p u + |u|^{p^* - 2} u = 0 \quad \text{in } \mathbb{R}^n \, .$$

plays a central role in the blow-up theories of critical equations. Problem (13) is also interesting from the point of view of the calculus of variations. Since the embedding  $W^{1,p}(\mathbb{R}^n) \hookrightarrow L^{p^*}(\mathbb{R}^n)$  is not compact, the classical tools of the calculus of variations (e.g. the Mountain Pass Lemma or the direct method of the calculus of variations) do not apply to

$$\mathcal{E}(u) = \frac{1}{p} \int_{\mathbb{R}^n} |\nabla u|^p \, dx - \frac{1}{p^*} \int_{\mathbb{R}^n} u^{p^*} \, dx \,,$$

which is the energy functional associated to (13).

Equation (13) is also related to the Yamabe problem. We recall that the Yamabe problem (see [10, 201, 220, 226]) is the following: given a compact Riemannian manifold  $(M, g_0)$  of dimension  $n \ge 3$ , is it possible to find a metric g conformal to  $g_0$  such that its scalar curvature  $R_g$  is equal to a prescribed constant R? One can show that this problem boils down to showing the existence of a positive solution u to the nonlinear PDE

$$\frac{4(n-1)}{n-2}\Delta_{g_0}u - R_{g_0}u + Ru^{\frac{n+2}{n-2}} = 0, \qquad (14)$$

where  $\Delta_{g_0}$  is the Laplace-Beltrami operator of the metric  $g_0$  and  $R_{g_0}$  denotes the scalar curvature of the metric  $g_0$ . Indeed, if u solves (14) then the metric  $g = u^{\frac{4}{n-2}}g_0$  is such that  $R_q = R$ .

When  $(M, g_0)$  is the round sphere then  $R_{g_0} = n(n-1)$  and (14) can be read on  $\mathbb{R}^n$  by means of the stereographic projection. More precisely, consider the inverse stereographic projection  $F : \mathbb{R}^n \to \mathbb{S}^n$ . Then  $v : \mathbb{S}^n \to \mathbb{R}$  solves (14) if and only if the function

$$u(x) = \left(\frac{2}{1+|x|^2}\right)^{\frac{n-2}{2}} v(F(x))$$

solves

$$\Delta u + \frac{n-2}{4(n-1)} R u^{\frac{n+2}{n-2}} = 0 \,,$$

which is, up to a multiplicative constant, equation (13) with p = 2.

We mention that an important task related to the Sobolev inequality (8) is the following: to compute the value of the best constant in (8), i.e. to obtain the *sharp Sobolev inequality*. The value of the best constant was obtained independently by Talenti [217] and Aubin [10]; moreover they showed that equality in the sharp Sobolev inequality is achieved by the following *Aubin-Talenti radial functions* (or bubbles):

$$\mathcal{U}(x) = \frac{a}{\left(b + |x - x_0|^{\frac{p}{p-1}}\right)^{\frac{n-p}{p}}} \quad \text{for some } a, b > 0 \text{ and } x_0 \in \mathbb{R}^n \,. \tag{15}$$

The idea to prove this result is to minimize the functional (11) in  $\mathcal{D}^{1,p}(\mathbb{R}^n)$ . By using the Schwarz symmetrization technique Talenti proved that the spherically symmetric radial rearrangement of u preserves  $||u||_{L^{p^*}(\mathbb{R}^n)}$  and makes  $||\nabla u||_{L^p(\mathbb{R}^n)}$  smaller. Hence, minimizers are forced to be spherically symmetric radial functions and, by computing the Gateaux differential of the functional  $\mathcal{J}$  for these type of functions, one obtains that minimizers of  $\mathcal{J}$  (we assume for simplicity that they are positive) satisfy the following ode:

$$\frac{1}{r^{n-1}}(r^{n-1}(u')^{p-1})' + u^{p^*-1} = 0,$$

where r = |x|. This yields to the Aubin-Talenti functions (15).

We observe that the results of Aubin and Talenti classify minima of the functional (11). An interesting issue in PDE's community is to look for a characterization of critical points of the functional (11), i.e. a characterization of solutions to (13). The classification of all solution to (13) started in the seminal papers [110] (see also [178]) and [48] for p = 2 and it has been the object of several studies. Recently in [224] and [203], solutions to (13) belonging to the class  $\mathcal{D}^{1,p}(\mathbb{R}^n)$  have been completely characterized. In particular, the following Liouville-type theorem holds:

**Theorem C** ([224, 203]): let  $n \ge 2$ , 1 and let <math>u be a solution to (13) such that  $u \in \mathcal{D}^{1,p}(\mathbb{R}^n)$ . Then there exist  $\lambda > 0$  and  $x_0 \in \mathbb{R}^n$  such that

$$u(x) = \mathcal{U}_{\lambda, x_0}(x) = \left(\frac{\lambda^{\frac{1}{p-1}} n^{\frac{1}{p}} \left(\frac{n-p}{p-1}\right)^{\frac{p-1}{p}}}{\lambda^{\frac{p}{p-1}} + |x - x_0|^{\frac{p}{p-1}}}\right)^{\frac{p-1}{p}}.$$
 (16)

We mention that the previous result is not true if u may change sign (see e.g. [84, 79, 80]). Actually, in the semilinear case p = 2 a more general version of Theorem C is available in literature, indeed the assumption  $u \in \mathcal{D}^{1,p}(\mathbb{R}^n)$ can be removed by using the Kelvin tranform. The other crucial ingredient to prove Theorem C is a generalization of the method of moving planes introduced by Serrin (see e.g. [27, 109]). In the quasilinear case  $p \neq 2$  the problem is more complicate because one has to tackle the nonlinear nature of the *p*-Laplace operator and Kelvin type transform is not available. A first version of Theorem C for  $\frac{2n}{n-2} \leq p < 2$  was proved in [76]; this result was extended to the range of exponent  $1 in [224]. Finally, in [203] the range <math>2 \leq p < n$  was considered and this concludes the picture. The key point in the proof of Theorem C is to show that the solutions to (13) are radial and then, from [123] one knows that the only positive, radially symmetric solutions of (13) are of the form (16).

Motivated by recent studies on the Sobolev inequality in convex cones and on the anisotropic Sobolev inequality in Chapter 3 we prove a generalization of Theorem C for critical anisotropic p-Laplace equations in convex cones. We mention that the Kelvin transform and the method of moving planes are not helpful neither for anisotropic problems nor inside cones. For this reason, in Chapter 3, we provide a new approach to the characterization of solutions to critical p-Laplacian equations, which is based on integral identities rather than moving planes. This approach takes inspiration from [206] where non-existence results are considered and also from [29, 30, 40, 215] where classical overdetermined problems for PDEs are considered (see also [63, 180] for analogous problems in convex cones). We mention that our approach is new also in the Euclidean space with the Euclidean norm (see Appendix C)

In Chapter 4 we investigate the validity of Sobolev-type inequalities in a particular class of Riemannian manifolds. The investigation of functional inequalities on Riemannian manifolds is a very wide and active research field (see e.g. [131, 129] where the discussion is mainly devoted to the rigorous definition of *Sobolev spaces* on Riemannian manifolds and the properties of the associated embeddings or functional inequalities). As already mentioned the first result dealing with the *optimal constant* of the Euclidean Sobolev inequality is due to the celebrated papers [12] and [217]. Then, T. Aubin continued the investigation of Sobolev-type inequalities as well as related optimality issues on Riemannian manifolds: see [10] (in the case of compact manifolds with applications to the Yamabe problem), [11] (where higher-order inequalities are also considered) and [13] (for estimates of the best constants of subcritical Sobolev embeddings). Some of the results of [12] were then improved in [130, 132], essentially by requiring bounds on the Ricci curvature in place of the sectional curvatures.

Functional-analytic issues on the hyperbolic space  $\mathbb{H}^n$  have drawn a lot of interest recently, the latter being in a sense the simplest example of a noncompact Riemannian manifold with negative curvatures. In this regard, we mention [25], where an improved version of the Poincaré (or spectral-gap) inequality is obtained with optimal remainder terms of Hardy type. In fact Hardy-type inequalities were also considered in [50], for non-standard weights satisfying suitable differential inequalities (with explicit applications to Cartan-Hadamard manifolds). Finally, in [175] the author establishes an inequality on  $\mathbb{H}^n$  yielding the optimal Sobolev and Poincaré inequalities simultaneously.

For a rather complete survey dealing with connections between the Poincaré

inequality, the logarithmic Sobolev inequality, measure-concentration issues and isoperimetric bounds (not only on Riemannian manifolds actually), we refer to [153], while in [18] the authors, starting from inequalities that interpolate between Poincaré and log-Sobolev, provide a method to obtain weighted inequalities of the same type for weights complying with optimal growth conditions.

Most of the results we have mentioned above are focused on proving the validity of Sobolev (or Poincaré, or Hardy) inequalities. One can also investigate topological consequences: in [51, 152] it is shown (under suitable curvature or volume-growth assumptions, respectively) that a Riemannian manifold supporting a Sobolev inequality with Euclidean constant is necessarily isometric to  $\mathbb{R}^n$ . For similar problems, but rather different methods of proof, see also [185].

More specifically, in Chapter 4 we study Sobolev-type inequalities on a particular class of Riemannian manifolds: the so-called Cartan-Hadamard man*ifolds* (i.e. Riemannian manifolds with negative sectional curvatures) under curvature bounds. The situation is the following: given a Riemannian manifold with everywhere nonpositive sectional curvatures then the Sobolev inequality holds true. Moreover if the sectional curvatures are bounded above by a negative constant then also the Poincaré inequality holds true. Motivated by these results in Chapter 4 we study what happens in between, explicitly we will assume that the sectional curvatures of the Riemannian manifold are bounded from above by a negative power of the distance from a fixed pole (times a negative constant), then all the  $L^p$  inequalities that interpolate between Poincarè and Sobolev hold in the radial setting, provided such power lies in the interval (-2,0), except the Poincaré inequality. Moreover, if the power is equal to -2then p must necessarily be strictly larger (in a quantitative way) than 2. Upon assuming similar bounds from below on the Ricci curvature, we show that the nonradial version of such Sobolev-type inequalities fails, except for the Sobolev one. Finally, in Section 4.6 we prove optimal smoothing effects for a porous medium equation.

### Part I

# Symmetry and quantitative stability results for problems in geometric analysis

### Introduction to Part I

Part I of this thesis deals with quantitative studies for hypersurfaces embedded in space forms with some curvature close to a constant (in Chapter 1) and symmetry results for overdetermined problems (in Chapter 2). This part collects the results obtained in the following papers: [63, 65, 196].

In this Introduction we describe in more details the results contained in Chapters 1 and 2.

**Chapter 1.** The first chapter of this thesis is dedicated to quantitative studies for hypersurfaces with almost constant curvature. The starting point that motivates the study in this chapter is Alexandrov's Theorem (Theorem A in the Introduction). Alexandrov's Theorem has attracted a lot of interest in the last decades and several possible generalizations have been considered. It is well-known that Alexandrov's Theorem is in general false for non-embedded submanifolds (see e.g. [139] and [216] for classical counterexamples in higher dimension and in  $\mathbb{R}^3$ , respectively). However, for immersed hypersurfaces, one can add some condition in order to guarantee that S is a sphere: in particular Hopf proved in [136] that every constant mean curvature  $C^2$ -regular sphere *immersed* in the 3-dimensional Euclidean space is necessary a round sphere (see also [1, 169, 170] for a very recent generalization of Hopf's theorem to simplyconnected homogeneous 3-manifolds), and Barbosa and DoCarmo [19] proved that every compact, orientable and stable hypersurface *immersed* in  $\mathbb{R}^n$  is a round sphere (see also [20] for generalizations of this result). It is also well-known that there exists *non-closed* constant mean curvature hypersufaces embedded in  $\mathbb{R}^3$  which are not diffeomorphic to a sphere, like for instance unduloids (see [81] and [140] for a possible generalization to higher dimensions).

Another interesting generalization was provided by Alexandrov himself. In [5] and [7] a generalization of Theorem A for hypersurfaces embedded in the hyperbolic space  $\mathbb{H}^n$  and in the hemisphere  $\mathbb{S}^n_+$ . Here and in the following, we will indicate with  $\mathbb{M}^n_+$  the three spaces:  $\mathbb{R}^n$ ,  $\mathbb{H}^n$ ,  $\mathbb{S}^n_+$  and we call them *space* forms. Moreover the theorem holds true for a large class of curvature operators, not only the mean curvature (see [7, 52, 125, 141, 145, 158, 199, 212, 227]). This is the type of generalization that we are interested in. In order to properly describe the result, we introduce some notation.

Let S be a  $C^2$ -regular, connected, closed hypersurface embedded in  $\mathbb{M}_+^n$ . Then S is always the boundary of a relatively compact connected open set  $\Omega \subset \mathbb{M}_+^n$  and we orient S by using the normal vector field to S inward with respect to  $\Omega$ . Let  $\{k_1, \ldots, k_{n-1}\}$  be the principal curvatures of S increasingly ordered.

We denote by  $H_S$  one of the following functions:

(I) the mean curvature H, with

$$H := \frac{1}{n-1} \sum_{i=1}^{n-1} k_i$$

 $(II) f(k_1, \ldots, k_{n-1})$ , where

$$f: \{x = (x_1, \dots, x_{n-1}) \in \mathbb{R}^{n-1} : x_1 \le x_2 \le \dots \le x_{n-1}\} \to \mathbb{R},\$$

is a  $C^2$ -function such that

$$f(x) > 0$$
, if  $x_i > 0$  for every  $i = 1, ..., n-1$ 

and f is concave on the component  $\Gamma$  of  $\{x \in \mathbb{R}^{n-1} : f(x) > 0\}$  containing  $\{x \in \mathbb{R}^{n-1} : x_i > 0\}$ .

For instance, if  $H_r$  denotes the *r*-higher order curvature of *S* defined as the elementary symmetric polynomial of degree *r* in the principal curvatures of *S*, then  $H_r^{1/r}$  satisfies (*ii*). By using this notation, Alexandrov's theorem can be stated as follows:

**Theorem I.A**: let S be a  $C^2$ -regular, connected, closed (i.e. compact and without boundary) hypersurface embedded in  $\mathbb{M}^n_+$  and let  $H_S$  be as in (I) or (II). Then  $H_S$  is constant if and only if S is a distance sphere.

The proof of this theorem can be done with the same technique introduced by Alexandrov, but we mention that for particular choices of the function fin (II) different proofs, based on integral inequalities and geometric identities, are available in literature (see e.g. [30, 127, 128, 172, 194, 198, 197]). For more recent generalizations of Alexandrov's Theorem we refer to [41], where constant mean curvature hypersurfaces are studied in rotationally symmetric Riemannian manifolds (e.g. the space forms, the Schwarzschild, the DeSitter-Schwarzschild and Anti-DeSitter-Schwarzschild manifolds), to [82] where the regularity assumptions on S are minimal and to [45, 56] where the non-local version of Alexandrov's Theorem is proved.

Theorems A, B and I.A have the same conclusion: a solution exists if and only if it is a sphere/ball. The rigidity of these problems is due to the following overdetermined conditions: the curvature  $H_S$  is constant in Alexandrov Theorems and the normal derivative  $\partial_{\nu} u$  is constant on  $\partial\Omega$  in Serrin Theorem. In view of this remark it is natural to investigate the stability of this results:

### if the overdetermined condition is slightly perturbed, can we say that the domain is close to a ball? Can we quantify the closedness?

Stated like that, the answer to the questions is in general negative. Indeed, it has been showed in [39] for Serrin Theorem and in [58] for Alexandrov Theorem that if the overdetermined condition is close to a constant then bubbling phenomena can appear, i.e. the domain can be close to a bunch of balls connected by small necks. We mention that both [39] and [58] do not use the moving planes method and they perturb proofs based on integral identities. Nevertheless if some hypothesis is introduced in order to prevent bubbling phenomena then the moving planes method can be studied in a quantitative way to obtain sharp quantitative information of the proximity of the solution to a single ball. In [67, 69] one condition that can prevent bubbling phenomena is the so-called touching ball condition (we will comment a little bit on it after the theorem). Our stability result related to Theorem I.A proved in [65] can be stated as follows

**Theorem I.B:** let S be a  $C^2$ -regular, connected, closed hypersurface embedded in  $\mathbb{M}^n_+$  satisfying a uniform touching ball condition of radius  $\rho$ . There exist constants  $\varepsilon$ , C > 0 such that if

$$\operatorname{osc}(\mathsf{H}_{\mathrm{S}}) := \max_{\mathrm{p} \in \mathrm{S}} \mathsf{H}_{\mathrm{S}}(\mathrm{p}) - \min_{\mathrm{p} \in \mathrm{S}} \mathsf{H}_{\mathrm{S}}(\mathrm{p}) \le \varepsilon \,, \tag{I.1}$$

then there are two concentric balls  $B_r$  and  $B_R$  in  $\mathbb{M}^n_+$  such that

$$S \subseteq \overline{B}_R \setminus B_r \tag{I.2}$$

and

$$R - r \le C \operatorname{osc}(\mathsf{H}_{\mathrm{S}}) \,. \tag{I.3}$$

The constants  $\varepsilon$  and C depend only on n, on upper bounds on  $\rho^{-1}$  and on the area of S.

The uniform touching ball condition of radius  $\rho$  in the previous theorem means that at any point of S there exist two balls of radius  $\rho$  tangent to S one from inside and one from outside. Since the constant  $\rho$  is fixed, a bubbling phenomenon can not appear as shown in [58, Theorem 1.1]. We emphasize that in Theorem I.B the stability estimate (I.3) is optimal and it is new in the general setting of Theorem I.B; moreover we mention that Theorem I.B remarkably improves the previous stability results available in literature (see Chapter 1 for a more detailed discussion).

**Chapter 2.** The second chapter of this thesis is dedicated to Serrin's overdetermined problem and related issues. We mention that the celebrated moving planes method introduced by Serrin can be used to prove a more general version of Serrin's Theorem involving uniformly elliptic quasilinear equation (see [204]). Moreover the moving planes method has been used to prove an analogue result in space forms (see [150, 171]) and in a nonlocal setting (see [90]). As for Alexandrov theorem, Serrin theorem can be proven by using different approaches, based on integral identities (see [215]). This allows to extend Serrin's theorem to possibly degenerate quasilinear equations and to fully nonlinear equations (see [3, 22, 29, 31, 33, 40, 42, 44, 55, 68, 70, 74, 75, 92, 102, 103, 106, 126, 133, 134, 135, 182, 190, 191, 196, 207, 209, 213]) and also for domains with Lipschitz singularities or contained in a convex cone (see [63, 180, 188]).

We now review Serrin's original proof and some alternative proofs of Serrin theorem, in particular the one due to Weinberger [215] and the more recent integral approach due to Brandolini-Nitsch-Salani-Trombetti in [40] (see also [164] and [176]).

The moving planes method. Let u be the solution to (5)-(6). For a fixed direction  $\omega \in \mathbb{R}^n$ , the moving plane procedure gives a critical hyperplane  $\pi^*$  such that

(i) either  $\Omega^*$  is tangent to  $\Omega$  at a point  $p \in \partial \Omega \setminus \pi^*$ ,

(*ii*) or  $\pi^*$  is orthogonal to  $\partial\Omega$  at some  $q \in \partial\Omega \cap \pi^*$ ,

where we recall that  $\Omega^*$  is the reflection with respect to  $\pi^*$  of the cap  $\Omega'$ . Now we define the reflection of u with respect to  $\pi^*$ , i.e.  $v(x) = u(x^*)$ , for  $x \in \Omega^*$ , where  $x^*$  is the reflection of x with respect to  $\pi^*$ . Is it easy to show that u - vsatisfies the following problem:

$$\begin{cases} \Delta(u-v) = 0 & \text{ in } \Omega^* \\ u-v = 0 & \text{ on } \partial \Omega^* \cap \pi^* \\ u-v \ge 0 & \text{ on } \partial \Omega^* \setminus \pi^* , \end{cases}$$

where the last inequality follows from the weak comparison principle. Then, by the strong maximum principle we have that either u - v > 0 in  $\Omega^*$  or  $u - v \equiv 0$ in  $\Omega^*$ . The first case leads to a contradiction, indeed in case (i) by the Hopf Lemma we have that

$$\partial_{\nu}(u-v)(p) < 0,$$

but from (6) we have that  $\partial_{\nu}u(p) = \partial_{\nu}v(p)$ . In case (*ii*) one shows that all the first and second derivatives of u and v coincide at q, and this is in contradiction with the so-called Serrin's corner Lemma [204, Lemma 1] (a refinement of the maximum principle).

Hence  $u \equiv v$  in  $\Omega^*$ , i.e.  $\Omega$  is symmetric with respect to  $\pi^*$ . This implies that for every direction  $\omega$ ,  $\Omega$  is symmetric with respect to the hyperplane orthogonal to  $\omega$ ; moreover, by construction,  $\Omega$  is simply connected, and then it has to be a ball.

Weinberger's proof. Let u be the solution to (5)-(6) and consider the following function:

$$P(u(x)) := |\nabla u(x)|^2 + \frac{2}{n}u(x).$$
 (I.4)

We compute the Laplacian of this function and, by using Cauchy-Schwarz inequality, we get  $\Delta P(u) \geq 0$ . Since  $P(u(x)) = c^2$  on  $\partial \Omega$  (due to the boundary conditions) from the strong maximum principle we conclude that either

$$P(u(x)) \equiv c^2 \quad \text{in } \Omega \,, \tag{I.5}$$

or

$$P(u(x)) < c^2 \quad \text{in } \Omega. \tag{I.6}$$

In the second case, by integrating (I.6) over  $\Omega$  and by using the divergence theorem, we get

$$\frac{n+2}{n} \int_{\Omega} u \, dx < c^2 |\Omega| \,. \tag{I.7}$$

The classical Pohozaev identity (see e.g. [186]) yields

$$\frac{n-2}{n}\int_{\Omega}|\nabla u|^2\,dx + \frac{c^2}{2}\int_{\partial\Omega}x\cdot\nu\,d\sigma = n\int_{\Omega}u\,dx\,,\tag{I.8}$$

and from (5) and (6) we obtain

$$(n+2)\int_{\Omega} u\,dx = c^2 n|\Omega|\,,\tag{I.9}$$

which contradicts (I.7). This means that (I.5) holds true, hence P(u) is a harmonic function and this implies that the equality in Cauchy-Schwarz inequality holds, i.e.

$$\nabla^2 u = -\frac{1}{n} \mathrm{Id} \,. \tag{I.10}$$

Hence

$$u(x) = \frac{1}{2n} \left( R^2 - |x - x_0|^2 \right) , \qquad (I.11)$$

for some R > 0 and  $x_0 \in \mathbb{R}^n$ ; from (I.11) and since u = 0 on  $\partial \Omega$  we immediately conclude that  $\Omega$  is the ball of radius R centred at  $x_0$ .

We mention that the approach of Weinberger inspired several works in the context of elliptic PDEs (see e.g. [48, 93, 181, 211]), in particular in [92, 103, 106] Weinberger's argument is used to prove symmetry result for overdetermined problems associated to more general PDE's.

An integral approach. Given a symmetric matrix  $A = \{a_{ij}\} \in \mathbb{R}^{n \times n}$ , for any  $k = 1, \ldots, n$  we denote by  $S_k(A)$  the sum of all the principal minors of A of order k. Observe that

$$S_1(A) = \text{Tr}(A), \quad S_2(A) = \sum_{1 \le i_1 < i_2 \le n} \lambda_{i_1} \lambda_{i_2}, \quad S_n(A) = \det(A), \quad (I.12)$$

where  $\operatorname{Tr}(A)$  is the trace of A,  $\det(A)$  is the determinant of A and  $\lambda_1, \dots, \lambda_n$  are the eigenvalues of A.

We are interested in  $S_1$  and  $S_2$ : suppose  $\text{Tr}(A) \ge 0$ , then the following Newton's inequality for symmetric matrices (see e.g. [124]),

$$S_2(A) \le \frac{n-1}{2n} (S_1(A))^2$$
 (I.13)

holds, and the equality in (I.13) holds if and only if

$$A = \frac{\operatorname{Tr}(A)}{n} \operatorname{Id}.$$
 (I.14)

Newton's inequality (I.13) can be used to prove Serrin overdetermined Theorem. Indeed, by taking  $A = \nabla^2 u$  we have that  $S_1(\nabla^2 u) = \Delta u$  and a direct computation shows that  $S_2(\nabla^2 u)$  can be written in the following divergence form

$$S_2(\nabla^2 u) = \frac{1}{2} \partial_i (S_{ij}^2(\nabla^2 u) \partial_j u) \quad \text{where } S_{ij}^2(\nabla^2 u) = -\partial_{ij}^2 u + \delta_{ij} \operatorname{Tr} (\nabla^2 u)$$
(I.15)

(here we are adopting the Einstein summation convention for repeated indices). Let u be the solution to (5)-(6). First we observe that, from (I.15) via an integration by parts,

$$\int_{\Omega} |\nabla u|^2 dx = -2 \int_{\Omega} |\nabla u|^2 dx + 2(n-1) \int_{\Omega} H |\nabla u|^2 dx + c^3 |\partial \Omega| \qquad (I.16)$$

where we used the fact that the mean curvature H of a level set of u (at a regular point) satisfies

$$-\Delta u = (n-1)H|\nabla u| - \frac{\partial_{ij}^2 u \partial_i u \partial_j u}{|\nabla u|^2} \,.$$

From (I.16) and the fact that the constant c in (6) is given by  $-|\Omega|/|\partial\Omega|$  we get

$$(n-1)\int_{\Omega} H|\nabla u|^3 \, dx = \frac{3}{2}\int_{\Omega} |\nabla u|^2 \, dx - \frac{c^2}{2}|\Omega|\,, \qquad (I.17)$$

Then, from (I.9) and (I.17) we obtain

$$\int_{\Omega} H|\nabla u|^3 \, dx = \frac{c^2|\Omega|}{n-2} \,, \tag{I.18}$$

where we used the divergence theorem and (5)-(6) to prove that

$$\int_{\Omega} |\nabla u|^2 \, dx = \int_{\Omega} u \, dx$$

From (I.18) we get

$$\frac{c^2|\Omega|}{n-2} = \frac{1}{n-1} \int_{\Omega} S_{ij}^2(\nabla^2 u) \partial_i u \partial_j u \, dx \,, \tag{I.19}$$

where we used the following pointwise identity

$$-(n-1)H = \frac{S_{ij}^2(\nabla^2 u)\partial_i u\partial_j u}{|\nabla u|^3}$$

Integrating by parts and by using (I.13), (5) (I.9) we obtain

$$\frac{c^2|\Omega|}{n-2} = \frac{2}{n-1} \int_{\Omega} S_2(\nabla^2 u) u \, dx \le \frac{1}{n} \int_{\Omega} u(\Delta u)^2 \, dx = \frac{c^2|\Omega|}{n-2} \,. \tag{I.20}$$

The conclusion follows because (I.20) implies that equality holds true in (I.13) so (I.10) holds true.

In Chapter 2 there are three symmetry results: the first one is contained in [196] and is related to a generalization of Weinberger's proof to a particular class of Riemannian manifolds; the remaining ones are contained in [63] and are related to a Serrin-type result for domains inside convex cones of the Euclidean space and in space forms. In what follows we will give the flavour of the results that we are going to prove in Chapter 2 and we refer to Chapter 2 for the precise definitions and statements.

Weinberger argument on Riemannian manifolds. The first symmetry result in Chapter 2 is a generalization of Weinberger's proof to the so-called *model* manifolds with non-negative Ricci curvature. Model manifolds are rotationally symmetric Riemannian manifolds  $(\mathbb{M}_{\sigma}^{n}, g_{\mathbb{M}_{\sigma}^{n}})$ , i.e.  $\mathbb{M}_{\sigma}^{n} = [0, R) \times \mathbb{S}^{n-1}$ , for some R > 0, and the metric is

$$g_{\mathbb{M}^n_{\sigma}} = dr \otimes dr + \sigma^2(r) g_{\mathbb{S}^{n-1}} ,$$

for some smooth function  $\sigma : [0, R) \to [0, +\infty)$  and where  $g_{\mathbb{S}^{n-1}}$  is the usual metric on the round sphere  $\mathbb{S}^{n-1}$  (see Definition 2.1 for the precise definition). We mention that important examples of model manifolds are the already celebrated space-forms. In particular

- the Euclidean space  $\mathbb{R}^n$  is isometric to the model manifold  $\mathbb{M}^n_{\sigma}$  with  $\sigma(r) = r: [0, +\infty) \to [0, +\infty);$
- the hyperbolic space  $\mathbb{H}^n$  is isometric to the model manifold  $\mathbb{M}^n_{\sigma}$  with  $\sigma(r) = \sinh(r) : [0, +\infty) \to [0, +\infty);$
- the standard sphere  $\mathbb{S}^n \setminus \{N\}$ , where N is the north pole, is isometric to the model manifold  $\mathbb{M}^n_{\sigma}$  with  $\sigma(r) = \sin(r) : [0, \pi) \to [0, +\infty)$ .

Under the assumption that the Ricci curvature of the model manifold is nonnegative and under a suitable "compatibility" assumption between the solution of the overdetermined problem and the geometry of the model, in Section 2.1, we will prove the following rigidity result

**Theorem I.C:** let  $(\mathbb{M}^n_{\sigma}, g_{\mathbb{M}^n_{\sigma}})$  be a model manifold such that

$$\operatorname{Ric}_{\mathbb{M}^n} \ge 0$$
, and  $\sigma' > 0$ 

Let  $\Omega \subset \mathbb{M}^n_{\sigma}$  be a smooth domain and assume that  $\Omega$  supports a solution u of

$$\begin{cases} \Delta u = -1 & \text{in } \Omega \\ u = 0 & \text{on } \partial \Omega \\ \partial_{\nu} u = c & \text{on } \partial \Omega . \end{cases}$$
(I.21)

(where  $\Delta$  is the Laplace-Beltrami operator). If u satisfies the following "compatibility" condition

$$\int_{\Omega} \frac{(\sigma''\sigma^{n-1})'}{\sigma^{n-1}} u^2 \ge 0, \qquad (I.22)$$

then  $\Omega$  is a Euclidean ball.

The proof of Theorem I.C is based on the Weinberger's approach. Explicitly we consider the P-function (I.4), and we show that it is a subharmonic function. The proof is achieved by using Bochner formula

$$\Delta |\nabla u|^2 = 2 \left| \nabla^2(u) \right|^2 + 2g(\nabla(\Delta u), \nabla u) + 2\operatorname{Ric}(\nabla u, \nabla u), \qquad (I.23)$$

and the inequality

$$(\Delta u)^2 \le n \left| \nabla^2(u) \right|^2 \,, \tag{I.24}$$

which can be easily obtained by using Cauchy-Schwarz inequality. Moreover, again by Cauchy-Schwarz inequality we have that the equality sign holds if and only if

$$\nabla^2(u) = \frac{\Delta u}{n}g. \tag{I.25}$$

We mention that (I.23), (I.24) and (I.25) hold in every n dimensional Riemannian manifolds (M, g) and for every  $C^2$ -map  $u : (M, g) \to \mathbb{R}$ .

Since P is subharmonic, the strong maximum principle yields that either (I.5) or (I.6) holds. By using a suitable generalization of the Pohozaev identity (2.58) we can exclude (I.6). Hence P is constant and harmonic, and from (I.25) the conclusion follows.

**Symmetry results in convex cones.** The second result that we present in Chapter 2 are two symmetry results for domains in convex cones. The starting

observation that motivates our study is the following. Let  $\Sigma$  be an open cone in  $\mathbb{R}^n$  with vertex at the origin O, i.e.

$$\Sigma := \{ tx : x \in \omega, t \in [0, +\infty) \},$$
 (I.26)

for some open domain  $\omega \subseteq \mathbb{S}^{n-1}$ . We notice that if the point  $x_0$  in (I.11) is chosen appropriately then u given by (I.11) is still the solution to the following problem:

$$\begin{cases} \Delta u = -1 & \text{in } B_R(x_0) \cap \Sigma \\ u = 0 \text{ and } \partial_{\nu} u = c & \text{on } \partial B_R(x_0) \setminus \overline{\Sigma} \\ \partial_{\nu} u = 0 & \text{on } B_R(x_0) \cap \partial \Sigma \end{cases}$$
(I.27)

More precisely,  $x_0$  may coincide with O or it may be just a point of  $\partial \Sigma \setminus \{O\}$ and, in this case,  $B_R(x_0) \cap \Sigma$  is half a sphere lying over a at portion of  $\partial \Omega$ . Hence, it is natural to look for a characterization of symmetry in this direction, as done in [180]. In order to properly describe the results, we introduce some notation. Given an open cone  $\Sigma$  such that  $\partial \Sigma \setminus \{O\}$  is smooth, we consider a bounded domain  $\Omega \subset \Sigma$  and denote by  $\Gamma_0$  its relative boundary, i.e.

$$\Gamma_0 = \partial \Omega \cap \Sigma \,,$$

and we set

$$\Gamma_1 = \partial \Omega \setminus \overline{\Gamma}_0 \,.$$

We assume that  $\mathcal{H}_{n-1}(\Gamma_1) > 0$ ,  $\mathcal{H}_{n-1}(\Gamma_0) > 0$  and that  $\Gamma_0$  is a smooth (n-1)-dimensional manifold, while  $\partial \Gamma_0 = \partial \Gamma_1 \subset \partial \Omega \setminus \{O\}$  is a smooth (n-2)-dimensional manifold. Following [180], such a domain  $\Omega$  is called a *sector-like domain*. In the following, we shall write  $\nu = \nu_x$  to denote the exterior unit normal to  $\partial \Omega$  wherever is defined (that is for  $x \in \Gamma_0 \cup \Gamma_1 \setminus \{O\}$ ). Under the assumption that  $\Sigma$  is a convex cone, in [180] it is proved that if  $\Omega$  is a sector-like domain and there exists a classical solution  $u \in C^2(\Omega) \cap C^1(\Omega \cup \Gamma_0 \cup \Gamma_1 \setminus \{O\})$  to

$$\begin{cases} \Delta u = -1 & \text{in } \Omega, \\ u = 0 \text{ and } \partial_{\nu} u = c & \text{on } \Gamma_0, \\ \partial_{\nu} u = 0 & \text{on } \Gamma_1 \setminus \{O\}, \end{cases}$$
(I.28)

and such that

$$u \in W^{1,\infty}(\Omega) \cap W^{2,2}(\Omega)$$

then

 $\Omega = B_R(x_0) \cap \Sigma$ 

for some  $x_0 \in \mathbb{R}^n$  and u is given by (I.11). Differently from the original paper of Serrin [204], the method of moving planes is not helpful (at least when applied in a standard way) and the rigidity result in [180] is proved by using two alternative approaches. One is based on integral identities and it is inspired from [40], the other one uses a *P*-function approach as in [215].

In [63], we generalize the rigidity result for Serrin's problem in [180] in two directions. The former is by considering more general operators than the Laplacian in the Euclidean space, where the operators may be of degenerate type. Here, the generalization is not trivial due to the lack of regularity of the solution (the operator may be degenerate) as well as to other technical details which are not present in the linear case. The operator that we are going to consider is the following:

$$L_f u = \operatorname{div} \left( \frac{f'(|\nabla u|)}{|\nabla u|} \nabla u \right) , \qquad (I.29)$$

where f is a positive, convex and super-linear function (we mention that overdetermined problems associated to the operator  $L_f$  has been considered in [92, 103, 106]).

The latter is by considering an analogous problem in space forms, i.e. the hyperbolic space and the (hemi)sphere. The operator that we consider is linear and it is interesting since it has been shown that it is a helpful generalization of the torsion problem to space forms (see e.g. [68, 189, 190]).

The first problem that we are going to consider is the following mixed boundary value problem:

$$\begin{cases} L_f u = -1 & \text{in } \Omega, \\ u = 0 & \text{on } \Gamma_0 \\ \partial_\nu u = 0 & \text{on } \Gamma_1 \setminus \{O\}, \end{cases}$$
(I.30)

where  $L_f$  is given by (I.29),  $\Omega$  is a sector like domain in  $\mathbb{R}^n$  and  $f: [0, +\infty) \to [0, +\infty)$  is such that

$$f \in C^{1}([0,\infty)) \cap C^{3}((0,\infty)) \text{ with } f(0) = f'(0) = 0, \ f''(s) > 0 \text{ for } s > 0$$
  
and 
$$\lim_{s \to +\infty} \frac{f(s)}{s} = +\infty.$$
 (L31)

The key observation that motivates our study is the following: we notice that the solution to  $L_f u = -1$  in  $B_R(x_0)$  (a ball of radius R centered at  $x_0$ ) such that u = 0 on  $\partial B_R(x_0)$  is radial and it is given by

$$u(x) = \int_{|x-x_0|}^{R} g'\left(\frac{s}{n}\right) \, ds \,, \tag{I.32}$$

where g denotes the Fenchel conjugate of f (see for instance [73] or [103]), i.e.

$$g = \sup\{st - f(s) : s \ge 0\}$$

(hence for us g' is the inverse function of f').

With these prelimitries, the second rigidity result in Chapter 2 is the following

**Theorem I.D:** let f satisfy (I.31). Let  $\Sigma$  be a convex cone such that  $\Sigma \setminus \{O\}$ is smooth and let  $\Omega$  be a sector-like domain in  $\Sigma$ . If there exists a solution  $u \in C^1(\Omega \cup \Gamma_0 \cup \Gamma_1 \setminus \{O\}) \cap W^{1,\infty}(\Omega)$  to (I.30) such that

$$\partial_{\nu} u = -c \quad on \quad \Gamma_0 \tag{I.33}$$

for some constant c, and satisfying

$$\frac{f'(|\nabla u|)}{|\nabla u|} \nabla u \in W^{1,2}(\Omega, \mathbb{R}^n), \qquad (I.34)$$

then there exists  $x_0 \in \mathbb{R}^n$  such that  $\Omega = \Sigma \cap B_R(x_0)$  with  $c = g'(|\Omega|/|\Gamma_0|)$ ,  $R = n|\Omega|/|\Gamma_0|$ . Moreover u is given by (I.32), where  $x_0$  is the origin or, if  $\partial \Sigma$ contains flat regions, it is a point on  $\partial \Sigma$ . We refer to Section 2.2 for a detailed discussion on the hypothesis of the Theorem. The proof of Theorem I.D s based on the integral approach introduced in [40] (see also [180] and [29]). The idea is the following: we set  $V(\xi) = f(|\xi|)$  and we define the matrix  $W(x) = (w_{ij}(x))_{ij}$  such that

$$w_{ij}(x) = \partial_j V_{\xi_i}(\nabla u(x)) \,.$$

The matrix W is such that  $L_f u = \text{Tr}(W)$ ; writing the operator as a trace has the advantage that we can use Newton's inequality (I.13). By using a Pohozaevtype identity, integral inequalities and the convexity of the cone we are able to prove that the equality holds in (I.13) and this implies the rigidity result.

The second result in Section 2.2 deals with an overdetermined problem in space forms: recall that a space form is a complete simply-connected Riemannian manifold (M, g) with constant sectional curvature K. Up to homotheties we may assume K = 0, 1, -1: the case K = 0 corresponds to the Euclidean space  $\mathbb{R}^n$ , K = -1 is the hyperbolic space  $\mathbb{H}^n$  and K = 1 is the unitary sphere with the round metric  $\mathbb{S}^n$ . More precisely, in the case K = 1 we consider the hemisphere  $\mathbb{S}^n_+$ . These three models can be described as warped product spaces  $M = I \times \mathbb{S}^{n-1}$  equipped with the rotationally symmetric metric

$$g = dr^2 + h(r)^2 g_{\mathbb{S}^{n-1}},$$

where  $g_{\mathbb{S}^{n-1}}$  is the round metric on the (n-1)-dimensional sphere  $\mathbb{S}^{n-1}$  and

- h(r) = r in the Euclidean case (K = 0), with  $I = [0, \infty)$ ;
- $h(r) = \sinh(r)$  in the hyperbolic case (K = -1), with  $I = [0, \infty)$ ;
- $h(r) = \sin(r)$  in the spherical case (K = 1), with  $I = [0, \pi/2)$  for  $\mathbb{S}^n_+$ .

By using the warping structure of the manifold, we denote by O the pole of the model and it is natural to define a *cone*  $\Sigma$  *with vertex at*  $\{O\}$  as the set

$$\Sigma = \{ tx \, : \, x \in \omega, \, t \in I \}$$

for some open domain  $\omega \subset \mathbb{S}^{n-1}$ . Moreover, we say that  $\Sigma$  is a *convex cone* if the second fundamental form II is nonnegative defined at every  $p \in \partial \Sigma$ .

The second problem that we are going to consider is the following mixed boundary value problem

$$\begin{cases} \Delta u + nKu = -1 & \text{in } \Omega, \\ u = 0 & \text{on } \Gamma_0 \\ \partial_{\nu} u = 0 & \text{on } \Gamma_1 \setminus \{O\}, \end{cases}$$
(I.35)

where  $\Omega$  is a sector-like domain in a smooth and convex cone  $\Sigma$  in the Euclidean space (K = 0), in the hyperbolic space (K = -1) or in the hemisphere (k = 1).

With these preliminaries, the third rididifity result in Chapter 2 is the following

**Theorem I.E:** let (M, g) be the Euclidean space, hyperbolic space or the hemisphere. Let  $\Sigma \subset M$  be a convex cone such that  $\Sigma \setminus \{O\}$  is smooth and let  $\Omega$  be a sector-like domain in  $\Sigma$ . Assume that there exists a solution  $u \in C^1(\Omega \cup \Gamma_0 \cup \Gamma_1 \setminus \{O\}) \cap W^{1,\infty}(\Omega) \cap W^{2,2}(\Omega)$  to (I.35) such that

$$\partial_{\nu} u = -c \quad on \quad \Gamma_0 \tag{I.36}$$

for some constant c. Then  $\Omega = \Sigma \cap B_R(x_0)$  where  $B_R(x_0)$  is a geodesic ball of radius R centered at  $x_0$  and u is given by

$$u(x) = \frac{H(R) - H(d(x, x_0))}{n\dot{h}(R)},$$

with

$$H(r) = \int_0^r h(s) ds$$

and where  $d(x, x_0)$  denotes the distance between x and  $x_0$ .

The strategy to prove Theorem I.E follows the Weinberger approach and uses the following P-function, introduced in [68] and in [190],

$$P(x) = |\nabla u(x)|^2 + \frac{2}{n}u(x) + Ku^2(x).$$

One can show that, if u solves (I.35)-(I.36) then P satisfies:

$$\begin{cases} \Delta P(x) \ge 0 & \text{in } \Omega, \\ P(x) = c^2 & \text{on } \Gamma_0 \\ \partial_{\nu} P(x) \le 0 & \text{on } \Gamma_1 \setminus \{O\}. \end{cases}$$

This implies that

$$P(x) \le c^2 \quad \text{in } \Omega.$$

We can exclude the case  $P(x) < c^2$  in  $\Omega$ , so

$$P(x) \equiv c^2 \quad \text{in } \Omega.$$

In particular,  $\Delta P(x) = 0$  in  $\Omega$  and this implies that

$$\nabla^2(u) = \left(-\frac{1}{n} - Ku\right)g$$
 in  $\Omega$ ,

and this equation implies the rigidity result.

#### Chapter 1

# Quantitative stability for almost constant mean curvature hypersurfaces

The main result of this chapter is the following sharp stability estimate of proximity to a single sphere for Alexandrov's theorem.

**Theorem 1.1.** Let S be a  $C^2$ -regular, connected, closed hypersurface embedded in  $\mathbb{M}^n_+$  satisfying a uniform touching ball condition of radius  $\rho$ . Let  $H_S$  given by (I) or (II). There exist constants  $\varepsilon$ , C > 0 such that if

$$\operatorname{osc}(\mathsf{H}_S) \le \varepsilon,$$
 (1.1)

then there are two concentric balls  $B^d_r$  and  $B^d_R$  of  $\mathbb{M}^n_+$  such that

$$S \subset \overline{B}_R^d \setminus B_r^d, \tag{1.2}$$

and

$$R - r \le C \operatorname{osc}(\mathsf{H}_S). \tag{1.3}$$

The constants  $\varepsilon$  and C depend only on n and upper bounds on  $\rho^{-1}$  and on the area of S.

For the sake of clarity we mention that the balls in the statement of Theorem 1.1 are distance or geodesic balls in  $\mathbb{M}^n_+$ , i.e. if we denote with d the distance induced by the metric g of the space form then  $B^d_r(p)$  denotes the ball centred at p of radius r with respect to the distance d. The center of the two balls in the statement will be specified later.

We recall that the uniform touching ball condition of radius  $\rho$  in Theorem 1.1 means that at any point of S there exist two balls of radius  $\rho$  both tangent to S one from inside and one from outside. As mentioned in the introduction the assumption that  $\rho$  is fixed implies that a bubbling phenomenon can not appear (see for instance [58]). As can be shown by a calculation for ellipsoids, the estimate in (1.3) is optimal and it is new in the general setting of Theorem 1.1. In the Euclidean space, quantitative results, in the spirit of Theorem 1.1, for constant mean curvature hypersurfaces are avalaible in literature only under the assumption that S bounds a convex domain. In particular, in [147, 151, 173] the case when the domain is an ovaloid is considered. Moreover in [144], where the results in [8, 200] are improved, an explicit Holder type stability estimate like (1.3) is proved. We emphasize that, in Theorem 1.1 we do not consider any convexity assumption and, as already mentioned, the estimate that we prove is optimal; hence (1.3) remarkably improves the previous stability results.

Theorem 1.1 follows the research line initiated in [67] and pursued in [69] and is contained in [65].

Theorem (1.1) has a quite interesting consequence which is a pinching result. It is well-known that, in the Euclidean space (see for instance [121]), if every principal curvature  $k_i$  of S is pinched between two positive numbers, i.e.

$$\frac{1}{r} \le k_i \le \frac{1+\delta}{r}, \quad \text{for } i = 1, \dots, n-1,$$

then S is close to a sphere of radius r. One can ask if this happens when only the mean curvature is pinched or, more in general if one consider the class of curvatures  $H_S$  defined in the introduction. As a consequence of Theorem 1.1 we have the following

**Corollary 1.1.** Let  $\rho_0$ ,  $A_0 > 0$  and  $n \in \mathbb{N}$  be fixed. There exists  $\varepsilon > 0$ , depending on n,  $\rho_0$  and  $A_0$ , such that if S is a connected closed  $C^2$  hypersurface embedded in  $\mathbb{M}^n_+$  having area bounded by  $A_0$ , satisfying a touching ball condition of radius  $\rho \geq \rho_0$ , and such that

$$\operatorname{osc}(\mathsf{H}_S) \le \varepsilon$$
, (1.4)

then S is C<sup>1</sup>-close to a sphere and there exists a C<sup>2</sup>-regular map  $\Psi:\partial B^d_r\to\mathbb{R}$  such that

$$F(p) = \exp_x(\Psi(p)N_p)$$

defines a  $C^2$ -diffeomorphism from  $B_r^d$  to S and

$$\|\Psi\|_{C^1(\partial B^d_x)} \le C \operatorname{osc}(\mathsf{H}_S)^{1/2},$$
(1.5)

where N is a normal vector field to  $\partial B_r^d$ .

Before explaining the proof of Theorem 1.1 and Corollary 1.1, we give a couple of remarks on the bounds on  $\rho$  and on the Area of S in Theorem 1.1 and Corollary 1.1. The upper bound on the Area of S is a control on the constants  $\varepsilon$  and C, which clearly change under dilatations. The upper bound on  $\rho^{-1}$  controls the  $C^2$ -regularity of the hypersurface, which is crucial for obtaining an estimate like (1.3). Indeed, if we assume that  $\rho$  is not bounded from below in the Euclidean case, it is possible to construct a family of closed surfaces embedded in  $\mathbb{R}^3$ , not diffeomorphic to a sphere, with  $\operatorname{osc}(H)$  arbitrarily small and such that (1.3) fails to hold (see Figure 1.1).

As already mentioned in the introduction, we tackle the problem by doing a quantitative study of the method of the moving planes, i.e. we study the original proof of Alexandrov from a quantitative point of view. There are other possible approaches for proving the symmetry result (i.e. of Alexandrov's Theorem) which are based on integral and geometric identities. The interested reader can refer to [172, 194, 197, 198], and to [41] for a recent generalization (see also [82] for minimal assumptions on the regularity of S). The approach in [194] has



Figure 1.1: counterexample obtained by gluing pieces of suitable small perturbations of unduloids.

been exploited in [58] to tackle the problem of bubbling (see also [83] and [148]). Symmetry and quantitative studies of proximity to a single sphere by using an integral approach can be found in [30, 45, 56, 96, 127, 128, 164, 165, 166, 189]. Coming back to our approach, it is based on a quantitative analysis of the method of moving planes and uses arguments from elliptic PDEs theory. Since, as explained in the Introduction, the proof of symmetry (i.e. of Alexandrov's Theorem) is based on the maximum principle, our proof of the stability result will make use of Harnack's and Carleson's (or boundary Harnack) inequalities and Hopf Lemma, which can be considered as the quantitative counterpart of the strong and boundary maximum principles.

For the sake of completeness we have to say that a quatitative study of the method of moving planes was first performed in [2], where the authors obtained a stability result for Serrin theorem (see the introduction), and it has been used in the following series of papers [60, 61, 62] to study the stability of radial symmetry for Serrin's and other overdetermined problems (see also the survey [64]).

In order to prove Theorem 1.1 we follow the approach of [2], but in our case the setting is complicated by the fact that we have to deal with Riemannian manifolds. As we will show in Section 1.4 the main task is to prove the approximate symmetry result for one direction. To prove this result we apply the method of moving planes and show that the union of the maximal cap and of its reflection provides a set that fits S well. This is done in Theorem 1.4; by the method of moving planes we know that the hypersurface and the reflected cap are tangent (internally or at the boundary) at some point  $p_0$ . We write the two hypersurfaces as graphs of two functions in neighbourhood of  $p_0$ . We set w the difference of these two functions and we have that it satisfies an elliptic equation

Lw = f,

where  $||f||_{\infty}$  is bounded by  $\operatorname{osc}(\mathsf{H}_S)$ . Then via Harnack's inequality and interior regularity estimates we are able to prove a bound on the  $C^1$  norm of w, which implies that the two surfaces are close in the following sense: not only the distance between points is controlled by a constant times  $\operatorname{osc}(\mathsf{H}_S)$ , but also the

two corresponding Gauss map are close (always by a constant times  $osc(H_S)$ ) in a neighbourhood of  $p_0$ . To propagate this estimate we connect any point of the reflected cap with  $p_0$  via a chain of balls and by using careful estimates and interior or boundary Harnack inequality we propagate the estimate (assuming that  $osc(H_S)$  is small).

The chapter is organized as follows. In Section 1.1 we review the method of the moving planes in space forms and set up some notation. In Section 1.2 we give technical local quantitative estimates in space forms. In Section 1.3 we find estimates on curvatures of projected surfaces in conformally Euclidean spaces. In Section 1.4 we prove the approximate symmetry in one direction. In Section 1.5 we show how to prove global approximate symmetry result by using the approximate symmetry in any direction. Finally, in Section 1.6 we complete the proof of main results.

We will use the following models of space forms:

•  $\mathbb{H}^n$  is the half-space  $\{x \in \mathbb{R}^n : x_n > 0\}$  with the Riemannian metric

$$g_x = \frac{1}{x_n} \langle \cdot, \cdot \rangle$$
, for every  $x \in \mathbb{R}^n$  (1.6)

where  $\langle \cdot, \cdot \rangle$  is the Euclidean product on  $\mathbb{R}^n$ ; for simplicity of notation, we will indicate it also with " $\cdot$ ";

•  $\mathbb{S}^n$  is the *n*-dimensional unitary sphere  $\{x \in \mathbb{R}^{n+1} : |x| = 1\}$  with the round metric *g* induced by the Euclidean metric in  $\mathbb{R}^{n+1}$ . Here we recall that if we consider the stereographic projection  $\mathbb{S}^n \setminus \{\text{one point}\} \to \mathbb{R}^n$ , then *g* is projected to the Riemann metric

$$g_x = \frac{4}{(1+|x|^2)^2} \langle \cdot, \cdot \rangle, \text{ for every } x \in \mathbb{R}^n.$$
(1.7)

We recall that  $\mathbb{M}^n$  denotes the space forms, i.e. the Euclidean space  $\mathbb{R}^n$ , the hyperbolic space  $\mathbb{H}^n$  and the sphere  $\mathbb{S}^n$ ; while  $\mathbb{M}^n_+$  denotes the Euclidean space  $\mathbb{R}^n$ , the hyperbolic space  $\mathbb{H}^n$  and the hemisphere  $\mathbb{S}^n_+$ . Moreover, in all the chapter S denotes a  $C^2$ -regular, connected, closed hypersurface embedded in  $\mathbb{M}^n_+$  and  $\Omega \subset \mathbb{M}^n_+$  denotes a relatively compact connected open set such that  $\partial\Omega = S$  (such an  $\Omega$  exists since S is an embedded hypersurface).

#### 1.1 The method of the moving planes

In this preliminary section we recall the method of the moving planes in  $\mathbb{M}^n_+$ .

We begin by recalling the definition of *center of mass* in the context of Riemannian geometry (see e.g. [142] for more details).

Let (M,g) be an oriented complete Riemannian manifold and let  $\Omega$  be a bounded domain (i.e. a bounded connected open set). Let  $P_{\Omega} \colon M \to \mathbb{R}$  be the function

$$P_{\Omega}(x) = \frac{1}{2 |\Omega|_g} \int_{\Omega} d(x, a)^2 \, da \,,$$

where  $|\Omega|_g$  is the volume of  $\Omega$  with respect to g. Then the gradient of  $P_{\Omega}$  takes the following expression

$$\nabla P_{\Omega}(x) = -\frac{1}{|\Omega|_g} \int_{\Omega} \exp_x^{-1}(a) \, da \,, \tag{1.8}$$

where  $\exp_x : T_x M \to M$  denotes the exponential map of the Riemannian manifold (M, g) at  $x \in M$   $(T_x M$  denotes the tangent space to the manifold M at  $x \in M$ ) and  $\exp_x^{-1}$  denotes its inverse. In some cases  $P_{\Omega}$  is a convex map and it attains the minimum at only one point  $\mathcal{O}$ , which is usually called the *center* of mass of  $\Omega$ . For instance this occurs in the following cases:

- all the sectional curvatures of M are non-positive;
- $\Omega$  is contained in a geodesic ball of radius  $r < \frac{1}{2} \min\left\{ \inf M, \frac{\pi}{2\sqrt{K}} \right\}$ , where *K* is an upper bound on the sectional curvatures of *M* and  $\inf M$  is the injectivity radius of *M*;
- $M = \mathbb{S}^n$  and  $\Omega$  is contained in  $\mathbb{S}^n_+$ .

We further recall that a Riemannian manifold (M, g) is a symmetric space if for every  $p \in M$  there exists an isometry  $f: M \to M$  such that f(p) = p and  $f_{*|p} = -\text{Id}$ .

**Lemma 1.1.** Let (M,g) be a symmetric space,  $\Omega$  a bounded domain in M and  $x \in M$  be such that  $\nabla P_{\Omega}(x) = 0$ . Assume that for every hyperplane  $\pi$  in M not containing x there exists a hyperplane  $\pi_1$  passing through x and such that  $\pi \cap \pi_1 \cap \Omega = \emptyset$ . Then every hyperplane of symmetry for  $\Omega$  contains x.

*Proof.* Assume by contradiction that there exists a hyperplane  $\pi$  of symmetry for  $\Omega$  not containing x. Let  $\pi_1$  be a hyperplane passing through x and disjoint from  $\pi$  inside  $\Omega$ , i.e.  $\pi \cap \pi_1 \cap \Omega = \emptyset$ . Since  $\pi_1$  and  $\pi$  are disjoint, they subdivide  $\Omega$  in three disjoint subsets  $\Omega_1$ ,  $\Omega_2$  and  $\Omega_3$ , with  $|\Omega_i|_g > 0$ , i = 1, 2, 3. Since  $\Omega$  is symmetric about  $\pi$ , we have that

$$|\Omega_1|_g + |\Omega_2|_g = |\Omega_3|_g.$$

Moreover formula (1.8) implies

$$\int_{\Omega_1} \exp_x^{-1}(a) \, da = -\int_{\Omega_2 \cup \Omega_3} \exp_x^{-1}(a) \, da \,. \tag{1.9}$$

Let  $f: M \to M$  be the isometry such that f(x) = x and  $f_{|*x} = -\text{Id}$ . Then

$$\exp_x^{-1} f(a) = -\exp_x^{-1}(a)$$

and

$$-\int_{\Omega_2\cup\Omega_3} \exp_x^{-1}(a) \, da = \int_{f(\Omega_2\cup\Omega_3)} \exp_x^{-1}(a) \, da \, .$$

Therefore (1.9) implies

$$|\Omega_1|_g = |\Omega_2|_g + |\Omega_3|_g.$$

which gives a contradiction since  $|\Omega_2|_q > 0$ .

Lemma 1.1 can be in particular applied in space forms  $\mathbb{M}^n_+$ , where we have the uniqueness of the center of mass, to prove the following result

**Proposition 1.1** (Characterization of the distance balls in  $\mathbb{M}_{+}^{n}$ ). Let  $S = \partial \Omega$ be a  $C^{2}$ -regular, connected, closed hypersurface embedded in  $\mathbb{M}_{+}^{n}$ , where  $\Omega$  is a relatively compact domain. Assume that for every geodesic path  $\gamma : \mathbb{R} \to \mathbb{M}^{n}$ there exists a hyperplane  $\pi$  orthogonal to  $\gamma$  such that S is symmetric about  $\pi$ . Then S is a distance sphere about  $\mathcal{O}$ .

The previous result is a generalization of the following characterization theorem, due to H. Hopf, of the sphere in the Euclidean space (see also [136, Chapter VII, Lemma 2.2])

**Theorem 1.2.** Let  $S = \partial \Omega$  be a  $C^2$ -regular, connected, closed hypersurface embedded in  $\mathbb{R}^n$ , where  $\Omega$  is a relatively compact domain. If for every direction  $\omega \in \mathbb{R}^n$  there exists a hyperplane of symmetry of  $\Omega$  orthogonal to  $\omega$ , then S is a round sphere.

Proof of Proposition 1.1. In view of Lemma 1.1 any hyperplane of symmetry of S contains the point  $\mathcal{O}$ . Therefore S is invariant by reflections about every hyperplane passing through  $\mathcal{O}$ . For all the three possible cases of space forms we can choose a model where  $\mathbb{M}^n_+$  is a ball in  $\mathbb{R}^n$ ,  $\mathcal{O}$  is the origin of  $\mathbb{R}^n$  and the hyperplanes passing through  $\mathcal{O}$  are Euclidean hyperplanes. In such a model S is invariant by reflections about every Euclidean hyperplane passing through the origin and hence it is an Euclidean ball, since every rotation in the Euclidean space can be obtained as the composition of two reflections about hyperplanes. Hence S is a round sphere in  $\mathbb{M}^n_+$  as required.  $\Box$ 

We describe the method of the moving planes in  $\mathbb{M}^n_+$  and we introduce some notation. The method consists in moving hyperplanes along a geodesic path orthogonal to a fixed direction and it is similar in  $\mathbb{R}^n$ ,  $\mathbb{H}^n$  and  $\mathbb{S}^n_+$ . The method can be described in terms of a point **o** that we fix. Since  $\mathbb{M}^n_+$  is a homogeneous space, the construction does not depend on the choice of the point we fix. In particular we choose **o** to be:

- the origin in  $\mathbb{R}^n$ ;
- $e_n$  in  $\mathbb{H}^n$ ;
- the north pole in  $\mathbb{S}^n_+$ .

For every direction  $v \in T_{o}\mathbb{M}^{n}_{+}$ , we consider the geodesic path  $\gamma_{v}: I \to \mathbb{M}^{n}_{+}$ satisfying  $\gamma_{v}(0) = \mathbf{o}$  and  $\dot{\gamma}_{v}(0) = v$ . The domain of  $\gamma_{v}$  is  $I = \mathbb{R}$  in  $\mathbb{R}^{n}$  and  $\mathbb{H}^{n}$ and  $I = (-\frac{\pi}{2}, \frac{\pi}{2})$  in  $\mathbb{S}^{n}_{+}$ . For any  $s \in I$  we denote by  $\pi_{v,s}$  the hyperplane passing through  $\gamma_{v}(s)$  and orthogonal to  $\dot{\gamma}_{v}(s)$ , and we define

 $S_{v,s} = \{ p \in S : p \in \pi_{v,t} \text{ for some } t > s \}.$ 

We denote by  $S_{v,s}^{\pi}$  be the reflection of  $S_{v,s}$  about  $\pi_{v,s}$ . Note that

- $S_{v,s} = \{ p \in S : p \cdot v > s \}$ , if  $\mathbb{M}^n_+ = \mathbb{R}^n$ ;
- $S_{v,s} = \{ p \in S : p \cdot \dot{\gamma}_v(s) > 0 \}, \text{ if } \mathbb{M}^n_+ = \mathbb{S}^n_+.$

In the hyperbolic case, giving an explicit description of  $S_{v,s}$  is more complicated, but it can be simplified by assuming  $v = e_1$ . This assumption is not restrictive since we can always rotate every direction v in  $e_1$  by using an isometry of  $\mathbb{H}^n$ . In this case we have

• 
$$S_{e_1,s} = \{ p \in S : p \cdot e_1 > s \}, \text{ if } \mathbb{M}^n_+ = \mathbb{H}^n.$$

Let

$$m_v = \inf\{s \in I : S_{v,t}^{\pi} \subset \Omega \text{ for every } t > s\}.$$

The hyperplane  $\pi_v := \pi_{v,m_v}$  is called the *critical hyperplane*. By construction  $S_{v,m_v}^{\pi}$  is contained in  $\overline{\Omega}$  and S and  $S_{v,m_v}^{\pi}$  are tangent at some point  $p_0$  which can be either interior to  $S_{v,m_v}^{\pi}$ , or on  $\partial S_{v,m_v}^{\pi}$  (and in this last case  $p_0 \in \pi_v$ ).

Now for the sake of completeness we recall the generalized version of Alexandrov's theorem (Alexandrov theorem II in the introduction) that we study in this chapter and its proof in  $\mathbb{M}^n_+$  (see [5, 145, 172, 194, 197, 198]).

**Theorem 1.3.** The only closed  $C^2$ -regular connected hypersurfaces embedded in  $\mathbb{M}^n_{\pm}$  and such that  $H_S$  is constant are the distance spheres.

*Proof.* The proof consists in showing that for every unitary vector  $v \in T_{o}\mathbb{M}_{+}^{n}$ , S is symmetric about  $\pi_{v}$ . This is obtained by showing that  $S \cap S_{v,m_{v}}^{\pi}$  is open and closed in  $S_{v,m_{v}}^{\pi}$ . Note that  $S \cap S_{v,m_{v}}^{\pi}$  is not empty since  $S_{v,m_{v}}^{\pi}$  is tangent to S at some point and  $S \cap S_{v,m_{v}}^{\pi}$  is closed in  $S_{v,m_{v}}^{\pi}$ . The only nontrivial step is that  $S \cap S_{v,m_{v}}^{\pi}$  is open, which is obtained by using maximum principles for solutions to elliptic equations.

Let  $p_0 \in S \cap S^{\pi}_{v,m_0}$ . By construction we have that

$$T_{p_0}S = T_{p_0}S^{\pi}$$

where  $S^{\pi}$  is the reflection of S about  $\pi_v$ . From the implicit function theorem, S and  $S^{\pi}$  are locally the Euclidean graph of  $C^2$ -regular functions u and  $\hat{u}$ , respectively, defined in a (Euclidean) ball  $B_r$  of radius r centered at the origin O in  $T_{p_0}S$ . The functions u and  $\hat{u}$  satisfy the elliptic equation  $Lu(x) = \mathsf{H}_S(x, u(x))$  for  $x \in B_r$ ; here the operator L is the mean curvature operator or, more generally, a fully nonlinear operator. The ellipticity of L is standard in the case of the mean curvature operator and it follows from [145] for the other cases considered.

Since  $\mathsf{H}_S$  is constant, the difference  $u - \hat{u}$  satisfies an elliptic equation of the form  $\mathcal{L}(u - \hat{u}) = 0$  with  $u(O) - \hat{u}(O) = 0$ .

If  $p_0$  is interior to  $S_{v,m_v}^{\pi}$  then we can choose r sufficiently small such that  $u - \hat{u} \ge 0$  in  $B_r$  and by the strong maximum principle we obtain that  $u - \hat{u} \equiv 0$  in  $B_r$ .

If  $p_0$  is on the boundary of  $S_{v,m_v}^{\pi}$ , then by construction  $u - \hat{u} \ge 0$  in a half ball  $B_r^+$ ,  $u(O) - \hat{u}(O) = 0$  and  $\nabla u(O) = \nabla u(O) = 0$ . By applying Hopf's boundary point lemma at the point O we obtain that  $u - \hat{u} \equiv 0$  in  $B_r^+$ .

Hence, we have proved that  $S \cap S^{\pi}_{v,m_v}$  is open, and the conclusion follows.  $\Box$ 

#### **1.2** Local quantitative estimates in space forms

In this section we prove some preliminary estimates which we will use in the proof of Theorem 1.1. We have the following preliminary lemma about the local equivalences of distances (we will denote the Euclidean norm with  $|\cdot|$ ).

**Lemma 1.2.** • Let d be the distance induced by the hyperbolic metric (1.6) in  $\mathbb{H}^n$  and let q be such that  $d(q, e_n) \leq R$ ; then

$$c|q - e_n| \le d(q, e_n) \le C|q - e_n|,$$
 (1.10)

for some positive constants c and C depending only on R.

• Let d be the distance induced by the round metric (1.7) in  $\mathbb{R}^n$  and let p, q in  $\mathbb{R}^n$  be such that  $|p|, |q| \leq R$ . Then

$$\frac{2}{1+R^2}|p-q| \le d(p,q) \le \pi |p-q|.$$
(1.11)

*Proof.* • We recall that the hyperbolic distance induced by the hyperbolic metric (1.6) is given, in terms of the Euclidean distance, by the following expression

$$d(p,q) = \operatorname{arccosh}\left(1 + \frac{|p-q|^2}{2p_n q_n}\right)$$
.

In particular

$$d(e_n, te_n) = |\log(t)|, \text{ for any } t \in (0, \infty).$$

This expression and the fact that  $d(q, e_n) \leq R$  imply that

$$e^{-R} \le q_n \le e^R \,. \tag{1.12}$$

For simplicity, we set

$$t=1+\frac{|q-e_n|^2}{2q_n}$$

hence, from (1.12),

$$1 + \frac{e^{-R}}{2}|q - e_n|^2 \le t \le 1 + \frac{e^R}{2}|q - e_n|^2, \qquad (1.13)$$

and, since  $|q - e_n| \le R(e^R - 1)$ , then

$$t \le A \,, \tag{1.14}$$

where A is a constant which depends only on R. Now we set  $\phi(t) = \operatorname{arccosh}(t)$ , for  $t \in [1, +\infty)$ . Since, from (1.14),  $1 \le t \le A$  we have

$$\frac{1}{\sqrt{A+1}}\frac{1}{\sqrt{t-1}} \leq \phi'(t) \leq \frac{1}{\sqrt{t-1}}$$

where we used the fact that  $\phi'(t) = (t^2 - 1)^{-1/2}$ , hence

$$\frac{1}{2\sqrt{A+1}}\sqrt{t-1} \le \phi(t) \le \frac{1}{2}\sqrt{t-1} \quad \text{for every } t \in [1, A].$$
(1.15)

The conclusion (1.10) follows combining (1.15) with

$$\frac{e^{-\frac{R}{2}}}{\sqrt{2}}|q-e_n| \le \sqrt{t-1} \le \frac{e^{\frac{R}{2}}}{\sqrt{2}}|q-e_n|$$

which follows from (1.13).
• We recall that the spherical distance induced by the round metric (1.7) is given, in terms of the Euclidean distance, by the following expression

$$d(p,q) = 2 \arcsin\left(\frac{|p-q|}{\sqrt{(1+|p|^2)(1+|q|^2)}}\right)$$

Since  $|p|, |q| \leq R$  and from the previous expression we get

$$2 \arcsin\left(\frac{|p-q|}{1+R^2}\right) \le d(p,q) \le 2 \arcsin(|p-q|)$$

i.e.

$$\frac{|p-q|}{1+R^2} \le \sin\left(\frac{d(p,q)}{2}\right) \le |p-q|\,. \tag{1.16}$$

The conclusion (1.11) follows combining (1.16) with the following trivial property of the sine function:

$$\frac{2t}{\pi} \le \sin(t) \le t \,, \quad \text{ for } 0 \le t \le \frac{\pi}{2};$$

recall that, being on the sphere  $d(p,q) \leq \pi$ .

Let us consider now, as in Alexandrov theorem, a  $C^2$ -regular, connected, closed hypersurface  $S = \partial \Omega$  embedded in  $\mathbb{M}^n_+$ , where  $\Omega$  is a relatively compact domain and denote by N the inward normal vector filed (inward with respect to  $\Omega$ ). For  $p \in S$  we denote by  $\varphi_p \colon \mathbb{M}^n_+ \to \mathbb{R}^n$  the following function whose definition depends on the geometry of  $\mathbb{M}^n$ :

- if  $\mathbb{M}^n$  is  $\mathbb{R}^n$ ,  $\varphi_p \in \mathrm{SO}(n) \rtimes \mathbb{R}^n$  and it is such that  $\varphi_p(p) = 0$  and  $\varphi_{p*|p}(T_pS) = \{x_n = 0\};$
- if  $\mathbb{M}^n$  is  $\mathbb{H}^n$ ,  $\varphi_p$  is an orientation preserving isometry of  $\mathbb{H}^n$  such that  $\varphi_p(p) = e_n$  and  $\varphi_{p*|p}(T_pS) = \{x_n = 0\};$
- if  $\mathbb{M}^n$  is  $\mathbb{S}^n$ ,  $\varphi_p$  is the stereographic projection form the antipodal point to p restricted to  $\mathbb{S}^n_+$  composed with a rotation of  $\mathbb{R}^n$  in order to have  $\varphi_{p*|p}(T_pS) = \{x_n = 0\}.$

Note that in all the three cases we have that  $\varphi_p(S)$  is a hypersurface embedded in  $\mathbb{R}^n$  and

$$\varphi_{p*|p}\left(T_{p}S\right) = \left\{x_{n} = 0\right\}.$$

For r > 0, we denote by  $\mathcal{U}_r(p)$  the open neighbourhood of p in S such that  $\varphi_p(\mathcal{U}_r(p))$  is the (Euclidean) graph of a  $C^2$ -function  $u: B_r \to \mathbb{R}$  defined in the ball of radius r of  $\mathbb{R}^{n-1}$  centred at the origin. Even if we don't have a canonical choice of  $\varphi_p$ , the subsets  $\mathcal{U}_r(p)$  do not depend on the choice of  $\varphi_p$ . Moreover, the implicit function theorem implies that any  $p \in S$  has a neighbourhood  $\mathcal{U}_r(p)$  for r sufficiently small. In order to establish the quantitative estimates we need in the proof of the main Theorem, we have to show that r can be uniformly bounded from below with a bound depending only on  $\rho$ . The following lemma also introduces the quantity  $\rho_1$  which will be largely used in what follows.

**Lemma 1.3.** Let S be a  $C^2$ -regular closed hypersurface embedded in  $\mathbb{M}^n_+$  and satisfying a touching ball condition of radius  $\rho$  and let  $\rho_1$  be defined in the following way:

- $\rho_1 = \rho$ , if  $\mathbb{M}^n = \mathbb{R}^n$ ;
- $\rho_1 = (1 e^{-\rho} \sinh \rho) e^{-\rho} \sinh \rho$ , if  $\mathbb{M}^n = \mathbb{H}^n$ ;

• 
$$\rho_1 = \frac{\rho}{\pi}$$
, if  $\mathbb{M}^n = \mathbb{S}^n$ .

Then

(i) any point  $p \in S$  admits a neighbourhood  $\mathcal{U}_{\rho_1}(p)$  and  $\varphi_p(\mathcal{U}_{\rho_1}(p))$  is the graph of a  $C^2$ -function  $u: B_{\rho_1} \to \mathbb{R}$  satisfying

$$|u(x) - u(O)| \le \rho_1 - \sqrt{\rho_1^2 - |x|^2}, \qquad (1.17)$$

and

$$|\nabla u(x)| \le \frac{|x|}{\sqrt{\rho_1^2 - |x|^2}};$$
(1.18)

where  $B_{\rho_1}$  denotes the Euclidean ball of radius  $\rho_1$  centred at the origin O of  $\mathbb{R}^{n-1}$ .

(ii) There exists a universal constant C such that for any  $0 < \alpha < \frac{1}{2}\min(1, \rho_1^{-1})$ and q in  $\mathcal{U}_{\alpha\rho_1}(p)$  we have

$$d_S(p,q) \le \alpha C \rho_1 \,, \tag{1.19}$$

where  $d_S$  is the geodesic distance on S.

*Proof.* We analyse each case separately and for every case we show that (i) holds true.

•  $\rho_1 = \rho$ , if  $\mathbb{M}^n = \mathbb{R}^n$ . By the implicit function theorem, we have that there exist r > 0,  $u: B_r \to \mathbb{R}$  and  $\mathcal{U}_r(p)$  as in (i), i.e.

$$\varphi_p(\mathcal{U}_r(p)) = \{ p + x + u(x)\nu_p : x \in B_r \}$$

where  $\nu_p$  denotes the (Euclidean) inward unitary normal vector at p to S. Moreover we may assume  $r \leq \rho$ . The bound (1.17) in  $B_r$  follows easily from the definition of the interior and exterior touching balls at p. We now prove the estimate (1.18) in  $B_r$ , which allows us to enlarge the domain of u to  $B_\rho$ . Let  $q \in \varphi_p(\mathcal{U}_r(p))$ , then  $q = p + x + u(x)\nu_p$ , with |x| < r. Since  $B_\rho(p + \rho\nu_p) \cap B_\rho(q - \rho\nu_q) = \emptyset$ , then

$$|p + \rho \nu_p - q + \rho \nu_q| \ge 2\rho.$$

Analogously, since  $B_{\rho}(p - \rho \nu_p) \cap B_{\rho}(q + \rho \nu_q) = \emptyset$ , then

$$|q + \rho\nu_q - p + \rho\nu_p| \ge 2\rho$$

By adding the squares of the last two inequalities we get

$$|p-q|^2 + 2\rho^2 \nu_p \cdot \nu_q \ge 2\rho^2$$
,

and from (1.17) we get

$$\nu_p \cdot \nu_q \ge \frac{1}{\rho} \sqrt{\rho^2 - |x|^2} \quad \text{and} \quad |\nu_p - \nu_q| \le \sqrt{2} \frac{|x|}{\rho}.$$
(1.20)

Moreover, since  $q = p + x + u(x)\nu_p$  then

$$\nu_q = \frac{\nu_p - \nabla u(x)}{\sqrt{1 + |\nabla u(x)|}}$$

and from (1.20) we get (1.18) in  $B_r$ . Since  $|\nabla u|$  is bounded in  $\overline{B}_r$ , we can extend u to a larger ball where (1.18) is still satisfied. It is clear that we can choose  $r = \rho$  and (1.17) and (1.18) hold.

- $\rho_1 = (1 e^{-\rho} \sinh \rho) e^{-\rho} \sinh \rho$ , if  $\mathbb{M}^n = \mathbb{H}^n$ . It is enough to observe that  $\varphi_p(S)$  satisfies an Euclidean touching ball condition of radius  $\rho_1$  (this motivates the choice of  $\rho_1$ ). This can be easily deduced by using Lemma 1.2. Hence the statement follows from the Euclidean case.
- $\rho_1 = \frac{\rho}{\pi}$ , if  $\mathbb{M}^n = \mathbb{S}^n$ . Also in this case it is enough to observe that  $\varphi_p(S)$  satisfies an Euclidean touching ball condition of radius  $\rho_1$  (by using Lemma 1.2). Hence, as before, the statement follows from the Euclidean case.

Now we show that (ii) holds true. Let  $q \in \mathcal{U}_{\rho_1}(p)$ . Then  $\varphi_p(q) = (x, v(x))$  for some  $|x| < \rho_1$ . Let  $\gamma : [0,1] \to \varphi_p(S)$  be the curve joining  $\varphi_p(p)$  to  $\varphi_p(q)$  defined as  $\gamma(t) = (tx, v(tx))$ . Then

$$\dot{\gamma}(t) = (x, \nabla v(tx) \cdot x)$$

and the Cauchy-Schwarz inequality implies

$$|\dot{\gamma}(t)| \le |x|\sqrt{1+|\nabla v(tx)|^2}.$$

Then (1.17) yields

$$|\dot{\gamma}(t)| \leq \frac{\rho_1 |x|}{\sqrt{\rho_1^2 - t^2 |x|^2}} \leq \frac{|x|}{\sqrt{1 - \alpha^2}} \leq \frac{2}{\sqrt{3}} |x|,$$

for  $0 \leq |x| \leq \alpha \rho_1$ . Since

$$d_S(p,q) \le l(\gamma)$$

 $\operatorname{and}$ 

$$\begin{split} l(\gamma) &= \int_0^1 |\dot{\gamma}| \, dt \,, \text{ in the Euclidean case,} \\ l(\gamma) &= \int_0^1 \frac{|\dot{\gamma}|}{v(tx)} \, dt \,, \text{ in the hyperbolic case,} \\ l(\gamma) &= \int_0^1 \frac{2}{1+|\gamma|^2} |\dot{\gamma}| \, dt \,, \text{ in the spherical case,} \end{split}$$

we obtain that

$$d_S(p,q) \le C|x|$$

for a universal constant C and (1.19) follows.

From Lemma 1.3 it follows the following

**Corollary 1.2.** Let S be a compact  $C^2$ -regular embedded hypersurfaces in  $\mathbb{M}^n_+$ satisfying a touching ball condition of radius  $\rho$ . Let  $q \in \mathcal{U}_{\alpha\rho_1}(p)$ , with  $0 < \alpha < \frac{1}{2}\min(1,\rho_1^{-1})$ . Then

$$d_S(p,q) \le Cd(p,q) \,,$$

where C depends only on  $\rho$ . In particular for every  $p \in S$  the geodesic ball  $\mathcal{B}_r(p)$  centred at p and with radius  $r < \frac{1}{2}\min(1, \rho_1^{-1})$  satisfies

$$\operatorname{Area}(\mathcal{B}_r(p)) \ge cr^{n-1}, \qquad (1.21)$$

and c is a constant depending only on n.

#### 1.2.1 Quantitative stability of the parallel transport

In this section we study quantitative estimates involving the parallel transport which will be useful in the proof of the main result. We recall that the parallel transport along a smooth curve  $\alpha : [t_0, t_1] \to \mathbb{M}^n$  is the linear map  $\tau : T_{\gamma(t_0)}\mathbb{M}^n \to T_{\gamma(t_1)}\mathbb{M}^n$  given by

$$\tau(v) = X(t_1)$$

where  $X : [t_0, t_1] \to \mathbb{M}^n$  is the solution to the following linear ODE:

$$\begin{cases} \dot{X}_k + \sum_{i,j=1}^n X_j \dot{\alpha}_j \Gamma_{ij}^k(\alpha) = 0 & k = 1, \dots, n\\ X_k(t_0) = v_k & k = 1, \dots, n \end{cases}$$

where  $\dot{X}(t) = (\dot{X}_1(t), \ldots, \dot{X}_n(t)), \dot{\alpha}(t) = (\dot{\alpha}_1(t), \ldots, \dot{\alpha}_n(t)), v = (v_1, \ldots, v_n)$  and  $\Gamma_{ij}^k$  are the Christoffel symbol in  $\mathbb{M}^n$ . In what follows we adopt the following notation: given  $p, q \in \mathbb{M}^n_+$  with  $p \neq q$ , we denote by

$$\tau_p^q \colon T_p \mathbb{M}^n \to T_q \mathbb{M}^n$$

the parallel transport along the unique geodesic path  $\sigma$  connecting p to q. We further assume that  $\tau_p^p$  is the identity of  $T_p\mathbb{M}^n$ . A direct computation shows that:

- if  $\mathbb{M}^n$  is  $\mathbb{R}^n$ , then  $\tau_p^q$  is the identity for every p, q;
- if  $\mathbb{M}^n = \mathbb{H}^n$  and if p and q belongs to the same vertical line, then we have

$$\tau_p^q(v) = \frac{q_n}{p_n} v;$$

moreover, if p and q do not belong to the same vertical line, then in the following lemma we provide the expression of the parallel transport in this general case (for simplicity we assume  $p = e_n$ ).

**Lemma 1.4.** Let  $q \in \mathbb{H}^n$  be such that  $q \in \text{Span} \{e_{n-1}, e_n\}$  and let  $v \in \mathbb{R}^n$ . Assume that  $q_{n-1} \neq 0$ , then

$$\tau_q^{e_n}(v) = \frac{1}{q_n}(v_1, \dots, v_{n-2}, \tilde{v}_{n-1}, \tilde{v}_n),$$

where

$$\begin{pmatrix} \tilde{v}_{n-1} \\ \tilde{v}_n \end{pmatrix} = \frac{1}{1+a^2} \begin{pmatrix} a(a-q_{n-1})+q_n & a-q_{n-1}-aq_n \\ -(a-q_{n-1}-aq_n) & a(a-q_{n-1})+q_n \end{pmatrix} \begin{pmatrix} v_{n-1} \\ v_n \end{pmatrix}$$

$$a = \frac{|q|^2-1}{2q_{n-1}} .$$

an

$$a = \frac{|q|^2 - 1}{2q_{n-1}}.$$

*Proof.* Let  $\alpha : [t_0, t_1] \to \mathbb{H}^n$  be defined in the following way

$$\alpha(t) = (\sqrt{1 + a^2}\cos(t) + a)e_{n-1} + \sqrt{1 + a^2}\sin(t)e_n,$$

such that  $\alpha(t_0) = q$  and  $\alpha(t_1) = e_n$ . Then, up to reparametrization,  $\alpha$  is a geodesic path connecting q to  $e_n$ . The parallel transport equation along  $\alpha$  yields

$$(\tau_q^{e_n}(v))_k = v_k, \quad k = 1, \dots, n-2,$$

while

$$(\tau_q^{e_n}(v))_{n-1} = X_{n-1}(t_1) \text{ and } (\tau_q^{e_n}(v))_n = X_n(t_1),$$

where the pair  $(X_{n-1}, X_n)$  solves the following system

$$\begin{pmatrix} \dot{X}_{n-1} \\ \dot{X}_n \end{pmatrix} = \begin{pmatrix} \cot an (t) & -1 \\ 1 & \cot an (t) \end{pmatrix} \begin{pmatrix} X_{n-1} \\ X_n \end{pmatrix}$$

and

$$\begin{pmatrix} X_{n-1}(t_0) \\ X_n(t_0) \end{pmatrix} = \begin{pmatrix} v_{n-1} \\ v_n \end{pmatrix} \,.$$

Therefore,

$$\begin{pmatrix} X_{n-1}(t) \\ X_n(t) \end{pmatrix} = A(t)A(t_0)^{-1} \begin{pmatrix} v_{n-1} \\ v_n \end{pmatrix}$$

where

$$A(t) := \begin{pmatrix} \cos(t)\sin(t) & -\sin^2(t) \\ \sin^2(t) & \cos(t)\sin(t) \end{pmatrix};$$

and the claim follows.

• if  $\mathbb{M}^n$  is  $\mathbb{S}^n_+$ , then  $\tau^q_p = (P_q \circ R_\alpha)_{|T_p \mathbb{S}^n}$ , where  $R_\alpha$  acts as the rotation of the angle  $\alpha = d(p,q)$  in the plane  $\pi$  containing  $\sigma$  and as the identity in the orthogonal complement of  $\pi$  and  $P_q$  is the orthogonal projection onto  $T_q \mathbb{S}^n$ .

Moreover, we introduce the following notation:

- given  $p \in \mathbb{M}^n$  and  $v \in T_p \mathbb{M}^n$ ,  $|v|_p := g_p(v, v)^{1/2}$ ;
- if  $S = \partial \Omega$  is a compact  $C^2$ -regular embedded hypersurface in  $\mathbb{M}^n_+$ , where  $\Omega$  is a relatively compact domain in  $\mathbb{M}^n_+,\,N$  is the inward unitary normal vector field on S.

The first result of this section is the following

**Proposition 1.2.** Let S be a compact  $C^2$ -regular embedded hypersurface in  $\mathbb{M}^n_+$  satisfying a touching ball condition of radius  $\rho$ . There exists  $\delta_0 = \delta_0(\rho)$  such that if  $p, q \in S$  with

$$d_S(p,q) \le \delta_0$$

then

$$g_p(N_p, \tau_q^p(N_q)) \ge \sqrt{1 - C^2 d_S(p,q)^2},$$
 (1.22)

and

$$|N_p - \tau_q^p(N_q)|_p \le C d_S(p,q),$$
 (1.23)

where C is a constant depending only on  $\rho$ .

*Proof.* We divide the proof in three parts where we consider the Euclidean, the hyperbolic and the spherical case. Let  $\delta_0 = \min(\rho_1, \frac{1}{C})$ , where C will be specified later.

- If  $\mathbb{M}^n_+ = \mathbb{R}^n$  then (1.22) (1.23) follow immediately from (1.20).
- If  $\mathbb{M}_{+}^{n} = \mathbb{H}^{n}$ , after applying the isometry  $\varphi_{p}$  we may assume that  $p = e_{n}$ and  $q = te_{n}$ . As already observed, from the definition of  $\rho_{1}$ , we have that S (actually  $\varphi_{p}(S)$ ) satisfies an Euclidean touching ball condition of radius  $\rho_{1}$ . Let  $\delta_{0} = \min(\rho_{2}, \frac{1}{C})$ , where  $\rho_{2}$  and C will be specified later. From Lemma (1.2) we get that there exists  $0 < r_{2} < r_{2}(\rho)$  such that if  $d(e_{n}, q) \leq r_{2}$  then  $|e_{n} - q| \leq \rho_{1}/2$ ; this implies that, being

$$d(p,q) \le d_S(p,q) \le r_2 \,,$$

we have

$$|1-t| = |p-q| \le \frac{\rho_1}{2}$$

Hence we can apply the Euclidean estimates (1.20) (with  $r_1$  in place of  $\rho$ ) and we get

$$\nu_p \cdot \nu_q \geq \sqrt{1 - \frac{|p-q|^2}{\rho_1^2}}$$

where, we recall that,  $\nu$  denotes the Euclidean inward normal vector field on S. Since  $d(p,q) \leq \rho$ , from (1.10) we have that  $|p-q| \leq C_1 d(p,q) \leq C_1 d_S(p,q)$  for some constant  $C_1 = C_1(\rho)$ , and hence

$$\nu_p \cdot \nu_q \ge \sqrt{1 - C^2 d_S(p, q)^2},$$
(1.24)

where C depends only  $\rho$  (actually  $C = C_1/\rho_1$ ) and provided that  $d_S(p,q) \leq \frac{1}{C}$  (observe that this implies that  $d(p,q) \leq \frac{1}{C}$ ). Moreover the inward g-unitary normal vector field N to S satisfies

$$N_p = \nu_p , \quad \nu_q = \frac{1}{t} N_q = \tau_q^p(N_q) ,$$

and (1.24) implies

$$g_p(N_p, \tau_q^p(N_q)) \ge \frac{1}{2}\sqrt{1 - C^2 d_S(p,q)^2},$$

which is (1.22). Inequality (1.23) follows by a direct computation.

• If  $\mathbb{M}^n_+ = \mathbb{S}^n_+$  It is convenient to regard S as a hypersurface of  $\mathbb{R}^n$  equipped with the spherical metric (1.7). We may further assume that p is the origin O of  $\mathbb{R}^n$  and q belongs to a straight line passing through O. Let  $\delta_0 = \min(\rho_1, \frac{1}{C})$ , where C will be specified later. If  $d(p, q) \leq \rho_1 = \frac{\rho}{\pi}$ , then we can apply the Euclidean estimates (1.20) and obtain

$$\nu_p \cdot \nu_q \ge \sqrt{1 - \frac{|p-q|^2}{\rho_1^2}}$$

where, we recall that,  $\nu$  denotes the Euclidean inward normal vector field on S. Since, from

$$d(p,q) \le \pi |p-q|,$$

we have

$$\nu_p \cdot \nu_q \ge \sqrt{1 - C^2 d_S(p,q)^2} \,, \tag{1.25}$$

where C depends only  $\rho$ . Moreover the inward g-unitary normal vector field N to S satisfies

$$N_p = \frac{1}{2}\nu_p \,, \quad \frac{2}{1+|q|^2}N_q = \nu_q = 2\tau_q^p(N_q) \,,$$

and (1.25) implies

$$g_p(N_p, \tau_q^p(N_q)) \ge \frac{1}{2}\sqrt{1 - C^2 d_S(p,q)^2}$$

which is (1.22). Inequality (1.23) follows by a direct computation and the claim follows.

The last result of this section will be used several times in the proof of the main result, for the sake of clarity we split the result in two lemmas: in the first one we state and prove the result in the hyperbolic space while in the second one we state its "spherical counterpart"; in the Euclidean space the result is trivial.

**Lemma 1.5.** Let  $\Sigma$  and  $\hat{\Sigma}$  be two compact embedded hypersurfaces in  $\mathbb{H}^n$  satisfying both a touching ball condition of radius  $\rho$ . Assume that  $e_n \in \Sigma$  and  $T_{e_n}\Sigma = \{x_n = 0\}$  and that there exist two local parametrizations  $u, \hat{u} : B_r \to \mathbb{R}$ of  $\Sigma$  and  $\hat{\Sigma}$ , respectively, with  $0 < r \leq \rho_1$  and such that  $u - \hat{u} \geq 0$ .

Let  $p_1 = (x_1, u(x_1))$  and  $\hat{p}_1^* = (x_1, \hat{u}(x_1))$ , with  $x_1 \in \partial B_{r/4}$ , and denote by  $\gamma$  the geodesic path starting from  $p_1$  and tangent to  $-\nu_{p_1}$  at  $p_1$ . Assume that

$$d(p_1, \hat{p}_1^*) + |\nu_{p_1} - \nu_{\hat{p}_1^*}| \le \theta, \qquad (1.26)$$

for some  $\theta \in [0, 1/2]$ , where  $\nu$  is the Euclidean unitary normal vector field to  $\Sigma$ . There exists  $\bar{r}$  depending only on  $\rho$  such that if  $r \leq \bar{r}$  we have that  $\gamma \cap \hat{\Sigma} \neq \emptyset$ 

and, if we denote by  $\hat{p}_1$  the first intersection point between  $\gamma$  and  $\hat{\Sigma}$ , then

$$d(p_1, \hat{p}_1) + |N_{p_1} - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \le C\theta,$$

where C is a constant depending only on n and  $\rho$ , and provided that  $C\theta < 1/2$ .

*Proof.* We first notice that, by choosing r small enough in terms of  $\rho$ , from Lemma 1.3 we have that  $|\nu_{p_1} - e_n| \leq 1/4$ . Let  $B^+$  and  $B^-$  be the exterior and interior touching balls of  $\hat{\Sigma}$  at  $\hat{p}_0 = (0, \hat{u}(0))$ , respectively. A standard geometrical argument shows that it is possible to choose  $\bar{r}$  small enough in terms of  $\rho$  such that  $\gamma$  intersects  $B^+$  and  $B^-$  at points which are distant from the origin less than  $\bar{r}$ . This implies the existence of the point  $\hat{p}_1$  in the assertion for any  $r \leq \bar{r}$ .

Now we estimate the distance between  $p_1$  and  $\hat{p}_1$  as follows. Let q be the unique point having distance  $2\varepsilon$  from  $p_1$  and lying on the geodesic path containing  $p_1$  and  $\hat{p}_1^*$ . Let T be the geodesic right-angle triangle having vertices  $p_1$  and q and hypotenuse contained in the geodesic passing through  $p_1$  and  $\hat{p}_1$  (see Figure 1.2). Since the angle  $\alpha$  at the vertex  $p_1$  is such that  $|\sin \alpha| \leq 1/4$ , then from the sine rule for hyperbolic triangles we have that

$$d(p_1, \hat{p}_1) \le C\theta. \tag{1.27}$$

Moreover, the cosine law in hyperbolic space gives that

$$d(\hat{p}_1^*, \hat{p}_1) \le C\theta \tag{1.28}$$

for some constant C, and from Proposition 1.2 we obtain that

$$|N_{p_1} - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \le |N_{p_1} - \tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*})|_{p_1} + |\tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*}) - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1}.$$
 (1.29)

Since  $p_1$  and  $\hat{p}_1^*$  are on the same vertical line, we have that

$$|N_{p_1} - \tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*})|_{p_1} = |\nu_{p_1} - \nu_{\hat{p}_1^*}| \le C\theta.$$
(1.30)

where the last inequality follows from (1.26). Now we show that

$$|\tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*}) - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \le C\theta.$$
(1.31)

We obtain (1.31) by showing that if q,  $\hat{q}$  and z in  $\mathbb{H}^n$  are such that q and  $\hat{q}$  belong to the Euclidean ball centred at z of radius R, then

$$|\tau_q^z(v) - \tau_{\hat{q}}^z(w)|_z \le C\left(d(z,q) + d(z,\hat{q}) + d(q,\hat{q}) + |v - \tau_{\hat{q}}^q(w)|_q\right), \quad (1.32)$$

for every  $v, w \in \mathbb{R}^n$  with  $|v|_q = |w|_{\hat{q}} = 1$ , where the constant C depends only on R. In order to prove (1.32) we may assume, up to apply the isometry  $\varphi_z$ , that  $z = e_n, q$  and  $\hat{q}$  belong to the same vertical line and z, q and  $\hat{q}$  belong to the same plane generated by  $e_{n-1}$  and  $e_n$ . Note that, by construction,  $q_{n-1} = \hat{q}_{n-1} \neq 0$ . Taking into account that

$$|v| = q_n$$
,  $|w| = \hat{q}_n$ ,  $|\tau_q^z(v)| = 1$ ,  $|\tau_{\hat{q}}^z(w)| = 1$ ,

where we used Lemma 1.4, and that

$$|v - \tau_{\hat{q}}^{q}(w)|_{q} = \left|v - \frac{\hat{q}_{n}}{q_{n}}w\right|_{q} = \frac{1}{q_{n}}\left|v - \frac{q_{n}}{\hat{q}_{n}}w\right| = \left|\frac{1}{q_{n}}v - \frac{1}{\hat{q}_{n}}w\right|,$$

where we used the fact that q and  $\hat{q}$  belong to the same vertical line, we have

$$\begin{split} |\tau_q^z(v) - \tau_{\hat{q}}^z(w)|_z &\leq \left|1 - \frac{1}{q_n}\right| |\tau_q^z(v)| + \left|\tau_q^z\left(\frac{1}{q_n}v - \frac{1}{\hat{q}_n}w\right)\right| \\ &+ \frac{1}{\hat{q}_n} |\tau_q^z(w) - \tau_{\hat{q}}^z(w)| + \left|\frac{1}{\hat{q}_n} - 1\right| |\tau_{\hat{q}}^z(w)| \\ &= \frac{1}{q_n} \left(|q_n - 1| + \left|\frac{1}{q_n}v - \frac{1}{\hat{q}_n}w\right|\right) + \frac{1}{\hat{q}_n} |\tau_q^z(w) - \tau_{\hat{q}}^z(w)| \\ &+ \frac{|\hat{q}_n - 1|}{\hat{q}_n} \\ &= \frac{1}{q_n} \left(|q_n - 1| + |v - \tau_{\hat{q}}^q(w)|_q\right) + \frac{1}{\hat{q}_n} |\tau_q^z(w) - \tau_{\hat{q}}^z(w)| \\ &+ \frac{|\hat{q}_n - 1|}{\hat{q}_n} \,. \end{split}$$

Moreover, from Lemma 1.4 we have that

$$\frac{1}{\hat{q}_n} |\tau_q^z(w) - \tau_{\hat{q}}^z(w)| \le C d(q, \hat{q}) \,,$$

where C is a constant depending only on R, and from Lemma 1.2 we get (1.32). Now we are ready to prove (1.31), indeed from (1.32) we get

$$\begin{aligned} |\tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*}) - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \\ &\leq C\left(d(p_1, \hat{p}_1) + d(p_1, \hat{p}_1^*) + d(\hat{p}_1, \hat{p}_1^*) + |N_{\hat{p}_1} - \tau_{\hat{p}_1^*}^{\hat{p}_1}(N_{\hat{p}_1^*})|_{\hat{p}_1}\right)\,, \end{aligned}$$

and from (1.26), (1.27), (1.28) and (1.30) we obtain (1.31). By using (1.31) and (1.30) in (1.29) we get

$$|N_{p_1} - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \le C\theta \,,$$

and therefore, from (1.27) we conclude.



Figure 1.2: Proof of Lemma 1.5.

**Lemma 1.6.** Let  $\Sigma$  and  $\hat{\Sigma}$  be two compact embedded hypersurfaces in  $\mathbb{R}^n$  with the round metric (1.7) satisfying a touching ball condition of radius  $\rho$ . Assume

 $O \in \Sigma$  and  $T_O \Sigma = \{x_n = 0\}$ , where O is the origin of  $\mathbb{R}^n$ , and that there exist two local parametrizations  $u, \hat{u} : B_r \to \mathbb{R}$  of  $\Sigma$  and  $\hat{\Sigma}$ , respectively, with  $0 < r \le \rho_1$  and such that  $u - \hat{u} \ge 0$ . Let  $p_1 = (x_1, u(x_1))$  and  $\hat{p}_1^* = (x_1, \hat{u}(x_1))$ , with  $x_1 \in \partial B_{r/4}$ , and denote by  $\gamma$  the geodesic path starting from  $p_1$  and tangent to  $-\nu_{p_1}$  at  $p_1$ . Assume that

$$d(p_1, \hat{p}_1^*) + |\nu_{p_1} - \nu_{\hat{p}_1^*}| \le \theta.$$
(1.33)

for some  $\theta \in [0, 1/2]$ . where  $\nu$  is the Euclidean unitary normal vector field to  $\Sigma$ . There exists  $\bar{r}$  depending only on  $\rho$  such that if  $r \leq \bar{r}$  we have that  $\gamma \cap \hat{\Sigma} \neq \emptyset$  and, if we denote by  $\hat{p}_1$  the first intersection point between  $\gamma$  and  $\hat{\Sigma}$ , then

$$d(p_1, \hat{p}_1) + |N_{p_1} - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \le C\theta,$$

where C is a constant depending only on n and  $\rho$ , and provided that  $C\theta < 1/2$ .

*Proof.* We first notice that, by choosing r small enough in terms of  $\rho$ , from Lemma 1.3 we have that  $|\nu_{p_1} - e_n| \leq 1/4$ . We observe that the geodesic  $\gamma$  is almost flat, i.e., viewed as an Euclidean circle its radius R satisfies

$$R = O\left(\frac{1}{|x_1|^2}\right) \text{ as } |x_1| \to 0.$$
 (1.34)

Indeed, up to apply a rotation, we may assume that both  $p_1$  and  $\nu_{p_1}$  belong to the plane  $\pi_1$  spanned by  $\{e_1, e_2\}$ . In this way, the geodesic path  $\gamma$  belongs to the plane  $\pi_1$  and we can work in a "bidimensional way". We can write  $p_1 = (x_1, y_1)$ and  $\nu_{p_1} = (\nu_1, \nu_2)$  and we compute the geodesic  $\gamma$  passing through  $p_1$  and tangent to  $\nu_{p_1}$ . We solve

$$\begin{cases} (x_1 + a)^2 + (y_1 + b)^2 = 1 + a^2 + b^2 \\ (x_1 + a, y_1 + b) \cdot (\nu_1, \nu_2) = 0 \end{cases}$$

and we find

$$a = \frac{(1 - |p_1|^2)\nu_2 + 2y_1(p_1 \cdot \nu)}{2(x_1\nu_2 - y_1\nu_1)}, \quad b = \frac{-(1 - |p_1|^2)\nu_1 - 2x_1(p_1 \cdot \nu)}{2(x_1\nu_2 - y_1\nu_1)}.$$

If  $|x_1| \to 0$ , according to Lemma 1.3, we have that  $y_1 = O(x_1^2)$ ,  $\nu_1 = O(x_1)$  and  $\nu_2 = 1 + o(1)$ ; so we get that  $a \sim \frac{1}{2x_1}$ , b is bounded and (1.34) follows.

Let  $B^+$  and  $B^-$  be the exterior and interior touching balls of  $\hat{\Sigma}$  at  $\hat{p}_0 = (0, \hat{u}(0))$ , respectively. A standard geometrical argument shows that it is possible to choose  $\bar{r}$  small enough in terms of  $\rho$  such that  $\gamma$  intersects  $B^+$  and  $B^-$  at points which are distant from the origin less than  $\bar{r}$ . This implies the existence of the point  $\hat{p}_1$  in the assertion for any  $r \leq \bar{r}$ .

Now we estimate the distance between  $p_1$  and  $\hat{p}_1$  as follows. Let q be the unique point having distance  $2\varepsilon$  from  $p_1$  and lying on the geodesic path containing  $p_1$  and  $\hat{p}_1^*$ . Let T be the geodesic right-angle triangle having vertices  $p_1$  and q and hypotenuse contained in the geodesic passing through  $p_1$  and  $\hat{p}_1$  (see Figure 1.3 and recall (1.34)). Since the angle  $\alpha$  at the vertex  $p_1$  is such that  $|\sin \alpha| \leq 1/4$ , then from the cosine rule for spherical triangles we have that

$$d(p_1, \hat{p}_1) \le C\theta. \tag{1.35}$$

Moreover, the triangle inequality gives that

$$d(\hat{p}_1^*, \hat{p}_1) \le C\theta \tag{1.36}$$

for some constant C, and from (1.22) we obtain that

$$|N_{p_1} - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \le |N_{p_1} - \tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*})|_{p_1} + |\tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*}) - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1}.$$
 (1.37)

Since  $p_1$  and  $\hat{p}_1^*$  are on the same vertical line (1.33) implies

$$|N_{p_1} - \tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*})|_{p_1} = |\nu_{p_1} - \nu_{\hat{p}_1^*}| \le C\theta.$$
(1.38)

As next step we show that

$$|\tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*}) - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \le C\theta.$$
(1.39)

We obtain (1.39) by showing that if  $p,q \in \mathbb{R}^n$  belong to the Euclidean ball centred at the origin and having radius s, then

$$2|\tau_p^O(v) - \tau_q^O(w)| \le \frac{(1+s)^2}{4} \left( d(p,O)^2 + d(q,O)^2 \right) + |v - \tau_q^p(w)|_p + \frac{9}{2} d(p,q) \,.$$
(1.40)

for every  $v, w \in \mathbb{R}^n$ ,  $|v|_q = |w|_{\hat{q}} = 1$ . We have

$$\tau^O_p(v) = \frac{1}{1+|p|^2}v\,,\quad \tau^O_q(w) = \frac{1}{1+|q|^2}w.$$

and using Cauchy-Schwarz inequality and taking into account Lemma 1.2 we have

$$||q|^2 - |p|^2| = |(q-p) \cdot (q+p)| \le |q-p||q+p| \le s(1+s^2) d(p,q)$$

 $\operatorname{since}$ 

$$|w| = \frac{1 + |\hat{q}|^2}{2}$$

we have

$$2\left|\frac{1}{1+|q|^2}\tau_p^O(w) - \frac{1}{1+|q|^2}\tau_q^O(w)\right| = 2\left|\frac{|q|^2 - |p|^2}{(1+|p|^2)(1+|q|^2)}\right| \\ \leq 2s(1+s^2)\,d(p,q)\,.$$

Now using

$$2\left|\tau_{p}^{O}(v)\right| = 1, \quad 2\left|\tau_{q}^{O}(w)\right| = 1;$$

and

$$|v| = \frac{1+|p|^2}{2}, \quad |w| = \frac{1+|q|^2}{2},$$

we compute

$$\begin{split} 2|\tau_p^O(v) - \tau_q^O(w)| &\leq 2 \left| \tau_p^O(v) - \frac{1}{1+|p|^2} \tau_p^O(v) \right| + 2 \left| \frac{1}{1+|p|^2} \tau_p^O(v) - \tau_q^O(w) \right| \\ &\leq 2 \left| 1 - \frac{1}{1+|p|^2} \right| \left| \tau_p^O(v) \right| + 2 \left| \frac{1}{1+|p|^2} \tau_q^O(v) - \frac{1}{1+|q|^2} \tau_p^O(w) \right| \\ &\quad + 2 \left| \frac{1}{1+|q|^2} \tau_p^O(w) - \frac{1}{1+|q|^2} \tau_q^O(w) \right| + 2 \left| 1 - \frac{1}{1+|q|^2} \right| \left| \tau_q^O(w) \right| \\ &\leq \left| 1 - \frac{1}{1+|p|^2} \right| + 2 \left| \frac{1}{1+|p|^2} \tau_q^O(v) - \frac{1}{1+|q|^2} \tau_p^O(w) \right| \\ &\quad + 4d(p,q) + \left| 1 - \frac{1}{1+|q|^2} \right| \\ &\leq |p|^2 + 2 \left| \frac{1}{1+|p|^2} \tau_q^O(v) - \frac{1}{1+|q|^2} \tau_p^O(w) \right| + 4d(p,q) + |q|^2 \\ &\leq \frac{(1+s)^2}{4} \left( d(p,O)^2 + d(q,O)^2 \right) \\ &\quad + 2 \left| \frac{1}{1+|p|^2} \tau_p^O(v) - \frac{1}{1+|q|^2} \tau_p^O(w) \right| + 4d(p,q) \,. \end{split}$$

Now we show that

$$2\left|\frac{1}{1+|p|^2}\tau_p^O(v) - \frac{1}{1+|q|^2}\tau_p^O(w)\right| \le |v-\tau_q^p(w)|_p + \frac{1}{2}d(p,q).$$

Let  $\sigma$  be the geodesic path connecting p with q. Then  $\sigma$  is contained in a circle of  $\mathbb{R}^n$  and denotes by C its center and by  $\alpha$  the angle between p - C and q - C. Then

$$\frac{1+|p|^2}{1+|q|^2}w=R_\alpha\tau^p_qw\,, \text{ for every } w\in\mathbb{R}^n\,,$$

where  $R_{\alpha}$  is the rotation (clockwise or anti-clockwise) about  $\alpha$  in the plane containing C and is the identity in the complement. Therefore we have

$$\begin{aligned} 2\left|\frac{1}{1+|p|^{2}}\tau_{p}^{O}(v)-\frac{1}{1+|q|^{2}}\tau_{p}^{O}(w)\right| &= \left|\frac{1}{1+|p|^{2}}v-\frac{1}{1+|q|^{2}}w\right|_{p} \\ &\leq \left|v-\frac{1+|p|^{2}}{1+|q|^{2}}w\right|_{p} \\ &= \left|v-R_{\alpha}\tau_{q}^{p}w\right|_{p} \\ &\leq \left|v-\tau_{q}^{p}w\right|_{p}+\left|\tau_{q}^{p}w-R_{\alpha}\tau_{q}^{p}w\right|_{p} \end{aligned}$$

and, consequently, we deduce,

$$\left|\tau_{q}^{p}w - R_{\alpha}\tau_{q}^{p}w\right|_{p} \le |\alpha| \le \frac{1}{2}d(p,q)$$

which implies (1.40).

Therefore, by applying (1.40) to  $|\tau_{\hat{p}_1^*}^{p_1}(N_{\hat{p}_1^*}) - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1}$ , we have

$$\begin{aligned} |\tau_{\hat{p}_{1}^{*}}^{p_{1}}(N_{\hat{p}_{1}^{*}}) - \tau_{\hat{p}_{1}}^{p_{1}}(N_{\hat{p}_{1}})|_{p_{1}} &\leq \frac{(1+C\theta)^{2}}{4} \left( d(p_{1},\hat{p}_{1}^{*})^{2} + d(p_{1},\hat{p}_{1})^{2} \right) \\ &+ |N_{\hat{p}_{1}^{*}} - \tau_{\hat{p}_{1}}^{\hat{p}_{1}^{*}}(N_{\hat{p}_{1}^{*}})|_{\hat{p}_{1}^{*}} \\ &+ \frac{9}{2} d(\hat{p}_{1},\hat{p}_{1}^{*}) \\ &\leq C\theta \,, \end{aligned}$$

where the last inequality follows from (1.33),(1.35),(1.36) and (1.22). This last inequality, (1.37) and (1.38) imply that

$$|N_{p_1} - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \le C\theta$$

and therefore from (1.35) we conclude.



Figure 1.3: Proof of Lemma 1.6.

# 1.3 Curvatures of projected surfaces in conformally Euclidean spaces

In this section we consider a connected open set  $\Omega$  in  $\mathbb{R}^n$  equipped with a metric  $g(\cdot, \cdot) = h^2 \langle \cdot, \cdot \rangle$  conformal to the Euclidean metric. We further assume the existence of an Euclidean hyperplane  $\pi$  of  $\mathbb{R}^n$  such that  $\Omega \cap \pi$  is a totally geodesic hypersurface in  $\Omega$ . This setting includes the Euclidean space, the hyperbolic space and  $\mathbb{R}^n$  with the round metric (1.7). For instance in the half-space model of the hyperbolic space we can take as  $\pi$  any vertical Euclidean hyperplane; in the spherical case we can consider Euclidean hyperplanes passing through the origin.

For our purposes, we consider a hypersurface U of class  $C^2$  embedded in  $\Omega$  which intersects  $\pi$  transversally. The implicit function theorem implies that  $U' = U \cap \pi$  is a  $C^2$ -submanifold of  $\pi$ . Furthermore if  $\nu_q$  is an Euclidean unit normal vector field to U and w is a unit normal vector to  $\pi$ , we have that

$$\nu_q' = \frac{(-1)^n \ast (\ast(\nu_q \land w) \land w)}{|\ast(\ast(\nu_q \land w) \land w)|}, \quad q \in U',$$

is an Euclidean unitary normal vector field to U' in  $\pi$ , where \* is the Euclidean Hodge star operator in  $\mathbb{R}^n$ . In particular U' is orientable in  $\pi$ . Let

$$N_q = \frac{1}{h(q)}\nu_q, \quad N'_q = \frac{1}{h(q)}\nu'_q$$

be the normal vectors with respect to the metric g and

$$\omega_p = \frac{1}{h(p)} w, \quad p \in \Omega.$$

**Proposition 1.3.** Let  $\kappa_j$ , j = 1, ..., n-1 be the principal curvatures of U with respect to metric g and to the orientation induced by N. Then the principal curvatures  $\kappa'_i$  of U' (viewed as submanifold of  $\pi$ ) with respect to the orientation induced by N' satisfy

$$\frac{1}{\sqrt{1 - g_q(\omega_q, N_q)^2}} \kappa_1(q) \le \kappa_i'(q) \le \frac{1}{\sqrt{1 - g_q(\omega_q, N_q)^2}} \kappa_{n-1}(q), \qquad (1.41)$$

for every  $q \in U'$ . Moreover, the principal curvatures  $\check{\kappa}'_i$  of U' seen as a hypersurface of U satisfy

$$|\check{\kappa}_{i}'(q)| \leq \frac{|g_{q}(\omega_{q}, N_{q})|}{\sqrt{1 - g_{q}(\omega_{q}, N_{q})^{2}}} \max\{|\kappa_{1}(q)|, |\kappa_{n-1}(q)|\}, \qquad (1.42)$$

for every  $q \in U'$ .

*Proof.* Let  $v \in T_q U'$  satisfy  $|v|_q = 1$  and

$$\kappa_q(v) = g_q(\nabla_v N, v) \,,$$

where  $\tilde{N}$  denotes an extension of N in  $\Omega$  and  $\nabla$  is the Levi-Civita connection of g. For  $p \in U'$ ,  $N_q$  is orthogonal to  $T_qU'$  and consequently it lies on the plane spanned by w and  $N'_q$  and hence

$$N = a w + b N',$$

where a is a function on U' and

$$b = q(N, N').$$

If  $\tilde{a}$ ,  $\tilde{b}$  and  $\tilde{N}'$  are extensions of a, b and N' in  $\Omega$ ,

$$\tilde{N} = \tilde{a} \, w + \tilde{b} \, \tilde{N}$$

defines an extension of N and a direct computation yields

$$\kappa_p(v) = a(p) g_p(\nabla_v w, v) + b(p) g_p(\nabla_v \tilde{N}', v) = b(p) g_p(\nabla_v \tilde{N}', v) ,$$

where we used that  $\pi \cap \Omega$  is totally geodesic. Therefore

$$\frac{1}{g_q(N_q,N_q')}\kappa_q(v) = g_q(\nabla_v \tilde{N}',v)$$

and consequently

$$\frac{1}{g_q(N_q, N'_q)} \kappa_1(q) \le \kappa'_i(q) \le \frac{1}{g_q(N_q, N'_q)} \kappa_{n-1}(q)$$

for every  $q \in U'$  and  $i = 1, \ldots, n-2$ . Now we show

$$g_q(N_q, N'_q) = \sqrt{1 - g_q(\omega_q, N_q)^2},$$
 (1.43)

which implies (1.41). We have

$$\nu_{q} \cdot \nu_{q}' = \frac{(-1)^{n} * (*(\nu_{q} \wedge w) \wedge w)}{|*(*(\nu_{q} \wedge w) \wedge w)|} \cdot \nu_{q}$$

$$= \frac{1}{|*(*(\nu_{q} \wedge w) \wedge w)|} (-1)^{n} * (\nu_{q} \wedge w) \wedge w \cdot *\nu_{q}.$$
(1.44)

Let  $\{\nu_q, e_1, \ldots, e_{n-1}\}$  be a positive-oriented orthonormal basis of  $\mathbb{R}^n$  such that

- $\{e_1, \ldots, e_{n-1}\}$  is a positive-oriented Euclidean-orthonormal basis of  $T_q U$ ;
- $\{e_2, \ldots, e_{n-1}\}$  is a basis of  $T_q U'$ .

In this way  $w \in \operatorname{span}\{\nu_q, e_1\},\$ 

$$*(\nu_q \wedge w) = (w \cdot e_1) e_2 \wedge \dots \wedge e_{n-1}, \quad *\nu_q = e_1 \wedge \dots \wedge e_{n-1},$$

and

$$|*(*(\nu_q \wedge w) \wedge w)| = |*(\nu_q \wedge w) \wedge w| = w \cdot e_1.$$

So, from (1.44) we get

$$\nu_{q} \cdot \nu_{q}' = \frac{(-1)^{n}}{\omega \cdot e_{1}} * (\nu_{q} \wedge w) \wedge w \cdot *\nu_{q}$$

$$= \frac{(-1)^{n}}{\omega \cdot e_{1}} (\omega \cdot e_{1}) e_{2} \wedge \dots \wedge e_{n-1} \wedge w \cdot e_{1} \dots \wedge e_{n-1}$$

$$= (-1)^{n} (\omega \cdot e_{1}) e_{2} \wedge \dots \wedge e_{n-1} \wedge e_{1} \cdot e_{1} \wedge \dots \wedge e_{n-1}$$

$$= (\omega \cdot e_{1}) e_{1} \wedge \dots \wedge e_{n-1} \cdot e_{1} \wedge \dots \wedge e_{n-1}$$

$$= \omega \cdot e_{1}.$$

Since |w| = 1, we have  $w \cdot e_1 = \sqrt{1 - (w \cdot \nu_q)^2}$  and so

$$\nu_q \cdot \nu'_q = \sqrt{1 - (w \cdot \nu_q)^2} \,. \tag{1.45}$$

Since

$$\nu_q \cdot \nu_q' = g_q(N_q, N_q'), \quad \text{and} \quad w \cdot \nu_q = g_q(\omega_q, N_q),$$

(1.43) follows.

Now we prove (1.42). In this case we regard U' as a submanifold of U. Let  $q \in U'$ ,  $v \in T_q U'$  such that  $|v|_q = 1$  and let  $\alpha: (-\delta, \delta) \to S$  be a unitary speed curve satisfying  $\alpha(0) = q$  and  $\dot{\alpha}(0) = v$ . Let  $\tilde{N}'$  be a unitary normal vector field of U' in U near q. We may complete v with an orthonormal basis  $\{v, v_2, \ldots, v_{n-2}\}$  of  $T_q U'$  such that

$$\check{N}'_{q} = *_{q} (N_{q} \wedge v \wedge v_{2} \wedge \dots \wedge v_{n-2}),$$

where  $*_q$  is the Hodge star operator at q in  $\Omega$  with respect to g and to the standard orientation. Let

$$\check{\kappa}_q'(v) = g_q(*_q(\check{N}_q \wedge v \wedge v_2 \wedge \dots \wedge v_{n-2}), D_t \dot{\alpha}_{|t=0}),$$

where  $D_t$  is the covariant derivative in  $(\Omega, g)$ . Since  $D_t \dot{\alpha}_{|t=0} \in \pi$ , we have

$$\check{\kappa}'_q(v) = g_q(N_q, \omega_q)g_q(*_q(\omega_q \wedge v \wedge v_2 \wedge \dots \wedge v_{n-2}), D_t\dot{\alpha}_{|t=0})$$

Now,  $*_q(\omega_q \wedge v \wedge v_2 \wedge \cdots \wedge v_{n-2})$  is a normal vector to  $T_qU'$  in  $\pi$  and so

$$\check{\kappa}_q'(v) = g_q(N_q, \omega_q)g_q(\nabla_v N, v),$$

where  $\tilde{N}$  is an arbitrary extension of N in a neighbourhood of q. From (1.41) we obtain

$$|\check{\kappa}'_q(v)| \le \frac{|g_q(N_q, \omega_q)|}{\sqrt{1 - g_q(\omega_q, N_q)^2}} \max\{|\kappa_1(q)|, |\kappa_{n-1}(q)|\},\$$

as required.

Remark 1.1. It may be convenient to explain the meaning of (1.45) when n = 3. In this case  $*(v \wedge w)$  is the vector product  $v \times w$ , so

$$\nu_q' = -\frac{(\nu_q \times w) \times w}{|(\nu_q \times w) \times w|} \quad \text{ and } \quad |(\nu_q \times w) \times w| = \sqrt{|\nu_q|^2 |w|^2 - (\nu_q \cdot w)^2} \,.$$

 $\operatorname{So}$ 

$$\begin{split} (\nu_q \cdot \nu'_q) &= -\frac{1}{\sqrt{|\nu_q|^2 |w|^2 - (\nu_q \cdot w)^2}} (\nu_q \times w) \times w \cdot \nu_q \\ &= -\frac{1}{\sqrt{|\nu_q|^2 |w|^2 - (\nu_q \cdot w)^2}} (\nu_q \times w) \cdot (w \times \nu_q) \\ &= \frac{|\nu_q \times w|^2}{\sqrt{|\nu_q|^2 |w|^2 - (\nu_q \cdot w)^2}} \\ &= \frac{|\nu_q|^2 |w|^2 - (\nu_q \cdot w)^2}{\sqrt{|\nu_q|^2 |w|^2 - (\nu_q \cdot w)^2}} \\ &= \sqrt{1 - (\nu_q \cdot w)^2} \,. \end{split}$$

Now we focus in a different setting. Let  $\overline{\Omega}$  be the projection of  $\Omega$  onto  $\{x_n = 0\}$  and let  $\pi \subseteq \Omega$  be the graph of a  $C^2$  function  $F: A \to \mathbb{R}$ , where  $A \subseteq \overline{\Omega}$  is a open subset.

**Proposition 1.4.** Let U' be a  $C^2$  regular oriented hypersurface of  $\pi$  and let U'' be the orthogonal projection of U' onto  $\{x_n = 0\}$ . Then the principal curvatures of U'' satisfy

$$\begin{aligned} |\kappa_i''(\bar{q})| &\leq \frac{h(q)}{\sqrt{1 + |\nabla F(\bar{q})|^2}} \left( (\nu_q' \cdot e_n)^2 + \frac{1}{1 + |\nabla F(\bar{q})|^2} \right)^{-3/2} \times \\ & \left( \max\{|\kappa_1'(q)|, |\kappa_{n-1}'(q)|\} + 4\frac{|\nabla h(q)|}{h(q)^2} \right), \end{aligned}$$
(1.46)

for every i = 1, ..., n-2, where  $\{\kappa''_i\}$  are the principal curvatures of U'' with respect to the Euclidean metric,  $q \equiv (\bar{q}, q_n) \in U'$  and  $\nu'_q = h(q)N'_q$ .

*Proof.* If X is a local positive oriented parametrization of U', then

$$\bar{X} = X - (X \cdot e_n)e_n$$

is a local parametrization of U'', and we can orient U'' with

$$\nu'' \circ \bar{X} := \operatorname{vers}(*(\bar{X}_1 \wedge \bar{X}_2 \wedge \dots \wedge \bar{X}_{n-2} \wedge e_n)), \qquad (1.47)$$

where  $\bar{X}_k$  is the  $k^{th}$  derivative of  $\bar{X}$  with respect to the coordinates of its domain and \* is the Hodge star operator in  $\mathbb{R}^n$  with respect to the the Euclidean metric and the standard orientation.

Now we prove inequalities (1.46). Fix a point  $q = (\bar{q}, q_n) \in U'$  and  $\bar{v} \in T_{\bar{q}}U'$ be nonzero. Let  $\beta : (-\delta, \delta) \to U''$  be an arbitrary regular curve contained in U''such that

$$\beta(0) = \bar{q}, \quad \dot{\beta}(0) = \bar{v}.$$

Then

$$\kappa_{\bar{q}}^{\prime\prime}(\bar{v}) = \frac{1}{|\bar{v}|^2} \nu_{\bar{q}}^{\prime\prime} \cdot \ddot{\beta}(0)$$

is the normal curvature of U'' at  $(\bar{q}, \bar{v})$ , viewed as hypersurface of  $\{x_n = 0\}$  with the Euclidean metric. We can write

$$\kappa_{\bar{q}}^{\prime\prime}(\bar{v}) = \frac{1}{|\bar{v}|^2} \nu_{\bar{q}}^{\prime\prime} \cdot \ddot{\alpha}(0)$$

where  $\alpha = (\beta, \alpha_n)$  whose projection onto U' is  $\beta$ . From

$$\bar{X}_k = X_k - (X_k \cdot e_n)e_n \,,$$

and the definition of  $\nu''$  (1.47) we have

$$\kappa_{\bar{q}}''(\bar{v}) = \frac{*(X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e_n) \cdot \ddot{\alpha}(0)}{|\dot{\beta}(0)|^2 |X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e_n|} \,.$$

We may assume that  $\{X_1(q), \ldots, X_{n-2}(q)\}$  is an orthonormal basis of  $T_qU'$  with respect to the Euclidean metric. Let

$$\mathsf{N}_{\mathsf{q}} = \frac{(-\nabla F(\bar{q}), 1)}{\sqrt{1 + |\nabla F(\bar{q})|^2}}$$

be the Euclidean normal vector to  $\pi$  at q and let

$$a = \sqrt{1 + |\nabla F(\bar{q})|^2} \,.$$

Therefore  $\{X_1(q), \ldots, X_{n-2}(q), \nu'_q, \mathsf{N}_q\}$  is an Euclidean orthonormal basis of  $\mathbb{R}^n$ and we can split  $\mathbb{R}^n$  in

$$\mathbb{R}^n = T_q U' \oplus \langle \nu_q' \rangle \oplus \langle \mathsf{N}_q \rangle, \qquad (1.48)$$

and  $e_n$  splits accordingly into

$$e_n = e'_n + e''_n + e'''_n.$$

Therefore

$$*(X_1(q) \wedge \cdots \wedge X_{n-2}(q) \wedge e_n) \cdot \ddot{\alpha}(0) = *(X_1(q) \wedge \cdots \wedge X_{n-2}(q) \wedge e_n'') \cdot \ddot{\alpha}(0),$$

i.e.

$$*(X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e_n) \cdot \ddot{\alpha}(0) = \frac{1}{a} * (X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge \mathsf{N}_q) \cdot \ddot{\alpha}(0).$$

Since

$$\nu'_q = * (X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge \mathsf{N}_q)$$

we obtain

$$\kappa_{\bar{q}}^{\prime\prime}(\bar{v}) = \frac{1}{a|\dot{\beta}(0)|^2} \frac{\nu_q^{\prime} \cdot \ddot{\alpha}(0)}{|X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e_n|}.$$

We may assume that  $\alpha$  is parametrized by arc length with respect to the metric g, i.e.

$$|\dot{\alpha}|^2 = h(\alpha)^{-2}$$

and so

$$|\dot{\beta}|^2 = h(\alpha)^{-2} - \dot{\alpha}_n^2$$
,

which implies

$$\kappa_{\bar{q}}^{\prime\prime}(\bar{v}) = \frac{1}{a(h(q)^{-2} - v_n^2)} \frac{\nu_q^{\prime} \cdot \ddot{\alpha}(0)}{|X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e_n|} \,. \tag{1.49}$$

Since

$$X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e_n$$
  
=  $X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e_n'' + X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e_n'''$ 

 $\quad \text{and} \quad$ 

$$X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e''_n = (\nu'_q \cdot e_n) X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge \nu'_q,$$
  
$$X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e'''_n = \frac{1}{a} X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge \mathsf{N}_q,$$

we obtain

$$|X_1(q) \wedge \dots \wedge X_{n-2}(q) \wedge e_n| = \left( (\nu'_q \cdot e_n)^2 + \frac{1}{a^2} \right)^{1/2}.$$

On the other hand

$$\kappa_q'(v) = g_q(N_q', D_t \dot{\alpha}_{|t=0})$$

where  $D_t$  is the covariant derivative in  $\pi$ . It is well-known that the Christoffel symbols of g are given by

$$\Gamma_{ij}^k = \delta_i^k \partial_j f + \delta_j^k \partial_i f - \delta_i^j \partial_k f \,,$$

where  $f = \log h$ . We have

$$D_t \dot{\alpha} = \ddot{\alpha} + \sum_{i,j,k=1}^n \Gamma_{ij}^k(\alpha) \dot{\alpha}_i \dot{\alpha}_j e_k$$
  
=  $\ddot{\alpha} + \sum_{i,j,k=1}^n (\delta_i^k \partial_j f(\alpha) + \delta_j^k \partial_i f(\alpha) - \delta_i^j \partial_k f(\alpha)) \dot{\alpha}_i \dot{\alpha}_j e_k$   
=  $\ddot{\alpha} + \sum_{i,k=1}^n (2\partial_i f(\alpha) \dot{\alpha}_i \dot{\alpha}_k - \dot{\alpha}_i^2 \partial_k f(\alpha)) e_k + \sum_{k=1}^n \partial_k f(\alpha) \dot{\alpha}_k^2 e_k$ 

 $\operatorname{and}$ 

$$D_t \dot{\alpha}_{|t=0} = \ddot{\alpha}(0) + \sum_{i,k=1}^n (2\partial_i f(q)v_i v_k - v_i^2 \partial_k f(q)) e_k + \sum_{k=1}^n \partial_k f(q) v_k^2 e_k.$$

Therefore

$$\begin{split} \kappa_{q}'(v) &= g_{q} \left( N_{q}', \ddot{\alpha}(0) + \sum_{i,k=1}^{n} (2\partial_{i}f(q)v_{i}v_{k} - v_{i}^{2}\partial_{k}f(q)) e_{k} + \sum_{k=1}^{n} \partial_{k}f(q)v_{k}^{2} e_{k} \right) \\ &= h(q)\nu_{q}' \cdot \ddot{\alpha}(0) + h(q) \sum_{i,k=1}^{n} (2\partial_{i}f(q)v_{i}v_{k} - v_{i}^{2}\partial_{k}f(q)) e_{k} \cdot \nu_{q}' \\ &+ h(q) \sum_{k=1}^{n} \partial_{k}f(q)v_{k}^{2}\nu_{q}' \cdot e_{k} \\ &= h(q)\nu_{q}' \cdot \ddot{\alpha}(0) + \sum_{i,k=1}^{n} (2\partial_{i}h(q)v_{i}v_{k} - v_{i}^{2}\partial_{k}h(q)) e_{k} \cdot \nu_{q}' \\ &+ \sum_{k=1}^{n} \partial_{k}h(q)v_{k}^{2}\nu_{q}' \cdot e_{k} \,, \end{split}$$

and we get

$$\nu_q' \cdot \ddot{\alpha}(0) = \frac{\kappa_q'(v)}{h(q)} - \frac{1}{h(q)} \sum_{i,k=1}^n (2\partial_i h(q) v_i v_k - v_i^2 \partial_k h(q)) e_k \cdot \nu_q'$$
$$- \frac{1}{h(q)} \sum_{k=1}^n \partial_k h(q) v_k^2 \nu_q' \cdot e_k \,.$$

From (1.49) we deduce

$$\begin{split} \kappa_{\bar{q}}^{\prime\prime}(\bar{v}) &= \frac{1}{a(h(q)^{-2} - v_n^2)} \left( \langle \nu_q^{\prime}, e_n \rangle^2 + \frac{1}{a^2} \right)^{-1/2} \times \\ & \left( \frac{\kappa_q^{\prime}(v)}{h(q)} - \frac{1}{h(q)} \sum_{i,k=1}^n (2\partial_i h(q) v_i v_k - v_i^2 \partial_k h(q)) \, e_k \cdot \nu_q^{\prime} - \frac{1}{h(q)} \sum_{k=1}^n \partial_k h(q) v_k^2 \nu_q^{\prime} \cdot e_k \right) \,, \end{split}$$

for every  $v\in T_qU'$  such that  $g_q(v,v)=1.$  Therefore

$$\kappa_1''(\bar{q}) = \frac{1}{ah(q)} \left( (\nu_q' \cdot e_n)^2 + \frac{1}{a^2} \right)^{-1/2} \inf_{\substack{v \in \mathbb{S}_q^{n-2} \\ q}} A_q(v) ,$$
  
$$\kappa_{n-2}''(\bar{q}) = \frac{1}{ah(q)} \left( (\nu_q' \cdot e_n)^2 + \frac{1}{a^2} \right)^{-1/2} \sup_{\substack{v \in \mathbb{S}_q^{n-2} \\ q}} A_q(v) ,$$

where

$$A_q(v) = \frac{1}{h(q)^{-2} - v_n^2} \times \left(\kappa'_q(v) - \sum_{i,k=1}^n (2\partial_i h(q)v_i v_k - v_i^2 \partial_k h(q)) e_k \cdot \nu'_q - \sum_{k=1}^n \partial_k h(q) v_k^2 \nu'_q \cdot e_k\right)$$

and  $\mathbb{S}_q^{n-2}=\{v\in T_qU'\ :\ |v|_q=1\}.$  Since  $|v|_q^2=1,$  then  $|v|^2=h(q)^{-2}$  and we can rewrite  $A_q(v)$  as

$$A_q(v) = \frac{1}{h(q)^{-2} - v_n^2} \times \left(\kappa'_q(v) - \nu'_q \cdot \left(2(\nabla h(q) \cdot v)v - h(q)^{-2}\nabla h(q) + \sum_{k=1}^n \partial_k h(q)v_k^2 e_k\right)\right).$$

Since  $|\kappa_i''(\bar{q})| \le \max\{|\kappa_1''(\bar{q})|, |\kappa_{n-2}''(\bar{q})|\}, i = 1, ..., n-2$ , we obtain

$$|\kappa_i''(\bar{q})| \le \frac{1}{ah(q)} \left( (\nu_q' \cdot e_n)^2 + \frac{1}{a^2} \right)^{-1/2} \sup_{v \in \mathbb{S}_q^{n-2}} |A_q(v)|.$$
(1.50)

We have

$$\begin{split} |A_{q}(v)| &= \frac{1}{h(q)^{-2} - v_{n}^{2}} \times \\ & \left| \kappa_{q}'(v) - \nu_{q}' \cdot \left( 2(\nabla h(q) \cdot v) \, v - \frac{1}{h(q)^{2}} \nabla h(q) + \sum_{k=1}^{n} \partial_{k} h(q) v_{k}^{2} e_{k} \right) \right| \\ & \leq \frac{1}{h(q)^{-2} - v_{n}^{2}} \times \\ & \left( |\kappa_{q}'(v)| + \left| 2(\nabla h(q) \cdot v) \, v - \frac{1}{h(q)^{2}} \nabla h(q) + \sum_{k=1}^{n} \partial_{k} h(q) v_{k}^{2} e_{k} \right| \right) \\ & \leq \frac{1}{h(q)^{-2} - v_{n}^{2}} \times \\ & \left( |\kappa_{q}'(v)| + 2 \left| (\nabla h(q) \cdot v) \, v \right| + \frac{1}{h(q)^{2}} \left| \nabla h(q) \right| + \left| \sum_{k=1}^{n} \partial_{k} h(q) v_{k}^{2} e_{k} \right| \right) \\ & \leq \frac{1}{h(q)^{-2} - v_{n}^{2}} \times \\ & \left( |\kappa_{q}'(v)| + \frac{2}{h(q)^{2}} \left| \nabla h(q) \right| + \frac{1}{h(q)^{2}} \left| \nabla h(q) \right| + \frac{1}{h(q)^{2}} \left| \nabla h(q) \right| \end{split}$$

i.e.,

$$|A_q(v)| \le \frac{1}{h(q)^{-2} - v_n^2} \left( |\kappa'_q(v)| + \frac{4}{h(q)^2} |\nabla h(q)| \right) \,.$$

Since  $\mathbb{R}^n = T_q U' \oplus \langle \mathsf{N}_q \rangle \oplus \langle \nu_q' \rangle$ , we can write

$$e_n = e''_n + \frac{1}{a} + (\nu_q \cdot e_n) \,\nu'_q \,,$$

where  $\tilde{e}_n$  is the orthogonal projection of  $e_n$  onto  $T_q U'$ . Therefore

$$1 - |e'_n|^2 = \frac{1}{a^2} + (\nu'_q \cdot e_n)^2$$

Since h(q)v lies in  $T_qU^\prime$  and it has unitary Euclidean norm, we have

$$|e'_n|^2 \ge h(q)^2 (e_n \cdot v)^2 = h(q)^2 v_n^2$$

and so

$$1 - h(q)^2 v_n^2 \ge \frac{1}{a^2} + (\nu_q' \cdot e_n)^2 \,,$$

i.e.

$$h(q)^{-2} - v_n^2 \ge \left(\frac{1}{a^2} + (\nu_q' \cdot e_n)^2\right) h(q)^{-2}$$

Hence

$$|A_q(v)| \le (|\kappa'_q(v)| + 4h(q)^{-2} |\nabla h(q)|) \left(\frac{1}{a^2} + (\nu'_q \cdot e_n)^2\right)^{-1} h(q)^2,$$

which yields

$$|\kappa_i''(\bar{q})| \le \frac{h(q)}{a} \left( (\nu_q' \cdot e_n)^2 + \frac{1}{a^2} \right)^{-3/2} \sup_{v \in \mathbb{S}_q^{n-2}} (|\kappa_q'(v)| + 4h(q)^{-2} |\nabla h(q)|), \quad (1.51)$$

which implies (1.46).

Now we use (1.46) in space forms.

In the Euclidean space we have  $\Omega = \mathbb{R}^n$  and h(q) = 1 and (1.46) reduces to

$$|\kappa_i''(\bar{q})| \le \frac{1}{\sqrt{1+|\nabla F(\bar{q})|^2}} \frac{\max\{|\kappa_1'(q)|, |\kappa_{n-1}'(q)|\}}{\left(\langle\nu_q', e_n\rangle^2 + \frac{1}{1+|\nabla F(\bar{q})|^2}\right)^{3/2}}.$$
 (1.52)

In particular, if  $\pi$  is an hyperplane and if we set  $\omega_1 = \frac{(\nabla F(\bar{q}), -1)}{\sqrt{1+|\nabla F(\bar{q})|^2}}$  and  $\omega_2 = e_n$ then we have

$$|\kappa_i''(\bar{q})| \le \frac{|\omega_1 \cdot \omega_2|}{\left((\omega_1 \cdot \omega_2)^2 + (\omega_2 \cdot \nu_q')\right)^{3/2}} \max\{|\kappa_1'(q)|, |\kappa_{n-1}'(q)|\}, \qquad (1.53)$$

for every i = 1, ..., n - 2.

In the Hyperbolic space we have  $\Omega = \{q_n > 0\}$  and  $h(q) = \frac{1}{q_n}$  and (1.46) reduces to

$$|\kappa_i''(\bar{q})| \le \frac{1}{q_n\sqrt{1+|\nabla F(\bar{q})|^2}} \frac{\left(\max\{|\kappa_1'(q)|, |\kappa_{n-1}'(q)|\} + 4\right)}{\left((\nu_q' \cdot e_n)^2 + \frac{1}{1+|\nabla F(\bar{q})|^2}\right)^{3/2}},$$
(1.54)

In particular, if  $\pi$  is a half-sphere of radius R with center in  $\pi_{\infty}$ , then we have

$$|\kappa_i''(\bar{q})| \le \frac{1}{R} \left( (\nu_q' \cdot e_n)^2 + \frac{q_n^2}{R^2} \right)^{-\frac{3}{2}} \left( \max\{|\kappa_1'(q)|, |\kappa_{n-1}'(q)|\} + 4 \right).$$
(1.55)

for every i = 1, ..., n - 2.

Now we focus on  $\mathbb{R}^n$  equipped with the spherical metric. In this case  $h(q) = \frac{2}{1+|q|^2}$  and (1.46) gives

$$|\kappa_i''(\bar{q})| \le \frac{2}{(1+|q|^2)\sqrt{1+|\nabla F(\bar{q})|^2}} \frac{\left(\max\{|\kappa_1'(q)|, |\kappa_{n-1}'(q)|\} + 4|q|\right)}{\left((\nu_q' \cdot e_n)^2 + \frac{1}{1+|\nabla F(\bar{q})|^2}\right)^{3/2}}.$$

In particular if  $\pi$  is the hemisphere of some hyperplane which does not contain the origin, then we have

$$|\kappa_i''(\bar{q})| \le \frac{h(q)|(q - O_\pi)_n|}{R} \frac{\left(\max\{|\kappa_1'(q)|, |\kappa_{n-1}'(q)|\} + 4|q|\right)}{\left(\left(\nu_q' \cdot e_n\right)^2 + \frac{(q - O_\pi)_n^2}{R^2}\right)^{3/2}},$$
(1.56)

for every i = 1, ..., n - 2, where  $O_{\pi}$  and R are the center and the radius of  $\pi$ , respectively.

## **1.4** Approximate symmetry in one direction

We consider the following set-up: let  $S = \partial \Omega$  be a  $C^2$ -regular connected closed hypersurface embedded in  $\mathbb{M}^n_+$ , where  $\Omega$  is a bounded domain. Assume that Ssatisfies a uniform touching ball condition of radius  $\rho > 0$ . We fix a direction vin  $T_{\mathbf{o}}\mathbb{M}^n$  and we apply the method of the moving planes as described in Section 1.1. Let  $\pi = \pi_{v,m_v}$  be the critical hyperplane and in order to simplify the notation we set

$$S_{+} = \{ p \in S : p \in \pi_{v,t} \text{ for some } t > m_{v} \},\$$
  
$$S_{-} = \{ p \in S : p \in \pi_{v,t} \text{ for some } t < m_{v} \}.$$

From the method of the moving planes we have that the reflection  $S^{\pi}_{+}$  of  $S_{+}$  with respect to  $\pi$  is contained in  $\Omega$  and it is tangent to  $S_{-}$  at a point  $p_{0}$  (internally or at the boundary). Let  $\Sigma$  and  $\hat{\Sigma}$  be the connected components of  $S^{\pi}_{+}$  and  $S_{-}$ containing  $p_{0}$ , respectively.

The main result in this section is the following

**Theorem 1.4.** There exists  $\varepsilon > 0$  such that if

$$\operatorname{osc}(\mathsf{H}_S) \leq \varepsilon$$
,

then for any  $p \in \Sigma$  there exists  $\hat{p} \in \hat{\Sigma}$  such that

$$d(p,\hat{p}) + |N_p - \tau^p_{\hat{p}}(N_{\hat{p}})|_p \le C \operatorname{osc}(\mathsf{H}_S).$$

Here, the constants  $\varepsilon$  and C depend only on n,  $\rho$  and the area of S. In particular  $\varepsilon$  and C do not depend on the direction v.

Moreover,  $\Omega$  is contained in a neighbourhood of radius  $\operatorname{Cosc}(\mathsf{H}_S)$  of  $\Sigma \cup \Sigma^{\pi}$ ( $\Sigma^{\pi}$  is the reflection of  $\Sigma$  about  $\pi$ ), i.e.

$$d(p, \Sigma \cup \Sigma^{\pi}) \le C \operatorname{osc}(\mathsf{H}_S),$$

for every  $p \in \Omega$ .

Before giving the proof of Theorem 1.4, we provide two preliminary results about the geometry of  $\Sigma$ . For t > 0 we set

$$\Sigma_t = \{ p \in \Sigma : d_{\Sigma}(p, \partial \Sigma) > t \}.$$

The following lemmas quantitatively show that  $\Sigma_t$  is connected for t small enough.

Here we use the results in Section (1.3) and we consider the unitary normal vector field  $\omega$  to  $\pi$  directed as the geodesic  $\gamma$  in  $\mathbb{M}^n_+$  satisfying  $\dot{\gamma}(0) = v$ .

Lemma 1.7. Assume

$$g_p(N_p, \omega_p) \le \mu \tag{1.57}$$

for every p on the boundary of  $\Sigma$ , for some  $\mu \leq 1/2$ , and let  $t_0 = \rho \sqrt{1-\mu^2}$ . Then  $\Sigma_t$  is connected for any  $0 < t < t_0$ .

*Proof.* We can work in  $\mathbb{R}^n$  for every space form considered, and we may assume that  $\pi$  is an Euclidean hyperplane of  $\mathbb{R}^n$  (in the spherical case we can consider the projection from a point antipodal to a point inside  $\pi$ ).

Let  $\Sigma'$  be the subset of  $\pi$  obtained by projecting  $\Sigma$  onto  $\pi$  (for any point  $p \in \Sigma$  we define the projection of p onto  $\pi$  as the point on  $\pi$  which realizes the distance d of p from  $\pi$ ).  $\Sigma'$  is an open set of  $\pi$  with  $\partial \Sigma' = \partial \Sigma$ . Proposition 1.3 gives

$$|\kappa_i'(p)| \le \frac{1}{\sqrt{1 - (g_p(N_p, \omega_p))^2}} \max\{|\kappa_1(p)|, |\kappa_{n-1}(p)|\},\$$

for any  $p \in \partial \Sigma$  and i = 1, ..., n - 1, where  $\kappa'_i$  are the principal curvatures of  $\partial \Sigma$  viewed as a hypersurface of  $\pi$ . Since S satisfies a touching ball condition of radius  $\rho$ , we have

$$\max\{|\kappa_1(p)|, |\kappa_{n-1}(p)|\} \le \frac{1}{\rho}$$

and, consequently,

$$|\kappa_i'(p)| \le \frac{1}{\rho \sqrt{1 - (g_p(N_p, \omega_p))^2}},$$
(1.58)

for i = 1, ..., n-1. From (1.57) and (1.58) we have that  $\partial \Sigma'$  satisfies a touching ball condition of radius

$$\rho' \ge \rho \sqrt{1 - (g_p(N_p, \omega_p))^2} \ge t_0.$$

Therefore if  $s < t_0$ ,

$$\mathcal{C}_s = \{ z \in \pi : \ d(z, \partial \Sigma) < s \}$$

is a collar neighbourhood of  $\partial \Sigma$  in  $\Sigma'$  of radius s. Since  $\pi$  is a critical hyperplane in the method of moving planes, if p belongs to the maximal cap  $S_+$  then any point on the geodesic path connecting p to its projection onto  $\pi$  is contained in the closure of  $\Omega$ . It follows that the preimage of  $C_s$  via the projection contains a collar neighbourhood of  $\partial \Sigma$  of radius s in  $\Sigma$ . This implies that  $\Sigma$  can be retracted in  $\Sigma_t$  for any  $t \leq s$  which completes the proof.

**Lemma 1.8.** There exists  $\overline{\delta} > 0$  depending only on  $\rho$  with the following property. Assume that there exists a connected component  $\Gamma_{\delta}$  of  $\Sigma_{\delta}$ , for some  $0 < \delta \leq \overline{\delta}$ , such that one of the following two assumptions is fulfilled:

- i)  $0 \leq g_q(N_q, \omega_q) \leq \frac{1}{8}$  for any  $q \in \partial \Gamma_{\delta}$ ,
- *ii)* for any  $q \in \partial \Gamma_{\delta}$  there exists  $\hat{q} \in \hat{\Sigma}$  such that

$$d(q,\hat{q}) + |N_q - \tau_{\hat{q}}^q(N_{\hat{q}})|_q \le \delta.$$

Then

$$0 \le g_q(N_q, \omega_q) \le \frac{1}{4} \tag{1.59}$$

for any  $q \in \partial \Sigma$  and  $\Sigma_{\delta}$  is connected.

*Proof. Case i*). The crucial observation is that we can choose  $\bar{\delta}$  small enough such that  $\bar{\delta} \leq \delta_0$ , where  $\delta_0$  is the bound appearing in Proposition 1.2, and the set  $\Sigma \setminus \Gamma_{\delta}$  is enclosed by  $\pi$  and the set obtained as the union of all the exterior and interior touching balls to the reflection of S about  $\pi$ ,  $S^{\pi}$ . This implies that for any  $p \in \Sigma \setminus \Gamma_{\delta}$  there exists  $q \in \partial \Gamma_{\delta}$  such that  $d_{\Sigma}(p,q) \leq \delta$  and we can apply the estimates in proposition 1.2. Indeed from (1.22) and (1.23) we have that

$$|N_p - au_q^p(N_q)|_p \le C\delta\,, \quad ext{ and } \quad g_p(N_p, au_q^p(N_q)) \ge \sqrt{1 - C^2\delta^2}\,,$$

where  $C = C(\rho)$ . Therefore

$$g_p(N_p, \omega_p) = g_p(N_p - \tau_q^p(N_q), \omega_p) + g_p(\tau_q^p(N_q), \omega_p)$$
  
$$\leq C\delta + g_p(\tau_q^p(N_q), \omega_p)$$

and by using

$$g_p(\tau_q^p(N_q),\omega_p) = g_q(N_q,\tau_p^q(\omega_q))$$

we obtain

$$g_p(N_p, \omega_p) \le C\delta + g_q(N_q, \omega_q) + g_q(N_q, \tau_p^q(\omega_p) - \omega_q)$$
$$\le C\delta + g_q(N_q, \omega_q) + |\tau_p^q(\omega_p) - \omega_q|_q.$$

Since

$$|\tau_p^q(\omega_p) - \omega_q|_q = 0\,,$$

we deduce

$$g_p(N_p, \omega_p) \le C\delta + g_q(N_q, \omega_q).$$

This last bound holds for every  $p \in \partial \Sigma$  and by choosing  $\delta$  small enough in terms of  $\rho$  we obtain (1.59), as required.

Case ii):  $\Gamma_{\delta}$  satisfies ii). Let  $q \in \partial \Gamma_{\delta}$ . By construction of the method of moving planes,  $g_q(N_q, \omega_q) \geq 0$ . We denote by  $q^{\pi}$  the reflection of q about  $\pi$  and we have

$$d(q^{\pi}, \hat{q}) \le d(q^{\pi}, q) + d(q, \hat{q}) \le 3\delta.$$

Up to consider a smaller  $\delta$  in terms of  $\rho$ , from Corollary 1.2 we find  $C = C(\rho)$  such that  $d_S(q^{\pi}, \hat{q}) \leq C\delta$  and  $q^{\pi} \in \mathcal{U}_{\rho_1}(\hat{q})$ . Hence we can apply (1.22) and obtain

$$g_{\hat{q}}(N_{\hat{q}}, \tau_{q^{\pi}}^{\hat{q}}(N_{q^{\pi}}) \ge \sqrt{1 - C^2 \delta^2} \quad \text{and} \quad |N_{\hat{q}} - \tau_{q^{\pi}}^{\hat{q}}(N_{q^{\pi}})|_{\hat{q}} \le C\delta.$$

Since  $N_{q^\pi}$  and  $q^\pi$  are the reflection of  $N_q$  and q about  $\pi,$  respectively, we have that

$$g_q(N_q,\omega_q) = -g(\tau_{q^{\pi}}^q(N_{q^{\pi}}),\omega_q),$$

and hence

$$2g_q(N_q, \omega_q) = g_q(N_q - \tau_{q^{\pi}}^q(N_{q^{\pi}}), \omega_q) = g_q(N_q - \tau_{\hat{q}}^q(N_{\hat{q}}), \omega_q) + g_q(\tau_{\hat{q}}^q(N_{\hat{q}}) - \tau_{q^{\pi}}^q(N_{q^{\pi}}), \omega_q).$$

This implies that

$$0 \le 2g_q(N_q, \omega_q) \le |N_q - \tau_{\hat{q}}^q(N_{\hat{q}})|_q + |\tau_{\hat{q}}^q(N_{\hat{q}}) - \tau_{q^{\pi}}^q(N_{q^{\pi}})|_q.$$

Next we observe that

$$|\tau_{\hat{q}}^{q}(N_{\hat{q}}) - \tau_{q^{\pi}}^{q}(N_{q^{\pi}})|_{q} = |N_{\hat{q}} - \tau_{q}^{\hat{q}}\tau_{q^{\pi}}^{q}(N_{q^{\pi}})|_{\hat{q}} \le c(\delta)|N_{\hat{q}} - \tau_{q^{\pi}}^{\hat{q}}(N_{q^{\pi}})|_{\hat{q}}$$

where  $c(\delta) \to 0$  when  $\delta \to 0$ . Hence for a suitable choice of  $\overline{\delta}$  we get

$$0 \le 2g_q(N_q, \omega_q) \le \frac{1}{8},$$
 (1.60)

and the claim follows from case i).

Now we can focus on the proof of the first part of Theorem 1.4, and show that there exist constants  $\varepsilon$  and C, depending only on n,  $\rho$  and  $|S|_g$  (the area of S with respect to g), such that if

$$\operatorname{osc}(\mathsf{H}_S) \leq \varepsilon,$$

then for any p in  $\Sigma$  there exists  $\hat{p}$  in  $\hat{\Sigma}$  satisfying

$$d(p,\hat{p}) + |N_p - \tau_{\hat{p}}^p(N_{\hat{p}})|_p \le C \operatorname{osc}(\mathsf{H}_S).$$
(1.61)

In the proof of Theorem 1.4 we are going to choose a number  $\delta > 0$  sufficiently small in terms of  $\rho$ , n and  $|S|_g$ . A first requirement on  $\delta$  is that the assumptions of Lemmas 1.7 and 1.8 are satisfied. Other restrictions on the value of  $\delta$  will be done in the development of the proof. We subdivide the proof of the first part of the statement in four cases depending on the whether the distances of  $p_0$  and p from  $\partial \Sigma$  are greater or less than  $\delta$ .

#### Case 1: $d_{\Sigma}(p_0, \partial \Sigma) > \delta$ and $d_{\Sigma}(p, \partial \Sigma) \ge \delta$ .

In this first case we assume that  $p_0$  and p are interior points of  $\Sigma$ , which are far from  $\partial \Sigma$  more than  $\delta$ . We first assume that  $p_0$  and p are in the same connected component of  $\Sigma_{\delta}$ ; then, Lemma 1.8 will be used in order to show that  $\Sigma_{\delta}$  is in fact connected.

Let  $r_0 > 0$  be such that  $\mathcal{U}_{r_0}(p_i) \subset \Sigma$  for every  $p_i \in \Sigma_{\delta}$ . The value of  $r_0$  follows from (ii), Lemma 1.3 by letting

$$r_0 = \min(\bar{r}, \alpha \rho_1), \qquad (1.62)$$

where  $\bar{r}$  is given by Lemmas 1.5 and 1.6,  $\alpha \in (0, \frac{1}{2}\min(1, \rho_1^{-1}))$  is such that  $\alpha C \rho_1 \leq \frac{\delta}{4}$ , and C is the constant appearing in (1.19).

**Lemma 1.9.** Let  $\varepsilon_0 \in [0, 1/2]$ ,  $p_0$  and p be in a connected component of  $\Sigma_{\delta}$  and  $r_i = (1 - \varepsilon_0^2)^i r_0$ . There exist an integer  $J \leq J_{\delta}$ , where

$$J_{\delta} := \max\left(4, \frac{2^{n-1}|S|_g}{\delta^{n-1}}\right), \qquad (1.63)$$

and a sequence of points  $\{p_1, \ldots, p_J\}$  in  $\Sigma_{\delta/2}$  such that

$$p_0, p \in \bigcup_{i=0}^{J} \overline{\mathcal{U}}_{r_i/4}(p_i),$$
  
$$\mathcal{U}_{r_0}(p_i) \subseteq \Sigma, \quad i = 0, \dots, J,$$
  
$$p_{i+1} \in \overline{\mathcal{U}}_{r_i/4}(p_i), \quad i = 0, \dots, J-1.$$

*Proof.* In view of Corollary 1.2, for every z in  $\Sigma$  and  $r \leq \rho_0$ , the geodesic ball  $\mathcal{B}_r(z)$  in  $\Sigma$  satisfies

$$\operatorname{Area}(\mathcal{B}_r(z)) \ge cr^{n-1}$$

where c is a constant depending only on n. A general result for Riemannian manifolds with boundary (see Proposition A.1) implies that there exists a piecewise geodesic path parametrized by arc length  $\gamma: [0, L] \to \Sigma_{\delta/2}$  connecting  $p_0$  to p and of length L bounded by  $\delta J_{\delta}$ , where  $J_{\delta}$  is given by (1.63).

We define  $p_i = \gamma(r_i/4)$ , for  $i = 1, \ldots, J - 1$  and  $p_J = p$ . Our choice of  $r_0$  guarantees that  $\mathcal{U}_{r_0}(p_i) \subset \Sigma$ , for every  $i = 0, \ldots, J$ , and the other required properties are satisfied by construction.

Since p and  $p_0$  are in a connected component of  $\Sigma_{\delta}$ , there exists a sequence of points  $p_1, \ldots, p_J$  in the connected component of  $\Sigma_{\delta/2}$  containing  $p_0$ , with  $J \in \mathbb{N}$  and  $p_J = p$ , and a chain of subsets  $\{\mathcal{U}_{r_0}(p_i)\}_{\{i=0,\ldots,J\}}$  of  $\Sigma$  as in Lemma 1.9. We notice that  $\Sigma$  and  $\hat{\Sigma}$  are tangent at  $p_0$  and that in particular the two normal vectors to  $\Sigma$  and  $\hat{\Sigma}$  at  $p_0$  coincide. Now we apply the map  $\varphi_{p_0}$  (see section 1.2). Then  $\varphi_{p_0}(\Sigma)$  and  $\varphi_{p_0}(\hat{\Sigma})$  can be locally parametrized near  $\varphi_{p_0}(p_0)$ as graphs of two functions  $u_0, \hat{u}_0 \colon B_{r_0} \subset \{x_n = 0\} \to \mathbb{R}$ . Lemma 1.3 implies that  $|\nabla u_0|, |\nabla \hat{u}_0| \leq M$  in  $B_{r_0}$ , where M is some constant which depends only on  $r_0$ , i.e. only on  $\rho$ . Hence the difference  $u_0 - \hat{u}_0$  solves a second-order linear uniformly elliptic equation of the form

$$\mathcal{L}(u_0 - \hat{u}_0)(x) = \mathsf{H}_{\mathrm{Graph}(u_0)}(x, u_0(x)) - \mathsf{H}_{\mathrm{Graph}(\hat{u}_0)}(x, \hat{u}_0(x))$$

with ellipticity constants uniformly bounded by a constant depending only on n and  $\rho$ . Since  $u_0(0) = \hat{u}_0(0)$  and  $u_0 \ge \hat{u}_0$ , Harnack's inequality (see Theorems 8.17 and 8.18 in [112]) yields

$$\sup_{B_{r_0/2}} \left( u_0 - \hat{u}_0 \right) \le C \operatorname{osc}(\mathsf{H}_S) \,,$$

and from interior regularity estimates (see e.g. [112, Theorem 8.32]) we obtain

$$\|u_0 - \hat{u}_0\|_{C^1(B_{\tau_0/4})} \le C \operatorname{osc}(\mathsf{H}_S), \tag{1.64}$$

where C depends only on  $\rho$  and n. Now we use Lemmas 1.5 and 1.6. Since  $p_1 \in \partial \mathcal{U}_{r_0/4}(p_0)$ , we can write  $\varphi_{p_0}(p_1) = (x_1, u_0(x_1))$ , with  $x_1 \in \partial B_{r_0/4}$ . Let  $\hat{p}_1^* \in \hat{\Sigma}$  be such that

$$\varphi_{p_0}(\hat{p}_1^*) = (x_1, \hat{u}(x_1)),$$

and let  $\hat{p}_1$  be the first intersection point between  $\hat{\Sigma}$  and the geodesic path  $\gamma$  starting from  $p_1$  and tangent to  $-N_{p_1}$  at  $p_1$ . From (1.64) we have

$$d(\varphi_{p_0}(p_1), \varphi_{p_0}(\hat{p}_1^*)) + |\nu_{\varphi_{p_0}(p_1)} - \nu_{\varphi_{p_0}(\hat{p}_1^*)}| \le C \operatorname{osc}(\mathsf{H}_S), \qquad (1.65)$$

which implies that the assumptions in Lemmas 1.5 and 1.6 are fulfilled, and we obtain

$$d(p_1, \hat{p}_1) + |N_{p_1} - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1} \le C \operatorname{osc}(\mathsf{H}_S), \qquad (1.66)$$

where C depends only on n and  $\rho$ .

Now we apply  $\varphi_{p_1}$ . By definition of  $\varphi_{p_1}$ , we have  $\varphi_{p_1}(\hat{p}_1) = te_n$  for some  $t \in \mathbb{R}$  (t depends on the geometry of the ambient space). A standard computation yields

$$|\nu_{\varphi_{p_1}(p_1)} - \nu_{\varphi_{p_1}(\hat{p}_1)}| = |N_{p_1} - \tau_{\hat{p}_1}^{p_1}(N_{\hat{p}_1})|_{p_1}$$

which in view of (1.66) implies

$$|\nu_{p_1} - \nu_{\hat{p}_1}| \le Cosc(\mathsf{H}_{\mathrm{S}})\,,$$

where C is a constant that depends only on  $\rho$  and n. Since  $\operatorname{osc}(\mathsf{H}_{\mathrm{S}}) \leq \varepsilon$  then  $|\nu_{p_1} - \nu_{\hat{p}_1}| < C\varepsilon$  and by choosing  $\varepsilon$  such that  $C\varepsilon < 1$ , we can use the implicit function theorem and obtain that  $\Sigma$  and  $\hat{\Sigma}$  are locally graphs of two functions

$$u_1, \hat{u}_1 : B_{r_1} \to \mathbb{R}^+,$$

such that  $u_1(0) = \varphi_{p_1}(p_1)$  and  $\hat{u}_1(0) = \varphi_{p_1}(\hat{p}_1)$ . Since both  $\Sigma$  and  $\hat{\Sigma}$  satisfy a touching ball condition of radius  $\rho_1$  then  $r_1 < \rho_1$ ; in particular one can show that

$$r_1 = (1 - C^2 \varepsilon^2) \rho < \rho_1$$
.

Now, we can iterate the argument we did before. Indeed, since

$$0 \le \inf_{B_{r_1/2}} (u_1 - \hat{u}_1) \le u_1(0) - \hat{u}_1(0) \le Cosc(\mathsf{H}_{\mathrm{S}}),$$

by applying Harnack's inequality we obtain that

$$\sup_{B_{r_1/2}} \left( u_1 - \hat{u}_1 \right) \le C \operatorname{osc}(\mathsf{H}_{\mathrm{S}})$$

and from interior regularity estimates we find

$$\|u_1 - \hat{u}_1\|_{C^1(B_{r_1/4})} \le C \operatorname{osc}(\mathsf{H}_{\mathrm{S}}), \qquad (1.67)$$

where C depends only on  $\rho$  and n. Hence, (1.67) is the analogue of (1.64), and we can iterate the argument. The iteration goes on until we arrive at  $p_J = p$ and obtain a point  $\hat{p}_J \in \hat{\Sigma}$  such that

$$d(p, \hat{p}_J) + |N_p - \tau^p_{\hat{p}_J}(N_{\hat{p}_J})|_p \le Cosc(\mathsf{H}_S).$$

In view of Lemma 1.8 we have that  $\Sigma_{\delta}$  is connected and the claim follows.

Case 2:  $d_{\Sigma}(p_0, \partial \Sigma) \ge \delta$  and  $d_{\Sigma}(p, \partial \Sigma) < \delta$ .

We extend the estimates found in case 1 to a point p which is far less than  $\delta$  from the boundary of  $\Sigma$ . Let  $q \in \Sigma$  and  $p_{min} \in \partial \Sigma$  be such that

$$d_{\Sigma}(q,\partial\Sigma) = \delta \,, \quad d_{\Sigma}(p,q) + d_{\Sigma}(p,\partial\Sigma) = \delta \,, \quad \text{and} \quad d_{\Sigma}(p,p_{min}) = d_{\Sigma}(p,\partial\Sigma) \,.$$

From case 1 we have that there exists  $\hat{q}$  in  $\hat{\Sigma}$  such that

$$d(q,\hat{q}) + |N_q - \tau^q_{\hat{q}}(N_{\hat{q}})|_q \le C \operatorname{osc}(\mathsf{H}_S).$$

Lemma 1.8 (case (ii)) yields that

$$0 \le g_z(N_z, \omega_z) \le \frac{1}{4},\tag{1.68}$$

for any  $z \in \partial \Sigma$  and  $\Sigma_{\delta}$  is connected.

Let  $q^{\pi} \in S$  be the reflection of q about  $\pi$  and fix  $r \leq \rho_1$  in order to define  $\mathcal{U}_r(q^{\pi})$ . We denote by  $U_r(q)$  the reflection of  $\mathcal{U}_r(q^{\pi}) \cap S$  about  $\pi$  and U' =

 $\mathcal{U}_r(q^{\pi}) \cap \pi$ . Proposition 1.3 implies that U' is a hypersurface of  $\pi$  with an induced orientation and its principal curvatures  $\kappa'_i$  satisfy the following bounds

$$\frac{1}{\sqrt{1 - g_z(N_z, \omega_z)^2}} \kappa_1(z) \le \kappa_i'(z) \le \frac{1}{\sqrt{1 - g_z(N_z, \omega_z)^2}} \kappa_{n-1}(z),$$

for every  $z \in U'$  and i = 1, ..., n - 1. From (1.68) and since  $|\kappa_i(z)| \leq \rho^{-1}$  for any  $z \in S$  (this follows from the touching ball condition), we have

$$|\kappa_i'(z)| \le \frac{2}{\rho},\tag{1.69}$$

for any  $z \in U'$ . Let U'' be the Euclidean orthogonal projection of  $\varphi_q(U')$  onto  $\{x_n = 0\}$ . In order to apply Carleson estimates in [26, Theorem 1.3], we need to prove the following

**Lemma 1.10.** Let  $\{\kappa''_1, \ldots, \kappa''_{n-2}\}$  be the Euclidean principal curvature of U'' viewed as a hypersurface of  $\mathbb{R}^{n-1}$ . Then

$$\|\kappa_i''\|_{\infty} \le C, \quad i = 1, \dots, n-2,$$
 (1.70)

for some constant  $C = C(\rho)$ .

*Proof.* Here we use the same notation as in Section 1.3 and we analyse each case separately.

•  $M_+^n = \mathbb{R}^n$ . Up to apply  $\varphi_q$  we may assume that  $\varphi_q(q) = 0$  and that the normal vector to  $\varphi_q(S)$  at  $\varphi_q(q)$  is  $e_n$ . Hence Proposition 1.4 (actually formula (1.53)) and (1.69) yield, for every  $i = 1, \ldots, n-2$ ,

$$|\kappa_i''(\bar{z})| \le \frac{2}{\rho} \frac{|\omega \cdot e_n|}{[(\omega \cdot e_n)^2 + (e_n \cdot \nu_z')^2]^{3/2}} \le \frac{2}{\rho} \frac{|\omega \cdot e_n|}{(e_n \cdot \nu_z')^3},$$
(1.71)

for every  $z = (\bar{z}, z_n) \in \varphi_q(U')$ . Now we estimate  $e_n \cdot \nu'_z$  in the following way: by writing

$$e_n \cdot \nu'_z = (e_n - \nu_z) \cdot \nu'_z + \nu_z \cdot \nu'_z$$

exploiting (1.45) and from (1.59) we get

$$\nu_z \cdot \nu'_z = \sqrt{1 - (\omega \cdot \nu_z)^2} \ge \frac{1}{2}.$$
 (1.72)

Moreover, from (1.20) we get

$$|e_n - \nu_z| \le \frac{1}{4} \,, \tag{1.73}$$

where we use the fact that  $\bar{z} \in U''$  where U'' is the orthogonal projection of  $\varphi_q(U')$  and  $U' = \mathcal{U}_r(q^{\pi}) \cap \pi$  with  $r \leq \rho_1 = \rho$ . So, from (1.72) and (1.73) we obtain

$$e_n \cdot \nu_z' \ge \frac{1}{2} \,. \tag{1.74}$$

The conclusion now follows from (1.74), (1.68) and (1.71).

•  $M_+^n = \mathbb{H}^n$ . Up to apply the isometry  $\varphi_q$  we may assume that  $\varphi_q(q) = e_n$ and the normal vector to  $\varphi_q(S)$  at  $\varphi_q(q)$  is  $e_n$ . It is clear that  $\varphi_q(\pi)$  is either a vertical plane or a half-sphere intersecting  $\varphi_q(S)$ . In the first case we immediately conclude since the curvatures of U'' vanish. Thus, we may assume that  $\varphi_q(\pi)$  is a half-sphere. A straightforward computation yields that the radius of  $\varphi_q(\pi)$  is

$$R = \frac{q_n(\Theta^2 + 1)}{2|\Theta||a\Theta + q_n|},$$

where

$$\Theta = -\frac{\sin(\theta)}{1 + \cos(\theta)}, \quad \cos(\theta) = \nu_q \cdot e_n$$

and a is the Euclidean distance of q from  $\pi$ . It is easy to see that

$$a \le q_n \sinh(\delta)$$

and so

$$\frac{1}{R} \leq \frac{2|\Theta|(\sinh(\delta)|\Theta|+1)}{\Theta^2 + 1}$$

which implies

$$\frac{1}{R} \le 1 + 2\sinh(\delta) \,. \tag{1.75}$$

Now we show (1.70). Proposition 1.4 (actually formula (1.55)), (1.69) (1.75) imply

$$\begin{aligned} |\kappa_i''(\bar{z})| &\leq \frac{1}{R} \left( (\nu_z' \cdot e_n)^2 + \frac{z_n^2}{R^2} \right)^{-3/2} \frac{4(1+\rho)}{\rho} \\ &\leq \frac{4(1+\rho)(1+2\sinh(\delta))}{\rho |\nu_z' \cdot e_n|^3} \,, \end{aligned} \tag{1.76}$$

for every  $z = (\bar{z}, z_n) \in \varphi_q(U')$  and  $i = 1, \ldots, n-2$ . Now we show a lower bound on  $\nu'_z \cdot e_n$ . We write

$$\nu'_{z} \cdot e_{n} = \nu'_{z} \cdot (e_{n} - \nu_{z}) + \nu'_{z} \cdot \nu_{z} = \nu'_{z} \cdot (e_{n} - \nu_{z}) + g_{z}(N'_{z}, N_{z}).$$
(1.77)

From (1.43) we get

$$g_z(N_z',N_z)=\sqrt{1-g_z(\omega_z,N_z)^2}$$

and from (1.59) we obtain

$$g_z(N'_z, N_z) \ge \frac{1}{2} \,.$$

Since  $\varphi_{q*|q}(\nu_q) = e_n$  from (1.20) we have that  $|e_n - \nu_z| \leq 1/4$ , by choosing r small enough in terms of  $\rho_1$ . Hence

$$\nu'_{z} \cdot e_{n} = \nu'_{z} \cdot (e_{n} - \nu_{z}) + g_{z}(N'_{z}, N_{z}) \ge \frac{1}{2} - |\nu'_{z} \cdot (e_{n} - \nu_{z})| \ge \frac{1}{4}.$$
 (1.78)

Therefore (1.70) follows from (1.76) and (1.78).

•  $M_+^n = \mathbb{S}_+^n$ . If we apply the stereographic projection  $\varphi_q$  then it is clear that  $\varphi_q(\pi)$  is a hemisphere in  $\mathbb{R}^n$  of center  $O_{\pi}$  and radius R. We notice that the radius of  $\varphi_q(\pi)$  is given by

$$R = \frac{1}{2a} + \frac{a}{2} \,,$$

where a is the Euclidean distance between  $\varphi_q(\pi)$  and the origin of  $\mathbb{R}^n$ . This follows from the proof of lemma 1.6: indeed, up to apply a rotation, we may assume that the point on  $\pi$  having minimal Euclidean distance from the origin is  $(a, 0, \ldots, 0)$ , with a > 0, and that the normal to  $\pi$  in  $(a, 0, \ldots, 0)$  is  $e_1$ . From a straightforward calculation we obtain the value of R. Since  $d(q, \pi) < \delta$  we have  $a < \delta$  which implies

$$R > \frac{1}{2\delta} \,. \tag{1.79}$$

Since  $U' \subset \mathcal{U}_r(q)$  and  $\varphi_q(q) = O$ , our choice of r implies that for any  $z \in \varphi_q(U')$  we have

$$|z| \le \rho_1 = \frac{\rho}{\pi} \tag{1.80}$$

 $\operatorname{and}$ 

$$|z - O_{\pi}| \le |z - q| + |q - O_{\pi}| \le \delta + R \le 2R.$$
(1.81)

Now we show (1.70). Proposition 1.4 (actually formula (1.56)), (1.79), (1.80) and (1.81) yield

$$|\kappa_i''(\bar{z})| \le 8(\nu_z' \cdot e_n)^{-3} \left( \max\{|\kappa_1'(z)|, |\kappa_{n-1}'(z)|\} + 4\rho_1 \right), \qquad (1.82)$$

for every  $z = (\overline{z}, z_n) \in \varphi_q(U')$  and for every  $i = 1, \ldots, n-2$ .

Now, as in the hyperbolic case, we give a lower bound of  $\nu'_z \cdot e_n$ . We can write (1.77) and arguing as before we obtain (1.78) (here we can change configuration by considering the stereographic projection from the antipodal point to  $\varphi_q^{-1}(z)$  in order to regard  $\pi$  as a vertical hyperplane of  $\mathbb{R}^n$ ). Also in this case (1.70) follows from (1.82), (1.78) and (1.69).

We denote by x and  $E_r$  the projections of  $\varphi_q(p_{min})$  and  $\varphi_q(U_r(q))$  onto  $\{x_n = 0\}$ , respectively. The Euclidean distance of x from  $\partial E_r$  is less than  $C\delta$  where C depends only on  $\rho$  and, up to chose a smaller  $\delta$  in terms of  $\rho$ , the projection of  $p_{min}$  stays close to  $U'' \subset \partial E_r$  and we can apply Theorem 1.3 in [26], Corollary 8.36 in [112] and Harnack's inequality (see e.g. [112, Corollary 8.36]) to obtain

$$\sup_{B_{2\delta C}(x)\cap E_{r}} (u-\hat{u}) \le C(u-\hat{u})(z) + \operatorname{osc}(\mathsf{H}_{S})$$
(1.83)

with  $z = x + 4C\delta\nu''_x$ , where  $\nu''_x$  is the interior normal to U" at x. Thanks to (1.70) and by choosing  $\delta$  small enough in terms of  $\rho$ , from (1.83) and Harnack's inequality we obtain

$$0 \le \|u - \hat{u}\|_{C^1(B_C\delta(x) \cap E_r)} \le C((u(0) - \hat{u}(0)) + \operatorname{osc}(\mathsf{H}_S)).$$
(1.84)

Since  $d_{\Sigma}(q, \partial \Sigma) = \delta$ , from Case 1 we know that

$$d(q,\hat{q}) + |N_q - \tau_{\hat{q}}^q(N_{\hat{q}})|_q \le \operatorname{Cosc}(\mathsf{H}_{\mathrm{S}}),$$

and from (1.84) we obtain that

$$0 \le \|u - \hat{u}\|_{C^1(B_{C\delta}(x) \cap E_r)} \le Cosc(\mathsf{H}_{\mathrm{S}}).$$
(1.85)

From Lemmas 1.5 and 1.6 we deduce

$$d(p,\hat{p}) + |N_p - \tau_{\hat{p}}^p(N_{\hat{p}})|_p \le \operatorname{Cosc}(\mathsf{H}_{\mathrm{S}}),$$

as required.

Case 3:  $0 < d_{\Sigma}(p_0, \partial \Sigma) < \delta$ .

We first prove the following preliminary lemma which implies via Lemma 1.7 that  $\Sigma$  is connected.

**Lemma 1.11.** By choosing  $\delta$  small enough in terms of  $\rho$ , the following inequality holds

$$0 \le g_{p_0}(N_{p_0}, \omega_{p_0}) \le \frac{1}{4}.$$
(1.86)

*Proof.* We first prove the statement in the Euclidean case: in this setting (1.86) reads as follows

$$0 \le \nu_{p_0} \cdot \omega \le \frac{1}{4} \,. \tag{1.87}$$

Since  $p_0$  is the tangency point, it is easy to show that the center of the interior touching sphere of radius  $\rho$  to S at  $p_0$  lies in the half-space  $\{q \in \mathbb{R}^n : q \cdot \omega \leq m\}$  (see e.g. [61, Lemma 2.1]). From this and since

$$|p_0 \cdot \omega - m| \le d_{\Sigma}(p_0, \partial \Sigma) < \delta$$

we obtain (1.87). Indeed let  $p_0^{\pi}$  be the reflection of  $p_0$  about  $\pi$  and let

$$t = \nu_{p_0} \cdot \omega \,.$$

By construction of the moving planes, it is clear that  $t \ge 0$  and the first inequality in (1.87) follows. We denote by  $\nu_{p_0^{\pi}}$  the inner normal vector to S at  $p_0^{\pi}$ . Since  $\nu_{p_0} \cdot \omega = -\nu_{p_0^{\pi}} \cdot \omega$  and  $\nu_{p_0} - \nu_{p_0^{\pi}} = 2t\omega$ , we have

$$\nu_{p_0} \cdot \nu_{p_0^{\pi}} = 1 - 2t^2 \,.$$

We notice that  $p_0^{\pi}$  and  $p_0$  both lie in S and  $|p_0^{\pi} - p_0| \leq 2\delta$ , which implies that  $p_0 \in \mathcal{U}_{\rho}(p_0^{\pi})$ , provided that  $2\delta < \rho$ . Hence, (1.20) yields

$$\nu_{p_0} \cdot \nu_{p_0^{\pi}} \ge \sqrt{1 - \frac{4\delta^2}{\rho^2}},$$

i.e.

$$1 - 2t^2 \ge \sqrt{1 - \frac{4\delta^2}{\rho^2}};$$

and by choosing  $\delta$  small enough in terms of  $\rho$  we obtain the second inequality in (1.87).

Now we show how to deduce, from the Eulidean case, the claim in the hyperbolic and in the spherical case. We first consider the Hyperbolic case. Up to apply an isometry we can assume that  $p_0 = e_n$  and  $\pi = \{x_1 = 0\}$ . Our assumptions on S imply that its diameter is bounded in terms of  $\rho$  and  $|S|_g$  (see Proposition A.2 in Appendix A). Therefore S is contained in an Euclidean ball about the origin and of radius depending only on  $\rho$  and  $|S|_g$ . Up to choose  $\delta$  small enough in terms of  $\rho$ , we have that (1.87) holds, i.e.

$$0 \le \nu_{p_0} \cdot e_1 \le \frac{1}{4} \,.$$

Since  $\nu_{p_0} \cdot e_1 = g_{p_0}(N_{p_0}, \omega_{p_0})$ , the claim follows. In the spherical case the proof is analogue once the setting is modified as follows: we work in  $(\mathbb{R}^n, g)$ , where g is the round metric (1.7), assuming that  $p_0 = O$  and  $\pi$  is an Euclidean hyperplane.

Then we prove the existence of a point  $q \in \Sigma$  such that

$$\begin{cases} d(q, \hat{q}) + |N_q - \tau_{\hat{q}}^q(N_{\hat{q}})|_q \le \operatorname{Cosc}(\mathsf{H}_{\mathrm{S}}) \\ d_{\Sigma}(q, \partial \Sigma) \ge \delta \end{cases}$$
(1.88)

and we apply cases 1 and 2 to conclude.

In the same fashion as in case 2, we can locally write  $\varphi_{p_0}(\Sigma)$  and  $\varphi_{p_0}(\Sigma)$ as graphs of function  $u, \hat{u}: E_r \to \mathbb{R}$  near  $\varphi_{p_0}(p_0)$ , respectively. Without loss of generality we can assume r < 1 (indeed r must be chosen small enough in terms of  $\delta$ ). Let  $U'' \subset \partial E_r$  be the projection of  $\varphi_{p_0}(U_r(p_0) \cap \pi)$  onto  $\{x_n = 0\}$ . Analogously to case 2, the Euclidean principal curvatures of U'' are bounded by a constant  $\mathcal{K}$  depending only on  $\rho$ . Then let  $\bar{x} \in U''$  be a point such that

$$|\bar{x}| = \min_{x \in U^{\prime\prime}} |x|.$$

Notice that  $|\bar{x}| \leq c_* d_{\Sigma}(p_0, \partial \Sigma) < c_* \delta$ , where  $c_*$  is 1 in the Euclidean and in the spherical case, while it is the constant c appearing in (1.10) in the hyperbolic case. Let  $\nu''_x$  be the interior normal to U'' at  $\bar{x}$ , and set

$$y = \bar{x} + 2c_* \delta \nu_{\bar{x}}''$$

(see Figure 1.4) By choosing  $\delta$  sufficiently small in terms of  $\rho$ , we have  $2c_*\delta \leq \mathcal{K}^{-1}$  and the ball  $B_{2c_*\delta}(y)$  is contained in  $E_r$  and it is tangent to U'' at  $\bar{x}$ , with  $\nu_{\bar{x}} = -\bar{x}/|\bar{x}|$ . Since  $u(0) = \hat{u}(0)$  from Harnack's inequality and from interior regularity estimates we find that

$$||u - \hat{u}||_{C^1(B_{C\delta/2}(y))} \le Cosc(\mathsf{H}_{\mathrm{S}})$$

which implies that

$$d(w, \hat{w}^*) + |\nu_w - \nu_{\hat{w}^*}| \le Cosc(\mathsf{H}_{\mathrm{S}})$$

where w = (y, u(y)) and  $\hat{w}^* = (y, \hat{u}(y))$ . Up to choose a smaller  $\delta$ , we can assume that  $2c_*\delta \leq \bar{r}$ , so that Lemmas 1.5 and 1.6 yield

$$d(q,\hat{q}) + |N_q - \tau_{\hat{q}}^q(N_{\hat{q}})|_q \le \operatorname{Cosc}(\mathsf{H}_{\mathrm{S}}),$$

where  $q = \varphi_{p_0}^{-1}(w)$ ,  $\hat{q}^* = \varphi_{p_0}^{-1}(\hat{w}^*)$  and  $\hat{q}$  the first intersection point between  $\hat{\Sigma}$  and the geodesic path starting from q and tangent to  $-N_q$  at q.

Let z be a point on  $\partial U_r(p_0)$  realizing  $d(q, \partial U_r(p_0))$ . By construction and from Lemma 1.2 we have

$$d_{\Sigma}(q,\partial\Sigma) \ge d(q,z) \ge c_* |\varphi_{p_0}(q) - \varphi_{p_0}(z)| \ge 2\delta.$$

Since  $d_{\Sigma}(q, \partial \Sigma) \geq \delta$  then q satisfies (1.88) and the claim follows.



Figure 1.4: Case 3 in the proof of Theorem 1.4.

Case 4:  $p_0 \in \partial \Sigma$ .

This is the limit configuration of case 3 when  $d_{\Sigma}(p_0, \partial \Sigma) \to 0$ . Indeed, here  $E_r$  is a half-ball in  $\mathbb{R}^n$  and the argument used in case 3 can be easily adapted. This completes the proof of the first part of Theorem 1.4.

Last step:  $d(x, \Sigma \cup \Sigma^{\pi}) \leq Cosc(\mathsf{H}_{\mathrm{S}})$  for every  $x \in \Omega$ .

Assume by contradiction that

$$d(x, \Sigma \cup \Sigma^{\pi}) > Cosc(\mathsf{H}_{S})$$

for some x in  $\Omega$ . Since  $\Omega$  is connected, it is possible to find  $y \in \Omega$ , such that

 $y \in \Omega_{-}$  and  $Cosc(\mathsf{H}_{S}) < d(y, \Sigma) \le 2Cosc(\mathsf{H}_{S})$ ,

where

$$\Omega_{+} = \{ p \in \Omega : p \in \pi_{v,t} \text{ for some } t > m_{v} \},\$$
  
$$\Omega_{-} = \{ p \in \Omega : p \in \pi_{v,t} \text{ for some } t < m_{v} \}.$$

Let p be a projection of y over  $\Sigma \cup \Sigma^{\pi}$ . If  $p \in \Omega_+$ , then y belongs to the exterior touching ball of S at p, which gives a contradiction. The same contradiction is obtained when  $p \in \pi$  since, in that case  $g_p(N_p, \omega_p) \leq 1/4$ . If  $p \in \Omega_-$ , we can

find a point  $\hat{p} \in S$  such that  $\hat{p}$  and p lies on the geodesic  $\gamma$  starting from p and orthogonal to  $\Sigma^{\pi}$  and such that

$$d(p, \hat{p}) + |N_p - \tau_{\hat{p}}^p(N_{\hat{p}})| \le Cosc(\mathsf{H}_{\mathrm{S}}).$$

By the smallness of  $osc(H_S)$  we obtain that y belongs to the exterior touching ball of S at p, which is a contradiction.

## 1.5 Global approximate symmetry

From the previous section we have that if a  $C^2$ -regular closed hypersurface  $S = \partial \Omega$  embedded in  $\mathbb{M}^n_+$  satisfies the assumptions of Theorem 1.4 then it is almost symmetric with respect to any direction, with the almost symmetry quantified by the deficit  $\operatorname{osc}(\mathsf{H}_S)$ . In this section we show how this result leads to the almost radial symmetry of S. Such procedure is not peculiar of the kind of deficit considered, but it can be applied whenever one has the approximate symmetry in any direction with respect to some deficit. More precisely we consider the following

**Definition 1.1.** Let  $\Upsilon$  be the space of open sets  $\Omega$  in  $\mathbb{M}^n_+$  whose boundary is a  $C^2$ -regular connected closed embedded hypersurface, with the topology induced by the Hausdorff distance. A *deficit function* is any continuous function  $\operatorname{def}(\Omega) : \Upsilon \to [0, +\infty)$  such that  $\operatorname{def}(\Omega) = 0$  if and only if  $\Omega$  is a ball.

Form now on we fix a deficit function def.

**Definition 1.2.** We say that a bounded open set  $\Omega$  satisfies the *approximate* symmetry property (ASP) if there exists a constant  $\mathcal{K} > 0$  satisfying the following condition: for every direction v there exists a connected component  $\Sigma$  of the maximal cap in the direction v such that

$$d(p, \Sigma \cup \Sigma^{\pi_v}) \le \mathcal{K} \operatorname{def}(\Omega),$$

for every  $p \in \Omega$ .

The main theorem in this section is the following

**Theorem 1.5.** Let  $S = \partial \Omega$  be a  $C^2$ -regular closed hypersurface embedded in  $\mathbb{M}^n_+$ , with  $\Omega$  satisfying (ASP) and

$$\operatorname{def}(\Omega) \le \frac{|\Omega|_g}{4\mathcal{K}} \,. \tag{1.89}$$

There exist  $\mathcal{O}$  in  $\mathbb{M}^n_+$  and two balls  $B^d_r(\mathcal{O})$  and  $B^d_R(\mathcal{O})$  centred at  $\mathcal{O}$  of radius r and R, respectively, with  $r \leq R$ , such that

$$B_r^d(\mathcal{O}) \subseteq \Omega \subseteq B_R^d(\mathcal{O})$$

and

$$R - r \le C \operatorname{def}(\Omega) \,, \tag{1.90}$$

where C depends on  $n, \rho, |S|_q$  and  $\mathcal{K}$ .

The following lemma is needed in order to prove Theorem 1.5; here we show that if def( $\Omega$ ) is small enough, then  $\pi_v$  is close to a point  $\mathcal{O}$  (we will call it the *approximate center of symmetry*), for every direction v.

**Lemma 1.12.** Let  $S = \partial \Omega$  be a  $C^2$ -regular closed hypersurface embedded in  $\mathbb{M}^n_+$ , with  $\Omega$  satisfying (ASP) and (1.89). Then there exists  $\mathcal{O}$  in  $\mathbb{M}^n_+$  such that

$$d(\mathcal{O}, \pi_v) \le C \mathrm{def}(\Omega) \,, \tag{1.91}$$

for every direction v in  $T_{o}\mathbb{M}^{n}$ , where C depends on  $n, \rho, |S|_{q}$  and  $\mathcal{K}$ .

Before proving it, we observe that: given a direction v, let  $\Omega_v$  be the corresponding maximal cap. If  $\Omega$  satisfies (ASP) then we have

$$|\Omega_v|_g \ge \frac{|\Omega|_g}{2} - C \operatorname{def}(\Omega), \qquad (1.92)$$

for some constant C depending only on  $n, \rho, |S|_q$  and  $\mathcal{K}$ . Moreover we have

$$|\Omega \triangle \Omega^{\pi}|_{g} = 2(|\Omega|_{g} - 2|\Omega_{v}|_{g}) \le 4C \operatorname{def}(\Omega), \qquad (1.93)$$

where  $\Omega \triangle \Omega^{\pi}$  denotes the symmetric difference between  $\Omega$  and  $\Omega^{\pi}$  and, we recall that  $\Omega^{\pi}$  denotes the reflection of  $\Omega$  about the critical hyperplane  $\pi$  in the direction v.

Proof of Lemma 1.12. We fix an orthonormal basis  $\{e_1, \ldots, e_n\}$  of the tangent space at the "origin" **o** and we consider the corresponding critical hyperplanes  $\pi_{e_i}$ . We define an approximate center of symmetry  $\mathcal{O}$  as follows:

$$\mathcal{O} := \bigcap_{i=1}^n \pi_{e_i} \,.$$

We notice that in the Euclidean case  $\mathcal{O}$  is well-defined. In  $\mathbb{S}_{+}^{n}$  the existence of  $\mathcal{O}$  is always guaranteed. Indeed every  $\pi_{e_{i}}$  is given by the intersection of a plane  $\Pi_{e_{i}}$  of  $\mathbb{R}^{n+1}$  with  $\mathbb{S}_{+}^{n}$  and the intersection of all the  $\Pi_{e_{i}}$ 's is a straight line r which, by construction, can not lie in the plane  $\{x_{n+1} = 0\}$ ; hence  $\mathcal{O} = r \cap \mathbb{S}_{+}^{n} \neq \emptyset$ . Although in the hyperbolic space n orthogonal hyperplanes do not always intersect, but we show that (1.89) implies the existence of  $\mathcal{O}$ . Indeed it is enough to show that

$$\pi_{e_i} \cap \pi_{e_j} \neq \emptyset$$
 for every  $i, j = 1, \dots, n.$  (1.94)

For simplicity, we may assume that  $e_n \in S$ . To simplify the notation we set, accordingly to the notation introduced in Section 1.1,

$$\pi_k^s = \pi_{e_k, m_{e_k} + s}$$
 for  $k \in \{1, \dots, n\}$  and  $s \in \mathbb{R}$ ,

so that the critical hyperplane in the direction  $e_k$  is denoted by  $\pi_k^0$ . We prove (1.94) by contradiction. Assume that  $\pi_i^0 \cap \pi_j^0 = \emptyset$  for some  $i \neq j$ . Then  $\pi_i^0$  and  $\pi_j^0$  divide  $\Omega$  in three disjoint sets which we denote by  $\Omega_1$ ,  $\Omega_2$  and  $\Omega_3$  and we may assume that  $\Omega_1$  is the maximal cap in the direction  $e_i$  and  $\Omega_1 \cup \Omega_2$  is the maximal cap in the direction  $e_i$  (see figure 1.5).



Figure 1.5: a picture of the proof of (1.94) in  $\mathbb{H}^2$ . Here  $e_j = e_1$  and  $e_i = e_2$ .

Moreover, in view of (1.92) we have that

$$|\Omega_1|_g \geq \frac{|\Omega|_g}{2} - C \mathrm{def}(\Omega)\,, \quad \text{ and } \quad |\Omega_1|_g + |\Omega_2|_g \geq \frac{|\Omega|_g}{2} - C \mathrm{def}(\Omega)\,.$$

From this, and since the reflection  $\Omega_1$  about  $\pi_i^0$  is contained in  $\Omega_2 \cup \Omega_3$  and the reflection of  $\Omega_1 \cup \Omega_2$  about  $\pi_j^0$  is contained in  $\Omega_3$  we have that

$$|\Omega_2|_g \le 2C \mathrm{def}(\Omega) \,.$$

We notice that for every k = 1, ..., n, we have that  $\pi_k^{s+t}$  and  $\pi_k^{s-t}$  are two connected components of the set of points which are far t from  $\pi_k^t$ . We define

$$\ell := \min\{d(\pi_i^0 \cap \Omega, \pi_j^0 \cap \Omega) : i, j = 1, \dots, n \text{ and } i \neq j\}.$$

Since  $\pi_i^0$  and  $\pi_j^0$  do not intersect  $S \subset B^d_{\operatorname{diam}(S)}(e_n)$ , we have that  $\ell > 0$  and Proposition A.2 implies that  $\ell$  depends only on  $n, \rho$  and  $|S|_g$ . Therefore

$$\Omega_2 \supseteq \mathcal{E}_1 := \bigcup_{s \in (0,l)} \Omega \cap \pi_j^s \,,$$

and hence  $|\mathcal{E}_1|_g \leq 2C \operatorname{def}(\Omega)$ . By reflecting  $\mathcal{E}_1$  about  $\pi_i^0$  we obtain that most of the mass of  $\Omega_1$  must be at distance more than  $\ell$  from  $\pi_i^0$ , i.e.

$$|\Omega_{e_i,\ell}|_g \ge \frac{|\Omega|_g}{2} - C \operatorname{def}(\Omega),$$

where

$$\Omega_{e_i,\ell} := \bigcup_{s \in (\ell, +\infty)} \Omega \cap \pi_i^s \, .$$

Since  $d(\Omega_{e_i,\ell}, \pi_j^0 \cap \Omega) \ge 2\ell$ , we have that most of the mass of  $\Omega_3$  is at distance  $2\ell$  from  $\pi_j^0$ . This implies that the set

$$\mathcal{E}_2 := \bigcup_{s \in (-2\ell,\ell)} \Omega \cap \pi_i^s \,,$$
is such that  $|\mathcal{E}_2|_g \leq 4C \operatorname{def}(\Omega)$ . By iterating the argument above we find  $m \in \mathbb{N}$  such that  $m\ell > \operatorname{diam}(S)$  and

$$0 = |\Omega_{e_i,m\ell}|_g \ge \frac{|\Omega|_g}{2} - (m+1)C\operatorname{def}(\Omega).$$

This leads to a contradiction provided that  $Cdef(\Omega)$  is small in terms of n,  $\rho$  and  $|S|_g$ . Therefore (1.94) holds true and the point  $\mathcal{O}$  is well-defined also in the hyperbolic case.

Now we prove (1.91). Let  $\mathcal{R}$  be the reflection about  $\mathcal{O}$ . Note that

$$\mathcal{R}(p) = \pi_{e_1} \circ \cdots \circ \pi_{e_n}(p) \,,$$

where we identify  $\pi_{e_i}$  with the reflection about the corresponding hyperplane. Next we work as in [56, Lemma 4.1]. Here we only sketch the argument referring to [56] for details.

Without lost of generalities, we may assume  $\mathcal{O} \in \pi_{v,m_v-\mu}$ , for some  $\mu > 0$ and for  $k \in \mathbb{N}$  we define

$$\mu_k = \left| \{ p \in \Omega \cap \pi_{v,s} : m_v + (k-1)\mu < s < m_v + k\mu \} \right|_a,$$

here we use the notation introduced in Section 1.1. By construction  $\mu_k$  is decreasing and, in particular,

$$\mu_k \le \mu_0 := \left| \{ \Omega \cap \pi_{v,s} : m_v - \mu < s < m_v \} \right|_g.$$

Moreover,  $\mu_0$  is bounded by  $Cdef(\Omega)$ . Indeed, formula (1.92) yields

$$|\Omega \triangle \mathcal{R}(\Omega)|_q \le C \mathrm{def}(\Omega)$$

and then we obtain

$$|\Omega \cap \mathcal{R}(\Omega_v)|_g \ge |\Omega_v|_g - |\Omega \triangle \mathcal{R}(\Omega)|_g \ge \frac{|\Omega|_g}{2} - C def(\Omega).$$

Since

$$\mathcal{R}(\Omega_v) \subset \bigcup_{s<0} \pi_{v,m_v-s},$$

we obtain that

$$\mu_0 := \left| \{ \Omega \cap \pi_{v,s} : m_v - \mu < s < m_v \} \right|_g \le C \operatorname{def}(\Omega) .$$

Therefore

$$\mu_k \le C \mathrm{def}(\Omega) \tag{1.95}$$

for every k in  $\mathbb{N}$ . Again from (1.92) we get

$$\begin{aligned} \frac{|\Omega|_g}{2} - C \mathrm{def}(\Omega) &\leq |\Omega_v|_g \\ &\leq \sum_{k=0}^{k_0} \mu_k \\ &\leq k_0 \mu_0 \\ &\leq \frac{\mathrm{diam}\left(\Omega\right)}{m_v} C \mathrm{def}(\Omega) \,, \end{aligned}$$

where  $k_0$  is the integer part of  $\frac{\operatorname{diam}(\Omega)}{m_v}$ . From Proposition A.1 we have

$$m_v \leq C \operatorname{def}(\Omega)$$
,

where C depends only on n,  $\rho$ ,  $|S|_q$  and  $\mathcal{K}$ , as required.

Proof of Theorem 1.5. Let  $\mathcal{O}$  be as in Lemma 1.12 and define

$$r = \sup\{s > 0: B_s^d(\mathcal{O}) \subset \Omega\}$$
 and  $R = \inf\{s > 0: B_s^d(\mathcal{O}) \supset \Omega\}$ ,

so that

$$B_r^d(\mathcal{O}) \subseteq \Omega \subseteq B_R^d(\mathcal{O})$$
.

Let  $p,q \in S$  be such that  $d(p,\mathcal{O}) = r$  and  $d(q,\mathcal{O}) = R$ . We can assume that  $p \neq q$  (otherwise r = R and S is a round sphere). Let  $v \in T_o \mathbb{M}^n$  be the direction

$$v := \frac{1}{d(p,q)} \tau_p^{\mathsf{o}}(\exp_p^{-1}(q))$$

and  $\pi_v$  the critical hyperplane in the v-direction. We denote by  $\gamma$  the geodesic path passing through p and q and let  $s_p$  and  $s_q$  in  $\mathbb{R}$  be such that

$$\gamma(s_p) = p \text{ and } \gamma(s_q) = q.$$

Let  $z \in \pi_v$  be such that  $d(z, \mathcal{O}) = d(\mathcal{O}, \pi_v)$ . We have

$$p \in \pi_{v,s_p}$$
,  $q \in \pi_{v,s_q}$ ,  $s_q = s_p + t$ ;

see Section 1.1 for the definition of  $\pi_{v,s_p}$  and  $\pi_{v,s_q}$ . We first show that  $d(q,z) \leq d(p,z)$ . Assume by contradiction that d(q,z) > d(p,z). Since q and p belong to a geodesic orthogonal to the hyperplanes  $\pi_{v,s}$  and  $s_p < s_q$ , then  $s_q > m_v$ . Since  $\pi_v = \pi_{v,m_v}$  corresponds to the critical position of the method of the moving planes in the direction v, we have that  $\gamma(s) \in \Omega$  for any  $s \in (m_v, s_q)$ . Since  $s_p < s_q$  we have that  $|s_p - m_v| \geq |s_q - m_v|$  and since  $\gamma$  is orthogonal to  $\pi_v$  we obtain  $d(q, z) \leq d(p, z)$ , which gives a contradiction. Since  $d(q, z) \leq d(p, z)$  we have

$$r \ge R - d(\mathcal{O}, z) = R - d(\mathcal{O}, \pi_v)$$

and Lemma 1.12 implies (1.90).

### **1.6** Proof of Theorem 1.1 and Corollary 1.1

We have all the ingredients to prove Theorem 1.1 and Corollary 1.1.

Proof of Theorem 1.1. Let  $S = \partial \Omega$  be a  $C^2$ -regular, connected, closed hypersurface embedded in  $\mathbb{M}^n_+$  satisfying a uniform touching ball condition of radius  $\rho$ , where  $\Omega$  is a relatively compact domain. Theorem 1.4 implies that there exit  $\epsilon$  and C positive such that if

 $\operatorname{osc}(\mathsf{H}_S) \leq \varepsilon,$ 

then

$$d(p, \Sigma \cup \Sigma^{\pi_v}) \le Cosc(\mathsf{H}_S),$$

for every  $p \in \Omega$ .

Proof of Corollary 1.1. The proof consists in one more application of the method of the moving planes and it is in the spirit of [56, Theorems 1.2 and 1.5]. Let  $B_r^d(\mathcal{O})$  and  $B_R^d(\mathcal{O})$  be as in Theorem 1.5 and let  $0 < t < r - C \operatorname{def}(\Omega)$ . We aim at proving that for any  $p \in S$ , there exist two cones with vertex at p and of fixed aperture, one contained in  $\Omega$  and one contained in the complementary of  $\Omega$ . The first cone  $C^-(p)$ , is obtained by considering all the geodesic path connecting p to the boundary of  $B_t^d(\mathcal{O})$  tangentially. The second cone  $C^+(p)$  is the reflection of  $\mathcal{C}^-(p)$  with respect to p. We show that  $C^-(p)$  is contained in  $\Omega$  and an analogous argument shows that  $\mathcal{C}^+(p)$  is contained in the complementary of  $\Omega$ . We assume, by contradiction, that  $p \notin B_r^d(\mathcal{O})$  (otherwise the claim is trivial) and that there exists a point  $q \in \mathcal{C}^-(p) \cap \partial B_t^d(\mathcal{O})$  such that the geodesic path  $\gamma$  connecting q to p is not contained in  $\Omega$ . We apply the method of the moving planes in the direction v defined by

$$v := \frac{1}{d(p,q)} \tau_q^{\mathsf{o}}(\exp_q^{-1}(p)) \,.$$

Since  $\gamma$  is not contained in  $\Omega$ , the method of the moving planes stops before reaching q and one can prove that

$$d(\mathcal{O}, \pi_{\omega}) \ge r - t$$

Since  $0 < t < r - Cdef(\Omega)$ , from Lemma 1.12, we obtain

$$C def(\Omega) < r - t \le d(\mathcal{O}, \pi_{\omega}) \le C def(\Omega),$$

which gives a contradiction. The argument above shows also that for any  $p \in S$  the geodesic path connecting p to  $\mathcal{O}$  is contained in  $\Omega$ . This implies that there exists a  $C^2$ -regular map  $\Psi : \partial B^d_r(\mathcal{O}) \to \mathbb{R}$  such that

$$F(p) = \exp_x(\Psi(p)N_p),$$

defines a  $C^2$ -diffeomorphism from  $B_r^d(\mathcal{O})$  to S. By choosing  $t = r - \sqrt{C \operatorname{def}(\Omega)}$ we have that for any  $p \in S$  there exists a uniform cone of opening  $\pi - \sqrt{C \operatorname{def}(\Omega)}$ with vertex at p and axis on the geodesic connecting p to  $\mathcal{O}$ . This implies that  $\Psi$ is locally Lipschitz and the bound (1.5) on  $\|\Psi\|_{C^1}$  follows (see also [56, Theorem 1.2]).

Remark 1.2. We observe that if  $H_S = H$  is the mean curvature of  $\partial\Omega$ , then (1.5) can be improved and we can obtain the optimal linear bound

$$\|\Psi\|_{C^{1,\alpha}} \leq Cosc(\mathbf{H})$$

by using elliptic regularity. Indeed, let  $\phi: U \to \partial B^d_r(\mathcal{O})$  be a local parametrization of  $\partial B^d_r(\mathcal{O})$ , where U is an open set of  $\mathbb{R}^{n-1}$ . From the proof of Corollary 1.1,  $F \circ \phi$  gives a local parametrization of S. A standard computation yields that

$$L(\Psi \circ \phi) = H(F \circ \phi) - H_{\mathsf{B}_{r}}$$

where  $H_{\mathsf{B}_r}$  is the mean curvature of  $\partial \mathsf{B}_r$  and L is an elliptic operator which, thanks to the bounds on  $\Psi$  above, can be seen as a second order linear operator acting on  $\Psi \circ \phi$ . Then [112, Theorem 8.32] implies the bound on the  $C^{1,\alpha}$ -norm of  $\Psi$ , as required.

### Chapter 2

# Symmetry results for Serrin-type overdetermined problems

In this chapter we consider symmetry results for Serrin-type overdetermined problems, in particular in Section 2.1 we prove a Serrin-type result for domains inside the so-called model manifolds by using a generalization of Weinberger's proof; in Section 2.2 we prove a Serrin-type result for domains inside convex cones of the Euclidean space and in space forms.

### 2.1 A symmetry result in model manifolds

As already mentioned, in [204], J. Serrin proved the following celebrated result: if there exists a positive solution  $u \in C^2(\overline{\Omega})$  to the following semilinear problem

$$\begin{cases} \Delta u = f(u) & \text{in } \Omega, \\ u = 0 & \text{on } \partial\Omega, \end{cases}$$
(2.1)

where  $\Omega \subset \mathbb{R}^n$  is a bounded domain with boundary of class  $C^2$  and f is a function of class  $C^1$ , such that

$$\partial_{\nu} u = c \quad \text{on } \partial\Omega,$$
 (2.2)

for some constant c, where  $\nu$  denotes the outward unit normal to  $\partial\Omega$ . Then  $\Omega$  must be a ball and u radially symmetric. We have also already mentioned that, in [215], H. Weinberger provided a simpler proof in the case  $\Delta u = -1$  based on what are nowadays called *P*-function and using integral identities.

In literature there are generalizations of Serrin's result for domains in the so-called space forms, i.e. complete, simply connected Riemannian manifolds with constant sectional curvature. Thanks to the Killing-Hopf theorem (see [137, 143]) it is well-known that space forms are isometric to the Euclidean space  $\mathbb{R}^n$ , to the hyperbolic space  $\mathbb{H}^n$  or to the sphere  $\mathbb{S}^n$ . In particular in [150] and [171] the moving planes method is used to prove the analogue of Serrin's result for the problem (2.1)-(2.2) for domains in  $\mathbb{H}^n$  and in  $\mathbb{S}^n_+$  (we mention that

in  $\mathbb{S}^n$  the theorem is not true, see e.g. [91]). In [196] we extend Weinberger's approach to a particular class of Riemannian manifolds: the already cited model manifolds, i.e. rotationally symmetric Riemannian manifolds. We recall the precise definition of model manifolds:

**Definition 2.1.** A *n*-dimensional Riemannian manifold  $(\mathbb{M}^n_{\sigma}, g_{\mathbb{M}^n_{\sigma}})$  is called a model manifold if

$$\mathbb{M}_{\sigma}^{n} := \frac{[0, R) \times \mathbb{S}^{n-1}}{\sim} \quad \text{and} \quad g_{\mathbb{M}_{\sigma}^{n}} := dr \otimes dr + \sigma^{2}(r)g_{\mathbb{S}^{n-1}},$$

where  $R \in (0, +\infty]$ ,  $\sim$  is the relation that identifies all the points of  $\{0\} \times \mathbb{S}^{n-1}$ and  $\sigma : [0, R) \to [0, +\infty)$  is a smooth function such that:

$$\sigma(r) > 0$$
, for all  $r > 0$ ,  $\sigma^{(2k)}(0) = 0$ , for all  $k = 0, 1, 2, \dots, \sigma'(0) = 1$ .

The point  $o \in \mathbb{M}^n_{\sigma}$  corresponding to r = 0 is called the *pole* of the model and  $\sigma$  is called the warping function.

The importance and the convenience of the model manifolds lies in the fact that their geometry and some natural differential operators (such as the Laplacian, see formula (2.8) below) have a particularly simple and explicit description. In particular we will use the following explicit expression of the Ricci curvature (see e.g. [184]): given  $x \in \mathbb{M}^n_{\sigma}$  and  $X \in \nabla r(x)^{\perp}$  in  $T_x \mathbb{M}^n_{\sigma}$  a unit vector we have

$$\operatorname{Ric}_{\mathbb{M}^n_{\sigma}}(X,X) = (n-2)\frac{1-(\sigma')^2}{\sigma^2} - \frac{\sigma''}{\sigma},$$

and

$$\operatorname{Ric}_{\mathbb{M}^n_{\sigma}}(\nabla r, \nabla r) = -(n-1)\frac{\sigma''}{\sigma}.$$

With these preliminaries, the main Theorem of the first section is the following (see also Theorem I.C in the introduction to Part I)

**Theorem 2.1.** Let  $\Omega \subset \mathbb{M}^n_{\sigma}$  be a smooth domain with  $o \in \Omega$ . Assume that  $\Omega \subseteq B_{\tilde{R}}(o)$  where the ray  $\tilde{R} > 0$  is such that the following conditions on  $\sigma$  are satisfied on the interval  $[0, \tilde{R})$ :

- (a)  $\operatorname{Ric}_{\mathbb{M}^n_{\sigma}} \ge 0$ , *i.e.*  $\sigma'' \le 0$  and  $(n-2)(1-(\sigma')^2) \sigma \sigma'' \ge 0$ ;
- (b)  $\sigma' > 0$ .

If  $\Omega$  supports a solution  $u \in C^2(\Omega) \cap C^1(\overline{\Omega})$  of

$$\begin{cases} \Delta u = -1 & \text{in } \Omega\\ u = 0 & \text{on } \partial\Omega \end{cases},$$
(2.3)

such that, for some constant c,

$$\partial_{\nu} u = c \quad on \; \partial\Omega \,, \tag{2.4}$$

and u satisfies the following "compatibility" condition

$$\int_{\Omega} \frac{(\sigma''\sigma^{n-1})'}{\sigma^{n-1}} u^2 \ge 0 \tag{2.5}$$

then we have that  $\Omega$  is a Euclidean ball of radius  $\rho$  centred in the pole o of the model and u has the specific form:

$$u(r) = \frac{1}{2n}(\rho^2 - r^2)$$
(2.6)

where r(x) = dist(x, o).

We briefly analyse the hypothesis of the Theorem: the condition  $\sigma' > 0$ appears in other articles on the subject (see e.g. [66]). The "compatibility" condition (2.5) describes a property of the solution in relation to the geometry of the model. It is automatically satisfied by any solution of (2.3)-(2.4) in the case of the Euclidean space and it can not be reduced to a simple condition on the model, like

$$(\sigma''\sigma^{n-1})' \ge 0.$$

Indeed, in this case, the three conditions are compatible only with the flat case: consider  $f(r) := \sigma''(r)\sigma^{n-1}(r)$ . Then f(0) = 0 and if  $f'(r) \ge 0$ , i.e. f(r) is non-decreasing, so  $f(r) \ge 0$  for r > 0. But  $\sigma''(r) \le 0$  according to (a), so we have that  $\sigma''(r) = 0$ . In this case the result is well known and is presented in Weinberger's article. Moreover, an analogue condition can be found in [4] where they consider a symmetry result for a overdetermined problem and they assume a "compatibility" condition as an integral on the boundary of the domain involving the solution and its gradient.

We mention that also in [189] and [68] a *P*-function approach is used to prove a symmetry result in space forms for the following equation

$$\Delta u + nKu = -1 \tag{2.7}$$

where K = 0 in  $\mathbb{R}^n$ , K = 1 in  $\mathbb{S}^n_+$  and K = -1 in  $\mathbb{H}^n$ . The big difference between the conclusion of Theorem 2.1 and the results in [189] and [68] is that in Theorem 2.1 the domain is a Euclidean ball, while in [189] and [68] the domain is a geodesic ball.

### 2.1.1 Explicit computations towards the proof of Theorem 2.1

The Laplace-Beltrami operator  $\Delta$  of  $\mathbb{M}^n_{\sigma}$  acts on  $C^2$ -functions  $u: \mathbb{M}^n_{\sigma} \to \mathbb{R}$  as follows:

$$\Delta u = \partial_r^2 u + (n-1)\frac{\sigma'}{\sigma}\partial_r u + \frac{1}{\sigma^2}\bar{\Delta}u = \frac{\partial_r(\sigma^{n-1}\partial_r u)}{\sigma^{n-1}} + \frac{1}{\sigma^2}\bar{\Delta}u, \qquad (2.8)$$

where  $\overline{\Delta}$  denotes the Laplacian on the standard sphere ( $\mathbb{S}^{n-1}, g_{\mathbb{S}^{n-1}}$ ). Using this expression we obtain:

Lemma 2.1. The following general formula holds:

$$\Delta(\sigma \,\partial_r u) = \sigma \,\partial_r \Delta u + 2\sigma' \,\Delta u + (2-n)\sigma'' \,\partial_r u. \tag{2.9}$$

Proof. We compute

$$\begin{split} \sigma\partial_r(\Delta u) =& \sigma \left\{ \partial_r^3 u + (n-1) \frac{\sigma''\sigma - (\sigma')^2}{\sigma^2} \partial_r u + (n-1) \frac{\sigma'}{\sigma} \partial_r^2 u - 2 \frac{\sigma'}{\sigma^3} \bar{\Delta} u + \frac{1}{\sigma^2} \partial_r (\bar{\Delta} u) \right\} \\ =& \sigma \partial_r^3 u + (n-2) \sigma'' \partial_r u + \sigma'' \partial_r u - 2(n-1) \frac{(\sigma')^2}{\sigma} \partial_r u + (n-1) \frac{(\sigma')^2}{\sigma} \partial_r u \\ &+ (n+1) \sigma' \partial_r^2 u - 2 \sigma' \partial_r^2 u - 2 \frac{\sigma'}{\sigma^2} \bar{\Delta} u + \frac{1}{\sigma} \partial_r (\bar{\Delta} u) + \frac{1}{\sigma^2} \bar{\Delta} (\sigma \partial_r u) \\ &- \frac{1}{\sigma^2} \bar{\Delta} (\sigma \partial_r u) \\ =& \Delta (\sigma \partial_r u) + (n-2) \sigma'' \partial_r u - 2 \sigma' \Delta u + \frac{1}{\sigma} \partial_r (\bar{\Delta} u) - \frac{1}{\sigma^2} \bar{\Delta} (\sigma \partial_r u), \end{split}$$

i.e.

$$\Delta(\sigma\partial_r u) = \sigma\partial_r(\Delta u) + (2-n)\sigma''\partial_r u + 2\sigma'\Delta u.$$

Now we focus on the solution u of (2.3)-(2.4) and we show the following **Lemma 2.2.** Let  $\Omega$  and u as in Theorem 2.1. Then:

$$(n+2)\int_{\Omega} u\,\sigma' = nc^2 \int_{\Omega} \sigma' + \frac{(n-2)}{2} \int_{\Omega} \frac{(\sigma''\sigma^{n-1})'}{\sigma^{n-1}} u^2.$$
(2.10)

Remark 2.1. In particular, if  $\sigma(r) = r$  and, hence,  $\mathbb{M}_{\sigma}^{n} = \mathbb{R}^{m}$ , we obtain

$$(n+2)\int_{\Omega} u = nc^2 |\Omega|, \qquad (2.11)$$

which is (I.9).

Proof. First of all we observe that, in this setting, formula (2.9) becomes

$$\Delta(\sigma \,\partial_r u) = -2\sigma' + (2-n)\sigma'' \,\partial_r u.$$

So by Green's Theorem

$$\begin{split} \int_{\Omega} \left[ -2\sigma' \, u + (2-n)\sigma'' \, \partial_r u \, u + \sigma \, \partial_r u \right] &= \int_{\Omega} \left[ \Delta(\sigma \, \partial_r u) u - \sigma \, \partial_r u \, \Delta u \right] \\ &= \int_{\partial\Omega} \left[ \partial_\nu (\sigma \, \partial_r u) u - \sigma \, \partial_r u \, \partial_\nu u \right] \\ &= -\int_{\partial\Omega} \sigma \left( \partial_\nu u \right)^2 \partial_\nu r \\ &= -c^2 \int_{\partial\Omega} \sigma \, \partial_\nu r \\ &= -c^2 \int_{\Omega} \left[ \sigma \, \Delta r + g_{\mathbb{M}^n_{\sigma}} (\nabla r, \nabla \sigma) \right] \\ &= -c^2 \int_{\Omega} \left[ \sigma \left( n-1 \right) \frac{\sigma'}{\sigma} + \sigma' \right] \\ &= -c^2 n \int_{\Omega} \sigma', \end{split}$$

where we have used the fact that u = 0 on  $\partial \Omega$  and that  $\partial_{\nu} u = c$  on  $\partial \Omega$ . Now note that

$$\begin{split} \int_{\Omega} \sigma \partial_r u &= \int_{\Omega} g_{\mathbb{M}^n_{\sigma}} (\nabla u, \nabla (\int_0^r \sigma(s) ds)) \\ &= - \int_{\Omega} u \Delta (\int_0^r \sigma(s) ds) \\ &= -n \int_{\Omega} u \sigma'. \end{split}$$

Using this and the previous computation we have

$$(n+2)\int_{\Omega} u\,\sigma' = nc^2 \int_{\Omega} \sigma' + (2-n) \int_{\Omega} \sigma'' \,u\partial_r u.$$
(2.12)

Finally we observe that

$$\int_{\Omega} \sigma'' u \partial_r u = \int_{\Omega} g_{\mathbb{M}^n_{\sigma}} (\nabla \sigma', \nabla(\frac{1}{2}u^2))$$

$$= -\frac{1}{2} \int_{\Omega} \Delta \sigma' u^2$$

$$= -\frac{1}{2} \int_{\Omega} \frac{(\sigma'' \sigma^{n-1})'}{\sigma^{n-1}} u^2,$$
(2.13)

where the second and the third equations are obtained using the condition u = 0on  $\partial\Omega$  and the expression (2.8), respectively.

Remark 2.2. Observe that, by the Strong Maximum Principle, a solution u of (2.3) is positive in  $\Omega$ . Moreover since  $\partial_{\nu} u = c \neq 0$  on  $\Omega$  we obtain that  $|\nabla u| \neq 0$  on  $\partial\Omega$  and the smooth hypersurface  $\partial\Omega = \{u = 0\}$  has exterior normal given by

$$\nu = -\frac{\nabla u}{|\nabla u|} \mid_{\partial\Omega}.$$

This implies that

$$\partial_{\nu} u = - |\nabla u|$$
 on  $\partial \Omega$ .

#### 2.1.2 Proof of Theorem 2.1

Now we are ready to prove Theorem 2.1.

Proof of Theorem 2.1. Let u and  $\Omega$  as in the statement of Theorem 2.1; by the Bochner formula (I.23) and the Cauchy-Schwarz inequality (I.24) we get

$$\begin{split} \Delta \left( |\nabla u|^2 + \frac{2}{n} u \right) &= 2 |\nabla^2(u)|^2 + 2 \operatorname{Ric}_{\mathbb{M}^n_{\sigma}} (\nabla u, \nabla u) + \frac{2}{n} \Delta u \quad (2.14) \\ &\geq 2 \left( |\nabla^2 u|^2 + \frac{1}{n} \Delta u \right) \\ &= 2 \left( |\nabla^2 u|^2 - \frac{1}{n} (\Delta u)^2 \right) \\ &\geq 0 \quad \text{on } \Omega, \end{split}$$

and the equality holds if and only if

$$\nabla^2(u) = \frac{\Delta u}{n} g_{\mathbb{M}^n_\sigma}$$

and

$$\operatorname{Ric}_{\mathbb{M}^n_{\sigma}}(\nabla u, \nabla u) = 0.$$

Since, according to Remark 2.2,

$$|\nabla u|^2 + \frac{2}{n}u = c^2 \text{ on } \partial\Omega, \qquad (2.15)$$

we conclude from the Strong Maximum Principle that either

$$|\nabla u|^2 + \frac{2}{n}u < c^2 \text{ on } \Omega \tag{2.16}$$

or

$$|\nabla u|^2 + \frac{2}{n}u \equiv c^2 \text{ on } \Omega.$$
(2.17)

By contradiction assume that condition (2.16) is satisfied. According to (b) we can multiply both the members of (2.16) by  $\sigma'$  and integrate in order to obtain

$$n \int_{\Omega} |\nabla u|^2 \sigma' + 2 \int_{\Omega} u \, \sigma' < nc^2 \int_{\Omega} \sigma'.$$
(2.18)

Now we use the identity (2.10) to deal with the second term i.e.

$$2\int_{\Omega} u\,\sigma' = nc^2 \int_{\Omega} \sigma' + \frac{(n-2)}{2} \int_{\Omega} \frac{(\sigma''\sigma^{n-1})'}{\sigma^{n-1}} u^2 - n \int_{\Omega} u\sigma'.$$
 (2.19)

Note that, by the divergence theorem,

$$n \int_{\Omega} \sigma' \operatorname{div}(u \nabla u) = -n \int_{\Omega} \sigma'' u \,\partial_r u.$$
(2.20)

Moreover,

$$n \int_{\Omega} \sigma' \operatorname{div}(u \nabla u) = n \int_{\Omega} \sigma' |\nabla u|^2 - n \int_{\Omega} \sigma' u.$$

 $\mathrm{So}$ 

$$n\int_{\Omega}\sigma'|\nabla u|^2 = n\int_{\Omega}\sigma' u - n\int_{\Omega}\sigma'' u\,\partial_r u.$$
(2.21)

Substituting (2.19) and (2.21) in (2.18) we obtain

$$\begin{split} nc^2 \int_{\Omega} \sigma' > &- n \int_{\Omega} \sigma'' u \, \partial_r u + n \int_{\Omega} \sigma' u + nc^2 \int_{\Omega} \sigma' \\ &+ \frac{(n-2)}{2} \int_{\Omega} \frac{(\sigma'' \sigma^{n-1})'}{\sigma^{n-1}} u^2 - n \int_{\Omega} u \sigma' \,. \end{split}$$

Lastly, we use the identity (2.13) to deduce

$$\frac{n}{2}\int_{\Omega}\frac{(\sigma^{\prime\prime}\sigma^{n-1})^{\prime}}{\sigma^{n-1}}u^2 + \frac{(n-2)}{2}\int_{\Omega}\frac{(\sigma^{\prime\prime}\sigma^{n-1})^{\prime}}{\sigma^{n-1}}u^2 < 0,$$

i.e.

$$-(n-1)\int_{\Omega} \frac{(\sigma''\sigma^{n-1})'}{\sigma^{n-1}}u^2 > 0; \qquad (2.22)$$

and this contradicts the "compatibility" condition (2.5). Therefore (2.17) holds true and  $|\nabla u|^2 + \frac{2}{n}u$  must be constant in  $\Omega$ . Since its

Laplacian then vanishes, we conclude from (2.14) that equality must hold in Cauchy-Schwarz inequality, i.e. we have proved that u is a solution of (recall that  $\Delta u = -1$  in  $\Omega$ )

$$\nabla^2(u) = -\frac{1}{n} g_{\mathbb{M}^n_{\sigma}} \text{ in } \Omega.$$
(2.23)

Now, let  $\rho := dist(o, \partial\Omega)$  and take  $B_{\rho}(o) \subset \Omega$ . Since  $\partial\Omega$  is compact, there exists  $p \in \partial\Omega$  such that  $p \in \partial\Omega \cap \partial B_{\rho}(o)$ . In particular, since u = 0 on  $\partial\Omega$ , we have that

u(p) = 0.

If we prove that u is a radial function in  $B_{\rho}(o)$  then

u = 0 on  $\partial B_{\rho}(o)$ .

On the other hand, by the Strong Maximum Principle,

$$u > 0$$
 in  $\Omega$ .

Therefore we can conclude that  $\partial B_{\rho}(o) \cap \Omega = \emptyset$  and, hence,  $\Omega = B_{\rho}(o)$ .

So the key point is to prove that  $u: B_{\rho}(o) \to \mathbb{R}$ , solution of (2.23), is a radial function in  $B_{\rho}(o)$ . To this end, take  $x \in B_{\rho}(o)$ . Since  $\mathbb{M}_{\sigma}^{n}$  is geodesically complete there exist a minimizing and normalized geodesic  $\gamma \subset B_{\rho}(o)$  from o to x. Let  $y(t) := u \circ \gamma(t)$  and note that, along  $\gamma$ , equation (2.23) implies

$$\begin{split} y''(t) &= \frac{d^2}{dt^2} (u \circ \gamma)(t) \\ &= \frac{d}{dt} g_{\mathbb{M}^n_{\sigma}}(\nabla u(\gamma(t)), \dot{\gamma}(t)) \\ &= g_{\mathbb{M}^n_{\sigma}}(D_{\dot{\gamma}} \nabla u(\gamma(t)), \dot{\gamma}(t)) + g_{\mathbb{M}^n_{\sigma}}(\nabla u(\gamma(t)), D_{\dot{\gamma}} \dot{\gamma}(t)) \\ &= g_{\mathbb{M}^n_{\sigma}}((D_{\dot{\gamma}(t)} \nabla u)(\gamma(t)), \dot{\gamma}(t)) \\ &= \nabla^2(u) \mid_{\gamma(t)} (\dot{\gamma}(t), \dot{\gamma}(t)) \\ &= -\frac{1}{n}. \end{split}$$

The solutions of  $y''(t) = -\frac{1}{n}$  are given by

$$y(t) = -\frac{1}{2n}t^2 + \alpha t + \beta$$

where  $\alpha, \beta \in \mathbb{R}$ . Now taking t = r(x) we get

$$u(x) = u \circ \gamma(r(x)) = y(r(x)) = -\frac{1}{2n}r(x)^2 + \alpha r(x) + \beta$$
(2.24)

which is radial. To determine the two constant in (2.24) we recall that u satisfies the following

$$\begin{cases} u(\rho) = 0\\ u(r) > 0 \quad \text{for } 0 < r < \rho \end{cases}$$

i.e., using the explicit formula of  $\boldsymbol{u}$  we obtain

$$\begin{cases} -\frac{1}{2n}\rho^2 + \alpha\rho + \beta = 0\\ -\frac{1}{2n}\left(\frac{\rho}{2}\right)^2 + \alpha\frac{\rho}{2} + \beta > 0 \quad \text{for } r = \frac{\rho}{2} \end{cases}$$

substituting the expression  $\beta = \frac{1}{2n}\rho^2 - \alpha\rho$  in the second equation we get

$$\alpha < \frac{3}{4n}\rho.$$

But, since u must be a  $C^2$ -function we have that  $\alpha = 0$ ; indeed, if we consider the Euclidean case where r(x) = d(x, 0) = |x| the gradient of u becomes

$$\nabla u(x) = -\frac{1}{n}x + \alpha \frac{x}{|x|}$$
(2.25)

which is not a  $C^1$  function in the origin (i.e. the pole of the Euclidean space) unless  $\alpha = 0$ . In a generic model the expression (2.25) holds in a system of normal coordinates in the pole. So the same conclusion holds and  $\beta = \frac{1}{2n}\rho^2$ ; with this constants the function u becomes

$$u(r) = -\frac{1}{2n}r^2 + \frac{1}{2n}\rho^2$$

which is exactly the expression (2.6); observe that, since u is radial,  $\partial_{\nu}u = u'(r)$ and the condition  $\partial_{\nu}u = constant$  in  $\partial\Omega = \partial B_{\rho}(o)$  is automatically satisfied. Moreover we recall that if  $f : \mathbb{M}^n_{\sigma} \to \mathbb{R}$  is a smooth radial function, then its Hessian takes the following expression

$$\nabla^2(f) = f'' dr \otimes dr + f' \sigma \sigma' g_{\mathbb{S}^{n-1}}.$$
(2.26)

Using this expression with the function u we get

$$\nabla^2(u) = -\frac{1}{n} dr \otimes dr - \frac{1}{n} r \sigma \sigma' g_{\mathbb{S}^{n-1}}, \qquad (2.27)$$

and using this latter in (2.23) we obtain

$$-\frac{1}{n}dr \otimes dr - \frac{1}{n}r\sigma\sigma' g_{\mathbb{S}^{n-1}} = -\frac{1}{n}(dr \otimes dr + \sigma^2 g_{\mathbb{S}^{n-1}})$$

i.e.

$$-\frac{1}{n}r\sigma\sigma' g_{\mathbb{S}^{n-1}} = -\frac{1}{n}\sigma^2 g_{\mathbb{S}^{n-1}}.$$

It follows that  $\sigma(r) = r$ , so in the ball  $B_{\rho}(o)$  not only the solution of (2.23) is radial but also the metric  $g_{\mathbb{M}_{\sigma}^{m}}$  is the Euclidean metric. This implies that the ball  $B_{\rho}(o)$  is a Euclidean ball and the claim follows.

*Remark* 2.3 (An alternative end of the proof). From the equality sign in the Bochner inequality (2.14) we get

$$\operatorname{Ric}_{\mathbb{M}_{\pi}^{m}}(\nabla u, \nabla u) = 0. \tag{2.28}$$

From the explicit expression of u (formula (2.6)) we see that the only critical point is in r = 0, i.e. in the pole o of the model. So the condition on the Ricci curvature becomes

$$\operatorname{Ric}_{\mathbb{M}^n_{\sigma}}(\nabla r, \nabla r) = 0 \text{ in } B_{\rho}(o) \setminus \{o\}.$$

$$(2.29)$$

From the explicit expression of  $\operatorname{Ric}_{\mathbb{M}_{\sigma}^{n}}(\nabla r, \nabla r)$  we get  $\sigma'' = 0$  in  $(0, \rho)$  and we conclude that  $\sigma(r) = r$ , i.e.  $B_{\rho}(o)$  is an Euclidean ball.

Remark 2.4. In [197] by Ros we can find a similar spirit where, using the Reilly's formula, he obtained a generalization of Alexandrov theorem for compact hypersurfaces with constant higher order mean curvatures; in this article equation (2.23) is used to prove a Euclidean symmetry result on a generic compact Riemannian manifold of non-negative Ricci curvature with smooth boundary with mean curvature positive everywhere.

Remark 2.5. In this remark we provide an example that shows that if the "compatibility" condition (2.5) is not satisfied then we can not have Euclidean symmetry. According to the result of Kumaresan and Prajapat result [150] we know that if we take a domain  $\Omega \subset \mathbb{S}^n$  such that  $\overline{\Omega} \subset \mathbb{S}^n_+$  and there exist a solution uto the Serrin's problem (2.3)-(2.4) then  $\Omega$  must be a geodesic ball and u must be radially symmetric. We know that the hemisphere is isometric to the model  $\mathbb{M}^n_{\sigma}$ with  $\sigma(r) = \sin(r) \mid_{[0,\pi/2]}$ ; so in this example conditions (a) and (b) of Theorem 2.1 are clearly satisfied and the "compatibility" condition (2.5) becomes:

$$\int_{\Omega} \frac{(\sigma'' \sigma^{n-1})'}{\sigma^{n-1}} u^2 = \int_{\Omega} -n\cos(r)u^2(r),$$

which is negative due to the monotonicity of the integral and to the fact that the function  $r \mapsto \cos(r)u^2(r)$  is positive in  $\Omega$ .

In conclusion the "compatibility" condition is not satisfied and the symmetry result is not Euclidean since the ball  $\Omega$  is a geodesic ball, i.e. the metric in this ball is the metric of the sphere.

#### 2.2 Symmetry results in convex cones

Before stating the results that we are going to prove, we brefly recall some notations that we already introduced in the introduction to Part I. Let  $\Sigma$  be an open cone in  $\mathbb{R}^n$  with vertex at the origin O, i.e.

$$\Sigma = \{ tx : x \in \omega, t \in (0, +\infty) \}$$

for some open domain  $\omega \subset \mathbb{S}^{n-1}$ . Given an open cone  $\Sigma$  such that  $\partial \Sigma \setminus \{O\}$  is smooth, we consider a bounded domain  $\Omega \subset \Sigma$  and denote by  $\Gamma_0$  its relative boundary, i.e.

$$\Gamma_0 = \partial \Omega \cap \Sigma$$

and we set

$$\Gamma_1 = \partial \Omega \setminus \overline{\Gamma}_0 \,.$$

We assume that  $\mathcal{H}_{n-1}(\Gamma_1) > 0$ ,  $\mathcal{H}_{n-1}(\Gamma_0) > 0$  and that  $\Gamma_0$  is a smooth (n-1)-dimensional manifold, while  $\partial \Gamma_0 = \partial \Gamma_1 \subset \partial \Omega \setminus \{O\}$  is a smooth (n-2)-dimensional manifold. Following [180], such a domain  $\Omega$  is called a *sector-like domain*. In the following, we shall write  $\nu = \nu_x$  to denote the exterior unit normal to  $\partial \Omega$  wherever is defined (that is for  $x \in \Gamma_0 \cup \Gamma_1 \setminus \{O\}$ ).

As already mentioned, under the assumption that  $\Sigma$  is a convex cone, in [180] it is proved that if  $\Omega$  is a sector-like domain and there exists a classical solution  $u \in C^2(\Omega) \cap C^1(\Omega \cup \Gamma_0 \cup \Gamma_1 \setminus \{O\})$  to

$$\begin{cases} \Delta u = -1 & \text{in } \Omega, \\ u = 0 \text{ and } \partial_{\nu} u = -c & \text{on } \Gamma_0, \\ \partial_{\nu} u = 0 & \text{on } \Gamma_1 \setminus \{O\}, \end{cases}$$
(2.30)

and such that

$$u \in W^{1,\infty}(\Omega) \cap W^{2,2}(\Omega)$$
,

then

$$\Omega = B_R(x_0) \cap \Sigma$$

for some  $x_0 \in \mathbb{R}^n$  and u is given by (I.11). Differently from the original paper of Serrin [204], the method of moving planes is not helpful (at least when applied in a standard way) and the rigidity result in [180] is proved by using two alternative approaches. One is based on integral identities and it is inspired by [40], the other one uses a *P*-function approach as in [215].

In this Section, we generalize the rigidity result for Serrin's problem in [180] in two directions. The former is by considering more general operators than the Laplacian in the Euclidean space, where the operators may be of degenerate type. Here, the generalization is not trivial due to the lack of regularity of the solution (the operator may be degenerate) as well as to other technical details which are not present in the linear case.

The latter is by considering an analogous problem in space forms, i.e. the hyperbolic space and the (hemi)sphere. The operator that we consider is linear and it is interesting since it has been shown that it is a helpful generalization of the torsion problem to space forms (see [68, 189, 190]).

Overdetermined problems for quasilinear and possible degenerate operators have attracted a lot of interest in the last decades, see for instance [92, 103, 106, 107, 192, 193]. As Fosdick and Serrin noticed in [204] and [100], Serrin's overdetermined problem for quasilinear elliptic operators is also interesting for possible applications to the study of steady rectilinear motion of viscous incompressible fluids and incompressible non-Newtonian fluids (see also [106]), and in the theory of torsion of a solid straight bar. Roughly speaking, a rigidity result as the one given by Serrin proves that the tangential stress on the pipe wall is the same at all points of the wall if and only if the pipe has a circular cross section or that when a solid straight bar is subject to torsion, the magnitude of the resulting traction which occurs at the surface of the bar is independent of posible applications for Serrin's type rigidity results, and we refer to [103, Introduction] for connections to capillarity theory, torsional creep, Born-Infeld theory and other applications to quantum-physics.

As explained in [180], the study of Serrin's overdetermined problem in convex cones is related to relative isoperimetric inequality and Alexandrov soap bubble theorem. In this manuscript we extend this study to non-Euclidean manifolds, in particular to space forms. The study of isoperimetric inequality and Alexandrov theorem in non-Euclidean manifolds has recently attracted a lot of interest in the geometric analysis community (see [41, 156, 189] and references therein). We believe that, by taking inspiration from our results and the ones in [156, 189], one can study Alexandrov theorem and relative isoperimetric inequalities for sector-like domains in more general Riemannian settings.

The study of rigidity problems in convex cones appears also in the context of critical points for Sobolev inequality (which in turns can be related to Yamabe problem), see [57, 160].

We also mention that the approach used in this paper originated from [40], which in turns has been later used for proving quantitative estimates for Serrin's overdetermined problem in [39]. As for the symmetry result, this approach is also useful when considering quantitative versions of Alexandrov soap bubble theorem, in particular to describe the appearance of bubbling [58].

More general operators in the Euclidean space. Let  $\Omega$  be a sector like domain in  $\mathbb{R}^n$  and let  $f: [0, +\infty) \to [0, +\infty)$  be such that

$$f \in C^{1}([0,\infty)) \cap C^{3}((0,\infty)) \text{ with } f(0) = f'(0) = 0, \ f''(s) > 0 \text{ for } s > 0$$
  
and 
$$\lim_{s \to +\infty} \frac{f(s)}{s} = +\infty.$$
(2.31)

We consider the following mixed boundary value problem

$$\begin{cases} L_f u = -1 & \text{in } \Omega, \\ u = 0 & \text{on } \Gamma_0 \\ \partial_\nu u = 0 & \text{on } \Gamma_1 \setminus \{O\}, \end{cases}$$
(2.32)

where the operator  $L_f$  is given by

$$L_f u = \operatorname{div}\left(f'(|\nabla u|)\frac{\nabla u}{|\nabla u|}\right),\tag{2.33}$$

and the equation  $L_f u = -1$  is understood in the sense of distributions

$$\int_{\Omega} \frac{f'(|\nabla u|)}{|\nabla u|} \nabla u \cdot \nabla \varphi \, dx = \int_{\Omega} \varphi \, dx$$

for any

$$\varphi \in T(\Omega) := \{ \varphi \in C^1(\Omega) : \varphi \equiv 0 \text{ on } \Gamma_0 \}.$$

Notice that the operator  $L_f$  may be of degenerate type.

We notice that the solution to  $L_f u = -1$  in  $B_R(x_0)$  (a ball of radius R centered at  $x_0$ ) such that u = 0 on  $\partial B_R(x_0)$  is radial and it is given by

$$u(x) = \int_{|x-x_0|}^{R} g'\left(\frac{s}{n}\right) \, ds \,, \qquad (2.34)$$

where g denotes the Fenchel conjugate of f (see for instance [73] or [103]), i.e.

$$g = \sup\{st - f(s) : s \ge 0\}$$

(hence for us g' is the inverse function of f'). Our first main result is the following.

**Theorem 2.2.** Let f satisfy (2.31). Let  $\Sigma$  be a convex cone such that  $\Sigma \setminus \{O\}$ is smooth and let  $\Omega$  be a sector-like domain in  $\Sigma$ . If there exists a solution  $u \in C^1(\Omega \cup \Gamma_0 \cup \Gamma_1 \setminus \{O\}) \cap W^{1,\infty}(\Omega)$  to (2.32) such that

$$\partial_{\nu} u = -c \quad on \quad \Gamma_0 \tag{2.35}$$

for some constant c, and satisfying

$$\frac{f'(|\nabla u|)}{|\nabla u|} \nabla u \in W^{1,2}(\Omega, \mathbb{R}^n), \qquad (2.36)$$

then there exists  $x_0 \in \mathbb{R}^n$  such that  $\Omega = \Sigma \cap B_R(x_0)$  with  $c = g'(|\Omega|/|\Gamma_0|)$ ,  $R = n|\Omega|/|\Gamma_0|$ . Moreover u is given by (2.34), where  $x_0$  is the origin or, if  $\partial \Sigma$ contains flat regions, it is a point on  $\partial \Sigma$ .

When  $L_f = \Delta$  (i.e.  $f(t) = t^2/2$ ), Theorem 2.2 is essentially Theorem 1.1 in [180]. Condition (2.36) holds locally in  $\Omega$  for a large class of elliptic operators, such as the mean curvature operator  $(f(t) = \sqrt{1 + t^2})$ , and for the *p*-Laplace operator  $(f(t) = t^p/p, p > 1)$ , see [14, Theorem 4.1] and [54, Theorem 2.1]. We stress that the validity of (2.36) up to the boundary is more subtle, since it depends strongly on how  $\Gamma_0$  and  $\Gamma_1$  intersect. Some global results may be obtained by following the approach in [54], where (2.36) is proved for Dirichlet or Neumann boundary value problems of *p*-Laplace type in domains which are convex or satisfying minimal regularity assumptions on the boundary.

We observe that the overdetermined problem (2.32) with the condition (2.35) can be seen as a partially overdetermined problem (see for instance [94] and [95]), since we impose both Dirichlet and Neumann conditions only on a part of the boundary, namely  $\Gamma_0$ , while a sole homogeneous Neumann boundary condition is assigned on  $\Gamma_1$  (where, however, there is the strong assumption that it is contained in the boundary of a cone).

We notice that the proof of Theorem 2.2 still works when  $\Gamma_1 = \emptyset$  (hence  $\partial \Omega = \Gamma_0$ ). In this case we obtain the celebrated result of Serrin [204] for the operator  $L_f$  (see also [29, 40, 59, 92, 103, 106, 196, 215]). Moreover, the proof is also suitable to be adapted to the anisotropic counterpart of the overdetermined problem (2.32) and (2.35) by following our approach and in [29] (see also [55] and [213]). We also mention that rigidity theorems in cones are related to the

study of relative isoperimetric and Sobolev inequalities in cones, and we refer to [180] for a more detailed discussion (see also [21, 46, 97, 122, 160, 159]).

Serrin's problem in cones in space forms. A space form is a complete simply-connected Riemannian manifold (M, g) with constant sectional curvature K. Up to homotheties we may assume K = 0, 1, -1: the case K = 0 corresponds to the Euclidean space  $\mathbb{R}^n$ , K = -1 is the hyperbolic space  $\mathbb{H}^n$  and K = 1 is the unitary sphere with the round metric  $\mathbb{S}^n$ . More precisely, in the case K = 1we consider the hemisphere  $\mathbb{S}^n_+$ . These three models can be described as warped product spaces  $M = I \times \mathbb{S}^{n-1}$  equipped with the rotationally symmetric metric

$$g = dr^2 + h(r)^2 g_{\mathbb{S}^{n-1}}$$

where  $g_{\mathbb{S}^{n-1}}$  is the round metric on the (n-1)-dimensional sphere  $\mathbb{S}^{n-1}$  and

- h(r) = r in the Euclidean case (K = 0), with  $I = [0, \infty)$ ;
- $h(r) = \sinh(r)$  in the hyperbolic case (K = -1), with  $I = [0, \infty)$ ;
- $h(r) = \sin(r)$  in the spherical case (K = 1), with  $I = [0, \pi/2)$  for  $\mathbb{S}^n_+$ .

By using the warping structure of the manifold, we denote by O the pole of the model and it is natural to define a *cone*  $\Sigma$  *with vertex at*  $\{O\}$  as the set

$$\Sigma = \{ tx \, : \, x \in \omega, \, t \in I \}$$

for some open domain  $\omega \subset \mathbb{S}^{n-1}$ . Moreover, we say that  $\Sigma$  is a *convex cone* if the second fundamental form II is nonnegative defined at every  $p \in \partial \Sigma$ .

Serrin's overdetermined problem for semilinear equations  $\Delta u + f(u) = 0$  in space forms has been studied in [150] and [171] by using the method of moving planes. If one considers the corresponding problem for sector-like domains in space forms, the method of moving planes can not be used and one has to look for alternative approaches. As already mentioned, in the Euclidean space these approaches typically use integral identities and *P*-functions (see [40, 215]) and have the common feature that at a crucial step of the proof they use the fact that the radial solution attains the equality sign in a Cauchy-Schwartz inequality, which implies that the Hessian matrix  $\nabla^2 u$  is proportional to the identity. Since the equivalent crucial step in space forms is to prove that the Hessian matrix of the solution is proportional to the metric, then the equation  $\Delta u = -1$  is no more suitable (one can easily verify that in the radial case the Hessian matrix of the solution is not proportional to the metric) and a suitable equation to be considered is

$$\Delta u + nKu = -1 \tag{2.37}$$

as done in [68] and [189], [190]. It is clear that for K = 0, i.e. in the Euclidean case, the equation reduces to  $\Delta u = -1$ . For this reason, we believe that, in this setting, (2.37) is the natural generalization of the Euclidean  $\Delta u = -1$  to space forms.

A Serrin's type rigidity result for (2.37) can be proved following Weinberger's approach by using a suitable *P*-function associated to (2.37) (see [68] and [190]). This approach is helpful for proving the following Serrin's type rigidity result for convex cones in space forms, which is the second main result of this section.

**Theorem 2.3.** Let (M, g) be the Euclidean space, hyperbolic space or the hemisphere. Let  $\Sigma \subset M$  be a convex cone such that  $\Sigma \setminus \{O\}$  is smooth and let  $\Omega$  be a sector-like domain in  $\Sigma$ . Assume that there exists a solution  $u \in C^1(\Omega \cup \Gamma_0 \cup \Gamma_1 \setminus \{O\}) \cap W^{1,\infty}(\Omega) \cap W^{2,2}(\Omega)$  to

$$\begin{cases} \Delta u + nKu = -1 & in \ \Omega, \\ u = 0 & on \ \Gamma_0 \\ \partial_{\nu} u = 0 & on \ \Gamma_1 \setminus \{O\}, \end{cases}$$
(2.38)

such that

$$\partial_{\nu}u = -c \quad on \quad \Gamma_0 \tag{2.39}$$

for some constant c. Then  $\Omega = \Sigma \cap B_R(x_0)$  where  $B_R(x_0)$  is a geodesic ball of radius R centered at  $x_0$  and u is given by

$$u(x) = \frac{H(R) - H(d(x, x_0))}{n\dot{h}(R)},$$

with

$$H(r) = \int_0^r h(s) ds$$

and where  $d(x, x_0)$  denotes the distance between x and  $x_0$ .

The Section is organized as follows: in Subection 2.2.1 we introduce some notation, we recall some basic facts about elementary symmetric function of a matrix and prove some preliminary result needed to prove Theorem 2.2. Theorems 2.2 and 2.3 are proved in Subsections 2.2.2 and 2.2.3, respectively.

#### 2.2.1 Preliminary results for Theorem 2.2

In this section we collect some preliminary results which are needed in the proof of Theorem 2.2. Let f satisfy (2.31) and consider problem (2.32)

$$\begin{cases} L_f u = -1 & \text{in } \Omega, \\ u = 0 & \text{on } \Gamma_0 \\ \partial_{\nu} u = 0 & \text{on } \Gamma_1 \setminus \{O\} \end{cases}$$

where the operator  $L_f$  is given by

$$L_f u = \operatorname{div}\left(f'(|\nabla u|) \frac{\nabla u}{|\nabla u|}\right)$$

**Definition 2.2.**  $u \in C^1(\Omega \cup \Gamma_0 \cup \Gamma_1 \setminus \{O\})$  is a solution to Problem (2.32) if

$$\int_{\Omega} \frac{f'(|\nabla u|)}{|\nabla u|} \nabla u \cdot \nabla \varphi \, dx = \int_{\Omega} \varphi \, dx \tag{2.40}$$

•

for any

$$\varphi \in T(\Omega) := \{ \varphi \in C^1(\Omega) : \varphi \equiv 0 \text{ on } \Gamma_0 \}.$$
(2.41)

We observe some facts that will be useful in the following. Since the outward normal  $\nu$  to  $\Gamma_0$  is given by

$$\nu = -\frac{\nabla u}{|\nabla u|}|_{\Gamma_0}, \qquad (2.42)$$

then (2.35) implies that

$$|\nabla u| = c \quad \text{on} \quad \Gamma_0. \tag{2.43}$$

Moreover we observe that the constant c in the statement is given by

$$c = g'\left(\frac{|\Omega|}{|\Gamma_0|}\right),\tag{2.44}$$

as it follows by integrating the equation  $L_f u = -1$ , by using the divergence theorem, formula (2.43) and the fact that  $\partial_{\nu} u = 0$  on  $\Gamma_1 \setminus \{O\}$ . We also notice that

$$x \cdot \nu = 0 \quad \text{on} \quad \Gamma_1. \tag{2.45}$$

It will be useful to write the operator  $L_f$  as the trace of a matrix. Let  $V : \mathbb{R}^n \to \mathbb{R}$  be given by

$$V(\xi) = f(|\xi|) \quad \text{for } \xi \in \mathbb{R}^n, \tag{2.46}$$

and notice that

$$\partial_{\xi_{i}}V(\xi) := V_{\xi_{i}}(\xi) = f'(|\xi|)\frac{\xi_{i}}{|\xi|} \quad \text{and} \\ \partial_{\xi_{i}\xi_{j}}^{2}V(\xi) := V_{\xi_{i}\xi_{j}}(\xi) = f''(|\xi|)\frac{\xi_{i}\xi_{j}}{|\xi|^{2}} - f'(|\xi|)\left(\frac{\xi_{i}\xi_{j}}{|\xi|^{3}} - \frac{\delta_{ij}}{|\xi|}\right). \quad (2.47)$$

Hence, by setting

$$W = (w_{ij})_{i,j=1,\dots,N}$$

where

$$w_{ij}(x) = \partial_j V_{\xi_i}(\nabla u(x)), \qquad (2.48)$$

we have

$$L_f(u) = \operatorname{Tr}(W). \tag{2.49}$$

Notice that at regular points, where  $\nabla u \neq 0$ , it holds that

$$W = \nabla_{\xi}^2 V(\nabla u) \nabla^2 u \,. \tag{2.50}$$

Our approach to prove Theorem 2.2 is to write several integral identities and just one pointwise inequality, involving the matrix W. Writing the operator  $L_f$  as trace of W has the advantage that we can use the generalization of the so-called Newton's inequalities, as explained in the following subsection.

#### Elementary symmetric functions of a matrix

Given a matrix  $A = (a_{ij}) \in \mathbb{R}^{n \times n}$ , for any  $k = 1, \ldots, n$  we denote by  $S_k(A)$  the sum of all the principal minors of A of order k. In particular,  $S_1(A) = \text{Tr}(A)$  is the trace of A, and  $S_n(A) = \det(A)$  is the determinant of A. We consider the case k = 2. By setting

$$S_{ij}^2(A) = -a_{ji} + \delta_{ij} \operatorname{Tr} (A), \qquad (2.51)$$

we can write

$$S_2(A) = \frac{1}{2} \sum_{i,j} S_{ij}^2(A) a_{ij} = \frac{1}{2} \left( (\operatorname{Tr}(A)^2 - \operatorname{Tr}(A^2)) \right).$$
 (2.52)

The elementary symmetric functions of a symmetric matrix A satisfy the so called Newton's inequalities. In particular,  $S_1$  and  $S_2$  are related by (cfr. (I.13) in the Introduction to Part I)

$$S_2(A) \le \frac{n-1}{2n} (S_1(A))^2.$$
 (2.53)

When the matrix A = W, with W given by (2.50), we have

$$S_{ij}^2(W) = -V_{\xi_j\xi_k}(\nabla u)\partial_{ki}^2 u + \delta_{ij}L_f u, \qquad (2.54)$$

and  $S_{ij}^2(W)$  is divergence free in the following (weak) sense

$$\frac{\partial}{\partial x_j} S_{ij}^2(W) = 0. \qquad (2.55)$$

If V and u are sufficiently smooth, (2.55) was proved in [55, Equation (4.14)]. In Lemma 2.7 below we will prove (2.55) under weaker regularity assumptions on V and u by approximation (notice that (2.55) is implicitly written in (2.70), as follows from (2.52)).

We will need a generalization of (2.53) to not necessarily symmetric matrices, which is given by the following lemma.

**Lemma 2.3** ([55], Lemma 3.2). Let B and C be symmetric matrices in  $\mathbb{R}^{n \times n}$ , and let B be positive semidefinite. Set A = BC. Then the following inequality holds:

$$S_2(A) \le \frac{n-1}{2n} \operatorname{Tr}(A)^2.$$
 (2.56)

Moreover, if  $Tr(A) \neq 0$  and equality holds in (2.56), then

$$A = \frac{\operatorname{Tr}\left(A\right)}{n} \operatorname{Id},$$

and B is, in fact, positive definite.

#### Some properties of solutions to (2.32)

In this subsection we collect some properties of the solutions to (2.32). We assume that the solution is of class  $C^1(\Omega) \cap W^{1,\infty}(\Omega)$ . From standard regularity elliptic estimates one has that u is of class  $C^{1,\alpha}(\Omega)$  and  $C^{2,\alpha}$  where  $\nabla u \neq 0$ .

In the following two lemmas we show that u > 0 in  $\Omega \cup \Gamma_1 \setminus \{O\}$  and we prove a Pohozaev-type identity.

Lemma 2.4. Let f satisfy (2.31) and let u be a solution of (2.32). Then

$$u > 0 \quad in \quad \Omega \cup \Gamma_1 \setminus \{O\}. \tag{2.57}$$

*Proof.* We write  $u = u^+ - u^-$  and use  $\varphi = u^-$  as test function in (2.40):

$$0 \ge -\int_{\Omega \cap \{u<0\}} \frac{f'(|\nabla u|)}{|\nabla u|} \, |\nabla u^-|^2 \, dx = \int_{\Omega \cap \{u<0\}} u^- \, dx \ge 0 \,,$$

which implies that  $u \ge 0$  in  $\Omega$ . Moreover, if one assumes that  $u(x_0) = 0$  at some point  $x_0 \in \Omega \cup \Gamma_1 \setminus \{O\}$ , then  $\nabla u(x_0) = 0$ . Since  $x_0 \in \Omega \cup \Gamma_1 \setminus \{O\}$  and  $\Gamma_1 \setminus \{O\}$ is smooth, there exists a ball  $B_r \subset \Omega$  such that  $x_0 \in \partial B_r$ . Let v be the solution of

$$\begin{cases} L_f v = -1 & \text{in } B_r, \\ v = 0 & \text{on } \partial B_r \end{cases}$$

By comparison principle we have that  $v \leq u$  in  $\overline{B}_r$ ; from  $\nabla u(x_0) = 0$  and since  $\nabla v(x_0) \neq 0$  we get a contradiction.

The following Pohozaev-type identity is a typical tool to prove symmetry results. In a similar setting as the one in this paper, a Pohozaev identity was proved in [106].

**Lemma 2.5** (Pohozaev-type identity). Let  $\Omega$  be a sector-like domain and assume that f satisfies (2.31). Let  $u \in C^1(\Omega \cup \Gamma_0 \cup \Gamma_1 \setminus \{O\}) \cap W^{1,\infty}(\Omega)$  be a solution to (2.32). Then the following integral identity

$$\int_{\Omega} \left[ (n+1)u - nf(|\nabla u|) \right] dx = \int_{\Gamma_0} \left[ f'(|\nabla u|) |\nabla u| - f(|\nabla u|) \right] x \cdot \nu \, d\sigma \qquad (2.58)$$

holds.

*Proof.* We argue by approximation. We first approximate f with functions  $f_{\varepsilon}$  such that

$$f_{\varepsilon} \in C^{\infty}([0,\infty))$$
 with  $f_{\varepsilon}(0) = f'_{\varepsilon}(0) = 0$ ,  $f''_{\varepsilon}(s) > 0$  for  $s \ge 0$ , (2.59)

and

$$f_{\varepsilon} \to f$$
 and  $f'_{\varepsilon} \to f'$  uniformly on compact sets of  $[0, +\infty)$ . (2.60)

We notice that such an approximation exists as shown in [106, Section 3].

We recall that  $V(\xi) = f(|\xi|)$  (see (2.46)) for  $\xi \in \mathbb{R}^n$ , and we define  $V^{\varepsilon}$ :  $\mathbb{R}^n \to \mathbb{R}$  as

$$V^{\varepsilon}(\xi) := f_{\varepsilon}(|\xi|).$$

We notice that  $\nabla_{\xi} V^{\varepsilon}$  and  $\nabla_{\xi} V$  can be continuously extended to 0 at  $\xi = 0$ .

We approximate  $\Omega$  by domains  $\Omega_{\delta}$  obtained by chopping off a  $\delta$ -tubular neighborhood of  $\partial \Gamma_0$  and a  $\delta$ -neighborhood of O. For  $j \in \mathbb{N}$ , we consider  $u_{\delta}^j \in C^{\infty}(\Omega_{\delta}) \cap C^1(\overline{\Omega}_{\delta})$  such that

$$u^j_{\delta} \to u \text{ in } C^1(\overline{\Omega}_{\delta}),$$

as j goes to infinity (see for instance [43, Section 2.6]). Since

$$\begin{split} \operatorname{div} \left( x \cdot \nabla u_{\delta}^{j} \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \right) &= x \cdot \nabla u_{\delta}^{j} \operatorname{div} \left( \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \right) \\ &+ \nabla (x \cdot \nabla u_{\delta}^{j}) \cdot \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \end{split}$$

and from

$$\begin{split} \nabla(x \cdot \nabla u_{\delta}^{j}) \cdot \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) = & \nabla u_{\delta}^{j} \cdot \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) + x \nabla^{2}(u_{\delta}^{j}) \cdot \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \\ = & \operatorname{div} \left( u_{\delta}^{j} \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \right) - u_{\delta}^{j} \operatorname{div} \left( \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \right) \\ & + \operatorname{div} (x V^{\varepsilon}(\nabla u_{\delta}^{j})) - n V^{\varepsilon}(\nabla u_{\delta}^{j}) \,, \end{split}$$

we obtain

$$\operatorname{div}\left(\varphi_{j}\nabla_{\xi}V^{\varepsilon}(\nabla u_{\delta}^{j}) - xV^{\varepsilon}(\nabla u_{\delta}^{j})\right) = \varphi_{j}\operatorname{div}\left(\nabla_{\xi}V^{\varepsilon}(\nabla u_{\delta}^{j})\right) - nV^{\varepsilon}(\nabla u_{\delta}^{j}), \quad (2.61)$$

where

$$\varphi_j(x) = x \cdot \nabla u^j_{\delta}(x) - u^j_{\delta}(x)$$

Moreover, from the divergence theorem we have

$$\int_{\Omega_{\delta}} \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \cdot \nabla \varphi_{j} \, dx = -\int_{\Omega_{\delta}} \varphi_{j} \operatorname{div} \left( \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \right) \, dx \\ + \int_{\partial \Omega_{\delta}} \varphi_{j} \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \cdot \nu \, d\sigma \,.$$

$$(2.62)$$

We are going to apply the divergence theorem in  $\Omega_{\delta}$ ; to this end we set

 $\Gamma_{0,\delta} = \Gamma_0 \cap \partial \Omega_\delta \,, \quad \Gamma_{1,\delta} = \Gamma_1 \cap \partial \Omega_\delta \quad \text{and} \quad \Gamma_\delta = \partial \Omega_\delta \setminus \left( \Gamma_{0,\delta} \cup \Gamma_{1,\delta} \right).$ 

From (2.62) and by integrating (2.61) in  $\Omega_{\delta}$  we obtain

$$\begin{split} \int_{\Omega_{\delta}} \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \cdot \nabla \varphi_{j} \, dx &= -n \int_{\Omega_{\delta}} V^{\varepsilon}(\nabla u_{\delta}^{j}) \, dx \\ &- \int_{\Omega_{\delta}} \operatorname{div} \left( \varphi_{j} \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \right) \, dx \\ &+ \int_{\Omega_{\delta}} \operatorname{div} \left( x V^{\varepsilon}(\nabla u_{\delta}^{j}) \right) \, dx \,, \end{split}$$

and from  $x \cdot \nu = 0$  on  $\Gamma_{1,\delta}$ , we find

$$\begin{split} \int_{\Omega_{\delta}} \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \cdot \nabla \varphi_{j} \, dx &= -n \int_{\Omega_{\delta}} V^{\varepsilon}(\nabla u_{\delta}^{j}) \, dx \\ &- \int_{\Gamma_{0,\delta} \cup \Gamma_{1,\delta}} \varphi_{j} \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) \cdot \nu \, d\sigma \\ &+ \int_{\Gamma_{0,\delta}} V^{\varepsilon}(\nabla u_{\delta}^{j}) x \cdot \nu \, d\sigma \\ &- \int_{\Gamma_{\delta}} [\varphi_{j} \nabla_{\xi} V^{\varepsilon}(\nabla u_{\delta}^{j}) - x V^{\varepsilon}(\nabla u_{\delta}^{j})] \cdot \nu \, d\sigma \end{split}$$

By taking the limit as  $\varepsilon \to 0$  and then as  $j \to \infty$ , using that  $\nabla u \cdot \nu = 0$  on  $\Gamma_{1,\delta}$ 

•

(since  $\partial_{\nu} u = 0$  on  $\Gamma_1$ ), we obtain

$$\int_{\Omega_{\delta}} \nabla_{\xi} V(\nabla u) \cdot \nabla \varphi \, dx = -n \int_{\Omega_{\delta}} V(\nabla u) \, dx$$

$$- \int_{\Gamma_{0,\delta}} \varphi \nabla_{\xi} V(\nabla u) \cdot \nu \, d\sigma$$

$$+ \int_{\Gamma_{0,\delta}} V(\nabla u) x \cdot \nu \, d\sigma$$

$$- \int_{\Gamma_{\delta}} [\varphi \nabla_{\xi} V(\nabla u) - x V(\nabla u)] \cdot \nu \, d\sigma$$
(2.63)

where we let

$$\varphi(x) = x \cdot \nabla u(x) - u(x). \qquad (2.64)$$

Now, we take the limit as  $\delta \to 0$ . Since  $u \in W^{1,\infty}(\Omega)$  and  $\mathcal{H}_{n-1}(\Gamma_{\delta})$  goes to 0 as  $\delta \to 0$ , we have that the last term in (2.63) vanishes and we obtain

$$\int_{\Omega} \nabla_{\xi} V(\nabla u) \cdot \nabla \varphi \, dx = -n \int_{\Omega} V(\nabla u) \, dx$$
$$- \int_{\Gamma_0} \varphi \nabla_{\xi} V(\nabla u) \cdot \nu \, d\sigma + \int_{\Gamma_0} V(\nabla u) x \cdot \nu \, d\sigma \,,$$

i.e. (in terms of f)

$$\int_{\Omega} \frac{f'(|\nabla u|)}{|\nabla u|} \nabla u \cdot \nabla \varphi \, dx = -n \int_{\Omega} f(|\nabla u|) \, dx$$
$$- \int_{\Gamma_0} \varphi \frac{f'(|\nabla u|)}{|\nabla u|} \partial_{\nu} u \, d\sigma + \int_{\Gamma_0} f(|\nabla u|) x \cdot \nu \, d\sigma.$$

Since u satisfies (2.40), we get

$$\int_{\Omega} \varphi \, dx = -n \int_{\Omega} f(|\nabla u|) \, dx$$
$$- \int_{\Gamma_0} \varphi \frac{f'(|\nabla u|)}{|\nabla u|} \partial_{\nu} u \, d\sigma + \int_{\Gamma_0} f(|\nabla u|) x \cdot \nu \, d\sigma. \quad (2.65)$$

From (2.64) and since u = 0 on  $\Gamma_0$  and  $\partial_{\nu} u = 0$  on  $\Gamma_1$ , we have

$$\int_{\Omega} \varphi \, dx = -(n+1) \int_{\Omega} u \, dx$$

 $\operatorname{and}$ 

$$\int_{\Gamma_0} \varphi \frac{f'(|\nabla u|)}{|\nabla u|} \partial_\nu u \, d\sigma = \int_{\Gamma_0} f'(|\nabla u|) |\nabla u| x \cdot \nu \, d\sigma \,, \tag{2.66}$$

where we used the expression of the unit exterior normal on  $\Gamma_0$  given by (2.42). From (2.66) and (2.65) we obtain

$$- (n+1) \int_{\Omega} u \, dx + n \int_{\Omega} f(|\nabla u|) \, dx = - \int_{\Gamma_0} f'(|\nabla u|) |\nabla u| x \cdot \nu \, d\sigma$$
$$+ \int_{\Gamma_0} f(|\nabla u|) x \cdot \nu \, d\sigma.$$

which is (2.58), and the proof is complete.

We conclude this subsection by exploiting the boundary condition  $\partial_{\nu} u = 0$ on  $\Gamma_1$ . Before doing this, we need to recall some notation from differential geometry (see also [88, Appendix A]). We denote by D the standard Levi-Civita connection. Recall that, given an (n-1)-dimensional smooth orientable submanifold M of  $\mathbb{R}^n$  we define the *tangential gradient* of a smooth function  $f: M \to \mathbb{R}$  with respect to M as

$$\nabla^T f(x) = \nabla f(x) - \nu \cdot \nabla f(x)\nu$$

for  $x \in M$ , where  $\nabla f$  denotes the usual gradient of f in  $\mathbb{R}^n$  and  $\nu$  is the outward unit normal at x to M. Moreover, we recall that the *second fundamental form* of M is the bilinear and symmetric form defined on  $TM \times TM$  as

$$II(v,w) = D\nu(v)w \cdot \nu;$$

a submanifold is called convex if the second fundamental form is non-negative definite.

Lemma 2.6. Let u be the solution to (2.32). Then

$$\nabla_{\xi} V(\nabla u) \cdot \nu = 0 \quad on \quad \Gamma_1 \,, \tag{2.67}$$

and

$$\nabla(\nabla_{\xi} V(\nabla u) \cdot \nu) \cdot \nabla u = 0 \quad on \quad \Gamma_1 .^1$$
(2.68)

*Proof.* Since  $\partial_{\nu} u = 0$  on  $\Gamma_1$ , we immediately find (2.67). By taking the tangential derivative in (2.67) we get

$$0 = \nabla^T (\nabla_{\xi} V(\nabla u) \cdot \nu) = \nabla (\nabla_{\xi} V(\nabla u) \cdot \nu) - \nu \cdot \nabla (\nabla_{\xi} V(\nabla u) \cdot \nu) \nu \quad \text{on} \quad \Gamma_1 \,.$$

By taking the scalar product with  $\nabla u$  we obtain

$$0 = \nabla (\nabla_{\xi} V(\nabla u) \cdot \nu) \cdot \nabla u - \nu \cdot \nabla (\nabla_{\xi} V(\nabla u) \cdot \nu) \partial_{\nu} u \,,$$

and since  $\partial_{\nu} u = 0$  on  $\Gamma_1$ , we find (2.68).

#### Integral Identities for $S_2$

In this Subsection we prove some integral inequalities involving  $S_2(W)$  and the solution to problem (2.32).

**Lemma 2.7.** Let  $\Omega \subset \mathbb{R}^n$  be a sector-like domain and assume that f satisfies (2.31). Let  $u \in W^{1,\infty}(\Omega)$  be a solution of (2.32) such that (2.36) holds. Then the following inequality

$$2\int_{\Omega} S_2(W)u\,dx \ge -\int_{\Omega} S_{ij}^2(W)V_{\xi_i}(\nabla u)\partial_j u\,dx \tag{2.69}$$

holds. Moreover the equality sign holds in (2.69) if and only if  $II(\nabla^T u, \nabla^T u) = 0$ on  $\Gamma_1$ .

<sup>&</sup>lt;sup>1</sup>We remark that (2.68) is understood to be zero at points where  $\nabla u = 0$ .

*Proof.* We split the proof in two steps.

Step 1: the following identity

$$2\int_{\Omega} S_2(W)\phi \, dx = -\int_{\Omega} S_{ij}^2(W) V_{\xi_i}(\nabla u) \partial_j \phi \, dx \tag{2.70}$$

holds for every  $\phi \in C_0^1(\Omega)$ .

For t > 0 we set  $\Omega_t = \{x \in \Omega : \text{dist}(x, \partial \Omega) > t\}$ . Let  $\phi \in C_0^1(\Omega)$  be a test function and let  $\varepsilon_0 > 0$  be such that  $\Omega_{\varepsilon_0} \subset \Omega$  and  $supp(\phi) \subset \Omega_{\varepsilon_0}$ . For  $\varepsilon < \varepsilon_0$  sufficiently small, we set

$$a^i(x) = V_{\xi_i}(\nabla u(x))$$
 for every  $i = 1, \dots, n, x \in \Omega$ .

From (2.36) we have that  $a^i \in W^{1,2}(\Omega)$ ,  $i = 1, \ldots, n$ . With this notation, the elements  $w_{ij} = \partial_j V_{\xi_i}(\nabla u)$  of the matrix W are given by

$$w_{ij} = \partial_j a^i$$

Let  $\{\rho_{\varepsilon}\}$  be a family of mollifiers and define  $a_{\varepsilon}^{i} = a^{i} * \rho_{\varepsilon}$ . Let  $W^{\varepsilon} = (w_{ij}^{\varepsilon})_{i,j=1,...,n}$ where  $w_{ij}^{\varepsilon} = \partial_{j} a_{\varepsilon}^{i}$ , and notice that

$$a^i_{\varepsilon} \to a^i \quad \text{in } W^{1,2}(\Omega_{\varepsilon_0}) \quad \text{and} \quad W^{\varepsilon} \to W \quad \text{in } L^2(\Omega_{\varepsilon_0}) \,,$$

as  $\varepsilon \to 0$ . Moreover, since

Tr 
$$W^{\varepsilon}(x) = \int_{\mathbb{R}^n} \rho_{\varepsilon}(y) \operatorname{Tr} W(x-y) \, dy$$

and Tr W = -1, we have that

$$\operatorname{Tr} W^{\varepsilon}(x) = -1 \tag{2.71}$$

for every  $x \in \Omega_{\varepsilon}$ .

Let  $i, j = 1, \ldots, n$  be fixed. We have

$$\begin{split} w_{ji}^{\varepsilon} w_{ij}^{\varepsilon} &= \partial_j (a_{\varepsilon}^i \partial_i a_{\varepsilon}^j) - a_{\varepsilon}^i \partial_j \partial_i a_{\varepsilon}^j \\ &= \partial_j (a_{\varepsilon}^i \partial_i a_{\varepsilon}^j) - a_{\varepsilon}^i \partial_i \partial_j a_{\varepsilon}^j \\ &= \partial_j (a_{\varepsilon}^i \partial_i a_{\varepsilon}^j) - a_{\varepsilon}^i \partial_i w_{jj}^{\varepsilon} \,, \end{split}$$

for every  $x \in \Omega_{\varepsilon}$ , and by summing up over j = 1, ..., n, using (2.71) (hence  $\partial_i \sum_j w_{jj}^{\varepsilon} = 0$ ), we obtain

$$\begin{split} \sum_{j} w_{ji}^{\varepsilon} w_{ij}^{\varepsilon} &= \sum_{j} \partial_{j} (a_{\varepsilon}^{i} \partial_{i} a_{\varepsilon}^{j}) \\ &= w_{ii}^{\varepsilon} \operatorname{Tr} W^{\varepsilon} - \sum_{j} \partial_{j} (S_{ij}^{2} (W^{\varepsilon}) a_{\varepsilon}^{i}), \quad x \in \Omega_{\varepsilon}. \end{split}$$

By summing over i = 1, ..., n, from (2.52) and (2.55) we have

$$2S_2(W^{\varepsilon}) = \sum_{i,j} \partial_j (S_{ij}^2(W^{\varepsilon})a_{\varepsilon}^i), \quad x \in \Omega_{\varepsilon}.$$
 (2.72)

Since

$$\begin{split} \int_{\Omega_{\varepsilon_0}} \partial_j (S_{ij}^2(W^{\varepsilon})a_{\varepsilon}^i)\phi\,dx + \int_{\Omega_{\varepsilon_0}} S_{ij}^2(W^{\varepsilon})a_{\varepsilon}^i\partial_j\phi\,dx \\ &= \int_{\partial\Omega_{\varepsilon_0}} S_{ij}^2(W^{\varepsilon})a_{\varepsilon}^i\nu_j\phi\,d\sigma = 0\,, \end{split}$$

from (2.72) and by letting  $\varepsilon$  to zero, we obtain (2.70).

Step 2. Let  $\delta > 0$  and consider a cut-off function  $\eta^{\delta} \in C_0^{\infty}(\Omega)$  such that  $\eta^{\delta} = 1$  in  $\Omega_{\delta}$  and  $|\nabla \eta^{\delta}| \leq \frac{C}{\delta}$  in  $\Omega \setminus \Omega_{\delta}$  for some constant C not depending on  $\delta$ . By taking  $\phi(x) = u(x)\eta^{\delta}(x)$  for  $x \in \Omega$  in (2.70) we obtain

$$2\int_{\Omega} S_2(W)u\eta^{\delta} dx = -\int_{\Omega} S_{ij}^2(W)V_{\xi_i}(\nabla u)\partial_j u\eta^{\delta} dx$$
$$-\int_{\Omega} S_{ij}^2(W)V_{\xi_i}(\nabla u)u\partial_j(\eta^{\delta}) dx. \quad (2.73)$$

From (I.34) we have that  $W \in L^2(\Omega)$  and the dominated convergence theorem yields

$$2\int_{\Omega} S_2(W) u\eta^{\delta} \, dx \to 2\int_{\Omega} S_2(W) u \, dx \tag{2.74}$$

as  $\delta \to 0$ . Analogously,

$$\int_{\Omega} S_{ij}^2(W) V_{\xi_i}(\nabla u) \partial_j u \eta^{\delta} \, dx \to \int_{\Omega} S_{ij}^2(W) V_{\xi_i}(\nabla u) \partial_j u \, dx \tag{2.75}$$

as  $\delta \to 0$ .

Now, we consider the last term in (2.73). We write  $\Omega$  in the following way:

$$\Omega = A_0^\delta \cup A_1^\delta \,, \tag{2.76}$$

where

$$A_0^{\delta} = \{x \in \Omega : \operatorname{dist}(x, \Gamma_0) \le \delta\}$$
 and  $A_1^{\delta} = \Omega \setminus A_0^{\delta}$ 

Since u = 0 on  $\Gamma_0$ , we get that

$$u(x) \le ||u||_{W^{1,\infty}(\Omega)} \operatorname{dist} (x, \Gamma_0) \le ||u||_{W^{1,\infty}(\Omega)} \delta$$

for every  $x \in A_0^{\delta}$  and we obtain

$$\left| \int_{A_0^{\delta}} S_{ij}^2(W) V_{\xi_i}(\nabla u) u \partial_j(\eta^{\delta}) \, dx \right| \le C_1 |A_0^{\delta}| \,,$$

where  $C_1$  is a constant depending on  $||u||_{W^{1,\infty}(\Omega)}$  and  $||W||_{L^2(\Omega)}$ , which implies that

$$\lim_{\delta \to 0} \int_{A_0^{\delta}} S_{ij}^2(W) V_{\xi_i}(\nabla u) u \partial_j(\eta^{\delta}) \, dx = 0 \,.$$
(2.77)

Now we show that

$$\lim_{\delta \to 0} \int_{A_1^{\delta}} S_{ij}^2(W(x)) V_{\xi_i}(\nabla u(x)) u(x) \partial_j(\eta^{\delta})(x) \, dx \ge 0 \,. \tag{2.78}$$

By choosing  $\delta$  small enough, a point  $x \in A_1^{\delta}$  can be written in the following way:  $x = \bar{x} + t\nu(\bar{x})$  where  $\bar{x} = \bar{x}(x) \in \Gamma_1$  and  $t = |x - \bar{x}|$  with  $0 < t < \delta$ . Moreover, by using a standard approximation argument,  $\eta^{\delta}$  can be chosen in such a way that  $\eta^{\delta}(x) = \frac{1}{\delta} \text{dist}(x, \Gamma_1)$  for any  $x \in A_1^{\delta}$ , so that

$$\nabla \eta^{\delta}(x) = -\frac{1}{\delta} \nu(\bar{x}) , \qquad (2.79)$$

for every  $x \in A_1^{\delta} \setminus \Omega_{\delta}$ . For simplicity of notation we set  $F = (F_1, \ldots, F_n)$ , where

$$F_j(x) = u(x)S_{ij}^2(W(x))V_{\xi_i}(\nabla u(x))$$
(2.80)

for  $j = 1, \ldots, n$ , and hence

$$\int_{A_1^{\delta}} S_{ij}^2(W) V_{\xi_i}(\nabla u) u \partial_j(\eta^{\delta}) \, dx = \int_{A_1^{\delta}} F(x) \cdot \nabla \eta^{\delta}(x) \, dx \,. \tag{2.81}$$

Since  $\nabla \eta^{\delta} = 0$  in  $\Omega_{\delta}$  and  $\nabla \eta^{\delta}(x) = -\frac{1}{\delta}\nu(\bar{x})$ , for every  $x \in A_1^{\delta} \setminus \Omega_{\delta}$ , we have

$$\begin{split} \int_{A_1^{\delta}} F(x) \cdot \nabla \eta^{\delta}(x) \, dx &= -\frac{1}{\delta} \int_{A_1^{\delta} \setminus \Omega_{\delta}} F(x) \cdot \nu(\bar{x}) \, dx \\ &= -\frac{1}{\delta} \int_0^{\delta} \, dt \int_{\{x \in A_1^{\delta} : \, \text{dist} \, (x, \Gamma_1) = t\}} F(x) \cdot \nu(\bar{x}) \, d\sigma \end{split}$$

where we used coarea formula. Since we are in a *small*  $\delta$ -tubular neighborhood of (part of)  $\Gamma_1$ , we can parametrize  $A_1^{\delta} \setminus \Omega_{\delta}$  over (part of)  $\Gamma_1$  as from [112, Formula 14.98] we obtain that

$$\int_{A_1^{\delta}} F(x) \cdot \nabla \eta^{\delta}(x) \, dx = -\frac{1}{\delta} \int_0^{\delta} dt \int_{\Gamma_1} F(\bar{x} + t\nu(\bar{x})) \cdot \nu(\bar{x}) |\det(Dg)| \, d\sigma \,. \tag{2.82}$$

We notice that, by using this notation, proving (2.78) is equivalent to prove

$$\lim_{\delta \to 0} \int_{A_1^{\delta}} F(x) \cdot \nabla \eta^{\delta}(x) \, dx \ge 0 \,, \tag{2.83}$$

for  $\delta > 0$  sufficiently small.

From (2.79), (2.80) and the definition of  $S_{ij}^2$  (2.51), we have

$$F(x) \cdot \nu(\bar{x}) = -\delta_{ij} V_{\xi_i}(\nabla u(x))u(x)\nu_j(\bar{x}) - w_{ji}(x)V_{\xi_i}(\nabla u(x))u(x)\nu_j(\bar{x})$$
  
$$= -\delta_{ij} V_{\xi_i}(\nabla u(x))u(x)\nu_j(\bar{x})$$
  
$$- u(x)\frac{f'(|\nabla u(x)|)}{|\nabla u(x)|}w_{ji}(x)\partial_i u(x)\nu_j(\bar{x})$$

for almost every  $x = \bar{x} + t\nu(\bar{x}) \in A_1^{\delta} \setminus \Omega_{\delta}$ , with  $0 \le t \le \delta$ . Since

$$w_{ij}\nu_i\partial_j u = \partial_j (V_{\xi_i}(\nabla u)\nu_i)\partial_j u - V_{\xi_i}(\nabla u)\partial_j\nu_i\partial_j u \,,$$

we have

$$F(x) \cdot \nu(\bar{x}) = -u(x)\nabla_{\xi}V(\nabla u(x)) \cdot \nu(\bar{x}) - u(x)\frac{f'(|\nabla u(x)|)}{|\nabla u(x)|} \times \left\{\nabla(\nabla_{\xi}V(\nabla u(x)) \cdot \nu(\bar{x})) \cdot \nabla u(x) - \frac{f'(|\nabla u(x)|)}{|\nabla u(x)|}\partial_{j}\nu_{i}(\bar{x})\partial_{j}u(x)\partial_{i}u(x)\right\}$$
(2.84)

for almost every  $x = \bar{x} + t\nu(\bar{x}) \in A_1^{\delta} \setminus \Omega_{\delta}$ , with  $0 \le t \le \delta$ . Let

$$\Gamma_1^{\delta,t} = \{ x \in A_1^\delta : \operatorname{dist}(x,\Gamma_1) = t \}.$$

We notice that if  $x \in \Gamma_1^{\delta,t}$  then  $\nu(\bar{x}) = \nu^t(x)$  where  $\nu^t(x)$  is the outward normal to  $\Gamma_1^{\delta,t}$  at x. Hence

$$\partial_j \nu_i(\bar{x}) \partial_j u(x) \partial_i u(x) = \Pi_x^{\delta, t}(\nabla^T u(x), \nabla^T u(x))$$
(2.85)

where  $\Pi_x^{\delta,t}$  is the second fundamental form of  $\Gamma_1^{\delta,t}$  at x. Since  $\Sigma$  is a convex cone then the second fundamental form of  $\Gamma_1 \setminus \{O\}$  is non-negative definite. This implies that the second fundamental form of  $\Gamma_1^{\delta,t}$  is non-negative definite for tsufficiently small (see e.g.[112, Appendix 14.6]) and hence

$$\partial_j \nu_i(\bar{x}) \partial_j u(x) \partial_i u(x) \ge 0.$$
 (2.86)

From (2.86) and (2.84) we obtain

$$F(x) \cdot \nu(\bar{x}) \ge -u(x)\nabla_{\xi}V(\nabla u(x)) \cdot \nu(\bar{x}) -u(x)\frac{f'(|\nabla u(x)|)}{|\nabla u(x)|}\nabla(\nabla_{\xi}V(\nabla u(x)) \cdot \nu(\bar{x})) \cdot \nabla u(x) \quad (2.87)$$

for almost every  $x = \bar{x} + t\nu(\bar{x}) \in A_1^{\delta} \setminus \Omega_{\delta}$ , with  $0 \le t \le \delta$ . We use (2.87) in the right-hand side of (2.82) and, by taking the limit as  $\delta \to 0$ , we obtain

$$\begin{split} \lim_{\delta \to 0} \int_{A_1^{\delta}} F(x) \cdot \nabla \eta^{\delta}(x) \, dx \\ \geq - \int_{\Gamma_1} u \left( \nabla_{\xi} V(\nabla u) \cdot \nu + \frac{f'(|\nabla u|)}{|\nabla u|} \nabla (\nabla_{\xi} V(\nabla u) \cdot \nu) \cdot \nabla u \right) d\sigma \, . \end{split}$$

From (2.67) and (2.68) we find (2.83), and hence (2.78). From (2.73), (2.74), (2.75), (2.76), (2.77) and (2.78), we obtain (2.69).

#### 2.2.2 Proof of Theorem 2.2

Proof of Theorem 2.2. We divide the proof in two steps. We first show that

$$W = -\frac{1}{n} \text{Id} \quad \text{a.e. in } \Omega. \tag{2.88}$$

 $\operatorname{and}$ 

$$II(\nabla^T u, \nabla^T u) = 0 \quad \text{on } \Gamma_1, \qquad (2.89)$$

and then we exploit (2.88) in order to prove that u is indeed radial.

Step 1. Let g be the Fenchel conjugate of f (in our case  $g' = (f')^{-1}$ ), using (2.47) we get that

$$\begin{aligned} \operatorname{div}\left(g(|\nabla_{\xi}V(\nabla u)|)\nabla_{\xi}V(\nabla u)\right) &= g'(|\nabla_{\xi}V(\nabla u)|)\nabla|\nabla_{\xi}V(\nabla u)|V_{\xi}(\nabla u) \\ &+ g(|\nabla_{\xi}V(\nabla u)|)\operatorname{Tr}\left(W\right) \\ &= g'(f'(|\nabla u|))\frac{V_{\xi_{i}}(\nabla u)}{|\nabla_{\xi}V(\nabla u)|}\partial_{j}(V_{\xi_{i}}(\nabla u))V_{\xi_{j}}(\nabla u) \\ &+ g(f'(|\nabla u|))\operatorname{Tr}\left(W\right), \end{aligned}$$

a.e. in  $\Omega$ , where we used (2.47). Since  $\partial_j V_{\xi_i}(\nabla u) = w_{ij}$  and  $g' = (f')^{-1}$ , we obtain

div 
$$(g(|\nabla_{\xi}V(\nabla u)|)\nabla_{\xi}V(\nabla u)) = \partial_{i}uw_{ij}V_{\xi_{j}}(\nabla u) + g(f'(|\nabla u|))\mathrm{Tr}(W)$$

a.e. in  $\Omega$ , and using again (2.47) we find

$$\operatorname{div}\left(g(|\nabla_{\xi}V(\nabla u)|)\nabla_{\xi}V(\nabla u)\right) = \frac{f'(|\nabla u|)}{|\nabla u|}\partial_{i}uw_{ij}\partial_{j}u + g(f'(|\nabla u|))\operatorname{Tr}(W)$$

a.e. in  $\Omega$ . Since

$$g(f'(t)) = tf'(t) - f(t)$$
(2.90)

and  $\operatorname{Tr}(W) = -1$ , we obtain

$$\operatorname{div}\left(g(|\nabla_{\xi}V(\nabla u)|)\nabla_{\xi}V(\nabla u)\right) = \frac{f'(|\nabla u|)}{|\nabla u|}\partial_{i}uw_{ij}\partial_{j}u + f(|\nabla u|) - |\nabla u|f'(|\nabla u|) \quad (2.91)$$

a.e. in  $\Omega.$ 

Since (2.54), (2.47) and (2.49) yield

$$-S_{ij}^{2}(W)V_{\xi_{i}}(\nabla u)\partial_{j}u = \frac{f'(|\nabla u|)}{|\nabla u|}w_{ji}\partial_{i}u\partial_{j}u + f'(|\nabla u|)|\nabla u|,$$

a.e. in  $\Omega$ , from (2.91) we obtain

$$-S_{ij}^{2}(W)V_{\xi_{i}}(\nabla u)\partial_{j}u = \operatorname{div}\left(g(|\nabla_{\xi}V(\nabla u)|)\nabla_{\xi}V(\nabla u)\right) + 2f'(|\nabla u|)|\nabla u| - f(|\nabla u|), \quad (2.92)$$

a.e. in  $\Omega$ .

From Lemma 2.7 and (2.92), we obtain

$$\begin{split} 2\int_{\Omega}S_2(W)u\,dx \geq &-\int_{\Omega}S_{ij}^2(W)V_{\xi_i}(\nabla u)\partial_j u\,dx\\ &=\int_{\partial\Omega}g(|\nabla_{\xi}V(\nabla u)|)\nabla_{\xi}V(\nabla u)\cdot\nu\,d\sigma\\ &+\int_{\Omega}\left[2f'(|\nabla u|)|\nabla u|-f(|\nabla u|)\right]\,dx\,. \end{split}$$

From (2.47) and (2.67) we find

$$2\int_{\Omega} S_2(W)u \, dx \ge \int_{\Gamma_0} g(|\nabla_{\xi} V(\nabla u)|) \frac{f'(|\nabla u|)}{|\nabla u|} \partial_{\nu} u \, d\sigma$$
$$+ \int_{\Omega} [2f'(|\nabla u|)|\nabla u| - f(|\nabla u|)] \, dx \, .$$

From (2.47) and (2.35) we have

$$2\int_{\Omega} S_2(W)u\,dx \ge -g(f'(c))f'(c)|\Gamma_0| + \int_{\Omega} \left[2f'(|\nabla u|)|\nabla u| - f(|\nabla u|)\right]\,dx$$

and from (2.90) we obtain

$$2\int_{\Omega} S_{2}(W)u \, dx \ge - [cf'(c) - f(c)]f'(c)|\Gamma_{0}| + \int_{\Omega} [2f'(|\nabla u|)|\nabla u| - f(|\nabla u|)] \, dx \,.$$
(2.93)

From the Pohozaev identity (2.58) and (2.43) we get

$$(n+1)\int_{\Omega} u\,dx - n\int_{\Omega} f(|\nabla u|)\,dx = (f'(c)c - f(c))n|\Omega|;$$

which we use in (2.93) to obtain

$$2\int_{\Omega} S_2(W)u \, dx \ge -\frac{f'(c)|\Gamma_0|}{n|\Omega|} \int_{\Omega} \left[ (n+1)u - nf(|\nabla u|) \right] dx + \int_{\Omega} \left[ 2f'(|\nabla u|)|\nabla u| - f(|\nabla u|) \right] dx.$$

$$(2.94)$$

We notice that from (2.44) we have

$$|\Omega| = f'(c)|\Gamma_0|,$$

and from (2.94) we obtain

$$2\int_{\Omega} S_2(W)u\,dx \ge -\frac{n+1}{n}\int_{\Omega} u\,dx + 2\int_{\Omega} f'(|\nabla u|)|\nabla u|\,dx\,.$$

$$(2.95)$$

By using u as a test function in (2.40) we have that

$$\int_{\Omega} u \, dx = \int_{\Omega} f'(|\nabla u|) |\nabla u| \, dx$$

and from (2.95) we find

$$2\int_{\Omega} S_2(W)u\,dx \ge \frac{n-1}{n}\int_{\Omega} u\,dx\,.$$
(2.96)

From (2.56) and using the fact that  $\operatorname{Tr}(W) = L_f u = -1$ , we get that also the reverse inequality

$$\frac{n-1}{n} \int_{\Omega} u \, dx \ge \int_{\Omega} 2S_2(W) u \, dx \tag{2.97}$$

holds. From (2.96) and (2.97), we conclude that the equality sign must hold in (2.96) and (2.97). From Lemma 2.3 we have that

$$W = \frac{\mathrm{Tr}\left(W\right)}{n}\mathrm{Id}$$

a.e. in  $\Omega$ , and since Tr (W) = -1 we obtain (2.88). Moreover, Lemma 2.7 yields (2.89).

Step 2: u is a radial function. From (2.88) we have that

$$-\frac{1}{n}\delta_{ij} = \partial_j V_{\xi_i}(\nabla u(x))\,,$$

for every i, j = 1, ..., n, which implies that there exists  $x_0 \in \mathbb{R}^n$  such that

$$\nabla_{\xi} V(\nabla u(x)) = -\frac{1}{n}(x - x_0),$$

i.e. according to (2.47)

$$\frac{f'(|\nabla u(x)|)}{|\nabla u(x)|}\nabla u(x) = -\frac{1}{n}(x-x_0).$$

Hence

$$abla u(x) = -g'\left(\frac{|x-x_0|}{n}\right)\frac{x-x_0}{|x-x_0|}$$
 in  $\Omega$ .

Since u = 0 on  $\Gamma_0$ , we obtain (2.34) and in particular u is radial with respect to  $x_0$ . Moreover, from (2.89) we find that  $x_0$  must be the origin or, if  $\partial \Sigma$  contains flat regions, a point on  $\partial \Sigma$ .

#### 2.2.3 Cones in space forms: proof of Theorem 2.3

The goal of this section is to give an easily readable proof of Theorem 2.3. More precisely we assume more regularity on the solution than the one actually assumed in Theorem 2.3 in order to give a concise and clear idea of the proof in this setting, and we omit the technical details which are, in fact, needed. A rigorous treatment of the argument described below can be done by adapting the (technical) details in Section 2.2.2 and in [180].

Before starting the proof we declare some notations we use in the statement of Theorem 2.3 and we are going to adopt in the following. Given a *n*-dimensional Riemannian manifold (M, g), we denote by D the Levi-Civita connection of g. Moreover given a  $C^2$ -map  $u: M \to \mathbb{R}$ , we denote by  $\nabla u$  the gradient of u, i.e. the dual field of the differential of u with respect to g, and by  $\nabla^2 u = Ddu$  the Hessian of u. We denote by  $\Delta$  the Laplace-Betrami operator induced by g;  $\Delta u$  can be defined as the trace of  $\nabla^2 u$  with respect to g. Given a vector field X on an oriented Riemannian manifold (M, g), we denote by divXthe divergence of X with respect to g. If  $\{e_k\}$  is a local orthonormal frame on (M, g), then

$$\operatorname{div} X = \sum_{k=1}^{n} g(D_{e_k} X, e_k);$$

notice that, if u is a  $C^1$ -map and if X is a  $C^1$  vector field on M, we have the following *integration by parts* formula

$$\int_{\Omega} g(\nabla u, \nu) \, dx = -\int_{\Omega} u \mathrm{div} X \, dx + \int_{\partial \Omega} u g(X, \nu) \, d\sigma \,,$$

where  $\nu$  is the outward normal to  $\partial\Omega$  and  $\Omega$  is a bounded domain which is regular enough. Here and in the following, dx and  $d\sigma$  denote the volume form of g and the induced (n-1)-dimensional Hausdorff measure, respectively.

Proof of Theorem 2.3. We divide the proof in four steps.

Step 1: the P-function. Let u be the solution to problem (2.38) and, as in [68], we consider the P-function defined by

$$P = |\nabla u|^2 + \frac{2}{n}u + Ku^2.$$

Following [68, Lemma 2.1], P is a subharmonic function and, since u = 0 on  $\Gamma_0$  and from (2.43), we have that  $P = c^2$  on  $\Gamma_0$ . Moreover,

$$\nabla P = 2\nabla^2 u \nabla u + \frac{2}{n} \nabla u + 2Ku \nabla u \,. \tag{2.98}$$

From the convexity assumption of the cone  $\Sigma$ , we have that

$$g(\nabla^2 u \nabla u, \nu) \le 0.$$
(2.99)

Indeed, since  $\partial_{\nu} u = 0$  on  $\Gamma_1$  and by arguing as done for (2.68), we obtain that

$$0 = g(\nabla(\partial_{\nu} u), \nabla u) = g(\nabla^2 u \nabla u, \nu) + \operatorname{II}(\nabla u, \nabla u) \ge g(\nabla^2 u \nabla u, \nu) \quad \text{on} \quad \Gamma_1 \,,$$

which is (2.99). From (2.98) and (2.99) we obtain

$$\partial_{\nu}P = 2g(\nabla^2 u \nabla u, \nu) + \frac{2}{n} \partial_{\nu} u + 2K u \partial_{\nu} u \le 0 \quad \text{in } \Gamma_1 \setminus \{O\}.$$

Hence, the function P satisfies:

$$\begin{cases} \Delta P \geq 0 & \text{in } \Omega, \\ P = c^2 & \text{on } \Gamma_0 \\ \partial_{\nu} P \leq 0 & \text{on } \Gamma_1 \setminus \{O\} \end{cases}$$

Moreover, again from [68, Lemma 2.1], we have that

$$\Delta P = 0$$
 if and only if  $\nabla^2 u = \left(-\frac{1}{n} - Ku\right)g$ . (2.100)

Step 2: we have

$$P \le c^2 \quad in \ \Omega. \tag{2.101}$$

Indeed, we multiply  $\Delta P \geq 0$  by  $(P-c^2)^+$  and by integrating by parts we obtain

$$0 \ge \int_{\Omega \cap \{P > c^2\}} |\nabla P|^2 \, dx - \int_{\partial \Omega} (P - c^2)^+ \partial_\nu P \, d\sigma \, .$$

Since  $P = c^2$  on  $\Gamma_0$  and  $\partial_{\nu} P \leq 0$  on  $\Gamma_1$  we obtain that

$$0 \ge \int_{\Omega \cap \{P > c^2\}} |\nabla P|^2 \, dx \ge 0$$

and hence  $P \leq c^2$ .

Step 3:  $P = c^2$ . By contradiction, we assume that  $P < c^2$  in  $\Omega$ . Since  $\dot{h} > 0$ , we have

$$c^2 \int_{\Omega} \dot{h} \, dx > \int_{\Omega} \dot{h} |\nabla u|^2 \, dx + \frac{2}{n} \int_{\Omega} \dot{h} u \, dx + K \int_{\Omega} \dot{h} u^2 \, dx \, .$$

Since

$$\operatorname{div}(\dot{h}u\nabla u) = \dot{h}|\nabla u|^2 + \dot{h}u\Delta u + \ddot{h}u\partial_r u$$

 $\operatorname{and}$ 

$$\ddot{h} = -Kh\,,$$

and from u = 0 on  $\Gamma_0$  and  $\partial_{\nu} u = 0$  on  $\Gamma_1 \setminus \{O\}$ , we have that

$$c^{2} \int_{\Omega} \dot{h} \, dx > -\int_{\Omega} \dot{h} u \Delta u \, dx - \int_{\Omega} \ddot{h} u \partial_{r} u \, dx + \frac{2}{n} \int_{\Omega} \dot{h} u \, dx + K \int_{\Omega} \dot{h} u^{2} \, dx$$
$$= (n+1)K \int_{\Omega} \dot{h} u^{2} \, dx + \left(1 + \frac{2}{n}\right) \int_{\Omega} \dot{h} u \, dx + K \int_{\Omega} h u \partial_{r} u \, dx \, .$$

From  $\operatorname{div}(h\partial_r) = n\dot{h}$  we have

$$\operatorname{div}(u^2h\partial_r) = n\dot{h}u^2 + 2hu\partial_r u$$

and from u = 0 on  $\Gamma_0$  and  $\partial_{\nu} u = 0$  on  $\Gamma_1 \setminus \{O\}$  we obtain

$$c^{2} \int_{\Omega} \dot{h} \, dx > \left(1 + \frac{2}{n}\right) \left(\int_{\Omega} \dot{h} u \, dx - K \int_{\Omega} h u \partial_{r} u \, dx\right) \,. \tag{2.102}$$

Now we show that if u is a solution of (2.38) satisfying (2.39) then the equality sign holds in (2.102). Indeed, let  $X = h\partial_r$  be the radial vector field and, by integrating formula (2.8) in [66], we get

$$-\frac{c^2}{n}\int_{\partial\Omega}g(X,\nu)\,d\sigma + \frac{n+2}{n}\int_{\Omega}\dot{h}u\,dx - (n-2)K\int_{\Omega}\dot{h}u^2\,dx + \left(\frac{2}{n}-3\right)K\int_{\Omega}ug(X,\nabla u)\,dx = 0\,.$$
 (2.103)

Since  $\operatorname{div} X = n\dot{h}$  we obtain

$$c^{2} \int_{\Omega} \dot{h} dx = \frac{n+2}{n} \int_{\Omega} \dot{h} u dx - (n-2)K \int_{\Omega} \dot{h} u^{2} dx + \left(\frac{2}{n} - 3\right)K \int_{\Omega} ug(X, \nabla u) dx,$$

i.e.

$$c^{2} \int_{\Omega} \dot{h} \, dx = \left(1 + \frac{2}{n}\right) \left(\int_{\Omega} \dot{h} u \, dx - K \int_{\Omega} h u \partial_{r} u \, dx\right) \, dx$$

where we used that u = 0 on  $\Gamma_0$ ,  $\partial_{\nu} u = 0$  on  $\Gamma_1 \setminus \{O\}$  and  $g(X, \nu) = 0$  on  $\Gamma_1$ . From (2.102) we find a contradiction and hence  $P \equiv c^2$  in  $\Omega$ .

Step 4: u is radial. Since P is constant, then  $\Delta P = 0$  and from (2.100) we find that u satisfies the following Obata-type problem

$$\begin{cases} \nabla^2 u = \left(-\frac{1}{n} - Ku\right)g & \text{in }\Omega, \\ u = 0 & \text{on }\Gamma_0, \\ \partial_{\nu} u = 0 & \text{on }\Gamma_1 \setminus \{O\}. \end{cases}$$
(2.104)

We notice that the maximum and the minimum of u can not be both achieved on  $\Gamma_0$  since otherwise we would have that  $u \equiv 0$ . Hence, at least one between the maximum and the minimum of u is achieved at a point  $p \in \Omega \cup \Gamma_1$ . Let  $\gamma : I \to M$  be a unit speed maximal geodesic satisfying  $\gamma(0) = p$  and let  $f(s) = u(\gamma(s))$ . From the first equation of (2.104) it follows

$$f''(s) = -\frac{1}{n} - Kf(s).$$

Moreover, the definition of f and the fact that  $\nabla u(p) = 0$  yield

$$f'(0) = 0$$
 and  $f(0) = u(p)$ ,

and therefore

$$f(s) = \left(u(p) - \frac{1}{n}\right)H(s) - \frac{1}{n}.$$

This implies that u has the same expression along any geodesic strating from p, and hence u depends only on the distance from p. This means that  $\Omega = \Sigma \cap B_R$  where  $B_R$  is a geodesic ball and u depends only on the distance from the center of  $B_R$ .

## Part II

# Symmetry and validity results related to functional inequalities

### Introduction to Part II

Part II is about symmetry results for critical anisotropic *p*-Laplace equations in convex cones (in Chapter 3) and functional inequalities on a particular class of Riemannian manifolds (in Chapter 4). This part collects the results obtained in the following papers: [57, 174].

In this Introduction we present the results that we are going to prove in the Chapters 3 and 4.

**Chapter 3.** In the last years, Sobolev inequality (8) has been studied for more general norms as well as in convex cones of  $\mathbb{R}^n$  (see [15, 46, 97, 98, 159, 160]), where it takes the following form

$$\left(\int_{\Sigma} u^{p^*} dx\right)^{p/p^*} \le S_{\Sigma,H} \int_{\Sigma} H(\nabla u)^p dx, \qquad (\text{II.1})$$

where H is a norm (i.e.  $H : \mathbb{R}^n \to \mathbb{R}$  is convex, positively one-homogeneous and positive) and  $\Sigma$  as in (I.26) is an open convex cone in  $\mathbb{R}^n$ , i.e.  $\Sigma = \mathbb{R}^k \times \mathcal{C}$  where  $k \in \{0, \ldots, n\}$  and  $\mathcal{C}$  is a convex cone in  $\mathbb{R}^{n-k}$  with only one vertex  $\{\mathcal{O}\}$ .

As for the usual Sobolev inequality in  $\mathbb{R}^n$ , the natural functional associated to (II.1) is the following:

$$\mathcal{J}^{H}(u) = \frac{\int_{\Sigma} H(\nabla u)^{p} dx}{\left(\int_{\Sigma} |u|^{p^{*}} dx\right)^{\frac{p}{p^{*}}}}.$$
 (II.2)

We observe that (II.2) is the natural generalization of (11).

We are interested in critical points of the functional  $\mathcal{J}$ . Let u be a (positive) critical point of (II.2) and we compute the first variation of the functional. Let  $\varepsilon > 0$  and  $\varphi \in C_c^{\infty}(\mathbb{R}^n)$  be a test function, then

$$\frac{d}{d\varepsilon}\Big|_{\varepsilon=0}\mathcal{J}^H(u+\varepsilon\varphi)=0\,.$$

This condition leads to the following boundary value problem:

$$\begin{cases} \operatorname{div} \left( a(\nabla u) \right) + u^{p^* - 1} = 0 & \text{in } \Sigma \\ u > 0 & \text{in } \Sigma \\ a(\nabla u) \cdot \nu = 0 & \text{on } \partial \Sigma \,, \end{cases}$$
(II.3)

where  $\nu$  is the outward normal to  $\partial \Sigma$ ,

$$a(\xi) = H^{p-1}(\xi)\nabla H(\xi) \qquad \forall \xi \in \mathbb{R}^n \,. \tag{II.4}$$

We will sometimes write

$$\Delta_n^H u = \operatorname{div}\left(a(\nabla u)\right),$$

where  $\Delta_p^H$  is called the *Finsler p-Laplace* (or *anisotropic p-Laplace*) operator. It is clear that when we consider the case  $\Sigma = \mathbb{R}^n$  no boundary conditions are given. In [57] we study problem (II.3) which is called *critical anisotropic p-Laplace equations in convex cones.* In [57] we prove the sharp version of (II.1) by suitably adapting the optimal transportation proof of the Sobolev inequality [72] to the case of cones (see Appendix B). It is interesting to observe that our proof applies also to the case of weighted Sobolev inequalities for the class of weights considered in [46], thus generalizing [46, Theorem 1.3] to the full range of exponents  $p \in (1, n)$ .

Hence, as shown in Appendix B, the extremals of (II.1) are of the form

$$\mathcal{U}_{\lambda,x_0}^{H}(x) := \left(\frac{\lambda^{\frac{1}{p-1}} n^{\frac{1}{p}} \left(\frac{n-p}{p-1}\right)^{\frac{p-1}{p}}}{\lambda^{\frac{p}{p-1}} + \tilde{H}_0(x-x_0)^{\frac{p}{p-1}}}\right)^{\frac{n-p}{p}}$$
(II.5)

for some  $\lambda > 0$  (see also [12, 72, 160, 217] and the references therein), where

$$\dot{H}_0(\zeta) := H_0(-\zeta) \qquad \forall \, \zeta \in \mathbb{R}^n \,, \tag{II.6}$$

and  $H_0$  denotes the dual norm associated to H, namely

$$H_0(\zeta) := \sup_{H(\xi)=1} \zeta \cdot \xi \qquad \forall \zeta \in \mathbb{R}^n.$$

Moreover, if  $\Sigma = \mathbb{R}^n$  then  $x_0$  may be any point of  $\mathbb{R}^n$ ; if  $\Sigma = \mathbb{R}^k \times \mathcal{C}$  with  $k \in \{1, \ldots, n-1\}$  and  $\mathcal{C}$  does not contain a line, then  $x_0 \in \mathbb{R}^k \times \{\mathcal{O}\}$ ; otherwise,  $x_0 = \mathcal{O}$ .

The main result of Chapter 3 is to provide a complete classification result for critical anisotropic *p*-Laplace equations in convex cones with the hypothesis that the solution belongs to the space  $\mathcal{D}^{1,p}(\Sigma)$  (as in Theorem C in the Introduction). More precisely, we consider the problem

$$\begin{cases} \operatorname{div} \left( a(\nabla u) \right) + u^{p^* - 1} = 0 & \operatorname{in} \Sigma \\ u > 0 & \operatorname{in} \Sigma \\ a(\nabla u) \cdot \nu = 0 & \operatorname{on} \partial \Sigma \\ u \in \mathcal{D}^{1, p}(\Sigma) \,, \end{cases}$$
(II.7)

the space  $\mathcal{D}^{1,p}(\Sigma)$  is defined as in (9) (with  $\mathbb{R}^n$  replaced by  $\Sigma$ ). The main goal of this Chapter 3 is to classify the critical points for (II.2), i.e. the classification of the solutions to (II.3). In [57] we prove the following result (which is the main result of Chapter 3)

**Theorem II.A:** let  $n \geq 2$ ,  $1 , and let <math>\Sigma = \mathbb{R}^k \times C$  be a convex cone, where C does not contain a line. Let H be a norm of  $\mathbb{R}^n$  such that  $H^2$  is of class  $C^2(\mathbb{R}^n \setminus \{\mathcal{O}\})$  and it is uniformly convex and  $C^{1,1}$  in  $\mathbb{R}^n$ , namely there exist constants  $0 < \lambda \leq \Lambda$  such that

$$\lambda \mathrm{Id} \le H(\xi) \, D^2 H(\xi) + \nabla H(\xi) \otimes \nabla H(\xi) \le \Lambda \, \mathrm{Id} \,, \qquad \forall \, \xi \in \mathbb{R}^n \setminus \{\mathcal{O}\} \,. \quad (\mathrm{II.8})$$

Let u be a solution to (II.3). Then  $u(x) = \mathcal{U}_{\lambda,x_0}^H(x)$  for some  $\lambda > 0$  and  $x_0 \in \overline{\Sigma}$ , where  $\mathcal{U}_{\lambda,x_0}^H$  is given by (II.5). Moreover,
- (i) if k = n then  $\Sigma = \mathbb{R}^n$  and  $x_0$  may be a generic point in  $\mathbb{R}^n$ ;
- (ii) if  $k \in \{1, \ldots, n-1\}$  then  $x_0 \in \mathbb{R}^k \times \{\mathcal{O}\}$ ;
- (iii) if k = 0 then  $x_0 = \mathcal{O}$ .

As already mentioned, case (i) in the previous Theorem has been already proved in [48, 76, 203, 224] when  $\Sigma = \mathbb{R}^n$  and H is the Euclidean norm. In that case, thanks to the symmetry of the problem, the authors can apply the method of moving planes. In the Euclidean case and for p = 2, the classification of solutions in convex cones was proved in [160, Theorem 2.4] by using the Kelvin transform and inspired by [108] and [178]. Unfortunately, the Kelvin transform and the method of moving planes are not helpful neither for anisotropic problems nor inside cones for a general  $p \in (1, n)$ . In this case non-existence results generalizing the one in [108] to  $p \in (1, n)$  are given in [206].

In Chapter 3 we provide a new approach to the characterization of solutions to critical p-Laplacian equations, which is based on integral identities rather than moving planes. This approach takes inspiration from [206] and also from [29, 30, 40, 215] where classical overdetermined problems for PDEs are considered (see also [63, 180] for analogous problems in convex cones).

The strategy of the proof can be explained as follows. First, using that  $u \in \mathcal{D}^{1,p}(\Sigma)$  we show that u is bounded and satisfies certain decay estimates at infinity (in particular it behaves as the fundamental solution both from above and below), so that one has optimal upper bounds on  $H(\nabla u)$  in terms of the fundamental solution. We notice that, differently from [203], we do not need asymptotic lower bounds on  $\nabla u$ ; instead, we use a Caccioppoli-type inequality to prove some asymptotic estimates on certain integrals involving higher order derivatives. Then we consider the auxiliary function  $v = u^{-\frac{p}{n-p}}$ . We find the elliptic equation satisfied by v and then, thanks to the asymptotic estimates on u, we show that v and  $\nabla v$  satisfy explicit growth conditions at infinity. By using integral identities, the convexity of  $\Sigma$ , and some suitable inequalities, we are able to prove that  $\nabla(a(\nabla v))$  is a multiple of the identity matrix, from which the symmetry result follows.

We mention that also in this case the hypothesis that u > 0 is fundamental, indeed in [71] the authors construct a nonradial sign-changing solution when p = 2 and H is the Euclidean norm.

**Chapter 4.** In Chapter 4 we focus on a particular class of Riemannian manifolds: the so-called Cartan-Hadamard manifolds M of dimension  $n \geq 3$ , namely complete, noncompact, simply connected Riemannian manifold with everywhere nonpositive sectional curvatures. On the one hand, it is well known that the Sobolev inequality (8) with p = 2 holds also on any Cartan-Hadamard manifold M; explicitly the following Sobolev inequality holds

$$||u||_{L^{2^*}(M)} \le C||\nabla u||_{L^2(M)} \tag{II.9}$$

for every function  $u \in C_c^1(M)$ . On the other hand, if the sectional curvatures are everywhere bounded from above by a negative constant k, then in addition to (II.9) also the following Poincarè inequality

$$||u||_{L^{2}(M)} \leq C||\nabla u||_{L^{2}(M)}$$
(II.10)

holds for every function  $u \in C_c^1(M)$ . We mention that (II.10) is equivalent to the fact that the infimum of the spectrum of (minus) the Laplace-Beltrami operator on M is bounded from below by  $k(N-1)^2/4$ , i.e.  $\Delta$  has a spectral gap (see the celebrated paper by McKeans [167]). For the sake of completness we mention that, in contrast with McKean's Theorem, in [162] (see also references therein) it was shown that, on any complete non-compact Riemannian manifold, the essential spectrum of the Laplace-Beltrami operator starts from zero as soon as the Ricci curvature vanishes at infinity. This generalizes [214], where the same thesis was established upon assuming an at-least-quadratic decrease of the negative curvatures.

So, the situation is the following: if the sectional curvatures are bounded from above by  $k \equiv 0$  then M supports (II.9), and as soon as such bound becomes strictly negative M also supports (II.10). In other words, there is a "jump" of the  $L^p$  exponent in the left-hand side of the inequality  $||f||_{L^p(M)} \leq C||\nabla u||_{L^2(M)}$ which depends on the curvatures. So the natural question is the following: what happens in between? That is, suppose that the sectional curvatures of M satisfy the following decay estimate

$$\operatorname{Sect}(x) \leq -\frac{K}{r(x)^{\beta}} \quad \text{for all } x \in M \setminus B_{R_0},$$
 (II.11)

for some  $\beta \in (0, 2]$  and  $K, R_0 > 0$  where r(x) denotes the geodesic distance from x to a fixed point  $o \in M$  (the pole of the manifold). Then, what kind of inequalities does a Cartan-Hadamard manifold M with sectional curvature satisfying (II.11) support? In [174] we find nontrivial answers and we will present them in Chapter 4. First of all one has to distinguish between two cases:  $\beta \in$ (0, 2) and  $\beta = 2$ , the first case is called the sub-hyperbolic range and the second is called the quasi-Euclidean range (this terminology was introduced in [120]). Another distinction is between radial and nonradial functions. In [174] we focus on the following Sobolev-type inequalities:

$$||u||_{L^{p}(M)} \leq C_{p}||\nabla u||_{L^{2}(M)}$$
 for some  $p \in (2, 2^{*}]$ . (II.12)

In the case  $\beta \in (0,2)$  we show that (II.12) holds in the radial setting for all  $p \in (2, 2^*]$ , for a positive constant  $C_p = C(n, p, K, R_0, \beta)$ ; moreover the result is optimal with respect to the dependence on p. In case  $\beta = 2$  we show that (II.12) starts to holds in the radial setting from a certain exponent  $\tilde{2} \in (2, 2^*)$ that depends on n and K, which tends to  $2^*$  as  $K \to 0$  and to 2 as  $K \to 1$ ; also in this case the result is optimal with respect to p. Finally, we prove that out of the radial setting all of the above results fail: it is enough to assume that (II.11) is satisfied with the reverse inequality to be able to construct a sequence of nonradial functions that make the constant  $C_p$  in (II.12) blow up for every  $p < 2^*$ . We mention that the case  $\beta > 2$  is not interesting because it is essentially Euclidean, i.e. the sole inequality of the type of (II.12) that holds, even if restricted to radial functions, is the standard Sobolev one: this is an immediate consequence of our results. The techniques of proof that we exploit take advantage of two main ingredients: one-dimensional weighted functional inequalities and Laplacian-comparison theorems; the idea is to first study the radial inequalities on the already mentioned model manifolds.

The main motivation to study inequalities like (II.12) in Cartan-Hadamard manifolds is related to the asymptotic behavior of nonnegative solutions of diffusion equation. The link between the Sobolev inequality (or Gagliardo-Nirenberg and Nash inequalities in low dimensions) and a sharp decay estimate for the heat kernel is a well-studied topic, which goes back to some pioneering works between the 50s and the 80s: see the monograph [77]. In this regard, let us also mention the recent paper [116], where the behaviour of Faber-Krahn inequalities (which are strictly related to the Sobolev ones) on Riemannian manifolds under removal of compact subsets or gluing of noncompact manifolds is investigated, with applications to heat-kernel bounds.

In the last decades, several results that connect the validity of functional inequalities of Sobolev, log-Sobolev or Poincaré type with smoothing effects for *nonlinear* diffusion equations (mostly modeled on the porous medium equation) have been established: see [119] for weighted porous medium equations in the presence of weights and Poincaré inequalities, [101, 117] for optimal short and long-time smoothing estimates for porous medium equations (or the more general *filtration equation*) on Euclidean domains in the case of homogeneous Neumann problems, and the above-mentioned paper [118] for similar analyses focused on the consequences of the validity of families of Sobolev-type inequalities of the type of (II.12). Previous results in this direction, having already in mind the manifold case, can be found in [34]. As a general reference on smoothing effects, we also quote [222]. It is worth pointing out that in order to prove some of the main theorems of these papers, the authors often exploit a very powerful equivalence tool between *families* of Gagliardo-Nirenberg-Sobolev inequalities and one *single* inequality, which is due to [16].

Among recent works that take advantage of connections between functional inequalities and *fast diffusion* flows we refer to [86, 87], where the latter have been thoroughly exploited in order to prove or disprove the achievement of optimality by radial functions in Caffarelli-Kohn-Nirenberg inequalities. For a functional-analytic investigation of fast diffusions on Cartan-Hadamard manifolds, see also [35]. Finally, the monograph [17] is devoted to a wide-scope discussion on the interplay between analytic, geometric and probabilistic features of Markov diffusion semigroups, which involves functional inequalities related to those treated here and curvature-dimension conditions in more general metric frameworks. More explicitly, it is well-known that in  $\mathbb{R}^n$  there is a close connection between the Sobolev inequality (II.9) and the heat equation:

$$\begin{cases} \partial_t u = \Delta u & \text{in } \mathbb{R}^n \times \mathbb{R}_+ \\ u = u_0 & \text{in } \mathbb{R}^n \times \{0\}. \end{cases}$$
(II.13)

Indeed, by classical results (mainly due to Nash, Varopoulos, Fabes and Stroock between 1958 and 1986), the Sobolev inequality is equivalent to the following smoothing estimate for solutions of (II.13):

$$||u(t)||_{\infty} \le Ct^{-\frac{n}{2}} ||u_0||_{L^1(\mathbb{R}^n)}.$$
 (II.14)

Similar results have recently been extended to suitable nonlinear diffusions like the porous medium equation:

$$\begin{cases} \partial_t u = \Delta(u^m) & \text{in } M \times \mathbb{R}_+ \\ u = u_0 & \text{in } M \times \{0\}. \end{cases}$$
(II.15)

for m > 1 and  $u_0 \in L^1(\mathbb{R}^n)$ . That is, if  $M = \mathbb{R}^n$  then the Sobolev inequality is equivalent to the smoothing estimate (see Bonforte, Grillo, Muratori between

2006 and 2013)

$$||u(t)||_{\infty} \le Ct^{-\frac{n}{2+n(m-1)}} ||u_0||_{L^1(\mathbb{R}^n)}^{\frac{2+n(m-1)}{2}}.$$
 (II.16)

When  $M = \mathbb{H}^n$ , the porous medium equation (II.15) had been also studied (see [223]). In particular in [118], as a sole consequence of the Sobolev inequality (II.9) and Poincaré inequality (II.10) which hold in  $\mathbb{H}^n$ , the following smoothing estimate is proved

$$||u(t)||_{\infty} \le C \left( \frac{\log(2+t)|u_0||_{L^1(\mathbb{R}^n)}^{m-1}}{t} \right)^{\frac{1}{m-1}}.$$
 (II.17)

Moreover, the porous medium equation (II.15) had been studied also when M is a Cartan-Hadamard manifold. In particular, since the Sobolev inequality (II.9) holds on any Cartan-Hadamrd manifold then one can prove an estimate like (II.16). In addition, if the sectional curvatures are negatively bounded away from zero, then (II.10) holds true and (II.16) can be improved for large times by an estimate like (II.17). A natural question is: what happens in between, i.e. for vanishing curvatures? In [118] they prove that if the sectional curvature of M satisfies (II.11) then, an estimate (II.17) does hold provided M supports (II.12), i.e. a family of Sobolev-type inequalities. That's why we study these type of inequality on Cartan-Hadamard manifolds and in [174] we have a partial result.

# Chapter 3

# Symmetry results for critical anisotropic *p*-Laplacian equations in convex cones

As mentioned in the introduction to Part II, in this chapter we consider the following problem

$$\begin{cases} \operatorname{div} \left( a(\nabla u) \right) + u^{p^* - 1} = 0 & \operatorname{in} \Sigma \\ u > 0 & \operatorname{in} \Sigma \\ a(\nabla u) \cdot \nu = 0 & \operatorname{on} \partial \Sigma \\ u \in \mathcal{D}^{1, p}(\Sigma) \,, \end{cases}$$
(3.1)

where: H is a norm<sup>1</sup>,  $\Sigma$  is a convex open cone in  $\mathbb{R}^n$  given by

$$\Sigma = \left\{ tx : x \in \omega, \, t \in (0, +\infty) \right\},\tag{3.2}$$

for some open domain  $\omega \subseteq \mathbb{S}^{n-1}$ ,  $\nu$  is the outward normal to  $\partial \Sigma$ ,

$$a(\xi) := H^{p-1}(\xi) \nabla H(\xi) \qquad \forall \xi \in \mathbb{R}^n,$$
(3.3)

and the space  $\mathcal{D}^{1,p}(\Sigma)$  is defined as follows

$$\mathcal{D}^{1,p}(\Sigma) := \left\{ u \in L^{p^*}(\Sigma) : \int_{\Sigma} |\nabla u|^p \, dx < \infty \right\}, \tag{3.4}$$

where  $p^*$  is the usual critical Sobolev exponent, i.e.

$$p^* = \frac{np}{n-p} \,. \tag{3.5}$$

Since it will be useful, we mention that, in general (up to a change of coordinates) every convex cone  $\Sigma$  of  $\mathbb{R}^n$  is of the form

$$\Sigma = \mathbb{R}^k \times \mathcal{C}$$

<sup>&</sup>lt;sup>1</sup>By abuse of notation, we say that  $H : \mathbb{R}^n \to \mathbb{R}$  is a norm if H is convex, positively one-homogeneous (namely,  $H(\ell\xi) = \ell H(\xi)$  for all  $\ell > 0$ ), and  $H(\xi) > 0$  for all  $\xi \in \mathbb{S}^{n-1}$ . Note that we do not require H to be symmetric, so it may happen that  $H(\xi) \neq H(-\xi)$ .

where  $k \in \{0, ..., n\}$  and  $C \subset \mathbb{R}^{n-k}$  is a convex cone containing no lines. Moreover we will sometimes write

$$\Delta_p^H u = \operatorname{div}\left(a(\nabla u)\right),\,$$

where  $\Delta_p^H$  is called the *Finsler p-Laplacian* (or *anisotropic p-Laplacian*) operator.

As already observed if  $u \in \mathcal{D}^{1,p}(\Sigma)$  is a positive critical point for the Sobolev functional

$$\mathcal{J}^{H}(u) = \frac{\int_{\Sigma} H(\nabla u)^{p} dx}{\left(\int_{\Sigma} |u|^{p^{*}} dx\right)^{\frac{p}{p^{*}}}},$$
(3.6)

then u satisfies (3.1). The main goal of this chapter is to classify the critical points for (3.6), i.e. the classification of the solutions to (3.1). The functional (3.6) is related to the sharp anisotropic Sobolev inequality that we prove in Appendix B where we also show that the analogue of the Aubin-Talenti bubbles are the following functions:

$$\mathcal{U}_{\lambda,x_{0}}^{H}(x) := \left(\frac{\lambda^{\frac{1}{p-1}}\left(n^{\frac{1}{p}}\left(\frac{n-p}{p-1}\right)^{\frac{p-1}{p}}\right)}{\lambda^{\frac{p}{p-1}} + \tilde{H}_{0}(x_{0}-x)^{\frac{p}{p-1}}}\right)^{\frac{n-p}{p}}$$
(3.7)

for some  $\lambda > 0$ , where

$$\tilde{H}_0(\zeta) := H_0(-\zeta) \qquad \forall \, \zeta \in \mathbb{R}^n \,, \tag{3.8}$$

and  $H_0$  denotes the dual norm associated to H, namely

$$H_0(\zeta) := \sup_{H(\xi)=1} \zeta \cdot \xi \qquad \forall \zeta \in \mathbb{R}^n$$

Moreover, if  $\Sigma = \mathbb{R}^n$  then  $x_0$  may be any point of  $\mathbb{R}^n$ ; if  $\Sigma = \mathbb{R}^k \times \mathcal{C}$  with  $k \in \{1, \ldots, n-1\}$  and  $\mathcal{C}$  does not contain a line, then  $x_0 \in \mathbb{R}^k \times \{\mathcal{O}\}$ ; otherwise,  $x_0 = \mathcal{O}$  (from now on,  $\mathcal{O}$  denotes the origin).

The main result that we prove in this chapter is the following Liouville-type Theorem (Liouville-type Theorem II.A in the introduction to part II)

**Theorem 3.1.** Let  $n \geq 2$ ,  $1 , and let <math>\Sigma = \mathbb{R}^k \times C$  be a convex cone, where C does not contain a line. Let H be a norm of  $\mathbb{R}^n$  such that  $H^2$  is of class  $C^2(\mathbb{R}^n \setminus \{\mathcal{O}\})$  and it is uniformly convex and  $C^{1,1}$  in  $\mathbb{R}^n$ , namely there exist constants  $0 < \lambda \leq \Lambda$  such that

$$\lambda \mathrm{Id} \le H(\xi) \, D^2 H(\xi) + \nabla H(\xi) \otimes \nabla H(\xi) \le \Lambda \, \mathrm{Id} \qquad \forall \, \xi \in \mathbb{R}^n \setminus \{\mathcal{O}\} \tag{3.9}$$

(note that  $D^2(H^2) = 2H D^2 H + 2\nabla H \otimes \nabla H$ ).

Let u be a solution to (3.1). Then  $u(x) = \mathcal{U}_{\lambda,x_0}^H(x)$  for some  $\lambda > 0$  and  $x_0 \in \overline{\Sigma}$ , where  $\mathcal{U}_{\lambda,x_0}^H$  is given by (3.7). Moreover,

- (i) if k = n then  $\Sigma = \mathbb{R}^n$  and  $x_0$  may be a generic point in  $\mathbb{R}^n$ ;
- (ii) if  $k \in \{1, \ldots, n-1\}$  then  $x_0 \in \mathbb{R}^k \times \{\mathcal{O}\}$ ;

(*iii*) if k = 0 then  $x_0 = \mathcal{O}$ .

The strategy of the proof can be explained as follows. First, using that  $u \in \mathcal{D}^{1,p}(\Sigma)$  we show that u is bounded (see Subsection 3.1.1). Then, in Subsection 3.1.2 we prove that u satisfies certain decay estimates at infinity (in particular it behaves as the fundamental solution both from above and below), so that one has optimal upper bounds on  $H(\nabla u)$  in terms of the fundamental solution. We notice that, differently from [203], we do not need asymptotic lower bounds on  $\nabla u$ ; instead, we use a Caccioppoli-type inequality to prove some asymptotic estimates on certain integrals involving higher order derivatives (see Subsection 3.1.3).

Then, in Section 3.2 we consider the auxiliary function  $v = u^{-\frac{p}{n-p}}$ . We find the elliptic equation satisfied by v and then, thanks to the asymptotic estimates on u, we show that v and  $\nabla v$  satisfy explicit growth conditions at infinity. By using integral identities, the convexity of  $\Sigma$ , and some suitable inequalities, we are able to prove that  $\nabla a(\nabla v)$ , recall that a is given by (3.3), is a multiple of the identity matrix; from this fact the symmetry result follows.

In Appendix B we prove, via the optimal transport approach introduced by [72], the sharp version of the following anisotropic Sobolev inequality in convex cones

$$\left(\int_{\Sigma} u^{p^*} dx\right)^{p/p^*} \le S_{\Sigma,H} \int_{\Sigma} H(\nabla u)^p dx, \qquad (3.10)$$

for general norms and cones, and even in a weighted setting (for the general inequality, not the sharp one we refer to [46]).

Most of the chapter will focus on the case in which  $\Sigma$  is a convex cone with nonempty boundary. Indeed our approach perfectly works also when  $\Sigma = \mathbb{R}^n$ . However, since the whole space case is simpler to be proven, we prefer to focus the exposition to the case when  $\Sigma$  has boundary (the case when  $\Sigma = \mathbb{R}^n$  can be found in Appendix C).

## 3.1 Preliminary results

In this section we collect some results that are well established when  $\Sigma = \mathbb{R}^n$ and H is the Euclidean norm. Since we are dealing with problem (3.1) and some modifications are needed, we report here their counterpart when  $\Sigma$  is a convex cone and H a general norm, and provide a sketch of the proofs emphasizing the main differences.

In the whole chapter we denote by  $B_r(x)$  the usual Euclidean ball, and by  $B_r$  the ball  $B_r(\mathcal{O})$  centered at the origin.

### **3.1.1** Boundeness of solutions

In the following lemma we prove that solutions to (3.1) are bounded. The result holds for more general Neumann problems, in particular for problems with a differential operator modelled on the *p*-Laplace operator. **Lemma 3.1.** Let  $\Sigma \subseteq \mathbb{R}^n$  be a convex cone as in (3.2) and let  $u \in \mathcal{D}^{1,p}(\Sigma)$  be a solution to

$$\begin{cases} \operatorname{div} \left( a(\nabla u) \right) + u^{p^* - 1} = 0 & \text{ in } \Sigma \\ u > 0 & \text{ in } \Sigma \\ a(\nabla u) \cdot \nu = 0 & \text{ on } \partial \Sigma \,, \end{cases}$$
(3.11)

where  $a : \mathbb{R}^n \to \mathbb{R}^n$  is a continuous vector field such that the following holds: there exist  $\alpha > 0$  and  $0 \le s \le 1/2$  such that

$$|a(\xi)| \le \alpha (|\xi|^2 + s^2)^{\frac{p-1}{2}} \quad and \quad \xi \cdot a(\xi) \ge \frac{1}{\alpha} \int_0^1 \left( t^2 |\xi|^2 + s^2 \right)^{\frac{p-2}{2}} |\xi|^2 \, dt \,, \quad (3.12)$$

for every  $\xi \in \mathbb{R}^n$ . Then there exists  $\delta > 0$  with the following property: let  $\rho > 0$  be such that

$$||u||_{L^{p^*}(B_{\rho}(x_0))} \le \delta \qquad \forall x_0 \in \mathbb{R}^n.$$

Then

$$||u||_{L^{\infty}(\Sigma \cap B_{R/2}(x_0))} \le CR^{-\frac{n}{p}}||u||_{L^p(\Sigma \cap B_R(x_0))} \qquad \forall R \le \rho,$$

where C depends only on n,  $\alpha$ , p and the Sobolev constant of  $\Sigma$ .

*Proof.* We closely follow [183, Theorem E.0.20] and [205, Theorem 1] and we only give a sketch of the proof. We first prove that  $u \in L^{qp^*}_{\text{loc}}(\overline{\Sigma})$  for any  $q < p^*/p$ . Given l > 0 and  $1 < q < \frac{p^*}{p}$ , we define

$$F(u) = \begin{cases} u^{q} & \text{if } u \leq l \\ ql^{q-1}(u-l) + l^{q} & \text{if } u > l , \end{cases}$$
(3.13)

and

$$G(u) = \begin{cases} u^{(q-1)p+1} & \text{if } u \le l \\ ((q-1)p+1)l^{(q-1)p}(u-l) + l^{(q-1)p+1} & \text{if } u > l \,. \end{cases}$$

Let  $\eta\in C_0^\infty(\mathbb{R}^n)$  and use

$$\xi = \eta^p G(u)$$

as a test-function in (3.11); then an integration by parts gives

$$\int_{\Sigma} a(\nabla u) \cdot \nabla(\eta^p G(u)) \, dx = \int_{\Sigma} u^{p^* - 1} \eta^p G(u) \, dx \,. \tag{3.14}$$

We aim at proving that

$$c \int_{\Sigma} \eta^{p} G'(u) |\nabla u|^{p} dx \leq \int_{\Sigma} \eta^{p-1} G(u) |a(\nabla u) \cdot \nabla \eta| dx$$
  
+ 
$$\int_{\Sigma} u^{p^{*}-1} \eta^{p} G(u) dx$$
  
+ 
$$s^{p} \int_{\Sigma} \eta^{p} G'(u) dx$$
 (3.15)

holds for  $0 \le s \le 1/2$ . We distinguish between the cases  $1 and <math>2 \le p < n$ .

If  $p \ge 2$ , then (3.12) implies

$$\xi \cdot a(\xi) \ge \frac{1}{\alpha} |\xi|^p \,,$$

and from (3.14) we get

$$\frac{1}{\alpha} \int_{\Sigma} \eta^p G'(u) |\nabla u|^p \, dx \le p \int_{\Sigma} \eta^{p-1} G(u) |a(\nabla u) \cdot \nabla \eta| \, dx + \int_{\Sigma} u^{p^*-1} \eta^p G(u) \, dx \,,$$

which implies (3.15).

If 1 then (3.15) is obtained by using a more careful argument. We claim that

$$\int_0^1 \left( t^2 |\xi|^2 + s^2 \right)^{\frac{p-2}{2}} |\xi|^2 \, dt \ge \frac{1}{2} \left( |\xi|^p - s^p \right). \tag{3.16}$$

To prove this we consider two cases. If  $s > |\xi|$  then the left-hand side of (3.16) is negative, and so the result is clearly true. Otherwise, if  $s \le |\xi|$  then

$$t^2 |\xi|^2 + s^2 \le 2|\xi|^2 \qquad \text{for } t \in [0,1],$$

and therefore

$$\int_0^1 \left( t^2 |\xi|^2 + s^2 \right)^{\frac{p-2}{2}} |\xi|^2 \, dt \ge \int_0^1 \left( 2|\xi|^2 \right)^{\frac{p-2}{2}} |\xi|^2 \, dt = 2^{\frac{p-2}{2}} |\xi|^p \ge \frac{1}{2} |\xi|^p,$$

that again implies (3.16).

Thanks to (3.14), (3.12), and (3.16), we obtain

$$\begin{split} \frac{1}{2\alpha} \int_{\Sigma} \eta^{p} G'(u) |\nabla u|^{p} \, dx &\leq p \int_{\Sigma} \eta^{p-1} G(u) |a(\nabla u) \cdot \nabla \eta| \, dx \\ &+ \int_{\Sigma} u^{p^{*}-1} \eta^{p} G(u) \, dx + \frac{s^{p}}{2} \int_{\Sigma} \eta^{p} G'(u) \, dx \,, \end{split}$$

and the proof of (3.15) is complete.

Note now that, by Young's inequality and (3.12), for any  $\epsilon \in (0,1)$  we have

$$\begin{split} \eta^{p-1} |a(\nabla u) \cdot \nabla \eta| &\leq \epsilon^{\frac{p}{p-1}} u^{-1} |a(\nabla u)|^{\frac{p}{p-1}} \eta^p + \epsilon^{-p} u^{p-1} |\nabla \eta|^p \\ &\leq C_0 \epsilon^{\frac{p}{p-1}} u^{-1} (|\nabla u|^p + s^p) \eta^p + \epsilon^{-p} u^{p-1} |\nabla \eta|^p, \end{split}$$

where  $C_0$  depends only on  $\alpha$  and p. Thanks to this inequality and recalling (3.15), since  $G(u) \leq uG'(u)$  (note that G is convex and G(0) = 0), for any  $\epsilon \in (0, 1)$  we obtain

$$\begin{split} c\int_{\Sigma}\eta^{p}G'(u)|\nabla u|^{p}\,dx \leq & C_{0}\epsilon^{\frac{p}{p-1}}\int_{\Sigma}\eta^{p}G'(u)|\nabla u|^{p}\,dx + (C_{0}+1)s^{p}\int_{\Sigma}\eta^{p}G'(u)\,dx \\ & +\epsilon^{-p}\int_{\Sigma}G(u)u^{p-1}|\nabla\eta|^{p}\,dx + \int_{\Sigma}u^{p^{*}-1}\eta^{p}G(u)\,dx\,. \end{split}$$

Hence, choosing  $\epsilon$  small enough so that  $C_0 \epsilon^{\frac{p}{p-1}} = c/2$ , we deduce that

$$c' \int_{\Sigma} \eta^{p} G'(u) |\nabla u|^{p} dx \leq s^{p} \int_{\Sigma} \eta^{p} G'(u) dx + \int_{\Sigma} G(u) u^{p-1} |\nabla \eta|^{p} dx + \int_{\Sigma} u^{p^{*}-1} \eta^{p} G(u) dx, \quad (3.17)$$

where c' > 0 depends only on  $n, \alpha$ , and p. Using now that  $G'(u) \ge c[F']^p$  and that  $u^{p-1}G(u) < C[F(u)]^p$ , we obtain

$$\begin{split} \hat{c} \int_{\Sigma} |\nabla(\eta F(u))|^p \, dx &\leq s^p \int_{\Sigma} \eta^p G'(u) \, dx + \int_{\Sigma} |\nabla \eta|^p F^p(u) \, dx \\ &+ \int_{\Sigma} \eta^p u^{p^* - p} F^p(u) \, dx \, . \end{split}$$

Hence, thanks to the Sobolev inequality (3.10) we get

$$\bar{c}\left(\int_{\Sigma}F^{p^*}(u)\eta^{p^*}\,dx\right)^{\frac{p}{p^*}} \leq s^p \int_{\Sigma}\eta^p G'(u)\,dx + \int_{\Sigma}|\nabla\eta|^p F^p(u)\,dx + \int_{\Sigma}\eta^p u^{p^*-p}F^p(u)\,dx\,,\quad(3.18)$$

where  $\bar{c} > 0$  depends only on  $n, \alpha, p$  and the Sobolev constant for  $\Sigma$ . Now, choose  $\delta = (\bar{c}/2)^{1/(p^*-p)}$ , so that for any  $R \leq \rho$  it holds

$$||u||_{L^{p^*}(B_R(x_0))}^{p^*-p} \le \frac{c}{2} \qquad \forall x_0 \in \mathbb{R}^n.$$

Then, if we choose  $\eta$  such that  $\operatorname{supp}(\eta) \subset B_R(x_0)$ , it follows from Holder's inequality that we can reabsorb the last term in (3.18), and we get

$$\frac{\overline{c}}{2} \left( \int_{\Sigma} F^{p^*}(u) \eta^{p^*} dx \right)^{\frac{p}{p^*}} \leq s^p \int_{\Sigma \cap B_R(x_0)} \eta^p G'(u) dx + \int_{\Sigma \cap B_R(x_0)} |\nabla \eta|^p F^p(u) dx.$$

Hence, taking the limit as  $l \to \infty$  in the definition of F and G, by monotone convergence we conclude

$$\frac{\bar{c}}{2} \left( \int_{\Sigma \cap B_R(x_0)} \eta^{p^*} u^{qp^*} dx \right)^{\frac{p}{p^*}} \leq s^p \int_{\Sigma \cap B_R(x_0)} u^{(q-1)p} dx + ||\nabla \eta||_{\infty}^p \int_{\Sigma \cap B_R(x_0)} u^{qp} dx.$$

Since  $qp < p^*$  it follows that the right hand side is finite, hence by the inequality above and the arbitrariness of  $x_0$  we conclude that  $u \in L_{\text{loc}}^{qp^*}(\overline{\Sigma})$ . Thanks to this information, we can rewrite the equation satisfied by u as

follows:

$$-\operatorname{div}\left(a(\nabla u)\right) = f(x)u^{p-1} + g(x)$$

where

$$f(x) = \begin{cases} 0 & \text{if } u < 1 \\ u^{p^* - p} & \text{if } u \ge 1 , \end{cases}$$

and

$$g(x) = \begin{cases} 0 & \text{if } u > 1 \\ u^{p^* - 1} & \text{if } u \le 1 \end{cases}$$

,

Since  $u \in L_{\text{loc}}^{qp^*}$  we get that  $f \in L^r$  with  $r > \frac{n}{p}$  and  $g \in L^{\infty}$ . Hence, as in the proof of [205, Theorem 1], a classical Moser iteration argument yields the result.

*Remark* 3.1. As observed in the proof of [183, Theorem E.0.20], the Moser iteration argument can also be used to show that u is uniformly  $C^{0,\theta}$  up to the boundary.

### **3.1.2** Asymptotic bounds on u and $\nabla u$

The main goal of this subsection is to prove Proposition 3.1 below. Proposition 3.1 is a generalization of [224, Theorem 1.1] to the conical-anisotropic setting. The proof of Proposition 3.1 follows the one given in [224], although the lack of smoothness of  $\Sigma$  creates some nontrivial extra difficulties.

**Proposition 3.1.** Let  $1 and let u be a solution to (3.1). Then there exist two positive constants <math>C_0$  and  $C_1$  such that

$$\frac{C_0}{1+|x|^{\frac{n-p}{p-1}}} \le u(x) \le \frac{C_1}{1+|x|^{\frac{n-p}{p-1}}} \quad and \quad |\nabla u(x)| \le \frac{C_1}{1+|x|^{\frac{n-1}{p-1}}}, \tag{3.19}$$

for all  $x \in \Sigma$ .

Before giving the proof of Proposition 3.1, we first introduce a useful definition.

**Definition 3.1.** Given L > 0, we say that a convex cone C is *L*-Lipschitz if for any point  $x \in \partial C$  there exist  $r_x > 0$  and a unit vector  $\nu_x$  such that

$$B_{r_x}(x + Lr_x\nu_x) \subset \mathcal{C}.$$

Note that, by convexity of C, also the convex hull of  $B_{r_x}(x + L\nu_x) \cup \{x\}$  is contained in C.

In the spirit of [224, Lemma 2.3], we now prove a general lower bound on the  $L^{p^*}$  norms of solutions to our equation in convex cones, with a bound depending only on the Lipschitz constant.

Lemma 3.2 (Lower bound on the mass). Let u be a nontrivial solution to

$$\begin{cases} \operatorname{div} \left( a(\nabla u) \right) + u^{p^* - 1} = 0 & \text{ in } \mathcal{C} \\ u > 0 & \text{ in } \mathcal{C} \\ a(\nabla u) \cdot \nu = 0 & \text{ on } \partial \mathcal{C} \\ u \in \mathcal{D}^{1, p}(\mathcal{C}) \,, \end{cases}$$
(3.20)

where C is a L-Lipschitz convex cone and  $a(\xi)$  is as in (3.3). Then there exists a constant  $k_0 > 0$ , depending only on n, p, L, and  $\min_{\mathbb{S}^{n-1}} H$ , such that

$$||u||_{L^{p^*}(\mathcal{C})} \ge k_0.$$

*Proof.* As in [224, Lemma 2.3], the proof is based on the Sobolev inequality in C, and on the integral identity that one obtains by multiplying (3.20) by u

and integrating in C. However in this case a bit more carefulness is needed, especially to quantify the dependencies.

First of all, up to a translation, we can assume that C has vertex at O. Then, since C is *L*-Lipschitz, there exist  $r_0 > 0$  and a unit vector  $\nu_0$  such that  $B_{r_0}(Lr_0\nu_0) \subset C$ . Therefore, since C is a convex cone, this implies that the cone

$$\hat{\mathcal{C}}_L := \bigcup_{r>0} B_r(Lr\nu_0)$$

is contained inside  $\mathcal{C}$ .

We now want to estimate the Sobolev constant of  $\mathcal{C}$ . To this aim we define the constant  $\mathcal{S}_L$  as

$$\inf\left\{\frac{\left(\int_{\Omega}|\nabla\varphi|^{p}dx\right)^{1/p}}{\left(\int_{\Omega}|\varphi|^{p^{*}}dx\right)^{1/p^{*}}}: \Omega \text{ is convex}, B_{1}\cap\hat{\mathcal{C}}_{L}\subset\Omega\subset B_{1}, \varphi\in C^{1}(\Omega), \varphi|_{\partial B_{1}\cap\hat{\mathcal{C}}_{L}}=0\right\}.$$

Since the set of convex domains  $\Omega \subset B_1$  containing  $B_1 \cap \hat{\mathcal{C}}_L$  are uniformly Lipschitz, standard arguments in the calculus of variations show that  $\mathcal{S}_L$  is positive.

We now notice that, given any function  $\psi \in C_c^1(\overline{\mathcal{C}})$ , there exists  $\lambda > 0$ large such that  $\psi_{\lambda}(x) := \psi(\lambda x)$  satisfies  $\psi_{\lambda} \in C^1(\mathcal{C})$  and  $\psi_{\lambda}|_{\partial B_1 \cap \widehat{\mathcal{C}}_L} = 0$  (since  $\partial B_1 \cap \widehat{\mathcal{C}}_L \subset \partial B_1 \cap \mathcal{C}$ ). Hence, we can bound

$$\frac{\left(\int_{\mathcal{C}} |\nabla\psi|^p dx\right)^{1/p}}{\left(\int_{\mathcal{C}} |\psi|^{p^*} dx\right)^{1/p^*}} = \frac{\left(\int_{\mathcal{C}} |\nabla\psi_\lambda|^p dx\right)^{1/p}}{\left(\int_{\mathcal{C}} |\psi_\lambda|^{p^*} dx\right)^{1/p^*}} \ge \mathcal{S}_L.$$

Since  $\psi \in C_c^1(\overline{\mathcal{C}})$  is arbitrary, it follows by approximation that

$$\left(\int_{\mathcal{C}} |\nabla \psi|^p dx\right)^{1/p} \ge \mathcal{S}_L \left(\int_{\mathcal{C}} |\psi|^{p^*} dx\right)^{1/p^*} \qquad \forall \, \psi \in \mathcal{D}^{1,p}(\mathcal{C}).$$

Applying this inequality to u and defining  $c_H := \min_{|\xi|=1} H(\xi)$ , we get

$$\int_{\mathcal{C}} H(\nabla u)^p dx \ge c_H^p \int_{\mathcal{C}} |\nabla u|^p dx \ge (c_H \mathcal{S}_L)^p \left( \int_{\mathcal{C}} u^{p^*} dx \right)^{p/p^*}.$$

On the other hand, multiplying (3.20) by u and integrating in C, we get

$$\int_{\mathcal{C}} H(\nabla u)^p dx = \int_{\mathcal{C}} u^{p^*} dx.$$

Combining the last two equations yield the desired lower bound.

Remark 3.2. An alternative proof of Lemma 3.2 can be obtained by computing the optimal Sobolev constant of C (using Appendix B) and noticing that this constant is bounded below in terms only of  $n, p, H_0$ , and the volume of  $C \cap B_1$ . In particular, whenever C is *L*-Lipschitz then  $\hat{C}_L \subseteq C$  and  $|C \cap B_1| \ge |\hat{C}_L \cap B_1|$ , and one concludes that the Sobolev constant of C is controlled by (actually, it is larger or equal than) the one of  $\hat{C}_L$ . We shall also need a doubling-type property on u which is proved in [187, Lemma 5.1] (see also [224, Lemma 3.1]). Below we state a version of this doubling property which is suitable for our setting.

Note that, by convexity, there exists a constant  $L_{\Sigma} > 0$  such that  $\Sigma$  is  $L_{\Sigma}$ -Lipschitz. Then we let  $k_0 > 0$  be the constant provided by Lemma 3.2 with  $L = L_{\Sigma}$ .

**Lemma 3.3** (Doubling property [187]). Let u be a solution to (3.20), let  $L_{\Sigma}$  be the Lipschitz constant of  $\Sigma$ , and let  $k_0 > 0$  be the constant provided by Lemma 3.2 with  $L = L_{\Sigma}$ .

Let  $k \in (0, k_0)$ , r > 0, and  $r' \in (0, r)$  be fixed, and set

$$r'' = \frac{r+r'}{2} \,.$$

Then for any  $x \in \overline{\Sigma} \setminus B_{r''}$  and  $\alpha > 0$  such that the distance d between x and  $\Sigma \cap B_{r''}$  satisfies

$$d(x, \Sigma \cap B_{r''})u(x)^{\frac{\nu}{n-p}} > 2\alpha, \qquad (3.21)$$

there exists a point  $y_0 \in \overline{\Sigma} \setminus B_{r''}$  such that

$$d(y_0, \Sigma \cap B_{r''})u(x)^{\frac{p}{n-p}} > 2\alpha, \quad u(x_0) \le u(y_0),$$
(3.22)

and

$$u(y) \le 2^{\frac{n-p}{p}} u(y_0) \quad \text{for all } y \in \Sigma \cap B_{\bar{r}}(y_0) , \qquad (3.23)$$

where  $\bar{r} = \alpha u(y_0)^{-\frac{p}{n-p}}$ .

Proof of Proposition 3.1. We divide the proof of Proposition 3.1 in three steps. In Step 1 we give a preliminary decay estimate on u (which is not sharp). In Step 2 we prove that  $u \in L^{\hat{p}-1,\infty}(\Sigma)$  for a suitable  $\hat{p}$ . Finally, in Step 3 we prove (3.19).

• Step 1: Let u be a solution of (3.1), and for  $k \in (0, k_0)$  define

$$r_k(u) := \inf\{r > 0 : ||u||_{L^{p^*}(\Sigma \setminus B_r)} < k\}.$$
(3.24)

Then, for any fixed  $k \in (0, k_0)$  and  $r > r_k(u)$ , there exists a constant  $K_0$  such that

$$|u(x)| \le K_0 H_0(x)^{\frac{p-n}{p}} \quad for \ all \ x \in \overline{\Sigma} \setminus B_r \,. \tag{3.25}$$

In order to prove the assertion, it suffices to show the existence of a constant  $K_1$  such that

$$d(x, \Sigma \cap B_{r''})u(x)^{\frac{p}{n-p}} \le K_1 \quad \text{for all } x \in \overline{\Sigma} \setminus B_r , \qquad (3.26)$$

where r'' = (r + r')/2 and  $r' \in (0, r)$  is fixed. We prove (3.26) by contradiction. Suppose there exists a sequence of points  $\{x_{\alpha}\}_{\alpha \in \mathbb{N}} \subset \overline{\Sigma} \setminus B_r$  such that

$$d(x_{\alpha}, \Sigma \cap B_{r''})u(x_{\alpha})^{\frac{p}{n-p}} > 2\alpha.$$
(3.27)

Since  $B_{r''} \subset B_r$ , it follows from (3.27) and Lemma 3.3 that there exists a sequence of points  $\{y_{\alpha}\}_{\alpha \in \mathbb{N}} \subset \overline{\Sigma} \setminus B_{r''}$  such that

$$d(y_{\alpha}, \Sigma \cap B_{r''})u(y_{\alpha})^{\frac{p}{n-p}} > 2\alpha, \quad u(x_{\alpha}) \le u(y_{\alpha}), \qquad (3.28)$$

 $\operatorname{and}$ 

$$u(y) \le 2^{\frac{n-p}{p}} u(y_{\alpha}) \quad \text{for all } y \in \Sigma \cap B_{\bar{r}}(y_{\alpha}).$$
(3.29)

We observe that, since u is bounded, the sequences  $\{x_{\alpha}\}_{\alpha \in \mathbb{N}}$  and  $\{y_{\alpha}\}_{\alpha \in \mathbb{N}}$  are both divergent as  $\alpha \to \infty$ .

For any  $\alpha \in \mathbb{N}$  and  $y \in \overline{\Sigma}$ , we define

$$\tilde{u}_{\alpha}(y) := u(y_{\alpha})^{-1}u(m_{\alpha}^{-1}y + y_{\alpha})$$
(3.30)

where  $m_{\alpha} := u(y_{\alpha})^{\frac{-p}{n-p}}$ . From (3.1) we obtain

$$\begin{cases} -\Delta_p^H \tilde{u}_{\alpha} = \tilde{u}_{\alpha}^{p^*-1} & \text{in } \Sigma_{\alpha} \\ \tilde{u}_{\alpha}(\mathcal{O}) = 1, \\ a(\nabla \tilde{u}_{\alpha}) \cdot \nu = 0 & \text{on } \partial \Sigma_{\alpha} , \end{cases}$$
(3.31)

where

$$\Sigma_{\alpha} := m_{\alpha}(\Sigma - y_{\alpha}) = \{ y \in \mathbb{R}^n : m_{\alpha}^{-1}y + y_{\alpha} \in \Sigma \}$$

is a convex cone.

It is immediate to check that the cones  $\Sigma_{\alpha}$  are  $L_{\Sigma}$ -Lipschitz. Furthermore, if we set  $\mu_{\alpha} := u(y_{\alpha})^{-1}$ , (3.29) and (3.30) yield that

$$\tilde{u}_{\alpha}(-y_{\alpha}m_{\alpha}) = \mu_{\alpha}u(\mathcal{O}) \neq 0 \text{ and } \tilde{u}_{\alpha}(y) \leq 2^{\frac{n-p}{p}} \text{ for all } y \in \Sigma_{\alpha} \cap B_{\alpha}.$$
 (3.32)

At this point we consider the ratio

$$q_{\alpha} := \frac{m_{\alpha}}{|y_{\alpha}|} \,.$$

Observe that (by (3.28))  $q_{\alpha} \to 0$  as  $\alpha \to \infty$ .

Since  $|y_{\alpha}| \to +\infty$ , the ratio between  $-y_{\alpha}m_{\alpha}$  and the scaling factor  $m_{\alpha}$  goes to infinity. Hence, one of the following two cases may occur as  $\alpha \to \infty$ :

- (i) the sequence of cones  $\{\Sigma_{\alpha}\}_{\alpha \in \mathbb{N}}$  converges to  $\mathbb{R}^n$  (this happens if the distance between  $m_{\alpha}y_{\alpha}$  and  $\partial \Sigma_{\alpha}$  goes to infinity);
- (*ii*) the sequence of cones  $\{\Sigma_{\alpha}\}_{\alpha \in \mathbb{N}}$  converges to a  $L_{\Sigma}$ -Lipschitz convex cone  $\mathcal{C}$ , not necessarily centered at the origin (this happens if the distance between  $m_{\alpha}y_{\alpha}$  and  $\partial\Sigma_{\alpha}$  remains bounded).

We now look in both cases at the behaviour of the functions  $\{u_{\alpha}\}_{\alpha \in \mathbb{N}}$ . We consider the two cases separately.

- Case (i): fix a ball  $B_R$ . Then there exists  $\overline{\alpha} \in \mathbb{N}$  such that  $\Sigma_{\alpha} \cap B_R = B_R$  for every  $\alpha \geq \overline{\alpha}$ ; moreover  $\tilde{u}_{\alpha}$  (for every  $\alpha \geq \overline{\alpha}$ ) is a solution of (3.31) in  $B_R$ . From (3.9), (3.32), and [85], there exist a constant C > 0 and a real number  $\theta \in (0, 1)$  such that

$$\|\tilde{u}_{\alpha}\|_{C^{1,\theta}(B_{R/2})} \le C \tag{3.33}$$

for any  $\alpha \geq \overline{\alpha}$ . Since R > 0 is arbitrary, Ascoli-Arzelà Theorem and a diagonal argument imply that  $\{\tilde{u}_{\alpha}\}_{\alpha \in \mathbb{N}}$  converges (up to subsequence) in  $C^{1}_{\text{loc}}(\mathbb{R}^{n})$  to some function  $\tilde{u}_{\infty}$ . By construction we have that  $\tilde{u}_{\infty} \in \mathcal{D}^{1,p}(\mathbb{R}^{n}), \tilde{u}_{\infty}(\mathcal{O}) = 1$ , and  $\tilde{u}_{\infty}$  is a weak solution of

$$-\Delta_p^H \tilde{u}_{\infty} = \tilde{u}_{\infty}^{p^*-1} \quad \text{in } \mathbb{R}^n \,. \tag{3.34}$$

- Case (*ii*): consider a ball  $B_R$ . Then for every compact set  $K \subset \subset B_R \cap C$ there exists  $\overline{\alpha} \in \mathbb{N}$  such that  $K \subset \Sigma_{\alpha} \cap B_R$  for every  $\alpha \geq \overline{\alpha}$ . As in Case (*i*), for every  $\alpha \geq \overline{\alpha}$  the function  $\tilde{u}_{\alpha}$  is a solution of (3.31) in K, and there exist a constant C > 0 and a real number  $\theta \in (0, 1)$  such that

$$||\tilde{u}_{\alpha}||_{C^{1,\theta}(K')} \le C \tag{3.35}$$

for any  $\alpha \geq \overline{\alpha}$  and  $K' \subset \mathbb{C} K$ . In addition, it follows by Remark 3.1 that the functions  $\tilde{u}_{\alpha}$  are uniformly  $C^{0,\theta}$  inside  $B_R \cap \overline{\mathcal{C}}$  for any R > 0. Hence, again Ascoli-Arzelà Theorem and a diagonal argument imply that  $\{\tilde{u}_{\alpha}\}_{\alpha \in \mathbb{N}}$  converges (up to subsequence) in  $C^0(B_R \cap \overline{\mathcal{C}}) \cap C^1_{\text{loc}}(B_R \cap \mathcal{C})$  to some function  $\tilde{u}_{\infty}$ , for any R > 0. Taking the limit in the weak formulation of the equation, we obtain that  $\tilde{u}_{\infty} \in \mathcal{D}^{1,p}(\mathcal{C}), \ \tilde{u}_{\infty}(\mathcal{O}) = 1$ , and  $\tilde{u}_{\infty}$  is a weak solution of

$$\begin{cases} -\Delta_p^H \tilde{u}_{\infty} = \tilde{u}_{\infty}^{p^*-1} & \text{in } \mathcal{C} \\ a(\nabla \tilde{u}_{\infty}) \cdot \nu = 0 & \text{on } \partial \mathcal{C} \,. \end{cases}$$
(3.36)

We now notice that, in both cases, for any  $\rho > 0$  we have

$$\|\tilde{u}_{\alpha}\|_{L^{p^*}(\Sigma_{\alpha}\cap B_{\rho})} = \|u\|_{L^{p^*}(\Sigma\cap B_{\rho m_{\alpha}}(y_{\alpha}))}.$$
(3.37)

Also, by (3.28), since  $r_k(u) < r''$  we get

$$B_{\rho m_{\alpha}}(y_{\alpha}) \cap B_{r_k(u)} = \emptyset \tag{3.38}$$

for  $\alpha$  large. Thus, from (3.37), (3.38), and by definition of  $r_k(u)$ , we obtain

$$\|\tilde{u}_{\alpha}\|_{L^{p^*}(\Sigma_{\alpha} \cap B_{\rho})} \le k \tag{3.39}$$

for  $\alpha$  large. Thus, taking the limit in (3.39) as  $\alpha \to \infty$  and then as  $\rho \to \infty$ , yields

$$||\tilde{u}_{\infty}||_{L^{p^{*}}(\mathbb{R}^{n})} \leq k \quad \text{or} \quad ||\tilde{u}_{\infty}||_{L^{p^{*}}(\mathcal{C})} \leq k,$$
 (3.40)

in Case (i) or Case (ii), respectively. Since  $k < k_0$  with  $k_0 > 0$  as in Lemma 3.2, it follows by (3.34) (resp. (3.36)) and (3.40) that  $\tilde{u}_{\infty} \equiv 0$  in Case (i) (resp. Case (ii)), a contradiction to the fact that  $\tilde{u}_{\infty}(\mathcal{O}) = 1$ . This completes the proof of the assertion of Step 1.

• Step 2: Let u be a solution of (3.20). Then  $u \in L^{\hat{p}-1,\infty}(\Sigma)$  for  $\hat{p} := \frac{p(n-1)}{n-p}$ .

Recall that, given a set  $\Omega$  and  $r \geq 1$ , one defines the space  $L^{r,\infty}(\Omega)$  as the set of all measurable functions  $v: \Omega \to \mathbb{R}$  such that

$$||v||_{L^{r,\infty}(\Omega)} := \sup_{h>0} \left\{ h \max\left(\{|u|>h\}\right)^{1/r} \right\} < \infty.$$
(3.41)

Using the Sobolev inequality in cones, the proof of this step can be easily adapted from the case of  $\mathbb{R}^n$  (see [224, Lemma 2.2]) and for this reason is omitted.

### • Step 3: Proof of (3.19).

The proof of this step closely follows the proof of [224, Theorem 1.1], which in turn uses [220, Theorem 1.3] and [205, Theorem 5]. Even if [220, Theorem 1.3] and [205, Theorem 5] are stated in a local setting, thanks to the homogeneous Neumann boundary condition they can be easily extended to our setting. For this reason we only give a sketch of the proof, following the argument of [224, Theorem 1.1].

Let k and r be as in Step 1. For any R > 0 and  $y \in \Sigma$ , we define

$$u_R(y) := R^{\frac{n-p}{p-1}} u(Ry) \,. \tag{3.42}$$

From (3.1) we obtain

$$-\Delta_p^H u_R = R^{-\frac{p}{p-1}} u_R^{p^*-1} \qquad \text{in } \Sigma \,. \tag{3.43}$$

Also, writing  $u_R^{p^*-1} = u_R^{p^*-p} u_R^{p-1}$  and using (3.25), we have

$$R^{-\frac{p}{p-1}}u_R^{p^*-1} \le K_0^{p^*-p}u_R^{p-1} \quad \text{in}\,\overline{\Sigma} \setminus B_1\,, \tag{3.44}$$

provided that  $R \ge r$ . Thus, it follows from (3.43), (3.44), and [220, Theorem 1.3], that for any  $\varepsilon > 0$  it holds

$$||u_R||_{L^{\infty}(\Sigma \cap (B_4 \setminus B_2))} \le C_{\varepsilon}||u_R||_{L^{p-1+\varepsilon}(\Sigma \cap (B_5 \setminus B_1))}$$
(3.45)

for some constant  $C_{\varepsilon} > 0$ . We fix  $\varepsilon_0 = \varepsilon_0(n, p)$  such that  $0 < \varepsilon_0 < \hat{p} - p$ , where  $\hat{p}$  is as in Step 2. Since

$$||u_R||_{L^{p-1+\varepsilon_0}(\Sigma\cap(B_5\setminus B_1))} \le C_0||u_R||_{L^{\hat{p}-1,\infty}(\Sigma\cap(B_5\setminus B_1))},$$

for  $C_0 = C_0(n, p)$ , recalling Step 2 we obtain that

$$||u_R||_{L^{\infty}(\Sigma \cap (B_4 \setminus B_2))} \le C_1 \tag{3.46}$$

for some constant  $C_1$ . Hence, by (3.43), (3.46), and elliptic regularity theory for *p*-Laplacian type equations [85, 218], we get

$$\|\nabla u_R\|_{L^{\infty}(\Sigma \cap (B_{7/2} \setminus B_{5/2}))} \le C_2 \tag{3.47}$$

for some constant  $C_2$ . Here we notice that, even if (3.47) is proved in [85, Section 3] in a local setting (see also [53], where the authors prove global Lipschitz regularity in convex domains for the case when H coincides with the Euclidean norm), the argument easily extends to our setting by an approximation argument. Indeed, as in the proof of Proposition 3.2 below, one can work in regularized domains and, because of the presence of the boundary, with respect to [85, Section 3] it appears an extra boundary term. However, this can be dropped since the second fundamental form of  $\partial \Sigma$  is nonnegative definite (compare with (3.57)-(3.60) below, or with [53, Proof of Theorem 1.2, Step 1]).

Finally, for any  $x \in \mathbb{R}^n \setminus B_{3r}$ , applying (3.46) and (3.47) with R = |x|/3 we obtain

$$u(x) \le C_3 |x|^{\frac{p-n}{p-1}}$$
 and  $|\nabla u(x)| \le C_3 |x|^{\frac{1-n}{p-1}}$  (3.48)

for some constant  $C_3$ . Since u and  $\nabla u$  are uniformly bounded in  $B_{3r}$ , (3.19) follows. Finally, to prove the lower bound in (3.19) one argues as in [224, pages 159-160].

### 3.1.3 Asymptotic estimates on higher order derivatives

By using a Caccioppoli-type inequality, in this subsection we prove Proposition 3.2 below which will be useful in the proof of Theorem 3.1. In particular it will avoid the use of an asymptotic lower bound on  $|\nabla u|$ , which is crucial in [203].

**Proposition 3.2.** Let  $\Sigma$  be a convex cone, and let u be a solution to (3.1) with  $a(\cdot)$  given by (3.3), where H satisfies the assumptions of Theorem 3.1. Then  $a(\nabla u) \in W^{1,2}_{\text{loc}}(\overline{\Sigma})$ , and for any  $\gamma \in \mathbb{R}$  the following asymptotic estimate holds:

$$\int_{B_r \cap \Sigma} |\nabla(a(\nabla u))|^2 u^{\gamma} \, dx \le C \left( 1 + r^{-n - \gamma \frac{n-p}{p-1}} \right) \qquad \forall r \ge 1, \tag{3.49}$$

where C is a positive constant independent of r.

*Proof.* The estimate (3.49) is obtained by using a Caccioppoli-type inequality. We argue by approximation, following the approach in [14, 54].

We approximate  $\Sigma$  by a sequence of convex cones  $\{\Sigma_k\}$  such that  $\Sigma_k \subseteq \Sigma$ and  $\partial \Sigma_k \setminus \{\mathcal{O}\}$  is smooth. Also, we fix a point  $\bar{x} \in \bigcap_k \Sigma_k$ , and for k fixed we let  $u_k$  be the solution of<sup>2</sup>

$$\begin{cases} \operatorname{div} \left( a(\nabla u_k) \right) + u^{p^* - 1} = 0 & \text{ in } \Sigma_k \\ u_k(\bar{x}) = u(\bar{x}) & \\ a(\nabla u_k) \cdot \nu = 0 & \text{ on } \partial \Sigma_k \,. \end{cases}$$
(3.50)

 $\operatorname{Set}$ 

$$a^{\ell}(z) := (a * \phi_{\ell})(z) \qquad \text{for } z \in \mathbb{R}^n \,, \tag{3.51}$$

where  $\{\phi_{\ell}\}$  is a family of radially symmetric smooth mollifiers. Standard properties of convolution and the fact  $a(\cdot)$  is continuous imply  $a^{\ell} \to a$  uniformly on compact subset of  $\mathbb{R}^n$ . From [99, Lemma 2.4] we have that  $a^{\ell}$  satisfies the first condition in (3.12) with s replaced by  $s_{\ell}$ , where  $s_{\ell} \to 0$  as  $\ell \to \infty$ . In addition, since

$$\frac{1}{\tilde{\alpha}}(|z|^2 + s_\ell^2)^{\frac{p-2}{2}}|\xi|^2 \le \nabla a^\ell(z)\xi \cdot \xi, \quad \text{for every } \xi, z \in \mathbb{R}^n,$$

for some  $\tilde{\alpha} > 0$ , we obtain that  $a^{\ell}$  satisfies also the second condition in (3.12). Let  $u_{k,\ell}$  be a solution of

$$\begin{cases} \operatorname{div}\left(a^{\ell}(\nabla u_{k,\ell})\right) + u^{p^* - 1} = 0 & \operatorname{in} \Sigma_k\\ a^{\ell}(\nabla u_{k,\ell}) \cdot \nu = 0 & \operatorname{on} \partial \Sigma_k \end{cases}$$
(3.52)

(this solution can be constructed analogously to  $u_k$ ).

We notice that  $u_{k,\ell}$  is unique up to an additive constant. Also, because u is locally bounded, the functions  $u_{k,\ell}$  are  $C^{1,\theta}_{\text{loc}}(\overline{\Sigma}_k \setminus \{\mathcal{O}\}) \cap C^{0,\theta}_{\text{loc}}(\overline{\Sigma}_k)$ , uniformly in  $\ell$ . In particular, assuming without loss of generality that  $u_{k,\ell}(\bar{x}) = u(\bar{x})$  for

$$\min_{v} \left\{ \int_{\Sigma_k \cap B_R} \left[ \frac{1}{p} H(\nabla v)^p - u^{p^* - 1} v \right] dx : v = 0 \text{ on } \Sigma_k \cap \partial B_R \right\},\$$

 $<sup>^2</sup>$  The function  $u_k$  can be found by considering first the minimizer  $v_{k,R}$  of the minimization problem

then setting  $u_{k,R}(x) := v_{k,R}(x) + u(\bar{x}) - v_{k,R}(\bar{x})$ , and finally taking the limit of  $u_{k,R}$  as  $R \to \infty$  (note that the functions  $\tilde{u}_{k,R}$  are uniformly  $C^{1,\theta}$  in every compact subset of  $\Sigma$ , and uniformly Hölder continuous up to the boundary).

some fixed point  $\bar{x} \in \Sigma_k$ , as  $\ell \to \infty$  one sees that  $u_{k,\ell}$  converges in  $C^1_{\text{loc}}$  to the unique solution  $\bar{u}_k$  of

$$\begin{cases} \operatorname{div} \left( a(\nabla \bar{u}_k) \right) + u^{p^* - 1} = 0 & \text{ in } \Sigma_k \\ \bar{u}_k(\bar{x}) = u(\bar{x}) & \\ a(\nabla \bar{u}_k) \cdot \nu = 0 & \text{ on } \partial \Sigma_k \,. \end{cases}$$
(3.53)

Since  $u_k$  is also a solution of the problem above, it follows by uniqueness that  $\bar{u}_k = u_k$  and therefore  $u_{k,\ell}$  converges to  $u_k$  as  $\ell \to \infty$ . Analogously,  $u_k \to u$  as  $k \to \infty$ .

Given R > 1 large, we define

$$\Omega_k := \Sigma_k \cap B_R, \quad \Gamma_{k,0} := \Sigma_k \cap \partial B_R, \quad \Gamma_{k,1} := \partial \Sigma_k \cap B_R.$$

Note that, since u is uniformly positive inside  $\Sigma$  (see Proposition 3.1), for k large enough (depending on R) also  $u_k$  is uniformly positive inside  $\Omega_k$ , and hence for  $\ell$ large enough we have that  $u_{k,\ell}$  is also uniformly positive inside  $\Omega_k$ . In the sequel we shall always assume that k and  $\ell$  are sufficiently large so that this positivity property holds. We now fix k and deal with the functions  $u_{k,\ell}$ . To simplify the notation, we shall drop the dependency on k and we write  $u_{\ell}, \Sigma, \Omega, \Gamma_0, \Gamma_1$ instead of  $u_{k,\ell}, \Sigma_k, \Omega_k, \Gamma_{k,0}, \Gamma_{k,1}$ , respectively.

The idea is to prove a Caccioppoli-type inequality for  $u_{\ell}$  and then let  $\ell \to \infty$ . Since  $u_{\ell}$  solves a non-degenerate equation, we have that  $u_{\ell} \in C^1 \cap W^{2,2}_{\text{loc}}(\overline{\Sigma})$  and furthermore we have  $a^{\ell}(\nabla u_{\ell}) \in W^{1,2}_{\text{loc}}(\overline{\Sigma})$ . In addition, since  $\Sigma$  is smooth outside the origin,  $u_{\ell}$  is of class  $C^2$  in  $\overline{\Omega}$  away from  $\Gamma_1 \cup \{\mathcal{O}\}$ .

Multiply (3.52) by  $\psi \in C_c^{\infty}(B_R \setminus B_{1/R})$  and integrate over  $\Omega$  to get

$$\int_{\Omega} \operatorname{div} \left( a^{\ell} (\nabla u_{\ell}) \right) \psi \, dx = - \int_{\Omega} u^{p^* - 1} \psi \, dx,$$

that together with the divergence theorem gives

$$-\int_{\Omega} a^{\ell}(\nabla u_{\ell}) \cdot \nabla \psi \, dx + \int_{\partial \Omega} \psi a^{\ell}(\nabla u_{\ell}) \cdot \nu \, d\sigma = -\int_{\Omega} u^{p^*-1} \psi \, dx \,. \tag{3.54}$$

Since

$$\int_{\partial\Omega} \psi a^{\ell}(\nabla u_{\ell}) \cdot \nu \, d\sigma = \int_{\Gamma_1} \psi a^{\ell}(\nabla u_{\ell}) \cdot \nu \, d\sigma + \int_{\Gamma_0} \psi a^{\ell}(\nabla u_{\ell}) \cdot \nu \, d\sigma$$

from the fact that  $\psi \in C_c^{\infty}(B_R \setminus B_{1/R})$  and from the boundary condition in (3.52), we obtain that the second term in (3.54) vanishes; hence (3.54) becomes

$$-\int_{\Omega} a^{\ell}(\nabla u_{\ell}) \cdot \nabla \psi \, dx = -\int_{\Omega} u^{p^*-1} \psi \, dx \,. \tag{3.55}$$

Let  $\varphi \in C_c^{\infty}(B_R \setminus B_{1/R})$ , and for  $\delta > 0$  small define the set

$$\Omega_{\delta} := \{ x \in \Omega : \operatorname{dist} (x, \partial \Omega) > \delta \}.$$

Since  $\Omega \cap \operatorname{supp}(\varphi)$  is smooth, for  $\delta$  small enough we see that  $\Omega_{\delta} \setminus \Omega_{2\delta}$  is of class  $C^{\infty}$  inside the support of  $\varphi$ . In particular, every point  $x \in (\Omega_{\delta} \setminus \Omega_{2\delta}) \cap \operatorname{supp}(\varphi)$  can be written as

$$x = y - |x - y|\nu(y)$$

where  $y = y(x) \in \partial \Omega_{\delta}$  is the projection of x on  $\partial \Omega_{\delta}$  and  $\nu(y)$  is the outward normal to  $\partial \Omega_{\delta}$  at y. Moreover the set  $(\Omega_{\delta} \setminus \Omega_{2\delta}) \cap \operatorname{supp}(\varphi)$  can be parametrized on  $\partial \Omega_{\delta}$  by a  $C^1$  function g (see [112, Formula 14.98]).

Let  $\zeta_{\delta} : \Omega \to [0,1]$  be a cut-off function such that  $\zeta_{\delta} = 1$  in  $\Omega_{2\delta}$ ,  $\zeta_{\delta} = 0$  in  $\Omega \setminus \Omega_{\delta}$ , and

$$abla \zeta_{\delta}(x) = -\frac{1}{\delta}\nu(y(x)) \qquad \text{inside } \Omega_{\delta} \setminus \Omega_{2\delta}$$

Using  $\psi = \partial_m(\varphi \zeta_{\delta})$  in (3.55) with  $m \in \{1, \ldots, n\}$  and integrating by parts, we get

$$\sum_{i=1}^{n} \left( \int_{\Omega} \partial_{m} a_{i}^{\ell}(\nabla u_{\ell}) \zeta_{\delta} \partial_{i} \varphi \, dx + \int_{\Omega} \partial_{m} a_{i}^{\ell}(\nabla u_{\ell}) \varphi \partial_{i} \zeta_{\delta} \, dx \right)$$
$$= \int_{\Omega} \partial_{m} (u^{p^{*}-1}) \varphi \zeta_{\delta} \, dx \,,$$

where we use the notation  $a^{\ell} = (a_1^{\ell}, \ldots, a_n^{\ell})$  to denote the components of the vector field  $a^{\ell}$ .

Observe that, from the definition of  $\zeta_{\delta}$ , we have

$$\lim_{\delta \to 0} \int_{\Omega} \partial_m a_i^{\ell}(\nabla u_{\ell}) \zeta_{\delta} \partial_i \varphi \, dx = \int_{\Omega} \partial_m a_i^{\ell}(\nabla u_{\ell}) \partial_i \varphi \, dx \, .$$

Also, if we set

$$f(x) = \partial_m a_i^\ell(\nabla u_\ell(x))\varphi(x),$$

by the coarea formula we have

$$\int_{\Omega_{\delta} \setminus \Omega_{2\delta}} f \partial_i \zeta_{\delta} \, dx = -\frac{1}{\delta} \int_{\Omega_{\delta} \setminus \Omega_{2\delta}} \nu_i(y(x)) f dx$$
  
$$= -\frac{1}{\delta} \int_{\delta}^{2\delta} dt \int_{\partial \Omega_{\delta}} \nu_i(y(x)) f(y - t\nu(y)) |\det(Dg)| d\sigma(y)$$
  
$$= -\int_{1}^{2} ds \int_{\partial \Omega_{s\delta}} f(y - s\delta\nu(y)) \nu_i(y) |\det(Dg)| d\sigma(y) \, .$$

Since  $f \in C^0$ , we can pass to the limit and obtain

$$\lim_{\delta \to 0} \int_{\Omega} \partial_m a_i^{\ell}(\nabla u_{\ell}) \varphi \partial_i \zeta_{\delta} \, dx = - \int_{\partial \Omega} \partial_m a_i^{\ell}(\nabla u_{\ell}) \varphi \nu_i d\sigma \, .$$

Hence, we proved that

$$\sum_{i=1}^{n} \left( \int_{\Omega} \partial_{m} a_{i}^{\ell}(\nabla u_{\ell}) \partial_{i} \varphi \, dx - \int_{\partial \Omega} \partial_{m} a_{i}^{\ell}(\nabla u_{\ell}) \varphi \nu_{i} d\sigma \right) = \int_{\Omega} \partial_{m} (u^{p^{*}-1}) \varphi \, dx \,.$$
(3.56)

Now, let

$$\Omega_{\delta}^{t} := \{ x \in \Omega_{\delta} : \operatorname{dist} (x, \partial \Omega_{\delta}) > t \}$$

We notice that, if  $x \in (\Omega_{\delta} \setminus \Omega_{2\delta}) \cap \operatorname{supp}(\varphi)$  with  $x = y - t\nu(y)$ , then  $x \in \partial \Omega_{\delta}^{t}$ and the outward normal to  $\partial \Omega_{\delta}^{t}$  at x coincides with the outward normal to  $\partial \Omega_{\delta}$ at y. Hence, by writing  $\nu(x)$  in place of  $\nu(y)$ , we have

$$\partial_m a_i^{\ell}(\nabla u_{\ell}(x))\varphi(x)\nu_i(x) = \varphi(x)\partial_m(a^{\ell}(\nabla u_{\ell}(x))\cdot\nu(x)) -\varphi(x)a_i^{\ell}(\nabla u_{\ell}(x))\partial_m\nu_i(x).$$
(3.57)

Now, we take a cut-off function  $\eta \in C_c^{\infty}(B_R \setminus B_{1/R})$ , and for  $m \in \{1, \ldots, n\}$  we set  $\varphi = a_m^{\ell}(\nabla u_{\ell})u_{\ell}^{\gamma}\eta^2$  where  $\gamma \in \mathbb{R}$ , and in (3.57) we obtain

$$\partial_m a_i^{\ell}(\nabla u_{\ell}(x))\varphi(x)\nu_i(x) = a_m^{\ell}(\nabla u_{\ell}(x))u_{\ell}^{\gamma}(x)\eta^2(x)\partial_m \left(a^{\ell}(\nabla u_{\ell}(x))\cdot\nu(x)\right) - a_m^{\ell}(\nabla u_{\ell}(x))u_{\ell}^{\gamma}(x)\eta^2(x)a_i^{\ell}(\nabla u_{\ell}(x))\partial_m\nu_i(x).$$
(3.58)

(3.58) We notice that  $\partial_m \nu_i(x)$  is the second fundamental form  $\Pi_x^t$  of  $\partial \Omega_{\delta}^t$  at x:

$$\sum_{i,m=1}^n \partial_m \nu_i(x) a_i^\ell(\nabla u_\ell(x)) a_m^\ell(\nabla u_\ell(x)) = \mathrm{II}_x^t(a^\ell(\nabla u_\ell(x)), a^\ell(\nabla u_\ell(x))).$$

Since the cone  $\Sigma$  is convex then  $\mathrm{II}^t_x$  is non-negative definite, which implies that

$$\sum_{i,m=1}^{n} \partial_m \nu_i(x) a_i^\ell(\nabla u_\ell(x)) a_m^\ell(\nabla u_\ell(x)) \ge 0.$$
(3.59)

Hence (3.58) becomes

$$\sum_{i,m=1}^{n} \partial_m a_i^{\ell}(\nabla u_{\ell}(x))\varphi(x)\nu_i(x)$$

$$\leq \sum_{i,m=1}^{n} a_m^{\ell}(\nabla u_n(x))u_{\ell}^{\gamma}(x)\eta^2(x)\partial_m \left(a^{\ell}(\nabla u_{\ell}(x))\cdot\nu(x)\right), \quad (3.60)$$

and so, with the choice  $\varphi = a_m^\ell (\nabla u_\ell) u_\ell^\gamma \eta^2$ , we obtain

$$\sum_{i,m=1}^{n} \int_{\partial\Omega} \partial_m a_i^{\ell}(\nabla u_{\ell}) \varphi \nu_i d\sigma \leq \sum_{i,m=1}^{n} \int_{\partial\Omega} u_{\ell}^{\gamma} \eta^2 a_m^{\ell}(\nabla u_{\ell}) \partial_m \left( a^{\ell}(\nabla u_{\ell}) \cdot \nu \right) dx$$
$$= \sum_{i=1}^{n} \int_{\partial\Omega} u_{\ell}^{\gamma} \eta^2 a^{\ell}(\nabla u_{\ell}) \cdot \nabla \left( a^{\ell}(\nabla u_{\ell}) \cdot \nu \right) dx = 0,$$

where the last equality follows from the condition  $a^{\ell}(\nabla u_{\ell}) \cdot \nu = 0$  on  $\partial \Sigma$ . Indeed, this condition implies that  $a^{\ell}(\nabla u_{\ell})$  is a tangent vector-field and that the tangential derivative of  $a^{\ell}(\nabla u_{\ell}) \cdot \nu$  vanishes on  $\partial \Sigma$ .

Hence, recalling (3.56), we proved that

$$\sum_{i,m=1}^{n} \int_{\Omega} \partial_m a_i^{\ell}(\nabla u_{\ell}) \partial_i \left( a_m^{\ell}(\nabla u_{\ell}) u_{\ell}^{\gamma} \eta^2 \right) \, dx \le n \int_{\Omega} |\nabla (u^{p^*-1})| |a^{\ell}(\nabla u_{\ell})| u_{\ell}^{\gamma} \eta^2 \, dx \,.$$

$$(3.61)$$

Inequality (3.61) can be used in place of Equation (4.11) in [14, Proof of Theorem 4.1], and by arguing as in [14] we obtain

$$\int_{\Omega} |\nabla(a^{\ell}(\nabla u_{\ell}))|^2 \eta^2 u_{\ell}^{\gamma} dx \leq C \int_{\Omega} |\nabla(a^{\ell}(\nabla u_{\ell}))| |a^{\ell}(\nabla u_{\ell})| \eta u_{\ell}^{\frac{\gamma}{2}} |\nabla(\eta u_{\ell}^{\frac{\gamma}{2}})| dx + C \int_{\Omega} |\nabla(u^{p^*-1})| |a^{\ell}(\nabla u_{\ell})| u_{\ell}^{\gamma} \eta^2 dx \,.$$

From Hölder and Young inequalities, for any  $\epsilon \in (0, 1)$  we can bound

$$\begin{split} C \int_{\Omega} |\nabla(a^{\ell}(\nabla u_{\ell}))| |a^{\ell}(\nabla u_{\ell})| \eta u_{\ell}^{\frac{\gamma}{2}} |\nabla(\eta u_{\ell}^{\frac{\gamma}{2}})| \, dx \\ & \leq C\epsilon \int_{\Omega} |\nabla(a^{\ell}(\nabla u_{\ell}))|^2 \eta^2 u_{\ell}^{\gamma} \, dx + \frac{C}{\epsilon} \int_{\Omega} |a^{\ell}(\nabla u_{\ell})|^2 |\nabla(\eta u_{\ell}^{\frac{\gamma}{2}})|^2 \, dx, \end{split}$$

so choosing  $\epsilon$  small enough such that  $C\epsilon = 1/2$ , we obtain

$$\begin{split} \int_{\Omega} |\nabla(a^{\ell}(\nabla u_{\ell}))|^2 \eta^2 u_{\ell}^{\gamma} \, dx &\leq C \int_{\Omega} |a^{\ell}(\nabla u_{\ell})|^2 |\nabla(\eta u_{\ell}^{\frac{\gamma}{2}})|^2 \, dx \\ &+ C \int_{\Omega} |\nabla(u^{p^*-1})| |a^{\ell}(\nabla u_{\ell})| u_{\ell}^{\gamma} \eta^2 \, dx \end{split}$$

Recall that here  $\eta \in C_c^{\infty}(B_R \setminus B_{1/R})$ . However, by approximation the same property holds for any  $\eta \in C_c^{\infty}(\mathbb{R}^n)$ .

Now, we recall that we were writing  $u_{\ell}$  in place of  $u_{k,\ell}$ . Then, since  $u_{k,\ell} \to u_k$  in  $C^1_{\text{loc}}$  and  $a^{\ell} \to a$  locally uniformly, we can let  $\ell \to \infty$  to deduce that

$$\int_{\Omega_k} |\nabla(a(\nabla u_k))|^2 \eta^2 u_k^{\gamma} dx$$
  
$$\leq C \int_{\Omega_k} |a(\nabla u_k)|^2 |\nabla(\eta u_k^{\frac{\gamma}{2}})|^2 dx + C \int_{\Omega_k} |\nabla(u^{p^*-1})| |a(\nabla u_k)| u_k^{\gamma} \eta^2 dx. \quad (3.62)$$

In particular, taking  $\gamma = 0$ , (3.62) proves that  $a(\nabla u_k) \in W^{1,2}_{\text{loc}}(\overline{\Sigma}_k)$ , and  $\{a(\nabla u_k)\}_{k \in \mathbb{N}}$  is uniformly bounded in  $W^{1,2}_{\text{loc}}$ . Hence, letting  $k \to \infty$  in (3.62) we obtain

$$\begin{split} \int_{\Omega} |\nabla(a(\nabla u))|^2 \eta^2 u^{\gamma} \, dx \\ &\leq C \int_{\Omega} |a(\nabla u)|^2 |\nabla(\eta u^{\frac{\gamma}{2}})|^2 \, dx + C \int_{\Omega} |\nabla(u^{p^*-1})| |a(\nabla u)| u^{\gamma} \eta^2 \, dx. \end{split}$$

Finally, the asymptotic estimate (3.49) follows from (3.19).

# 3.2 Proof of Theorem 3.1

As already mentioned in the introduction, we consider the auxiliary function

$$v = u^{-\frac{p}{n-p}} \tag{3.63}$$

where u is a solution of (3.1). A straightforward computation shows that v > 0 satisfies the following problem

$$\begin{cases} \Delta_p^{\tilde{H}} v = f(v, \nabla v) & \text{in } \Sigma\\ \tilde{a}(\nabla v) \cdot \nu = 0 & \text{on } \partial \Sigma \,, \end{cases}$$
(3.64)

where  $\Delta_p^{\tilde{H}} v = \operatorname{div} \left( \tilde{a}(\nabla v) \right)$  with

$$\tilde{a}(\xi) = \tilde{H}^{p-1}(\xi)\nabla\tilde{H}(\xi) \qquad \forall \xi \in \mathbb{R}^n , \qquad (3.65)$$

and where we set

$$f(v,\nabla v) = \left(\frac{p}{n-p}\right)^{p-1} \frac{1}{v} + \frac{n(p-1)}{p} \frac{\tilde{H}^p(\nabla v)}{v}, \qquad (3.66)$$

 $\operatorname{with}$ 

$$\widetilde{H}(\xi) = H(-\xi) \qquad \forall \xi \in \mathbb{R}^n.$$
(3.67)

It is clear that v inherits some properties from u. In particular  $v \in C_{\text{loc}}^{1,\theta}$ , and it follows from Proposition 3.1 that there exist constants  $C_0, C_1 > 0$  such that

$$C_0|x|^{-\frac{p}{p-1}} \le v(x) \le C_1|x|^{-\frac{p}{p-1}}$$
(3.68)

 $\operatorname{and}$ 

$$|\nabla v(x)| \le C_1 |x|^{-\frac{1}{p-1}} \tag{3.69}$$

for  $|\boldsymbol{x}|$  sufficiently large. Higher regularity results for v are summarized in the following lemma.

**Lemma 3.4.** Let v be given by (3.63). Then, for every  $\sigma \in \mathbb{R}$ , the asymptotic estimate

$$\int_{B_r \cap \Sigma} |\nabla(\tilde{a}(\nabla v))|^2 v^{\sigma} \, dx \le C \left(1 + r^{n + \frac{\sigma p}{p-1}}\right) \qquad \forall r \ge 1 \tag{3.70}$$

holds.

*Proof.* We notice that

$$\tilde{a}(\nabla v) = -\left(\frac{p}{n-p}\right)^{p-1} u^{-\frac{n(p-1)}{n-p}} a(\nabla u)$$

 $\operatorname{and}$ 

$$\nabla(\tilde{a}(\nabla v)) = -\left(\frac{p}{n-p}\right)^{p-1} \times \left[u^{-\frac{n(p-1)}{n-p}} \nabla(a(\nabla u)) - \frac{n(p-1)}{n-p} u^{\frac{p(1-n)}{n-p}} \nabla u \otimes a(\nabla u)\right],$$

so it follows from Proposition 3.2 that

$$\tilde{a}(\nabla v) \in W^{1,2}_{\text{loc}}(\overline{\Sigma}).$$
 (3.71)

Finally, the asymptotic estimate (3.70) follows from (3.49) and (3.19).

### 3.2.1 An integral inequality

In this subsection, by using the convexity of the cone, we show that v satisfies an integral inequality.

We recall that the second symmetric function  $S^2(M)$  of a  $n \times n$  matrix  $M = (m_{ij})$  is the sum of all the principal minors of A of order two, and we have

$$S^{2}(M) = \frac{1}{2} \sum_{i,j} S^{2}_{ij}(M) m_{ij}, \qquad (3.72)$$

where

$$S_{ij}^2(M) = -m_{ji} + \delta_{ij} \operatorname{Tr}(M) \,.$$

As proved in [55, Lemma 3.2] (see also Lemma 2.3), given two symmetric matrices  $B, C \in \mathbb{R}^{n \times n}$  with B positive semidefinite, and by setting M = BC, we have the following Newton's type inequality:

$$S^{2}(M) \le \frac{n-1}{2n} \operatorname{Tr}(M)^{2}$$
. (3.73)

Moreover, if  $Tr(M) \neq 0$  and equality holds in (3.73), then

$$M = \frac{\operatorname{Tr}\left(M\right)}{n} \operatorname{Id}_{*}$$

and B is positive definite. As we will describe later, we will apply (3.73) to the matrix  $M = \nabla[\tilde{a}(\nabla v)]$ .

We start from the following differential identity (see [29]). We use the Einstein convention of summation over repeated indices.

**Lemma 3.5.** Let v be a positive function of class  $C^3$  and let  $V : \mathbb{R}^n \to \mathbb{R}^+$  be of class  $C^3(\mathbb{R}^n)$  and such that  $V(\nabla v) \operatorname{div} (\nabla V(\nabla v))$  can be continuously extended to zero at  $\nabla v = 0$ . Let

$$W = \nabla [\nabla_{\xi} V(\nabla v)] = V_{\xi_i \xi_j} (\nabla v) \partial_{ij}^2 v. \qquad (3.74)$$

Then, for any  $\gamma \in \mathbb{R}$  we have

$$2v^{\gamma}S^{2}(W) = \operatorname{div}\left(v^{\gamma}S_{ij}^{2}(W)V_{\xi_{i}}(\nabla v)\right) - \gamma v^{\gamma-1}S_{ij}^{2}(W)V_{\xi_{i}}(\nabla v)\partial_{j}v \qquad (3.75)$$

and

$$\operatorname{div} \left( v^{\gamma} S_{ij}^{2}(W) V_{\xi_{i}}(\nabla v) + \gamma(p-1) v^{\gamma-1} V(\nabla v) V_{\xi_{j}}(\nabla v) \right)$$

$$= 2v^{\gamma} S^{2}(W) + \gamma(\gamma-1)(p-1) v^{\gamma-2} V(\nabla v) V_{\xi_{i}}(\nabla v) \partial_{i} v$$

$$+ \gamma v^{\gamma-1} \left( (p-1) V(\nabla v) + V_{\xi_{i}}(\nabla v) \partial_{i} v \right) \operatorname{Tr} (W)$$

$$+ \gamma v^{\gamma-1} \left( (p-1) V_{\xi_{i}}(\nabla v) V_{\xi_{j}}(\nabla v) \partial_{ij}^{2} v + V_{\xi_{j}\xi_{l}}(\nabla v) \partial_{li}^{2} v V_{\xi_{i}}(\nabla v) \partial_{j} v \right) .$$

$$(3.76)$$

In particular, if

$$V(\xi) = \frac{\tilde{H}^p(\xi)}{p} \quad \text{for } p > 1 \text{ and } \xi \in \mathbb{R}^n , \qquad (3.77)$$

where  $\tilde{H}$  is a norm, then

$$2v^{\gamma}S^{2}(W) = \operatorname{div}\left(v^{\gamma}S_{ij}^{2}(W)V_{\xi_{i}}(\nabla v) + \gamma(p-1)v^{\gamma-1}V(\nabla v)\nabla_{\xi}V(\nabla v)\right)$$
  
$$-\gamma(\gamma-1)p(p-1)v^{\gamma-2}V^{2}(\nabla v)$$
  
$$-\gamma(2p-1)v^{\gamma-1}V(\nabla v)\Delta_{p}^{\tilde{H}}v,$$
(3.78)

where  $\Delta_p^{\tilde{H}}v = \operatorname{div}\left(\tilde{a}(\nabla v)\right)$  and  $\tilde{a}(\cdot)$  is given by (3.65). Observe that, in this particular case,

$$W(x) := \nabla[\tilde{a}(\nabla v(x))].$$

*Proof.* See [29, Lemma 4.1].

The idea is to apply the above lemma to the function v solving (3.64) and integrate the identity above on  $\Sigma$ . Due to the lack of regularity of v, Lemma 3.5 cannot be applied directly but we can still prove its integral counterpart.

**Lemma 3.6.** Let v be given by (3.63), let V be as in (3.77), and W as in (3.74). Then, for any  $\varphi \in C_c^{\infty}(\Sigma)$ , we have

$$\int_{\Sigma} \left( 2v^{\gamma} S^{2}(W) + \gamma(\gamma - 1)p(p - 1)v^{\gamma - 2}V^{2}(\nabla v) + \gamma(2p - 1)v^{\gamma - 1}V(\nabla v)\Delta_{p}^{\tilde{H}}v \right) \varphi \, dx$$
  
$$= -\int_{\Sigma} \partial_{j}\varphi \left( v^{\gamma} S_{ij}^{2}(W)V_{\xi_{i}}(\nabla v) + \gamma(p - 1)v^{\gamma - 1}V(\nabla v)V_{\xi_{j}}(\nabla v) \right) dx \,.$$
(3.79)

*Proof.* We argue by approximation. So, first we extend v as 0 outside  $\Sigma$ , and then for  $\varepsilon > 0$  we define  $v^{\varepsilon} = v * \rho^{\varepsilon}$  and  $V^{\varepsilon} = V * \rho^{\varepsilon}$ , where  $\rho^{\varepsilon}$  is a standard mollifier. Also, we set  $\tilde{a}^{\varepsilon} = \nabla V^{\varepsilon}$  and  $W^{\varepsilon} = (w_{ij}^{\varepsilon})_{i,j=1,\dots,n}$  where  $w_{ij}^{\varepsilon} = \partial_j (\tilde{a}_i^{\varepsilon} (\nabla v^{\varepsilon}))$ .

Since  $V \in C^1(\mathbb{R}^n)$  then  $\tilde{a}_i^{\varepsilon} = \tilde{a}_i * \rho^{\varepsilon}$  for i = 1, ..., n, where  $\tilde{a}$  is given by (3.65). Also, since  $\tilde{a}(\nabla v) \in W^{1,2}_{\text{loc}}(\Sigma)$ , then  $\tilde{a}_i^{\varepsilon}(\nabla v^{\varepsilon}) \to \tilde{a}_i(\nabla v)$  and  $w_{ij}^{\varepsilon} \to w_{ij}$  in  $L^2_{\text{loc}}(\Sigma)$ . Moreover, since

$$\widetilde{H}_0(\nabla \widetilde{H}(\xi)) = 1 \qquad \forall \xi \in \mathbb{R}^n \setminus \{0\},$$

we have that

$$\tilde{H}_0(\tilde{a}(\xi)) = \tilde{H}^{p-1}(\xi) \qquad \forall \xi \in \mathbb{R}^n \setminus \{0\},\$$

which implies that

$$pV(\xi) = \tilde{H}_0^{\frac{p}{p-1}}(\tilde{a}(\xi)) \qquad \forall \xi \in \mathbb{R}^n \setminus \{0\}.$$

Since  $\tilde{H}_0^{\frac{p}{p-1}}$  is locally Lipschitz and  $\tilde{a}(\nabla v) \in W^{1,2}_{\text{loc}}(\Sigma)$  then  $V(\nabla v) \in W^{1,2}_{\text{loc}}(\Sigma)$  and we have that

$$\partial_{x_j}(V^{\varepsilon}(\nabla v^{\varepsilon})) \to \partial_{x_j}(V(\nabla v)) \quad \text{ in } L^2_{\text{loc}}(\Sigma).$$

Now we write (3.76) for the approximating functions  $v^{\varepsilon}$ ,  $V^{\varepsilon}$  and  $W^{\varepsilon}$ , we multiply by  $\varphi \in C_c^{\infty}(\Sigma)$  and integrate over  $\Sigma$ . Since  $\varphi$  has compact support inside  $\Sigma$ , it follows from the divergence theorem that

$$\begin{split} &\int_{\Sigma} \left( 2(v^{\varepsilon})^{\gamma} S^{2}(W^{\varepsilon}) + \gamma(\gamma - 1)(p - 1)(v^{\varepsilon})^{\gamma - 2} V^{\varepsilon}(\nabla v^{\varepsilon}) V_{\xi_{i}}^{\varepsilon}(\nabla v^{\varepsilon}) \partial_{i} v^{\varepsilon} \right) \varphi \, dx \\ &+ \int_{\Sigma} \gamma(v^{\varepsilon})^{\gamma - 1} \left( (p - 1) V^{\varepsilon}(\nabla v^{\varepsilon}) + V_{\xi_{i}}^{\varepsilon}(\nabla v^{\varepsilon}) \partial_{i} v^{\varepsilon} \right) \operatorname{Tr} (W^{\varepsilon}) \varphi \, dx \\ &+ \int_{\Sigma} \gamma(v^{\varepsilon})^{\gamma - 1} \left( (p - 1) V_{\xi_{i}}^{\varepsilon}(\nabla v^{\varepsilon}) V_{\xi_{j}}^{\varepsilon}(\nabla v^{\varepsilon}) \partial_{ij}^{2} v^{\varepsilon} + V_{\xi_{j}\xi_{l}}^{\varepsilon}(\nabla v^{\varepsilon}) \partial_{li}^{2} v^{\varepsilon} V_{\xi_{i}}^{\varepsilon}(\nabla v^{\varepsilon}) \partial_{j} v^{\varepsilon} \right) \varphi \, dx \\ &= - \int_{\Sigma} \partial_{j} \varphi \left( (v^{\varepsilon})^{\gamma} S_{ij}^{2}(W^{\varepsilon}) V_{\xi_{i}}^{\varepsilon}(\nabla v^{\varepsilon}) + \gamma(p - 1)(v^{\varepsilon})^{\gamma - 1} V^{\varepsilon}(\nabla v^{\varepsilon}) V_{\xi_{j}}^{\varepsilon}(\nabla v^{\varepsilon}) \right) \, dx \,. \end{split}$$

$$(3.80)$$

Since  $V_{\xi_i}^{\varepsilon}(\nabla v^{\varepsilon})\partial_{ij}^2 v^{\varepsilon} = \partial_{x_j}(V^{\varepsilon}(\nabla v^{\varepsilon}))$ , recalling (3.78) we conclude easily by letting  $\varepsilon \to 0$ .

Now we extend Lemma 3.6 to cut-off functions defined on a ball centered at the origin. Here, the convexity of  $\Sigma$  plays a crucial role.

**Lemma 3.7.** Let v be given by (3.63), let V be as in (3.77), and W as in (3.74). Consider a non-negative cut-off function  $\eta \in C_c^{\infty}(\mathbb{R}^n)$ . Then

$$\int_{\Sigma} \left( 2v^{\gamma} S^{2}(W) + \gamma(\gamma - 1)p(p - 1)v^{\gamma - 2}V^{2}(\nabla v) + \gamma(2p - 1)v^{\gamma - 1}V(\nabla v)\Delta_{p}^{\tilde{H}}v \right) \eta \, dx$$
  

$$\geq -\int_{\Sigma} \partial_{j}\eta \left( v^{\gamma} S_{ij}^{2}(W)V_{\xi_{i}}(\nabla v) + \gamma(p - 1)v^{\gamma - 1}V(\nabla v)V_{\xi_{j}}(\nabla v) \right) \, dx \,.$$
(3.81)

*Proof.* As in the proof of Proposition 3.2, this proof requires a regularization argument considering the solutions of the approximating problems

$$\begin{cases} \operatorname{div}\left(\tilde{a}^{\ell}(\nabla v_{k,\ell})\right) = f(v,\nabla v) & \text{ in } \Sigma_k\\ \tilde{a}^{\ell}(\nabla v_{k,\ell}) \cdot \nu = 0 & \text{ on } \partial \Sigma_k \end{cases}$$

where  $\tilde{a}^{\ell}$  are defined as in (3.51) with *a* replaced by  $\tilde{a}$  given by (3.65) and  $f(v, \nabla v)$  is given by (3.66). Note that, since  $v \in C_{\text{loc}}^{1,\theta}(\Sigma \setminus \{\mathcal{O}\})$ , the functions  $v_{k,\ell}$  are of class  $C_{\text{loc}}^{2,\theta}$  in  $\overline{\Sigma}_k \setminus \{\mathcal{O}\}$ , and this allows one to perform all the desired computations on the functions  $v_{k,\ell}$ , and then let  $\ell$  and k to infinity. Since this approximation argument is very similar to the one in the proof of Proposition 3.2, to simplify the notation and emphasize the main ideas we shall work directly with v, assuming that v is of class  $C_{\text{loc}}^{2,\theta}$  in  $\overline{\Sigma} \setminus \{\mathcal{O}\}$  in order to justify all the computations.

 $\mathbf{Set}$ 

$$F = 2v^{\gamma}S^{2}(W) + \gamma(\gamma - 1)p(p - 1)v^{\gamma - 2}V^{2}(\nabla v)$$
$$+ \gamma(2p - 1)v^{\gamma - 1}V(\nabla v)\Delta_{p}^{\tilde{H}}v \quad (3.82)$$

and  $L = (L_1, \ldots, L_n)$  with

$$L_j = v^{\gamma} S_{ij}^2(W) V_{\xi_i}(\nabla v) + \gamma (p-1) v^{\gamma-1} V(\nabla v) V_{\xi_j}(\nabla v)$$

for j = 1, ..., n. Then we apply Lemma 3.6 with  $\varphi = \eta \zeta_{\delta}$ , where  $\eta \in C_c^{\infty}(\mathbb{R}^n)$ is a cut-off function as in the statement, and  $\zeta_{\delta} \in C_c^{\infty}(\Sigma)$  is a cut-off function of the distance from  $\partial \Sigma$  that converges to 1 inside  $\Sigma$  as  $\delta \to 0$ . In this way, as in the proof of (3.56), letting  $\delta \to 0$  the term involving  $\nabla \zeta_{\delta}$  gives rise to a boundary term: more precisely, we obtain

$$\int_{\Sigma} F\eta \, dx = -\int_{\Sigma} \nabla\eta \cdot L \, dx + \int_{\partial\Sigma} \eta L \cdot \nu d\sigma \,. \tag{3.83}$$

Now, to conclude the proof, we need to show that the last integral in (3.83) is non-negative; indeed, for  $x \in \partial \Sigma \setminus \{\mathcal{O}\}$ , by using the explicit expression of Land of  $S_{ij}^2(W)$  we get

$$L(x) \cdot \nu(x) = v^{\gamma}(x)\tilde{a}(\nabla v(x)) \cdot \nu(x) \left[ \operatorname{Tr}(W)(x) + \gamma(p-1)v^{-1}(x)V(\nabla v(x)) \right]$$
  
-  $v^{\gamma}(x)\partial_{i}(\tilde{a}_{j}(\nabla v(x)))\tilde{a}_{i}(\nabla v(x))\nu_{\ell}(x),$  (3.84)

where we used that  $w_{ji}(x) = \partial_i \tilde{a}_j(\nabla v(x))$  and  $V_{\xi_i} = \tilde{a}_i$ .

We notice now that  $\partial_i \nu_\ell(x)$  is the second fundamental form of  $\partial \Sigma$  at x, which is non-negative definite by the convexity of  $\Sigma$ . Hence

$$\partial_i \nu_\ell(x) \tilde{a}_j(\nabla v(x)) \tilde{a}_i(\nabla v(x)) \ge 0. \tag{3.85}$$

From (3.84) and (3.85) we get

$$L(x) \cdot \nu(x) \ge v^{\gamma}(x)\tilde{a}(\nabla v(x)) \cdot \nu(y) \left[ \operatorname{Tr} (W)(x) + \gamma(p-1)v^{-1}(x)V(\nabla v(x)) \right] - v^{\gamma}(x)\nabla(\tilde{a}(\nabla v(x)) \cdot \nu(y)) \cdot \tilde{a}(\nabla v(x)).$$

Now, since  $\tilde{a}(\nabla v) \cdot \nu = 0$  on  $\partial \Sigma$ , the first term on the right-hand side vanishes. Moreover, since the tangential derivative of  $\tilde{a}(\nabla v) \cdot \nu$  vanishes on  $\partial \Sigma$  and  $\tilde{a}(\nabla v)$ is a tangential vector-field, also the second term vanishes. This proves that  $L \cdot \nu \geq 0$  on  $\partial \Sigma \setminus \{\mathcal{O}\}$ , that together with (3.83) (recall that  $\eta \geq 0$ ) concludes the proof.

**Proposition 3.3.** Let v be given by (3.63), let V be as in (3.77), and W as in (3.74). Then

$$2\int_{\Sigma} v^{\gamma} S^{2}(W) dx + \gamma(\gamma - 1)p(p - 1) \int_{\Sigma} v^{\gamma - 2} V^{2}(\nabla v) dx + \gamma(2p - 1) \int_{\Sigma} v^{\gamma - 1} V(\nabla v) \Delta_{p}^{\tilde{H}} v dx \ge 0, \quad (3.86)$$

for any  $\gamma < -\frac{n(p-1)}{p}$ .

*Proof.* From (3.64), (3.68), and (3.69) we know that  $|\Delta_p^{\tilde{H}}v| \leq C$  in  $\Sigma$ , and from Newton's inequality (3.73) we also have  $|S^2(W)| \leq C$  (recall that  $\operatorname{Tr}(W) = \Delta_p^{\tilde{H}}v$ ).

Now, let  $\eta$  be a non-negative radial cut-off function such that  $\eta = 1$  in  $B_R$ ,  $\eta = 0$  outside  $B_{2R}$ , and  $|\nabla \eta| \leq \frac{2}{R}$ . Thanks to (3.68) and (3.69), we can take the limit as  $R \to \infty$  in the left-hand side of (3.81) to obtain the left-hand side of (3.86). Hence, in order to prove (3.86) it is enough to show that

$$\lim_{R \to \infty} \int_{E_R} \partial_j \eta \left( v^{\gamma} S_{ij}^2(W) V_{\xi_i}(\nabla v) + \gamma (p-1) v^{\gamma-1} V(\nabla v) V_{\xi_j}(\nabla v) \right) dx = 0, \quad (3.87)$$

where we set for simplicity

$$E_R := \Sigma \cap (B_{2R} \setminus B_R).$$

Since  $|S_{ij}^2(W)| \leq |W|$ , using Holder's inequality we get

$$\begin{split} \left| \int_{E_R} \partial_j \eta v^{\gamma} S_{ij}^2(W) V_{\xi_i}(\nabla v) \, dx \right| &\leq \frac{c(n)}{R} \|W\|_{L^2(E_R)} \times \\ & \left( \int_{E_R} v^{2\gamma} |\nabla V(\nabla v)|^2 \, dx \right)^{\frac{1}{2}} \, . \end{split}$$

Observe that (3.70) yields

$$||W||^2_{L^2(E_R)} \le CR^n$$
.

Also, from (3.68) and (3.69) we have

$$\int_{E_R} v^{2\gamma} |\nabla V(\nabla v)|^2 \, dx \le C R^{\frac{2\gamma p}{p-1} + n + 2}.$$

Hence, since by assumption  $\gamma < -\frac{n(p-1)}{p}$ , this proves that

$$\lim_{R \to \infty} \int_{E_R} \partial_j \eta v^{\gamma} S_{ij}^2(W) V_{\xi_i}(\nabla v) \, dx = 0 \, .$$

Analogously, using (3.68) and (3.69), the second term in (3.87) can be bounded as

$$\left| \int_{E_R} \partial_j \eta v^{\gamma-1} V(\nabla v) V_{\xi_j}(\nabla v) \, dx \right| \le C R^{\frac{p\gamma}{p-1}+n} \,, \tag{3.88}$$

which also goes to zero as  $R \to \infty$  since  $\gamma < -\frac{n(p-1)}{p}$ . This proves (3.87) and hence (3.86).

### 3.2.2 Conclusion

We multiply (3.64) by  $v^{-n}$  and integrate over  $\Sigma$ . By using the divergence theorem, the boundary condition in (3.64), and the decay estimates (3.68) and (3.69), we get

$$\left(\frac{p}{n-p}\right)^{p-1} \int_{\Sigma} v^{-n-1} \, dx - \frac{n}{p} \int_{\Sigma} v^{-n-1} \tilde{H}^p(\nabla v) \, dx = 0 \,. \tag{3.89}$$

Now we use Newton's inequality applied to W in (3.86). More precisely, since  $\operatorname{Tr}(W) = \Delta_p^{\tilde{H}} v$ , we have

$$2S^{2}(W) \le \frac{n-1}{n} (\Delta_{p}^{\tilde{H}} v)^{2}, \qquad (3.90)$$

and from (3.86) we obtain

$$\frac{n-1}{n} \int_{\Sigma} v^{\gamma} (\Delta_p^{\tilde{H}} v)^2 \, dx + \gamma(\gamma-1) p(p-1) \int_{\Sigma} v^{\gamma-2} V^2(\nabla v) \, dx + \gamma(2p-1) \int_{\Sigma} v^{\gamma-1} V(\nabla v) \Delta_p^{\tilde{H}} v \, dx \ge 0 \,, \quad (3.91)$$

for any  $\gamma < -\frac{n(p-1)}{p}$ . Since p < n we can choose  $\gamma = 1 - n$  in (3.91), and using (3.64), (3.66), and (3.77), we obtain

$$\left(\frac{p}{n-p}\right)^{p-1} \int_{\Sigma} v^{-n-1} dx - \frac{n}{p} \int_{\Sigma} v^{-n-1} \tilde{H}^p(\nabla v) dx \ge 0.$$
(3.92)

Recalling (3.89), this implies that the equality case must hold in (3.92). Hence the equality case must hold in (3.90) a.e., which implies that

$$W(x) = \lambda(x) \operatorname{Id}$$
 for a.e.  $x \in \Sigma$ , (3.93)

for some function  $\lambda : \Sigma \to \mathbb{R}$ , where Id is the identity matrix.

Now we show that the function  $\lambda$  is constant. Since

$$\lambda(x) = \frac{1}{n} \operatorname{Tr} \left( W \right) = \frac{1}{n} \Delta_p^{\tilde{H}} v(x) = \frac{1}{n} f(v, \nabla v)$$

(see (3.64)), and since  $v \in C_{\text{loc}}^{1,\theta}(\Sigma)$ , we get that  $\lambda \in C_{\text{loc}}^{0,\theta}(\Sigma)$ . Moreover, elliptic regularity theory yields that  $v \in C_{\text{loc}}^{2,\theta}(\Sigma \cap \{\nabla v \neq 0\})$ , which implies that  $\lambda \in C_{\text{loc}}^{1,\theta}(\Sigma \cap \{\nabla v \neq 0\})$ . From (3.93) we have that

$$\partial_i(\tilde{a}_j(\nabla v(x))) = \lambda(x)\delta_{ij} \tag{3.94}$$

for  $i, j \in \{1, \ldots, n\}$ , which implies that  $\tilde{a}(\nabla v) \in C^{2,\theta}_{\text{loc}}(\Sigma \cap \{\nabla v \neq 0\})$ . Then, given  $i \in \{1, \ldots, n\}$ , choosing  $j \neq i$  and using (3.94) we obtain

$$\partial_i \lambda(x) = \partial_i \left( \partial_j (\tilde{a}_j(\nabla v(x))) \right) = \partial_j \left( \partial_i (\tilde{a}_j(\nabla v(x))) \right) = 0$$

for any  $x \in \Sigma \cap \{\nabla v \neq 0\}$ , which implies that  $\lambda$  is constant on each connected component of  $\Sigma \cap \{\nabla v \neq 0\}$ . Since  $\lambda$  is continuous in  $\Sigma$  and  $\{\nabla v = 0\}$  has no interior points (this follows easily from (3.64)), we deduce that  $\lambda$  is constant. In particular, recalling (3.93), we get

$$\nabla[\tilde{a}(\nabla v(x))] = W(x) = \lambda \operatorname{Id}$$
 in  $\Sigma$ .

Hence  $\tilde{a}(\nabla v(x)) = \lambda(x-x_0)$  for some  $x_0 \in \overline{\Sigma}$ , and from the boundary condition in (3.64) we obtain that  $x_0 \in \partial \Sigma$ . This implies that  $v(x) = c_1 + c_2 \tilde{H}_0(x-x_0)^{\frac{p}{p-1}}$ , or equivalently (recalling (3.63))  $u(x) = \mathcal{U}_{\mu,x_0}^H(x)$  for some  $\mu > 0$ . Finally, it is clear that:

- if  $\Sigma = \mathbb{R}^n$  and  $x_0$  may be a generic point in  $\mathbb{R}^n$ ;

- if  $k \in \{1, \ldots, n-1\}$  then  $x_0 \in \mathbb{R}^k \times \{\mathcal{O}\};$ 

- if k = 0 then  $x_0 = \mathcal{O}$ .

This completes the proof of Theorem 3.1.

# Chapter 4

# Sobolev-type inequalities on Cartan-Hadamard manifolds

## 4.1 Statements of the main results

As already mentioned in the Introduction, in this Chapter we consider Sobolevtype inequalities on *Cartan-Hadamard* manifold M of dimension  $n \ge 3$ , namely on complete, noncompact, simply connected Riemannian manifold with everywhere nonpositive sectional curvatures. The starting remark which motivates our study is the following: from one hand it is well known that on any *Cartan-Hadamard* manifold M the Euclidean Sobolev inequality

$$\|f\|_{L^{2^*}(M)} \le C_S \|\nabla f\|_{L^2(M)} \quad \forall f \in C_c^1(M), \qquad 2^* := \frac{2n}{n-2}$$
(4.1)

holds. On the other hand, if the sectional curvatures are everywhere bounded from above by a negative constant -k, then in addition to (4.1) also the *Poincaré* inequality

$$\|f\|_{L^{2}(M)} \leq \frac{2}{\sqrt{k}(n-1)} \|\nabla f\|_{L^{2}(M)} \qquad \forall f \in C^{1}_{c}(M)$$
(4.2)

holds, or equivalently the infimum of the spectrum of (minus) the Laplace-Beltrami operator on M is bounded from below by  $k(n-1)^2/4$ , i.e.  $\Delta$  has a spectral gap. This is a celebrated result due to H.P. McKean [167], which we discuss extensively in Section 4.4 (Theorem 4.4). Note that the spectral bound is optimal since it is attained by the hyperbolic space  $\mathbb{H}^n$  of curvature -k.

As already mentioned in the Introduction, in this Chapter, we suppose that the sectional curvatures of the Cartan-Hadamard manifold satisfy a bound of the following type

$$\operatorname{Sect}(x) \le -K r^{-\beta} \qquad \forall x \in M \setminus B_{R_0}, \qquad (4.3)$$

for some  $\beta \in (0,2]$  and  $K, R_0 > 0$ , where r = r(x) := dist(x, o) denotes the geodesic distance from x to a fixed point o (the *pole* of the manifold). Then the question is:

#### what kind of inequalities does M support?

The answers we show in the Chapter are nontrivial. First of all, one has to distinguish between the so-called *sub-hyperbolic* range ( $\beta \in (0, 2)$ ) and the *quasi-Euclidean* range ( $\beta = 2$ ), following a terminology originally introduced in [120]. Another crucial difference is between *radial* and *nonradial* functions. We shall focus on the following *Sobolev-type inequalities* 

$$\|f\|_{L^{p}(M)} \le C_{p} \|\nabla f\|_{L^{2}(M)}, \qquad (4.4)$$

where p is a suitable exponent belonging to the interval  $(2, 2^*]$ .

Let us briefly describe the results we prove, which are stated precisely later. In the case  $\beta \in (0,2)$  we show that (4.4) holds in the *radial* setting for all  $p \in (2,2^*]$ , for a positive constant  $C_p$  of the form

$$C_{p} \equiv \frac{C p^{\frac{2+\beta}{2(2-\beta)}}}{(p-2)^{\frac{\beta}{2-\beta}}},$$
(4.5)

C being another positive constant that depends only on the constants  $n, \beta, K, R_0$ appearing in (4.3). The result is optimal with respect to the dependence on p(see Theorem 4.1). In the case  $\beta = 2$ , namely negative curvatures that can decay quadratically at infinity, inequalities (4.4) (still in the radial setting) start to hold from a certain exponent  $\tilde{2} \in (2, 2^*)$  that depends on n and K, which tends to  $2^*$  as  $K \to 0$  and to 2 as  $K \to \infty$ . Hence, one is no more allowed to let  $p \downarrow 2$ . This result is also optimal with respect to p, see Theorem 4.2 for the details. Finally, we prove in Theorem 4.3 that out of the radial setting all of the above results *fail*: namely, it is enough to assume that (4.3) is satisfied with reverse inequality (in fact it suffices to require the same bound on the *Ricci* curvature) to be able to construct a sequence of nonradial functions that make the constant  $C_p$  in (4.4) blow up for every  $p < 2^*$ . We point out that the case  $\beta > 2$  is not interesting because it is essentially Euclidean, i.e. the sole inequality of the type of (4.4) that holds, even if restricted to radial functions, is the standard Sobolev one: this is an immediate consequence of our results, see Remark 4.2.

The techniques of proof that we exploit take advantage of two main ingredients: *one-dimensional weighted* functional inequalities and *Laplacian-comparison* theorems, which are recalled in Subsections 4.2.4 and 4.2.2, respectively. The idea is to first study the radial inequalities on *model manifolds*, namely sphericallysymmetric Riemannian manifolds whose metric g can be written as

$$g \equiv dr^2 + \psi(r)^2 \, d\theta_{\mathbb{S}^{n-1}}^2$$

for some regular "model" function  $\psi : \mathbb{R}^+ \to \mathbb{R}^+$  (see Definition 4.7), where  $d\theta_{\mathbb{S}^{n-1}}^2$  is the usual metric on the (n-1)-dimensional unit sphere. In this context, (4.4) becomes a family of one-dimensional weighted inequalities, where the associated weight is just  $\psi(r)^{n-1}$ . Then, by resorting again to Laplacian comparison, as well as to surface-measure comparison (see also Subsection 4.2.1), one can extend the results to general Cartan-Hadamard manifolds.

As we mentioned in the Introduction an important motivation to study the validity of (4.4) on Cartan-Hadamard manifolds under curvature bounds like (4.3) came from the recent work [120], where the authors investigate the asymptotic behavior of nonnegative solutions to the *porous medium equation* (see the

monograph [223])

$$\begin{cases} u_t = \Delta(u^m) & \text{in } M \times \mathbb{R}^+, \\ u = u_0 & \text{on } M \times \{0\}, \end{cases}$$
(4.6)

where m > 1. They prove (in particular) that if  $\text{Sect}(x) \sim r^{-\beta}$  for some  $\beta \in (0, 2)$  then solutions to (4.6) starting from nontrivial compactly-supported data satisfy

$$u(x,t) \sim t^{-\frac{1}{m-1}} \left[ \gamma \left( \log t \right)^{\frac{2+\beta}{2-\beta}} - r^{\frac{2+\beta}{2}} \right]_+^{\frac{1}{m-1}} \qquad \text{as } t \to \infty \,,$$

for a suitable positive constant  $\gamma$ . Such bounds are compatible with the  $L^{1}-L^{\infty}$ smoothing effects proved in [118] under the sole assumption that inequalities (4.4) are satisfied with a constant  $C_p$  as in (4.5). It was by combining these results that we conjectured that, at least in the radial framework, the above inequalities might hold indeed. Completely analogous connections can be established in the quasi-Euclidean case. For more details on this discussion, we refer to Section 4.6.

Concerning the organization, the Chapter is organized as follows: in Section 4.1 we state our main results, after a brief introduction to notations. Section 4.2 recalls some preliminary tools in Riemannian geometry and functional inequalities, which are key concepts in order to carry out our methods of proof. In Section 4.3 we provide the proofs of the radial inequalities (Theorems 4.1 and 4.2), while in Section 4.4 we focus on the Poincaré case p = 2 and give an alternative proof of McKean's Theorem, under somewhat weaker assumptions (see Theorems 4.5 and 4.6). Section 4.5 deals with the nonradial case, namely the disproof of the analogues of Theorems 4.1 and 4.2 for nonradial functions (see Theorem 4.3 and Remark 4.2). The last section shows how the functional inequalities established here yield optimal smoothing estimates for the (radial) porous medium equation flow on the manifold at hand (Theorems 4.7 and 4.8).

### 4.1.1 Basic notations

Given an *n*-dimensional  $(n \ge 2)$  Riemannian manifold (M, g) and  $x \in M$ , we shall denote by Sect(x) the sectional curvature w.r.t. any 2-dimensional tangent subspace at  $x \in M$  and by Ric(x) the Ricci curvature at  $x \in M$ , as a quadratic form on  $T_x M$ , the latter being the tangent space at x.

We shall denote by  $C^1_{c:rad}(M)$  the space of all  $C^1$  functions on M, with compact support, which are radial with respect to some (fixed) point  $o \in M$ , i.e.

$$C^1_{c:\mathrm{rad}}(M) := \left\{ f \in C^1_c(M): \ f(x) \equiv f(\mathrm{d}(x,o)) \quad \forall x \in M \right\},$$

where r = r(x) := d(x, o) is the geodesic distance from o, which is called *pole*, to x. Furthermore, for all  $1 \le p < \infty$ 

$$||f||_{L^p(M)} := \left(\int_M |f|^p\right)^{\frac{1}{p}},$$

where the integral is computed with respect to  $d\nu$ : the Riemannian measure of M; while  $||f||_{L^{\infty}(M)}$  stands for the essential supremum of |f|. The Sobolev critical exponent is, as usual,

$$2^* := \begin{cases} \frac{2n}{n-2} & \text{if } n \ge 3, \\ \infty & \text{if } n = 2. \end{cases}$$

In many parts of the Chapter we shall need to deal with spherically symmetric, complete and noncompact manifolds M for which the Riemannian metric has the following special structure:

$$g \equiv dr^2 + \psi(r)^2 \, d\theta_{\mathbb{S}^{n-1}}^2 \, ,$$

where  $d\theta_{\mathbb{S}^{n-1}}^2$  is the standard metric on  $\mathbb{S}^{n-1}$  and  $\psi : \mathbb{R}^+ \to \mathbb{R}^+$  is a function belonging to the class  $\mathcal{A}$ , which is defined as

$$\mathcal{A} := \left\{ \psi \in C^{\infty}((0,\infty)) \cap C^{1}([0,\infty)) : \psi(0) = 0, \ \psi(r) > 0 \ \forall r > 0, \ \psi'(0) = 1 \right\}$$

$$(4.7)$$

Such Riemannian manifolds are referred to as model manifolds, and will be denoted by  $\mathbb{M}_{\psi}^{n}$ . For example, the Euclidean space corresponds to  $\psi(r) = r$ , while the hyperbolic space corresponds to  $\psi(r) = \sinh r$ .

By an *n*-dimensional Cartan-Hadamard manifold  $M \equiv \mathbb{M}^n$  we mean a complete, noncompact, simply connected Riemannian manifold with everywhere nonpositive sectional curvatures. First of all, we observe that on Cartan-Hadamard manifolds the cut locus of any point o is empty; hence, for every  $x \in M \setminus \{o\}$ one can define polar coordinates with pole at o, namely r(x) = d(x, o) and  $\theta \in \mathbb{S}^{N-1}$ . We then denote by  $B_r$  the Riemannian ball of radius r centered at oand  $S_r := \partial B_r$ . In the case of Cartan-Hadamard manifolds, we also denote by  $\operatorname{Sect}_{\omega}(x)$  the sectional curvature w.r.t. to any 2-dimensional tangent subspace  $\omega$  at x containing a radial direction, and by  $\operatorname{Ric}_o(x)$  the Ricci curvature at xevaluated in the radial direction corresponding to the pole o.

### 4.1.2 The sub-hyperbolic radial case

In the first result of this paper, we prove that if the (radial) sectional curvatures of a Cartan-Hadamard manifold do not decay too fast at infinity, namely slower than an inverse-quadratic rate with respect to r, then a suitable family of Sobolev-type inequalities holds in the *radial* framework. The terminology *sub-hyperbolic* is borrowed from [120].

**Theorem 4.1.** Let  $\mathbb{M}^n$  be a Cartan-Hadamard manifold such that

$$\operatorname{Sect}_{\omega}(x) \le -C_0 r^{-\beta} \qquad \forall x \in \mathbb{M}^n \setminus B_{R_0},$$

$$(4.8)$$

for some  $\beta \in (0,2)$  and  $C_0, R_0 > 0$ . Then there exists a positive constant C, depending only on  $n, \beta, C_0, R_0$ , such that for every  $p \in (2, 2^*] \cap (2, \infty)$  the following radial Sobolev-type inequalities

$$\|f\|_{L^{p}(\mathbb{M}^{n})} \leq \frac{C p^{\frac{2+\beta}{2(2-\beta)}}}{(p-2)^{\frac{\beta}{2-\beta}}} \|\nabla f\|_{L^{2}(\mathbb{M}^{n})} \qquad \forall f \in C^{1}_{c:\mathrm{rad}}(\mathbb{M}^{n})$$
(4.9)

hold. Moreover, the dependence on p of the multiplying constant in (4.9) is optimal, in the sense that for each  $\beta \in (0,2)$  there exists a model manifold  $\mathbb{M}^n_{q_0}$ , complying with (4.8), such that

$$\inf_{f \in C^{1}_{c:\mathrm{rad}}(\mathbb{M}^{n}_{\psi}), \, f \neq 0} \frac{\|\nabla f\|_{L^{2}(\mathbb{M}^{n}_{\psi})}}{\|f\|_{L^{p}(\mathbb{M}^{n}_{\psi})}} \leq \frac{(p-2)^{\frac{p}{2-\beta}}}{C p^{\frac{2+\beta}{2(2-\beta)}}} \qquad \forall p \in (2,2^{*}] \cap (2,\infty) \quad (4.10)$$

for another positive constant C depending on  $n, \beta, C_0, R_0$ .

### 4.1.3 The quasi-Euclidean radial case

In the case of curvatures that decay with a rate which is *at most* quadratic, we still have radial Sobolev-type inequalities: however, in this case, they start to hold from a certain exponent which is *strictly* larger than 2. Again, the terminology *quasi-Euclidean* is in agreement with [120].

**Theorem 4.2.** Let  $\mathbb{M}^n$  be a Cartan-Hadamard manifold such that

$$\operatorname{Sect}_{\omega}(x) \leq -C_1 r^{-2} \qquad \forall x \in \mathbb{M}^n \setminus B_{R_0}$$

$$(4.11)$$

for some  $C_1$  and  $R_0 > 0$ . Then there exists a positive constant C, depending only on  $n, C_1, R_0$ , such that for every  $p \in [\tilde{2}, 2^*] \cap [\tilde{2}, \infty)$  the following radial Sobolev-type inequalities

$$\|f\|_{L^p(\mathbb{M}^n)} \le C\sqrt{p} \,\|\nabla f\|_{L^2(\mathbb{M}^n)} \qquad \forall f \in C^1_{c:\mathrm{rad}}(\mathbb{M}^n) \tag{4.12}$$

hold, where

$$\tilde{2} := \frac{2\tilde{n}}{\tilde{n}-2}, \qquad \tilde{N} := \frac{n+1+\sqrt{1+4C_1}(n-1)}{2}.$$
(4.13)

Moreover, the result is optimal w.r.t. to the exponent p, in the sense that for each  $C_1 > 0$  there exists a model manifold  $\mathbb{M}^n_{\psi}$ , complying with (4.11), such that (4.12) fails for all  $p \in [2, \tilde{2})$  and, in the case n = 2, the optimal constant in (4.12) does behave like  $\sqrt{p}$  (up to multiplicative constants) as  $p \to \infty$ .

Remark 4.1 (Ricci bounds from above). It is worth pointing out that actually the thesis of Theorem 4.1 still holds if one replaces  $\operatorname{Sect}_{\omega}(x)$  with  $\operatorname{Ric}_o(x)$  in assumption (4.8): this is due to the fact that Laplacian-comparison results with model manifolds, which we exploit extensively throughout the paper, can also be established under such a weaker hypothesis. The argument applies to Theorem 4.2 as well, except that in this case the analogue of exponent  $\tilde{2}$  in (4.13) is no more optimal, just because one has (4.24) in place of (4.20). Same observations can be made with regards to Theorems 4.5 (no optimal constant however) and 4.6, both dealing with the Poincaré inequality. The key result on which these extensions rely can be found in the monograph [228]: see Subsection 4.2.2.

### 4.1.4 Failure of the nonradial inequalities

Surprisingly enough, all of the above inequalities (except the *Euclidean* Sobolev inequality) *fail* in the nonradial framework under reverse curvature bounds. Indeed, we have the following.

**Theorem 4.3.** Let  $\mathbb{M}^n$  be a Cartan-Hadamard manifold such that

$$\operatorname{Ric}(x) \ge -C_2 r^{-\beta} \qquad \forall x \in \mathbb{M}^n \setminus B_{R_0}, \qquad (4.14)$$

for some  $\beta \in (0,2)$  and  $R_0, C_2 > 0$ . Let  $p \in [2,2^*)$ . Then there exists no positive constant C for which the following Sobolev-type inequality

$$\|f\|_{L^p(\mathbb{M}^n)} \le C \,\|\nabla f\|_{L^2(\mathbb{M}^n)} \qquad \forall f \in C_c^1(\mathbb{M}^n) \tag{4.15}$$

holds.

Remark 4.2 (Optimality of the curvature bounds). It is plain that Theorem 4.3 implies that the conclusions of Theorems 4.1 and 4.2 cannot hold, in general, in the nonradial framework: indeed, for each given  $\beta \in (0, 2]$ , it is enough to consider a Cartan-Hadamard manifold (e.g. a model) satisfying

$$\operatorname{Sect}(x) \sim r^{-\beta}$$
 as  $r \to \infty$ 

Any such a manifold clearly complies with both the assumptions of Theorem 4.1 (or Theorem 4.2 if  $\beta = 2$ ) and the ones of Theorem 4.3, so that the thesis of the latter prevents the validity of (4.9) (or (4.12)) extended to *nonradial* functions.

For analogous reasons, we do not treat the case  $\beta > 2$ : by following a strategy similar to proof of optimality of Theorem 4.2, it is not difficult to check that for any  $\beta > 2$  one can construct a model manifold such that  $\text{Sect}(x) \sim r^{-\beta}$  as  $r \to \infty$ , for which any of the radial inequalities (4.9) fails as long as  $p < 2^*$  (we omit details and refer to [120, Section 2.3, Type IV]). Hence, in general no inequality of the type of (4.9) different from the classical Sobolev one can hold.

Remark 4.3 (Validity of the inequalities for more general functions). For simplicity, we have stated the above results of Theorems 4.1–4.2 (as well as those of Theorems 4.4, 4.5 and 4.6 below) for functions in  $C_c^1(\mathbb{M}^n)$ . Nevertheless, by means of standard density arguments, it is apparent that they also hold for compactly supported *Lipschitz* functions or, more in general, for all functions belonging to the closure of  $C_c^1(\mathbb{M}^n)$  w.r.t. the  $L^2$  norm of the gradient.

# 4.2 Geometric and functional preliminaries

In the following, we recall some basic facts in Riemannian geometry concerning volume, surface and Laplacian comparison (Subsections 4.2.1–4.2.3), along with a key result related to weighted one-dimensional Sobolev-type inequalities (Subsection 4.2.4).

### 4.2.1 Notations from Riemannian geometry

We shall adopt the same notations as in Section 4.1.1. If  $\mathbb{M}^n$  is an *n*-dimensional Cartan-Hadamard manifold, then one the surface measure of spheres reads

meas 
$$(S_r) = \int_{\mathbb{S}^{n-1}} A(r,\theta) \, d\theta$$
, where  $d\theta := d\theta_1 \dots d\theta_{n-1}$  (4.16)

and  $A(r, \theta)$  is the weight associated with the volume measure of  $\mathbb{M}^n$  w.r.t. polar coordinates, which turns out to be the square root of the determinant of the

metric matrix written in such coordinates (see e.g. [115, Section 3]). In particular, the latter is a nonnegative  $C^{\infty}(\mathbb{R}^+ \times \mathbb{S}^{n-1})$  function. Hence, if f is a radial function as in (4.9) or in (4.12), then by Fubini's Theorem

$$||f||_{L^{p}(M)} = \left(\int_{M} |f|^{p}\right)^{\frac{1}{p}} = \left(\int_{0}^{\infty} \int_{\mathbb{S}^{n-1}} |f(r)|^{p} A(r,\theta) \, d\theta \, dr\right)^{\frac{1}{p}} = \left(\int_{0}^{\infty} |f(r)|^{p} \operatorname{meas}\left(S_{r}\right) dr\right)^{\frac{1}{p}},$$
(4.17)

so that radial Sobolev-type inequalities can be rewritten as one-dimensional weighted inequalities (see Subsection 4.2.4).

It is direct to see that the Laplace-Beltrami operator on  $\mathbb{M}^n$  in polar coordinates has the form

$$\Delta = \frac{\partial^2}{\partial r^2} + \mathsf{m}(r,\theta) \, \frac{\partial}{\partial r} + \Delta_{S_r} \,,$$

where  $\Delta_{S_r}$  is the Laplace-Beltrami operator on the submanifold  $S_r$  and

$$\mathsf{m}(r,\theta) = \frac{\partial}{\partial r} \left( \log A(r,\theta) \right) \qquad \forall x \equiv (r,\theta) \in \mathbb{R}^+ \times \left( \mathbb{S}^{n-1} \setminus \mathcal{P} \right), \tag{4.18}$$

where  $\mathcal{P}$  is the (finite) set of poles on  $\mathbb{S}^{n-1}$ , namely all angles  $\theta \in \mathbb{S}^{n-1}$  at which  $A(r,\theta)$  vanishes identically. Elsewhere,  $A(r,\theta)$  is always strictly positive. Note that  $\mathsf{m}(r,\theta)$  is just the Laplacian of the distance function  $x \equiv (r,\theta) \mapsto r$ . So, by integrating (4.18) from a fixed  $r_0 > 0$  to  $r > r_0$  we deduce that

$$\int_{r_0}^r \mathsf{m}(s,\theta) \, ds = \log A(r,\theta) - \log A(r_0,\theta) \,,$$

i.e.

$$A(r,\theta) = e^{\int_{r_0}^r \mathsf{m}(s,\theta) \, ds + c_\theta}$$

with  $c_{\theta} := \log A(r_0, \theta)$ . As a result, recalling (4.16), we get that

meas 
$$(S_r) = \int_{\mathbb{S}^{n-1}} e^{\int_{r_0}^r \mathfrak{m}(s,\theta) \, ds + c_\theta} \, d\theta$$

### 4.2.2 Laplacian comparison

We recall here some classical results which allow one to compare the Laplacian (as well as the Hessian in some cases) of the distance function of a Cartan-Hadamard manifold with the Laplacian of the distance function of a suitable model manifold corresponding to the curvature bounds (as a reference see e.g. [114, Section 2] or [115, Section 15]). More precisely, if

$$\operatorname{Sect}_{\omega}(x) \leq -\frac{\psi''(r)}{\psi(r)} \qquad \forall x \equiv (r, \theta) \in M \setminus \{o\}$$

$$(4.19)$$

for some function  $\psi \in \mathcal{A}$ , then

$$\mathsf{m}(r,\theta) \ge (n-1)\,\frac{\psi'(r)}{\psi(r)} \qquad \forall (r,\theta) \in \mathbb{R}^+ \times \mathbb{S}^{n-1}\,. \tag{4.20}$$

Similarly, if

$$\operatorname{Ric}_{o}(x) \ge -(n-1) \frac{\psi''(r)}{\psi(r)} \qquad \forall x \equiv (r,\theta) \in M \setminus \{o\}$$

for another function  $\psi \in \mathcal{A}$ , then

$$\mathsf{m}(r,\theta) \le (n-1) \frac{\psi'(r)}{\psi(r)} \qquad \forall (r,\theta) \in \mathbb{R}^+ \times \mathbb{S}^{n-1}.$$

Though we shall mostly use them in the Cartan-Hadamard setting, let us point out that the above inequalities are true in more general Riemannian manifolds (at least manifolds with a pole).

As a simple consequence of Laplacian comparison with the Euclidean space (just use (4.19)–(4.20) with  $\psi(r) = r$ ), on any Cartan-Hadamard manifold there holds

$$\mathsf{m}(r,\theta) \ge \frac{n-1}{r} \qquad \forall (r,\theta) \in \mathbb{R}^+ \times \mathbb{S}^{n-1}.$$
 (4.21)

In particular, thanks to (4.18) and the fact that  $\mathbb{M}^n$  is locally Euclidean, we immediately deduce that

the function  $r \mapsto A(r, \theta)$  is nondecreasing for all  $\theta \in \mathbb{S}^{n-1} \setminus \mathcal{P}$  (4.22)

and

$$\frac{\partial A}{\partial r}(0,\theta) > 0 \qquad \forall \theta \in \mathbb{S}^{n-1} \setminus \mathcal{P}.$$
(4.23)

Actually, in the special case of Cartan-Hadamard manifolds, a comparison result similar to the first one can be deduced by replacing the (radial) sectional curvatures  $\operatorname{Sect}_{\omega}(x)$  with the Ricci curvature evaluated in the radial direction w.r.t. the pole o, which we denote  $\operatorname{Ric}_o(x)$ . Namely, if

$$\operatorname{Ric}_{o}(x) \leq -(n-1)\frac{\psi''(r)}{\psi(r)} \qquad \forall x \equiv (r,\theta) \in \mathbb{M}^{n} \setminus \{o\}$$

for some function  $\psi \in \mathcal{A}$ , then

$$\mathsf{m}(r,\theta) \ge \sqrt{n-1} \, \frac{\psi'(\sqrt{n-1}\,r)}{\psi(\sqrt{n-1}\,r)} \qquad \forall (r,\theta) \in \mathbb{R}^+ \times \mathbb{S}^{n-1} \,. \tag{4.24}$$

This is basically due to the fact that the *Hessian* of the distance function on  $\mathbb{M}^n$  has nonnegative eigenvalues: we refer to [228, Theorem 2.15].

By exploiting Laplacian comparison with carefully chosen model functions  $\psi \in \mathcal{A}$ , one can easily prove the following (for the details see e.g. [120, Lemma 4.1]).

**Lemma 4.1.** Let  $\mathbb{M}^n$  be a Cartan-Hadamard manifold satisfying (4.8) for some  $\beta \in [0,2)$  and  $C_0, R_0 > 0$ . Then there exist  $r_0 = r_0(\beta, C_0, R_0) > 0$  and  $c = c(n, \beta, C_0, R_0) > 0$  such that

$$\mathsf{m}(r,\theta) \ge c r^{-\frac{\beta}{2}} \qquad \forall (r,\theta) \in [r_0,\infty) \times \mathbb{S}^{n-1}.$$
#### 4.2.3 Geometric interpretation of the scalar curvature

Given a generic *n*-dimensional Riemannian manifold (M, g), it is well known (see for instance [28, Introduction] or [49, p. 133]) that the scalar curvature  $S_g$ is linked to the volume of balls: in particular, when  $S_g$  is positive the volume of the balls in M is smaller than the volume of the balls of the same radius in the Euclidean space. On the other hand, when  $S_g$  is negative we have the opposite relation. This facts can be made quantitative, at least for small balls. More precisely, the ratio between the volume of a ball  $B_{\varepsilon}(p) \subset M$  of radius  $\varepsilon > 0$ centered at  $p \in M$  and the Euclidean volume of the corresponding ball  $\mathbb{B}_{\varepsilon} \subset \mathbb{R}^n$ centered at the origin is given by

$$\frac{\operatorname{Vol}(B_{\varepsilon}(p))}{|\mathbb{B}_{\varepsilon}|} = 1 - \frac{S_g}{6(n+2)} \,\varepsilon^2 + O(\varepsilon^4)\,,$$

where  $|\cdot|$  stands for the Euclidean volume. Moreover, the boundaries of these balls, i.e.  $\partial B_{\varepsilon}(p) = S_{\varepsilon}(p)$  and  $\partial \mathbb{B}_{\varepsilon} = \mathbb{S}_{\varepsilon}$ , are (n-1)-dimensional geodesic spheres of radius  $\varepsilon > 0$  whose ratio of the corresponding surface (or Hausdorff) measures satisfies

$$\frac{\operatorname{meas}\left(S_{\varepsilon}(p)\right)}{\operatorname{meas}\left(\mathbb{S}_{\varepsilon}\right)} = 1 - \frac{S_g}{6n}\,\varepsilon^2 + O(\varepsilon^4)\,. \tag{4.25}$$

#### 4.2.4 One-dimensional weighted inequalities

In the following, by a *weight* in  $\mathbb{R}^+$  we simply mean any positive  $L^1_{loc}([0,\infty))$  function, even though in the rest of the paper we shall in fact deal with more regular functions. The techniques we exploit in Sections 4.3 and 4.4 take advantage of some results for *one-dimensional* weighted Sobolev-type inequalities (or Hardy-type inequalities according to the terminology of [149]), which have been known for a long time. In this regard, we shall mainly refer to the monograph [149] by A. Kufner and P. Opic, which collects several results from this perspective (not only in the one-dimensional framework by the way).

**Proposition 4.1** ([149, Theorem 6.2]). Let w be a weight in  $\mathbb{R}^+$ . Let  $p \in [2, \infty)$ . Then the Sobolev-type inequality

$$\left(\int_{0}^{\infty} |g(r)|^{p} w(r) dr\right)^{\frac{1}{p}} \leq C \left(\int_{0}^{\infty} |g'(r)|^{2} w(r) dr\right)^{\frac{1}{2}} \qquad \forall g \in C_{c}^{1}([0,\infty))$$
(4.26)

holds for some C > 0 if and only if

$$\mathcal{B}(w,p) := \sup_{r \in (0,\infty)} \left( \int_0^r w(s) \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{w(s)} \, ds \right)^{\frac{1}{2}} < \infty \,, \tag{4.27}$$

and the optimal constant C appearing in (4.26) satisfies the two-sided bound

$$\mathcal{B}(w,p) \le C \le \left(1 + \frac{p}{2}\right)^{\frac{1}{p}} \left(1 + \frac{2}{p}\right)^{\frac{1}{2}} \mathcal{B}(w,p).$$

$$(4.28)$$

### 4.3 Proofs of the radial results

We devote this section to the proof of Theorems 4.1 and 4.2, so that here we shall focus only on radial functions. Nonradial issues will then be addressed in Sections 4.4 and 4.5.

#### 4.3.1 The sub-hyperbolic case

We start this subsection by a key result showing that, under suitable assumptions on the weight  $w(r) \equiv \psi(r)^{n-1}$ , where  $\psi \in \mathcal{A}$  (recall (4.7)) is any function corresponding to sub-hyperbolic model manifolds (according to the terminology adopted in Subsection 4.1.2), the supremum appearing in the statement of Theorem 4.1 can be bounded from above in a quantitative way.

**Lemma 4.2.** Let  $\psi \in A$  satisfy the following assumptions:

$$\psi(r) \ge r \quad \forall r \ge 0, \qquad \psi'(r) \ge 0 \quad \forall r \ge 0, \qquad \frac{\psi'(r)}{\psi(r)} \ge \frac{c}{r^{\alpha}} \quad \forall r \ge r_0, \quad (4.29)$$

for some  $\alpha \in (0,1)$  and  $c, r_0 > 0$ . Let  $n \in \mathbb{N}$  be larger than or equal to 2. Then there exists a positive constant C, depending only on  $n, \alpha, c, r_0$ , such that for every  $p \in (2, 2^*)$  there holds

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \le \frac{C \, p^{\frac{1+\alpha}{2(1-\alpha)}}}{(p-2)^{\frac{\alpha}{1-\alpha}}} \,. \tag{4.30}$$

*Proof.* First of all, let us establish that the l.h.s. of (4.30) is finite. To this end, set

$$Q(r) := \left(\int_0^r \psi(s)^{n-1} \, ds\right)^{\frac{1}{p}} \left(\int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds\right)^{\frac{1}{2}} \qquad \forall r > 0 \,. \tag{4.31}$$

The integration of the last inequality in (4.29) from  $r_0$  to  $r > r_0$ , along with the first inequality, yields the bound from below

$$\psi(r) \ge k e^{\frac{c}{1-\alpha}r^{1-\alpha}} \quad \forall r \ge r_0, \qquad \text{where } k = k(\alpha, c, r_0) := r_0 e^{-\frac{c}{1-\alpha}r_0^{1-\alpha}},$$
(4.32)

which readily ensures that Q(r) is a smooth function on  $(0,\infty)$ . In addition, because  $\psi(r) \sim r$  as  $r \to 0$  and  $p < 2^*$ , it is immediate to check that  $\lim_{r\to 0} Q(r) = 0$ . In order to deal with the behaviour of Q(r) at infinity, we need some more integral estimates. To this aim, upon rewriting the last inequality in (4.29) as

$$\psi(r)^{n-1} \le \frac{r^{\alpha}}{c(n-1)} \frac{d}{dr} \left(\psi^{n-1}\right)(r) \qquad \forall r \ge r_0$$

and integrating (by parts) between  $r_0$  and  $r > r_0$ , we obtain:

$$\int_{r_0}^r \psi(s)^{n-1} ds \leq \frac{1}{c(n-1)} \int_{r_0}^r s^\alpha \frac{d}{ds} \left(\psi^{n-1}\right)(s) ds$$
  
=  $\frac{1}{c(n-1)} \left[ r^\alpha \psi(r)^{n-1} - r_0^\alpha \psi(r_0)^{n-1} - \alpha \int_{r_0}^r \frac{\psi(s)^{n-1}}{s^{1-\alpha}} ds \right]$   
$$\leq \frac{1}{c(n-1)} r^\alpha \psi(r)^{n-1}.$$
 (4.33)

By plugging estimate (4.33) in (4.31), exploiting (4.32) and the fact that  $\psi(r)$  is nondecreasing, we deduce that

$$\begin{split} Q(r) &= \left(\int_{0}^{r_{0}} \psi(s)^{n-1} \, ds + \int_{r_{0}}^{r} \psi(s)^{n-1} \, ds\right)^{\frac{1}{p}} \left(\int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds\right)^{\frac{1}{2}} \\ &\leq \left(\int_{0}^{r_{0}} \psi(s)^{n-1} \, ds + \frac{1}{c(n-1)} \, r^{\alpha} \, \psi(r)^{n-1}\right)^{\frac{1}{p}} \left(\int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds\right)^{\frac{1}{2}} \\ &= \left(\frac{1}{r^{\alpha} \, \psi(r)^{n-1}} \int_{0}^{r_{0}} \psi(s)^{n-1} \, ds + \frac{1}{c(n-1)}\right)^{\frac{1}{p}} \times \\ &\left[\left(r^{\alpha} \, \psi(r)^{n-1}\right)^{\frac{2}{p}} \int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds\right]^{\frac{1}{2}} \\ &\leq \left(\frac{1}{r^{\alpha} \, \psi(r)^{n-1}} \int_{0}^{r_{0}} \psi(s)^{n-1} \, ds + \frac{1}{c(n-1)}\right)^{\frac{1}{p}} \left(\int_{r}^{\infty} \frac{s^{\frac{2\alpha}{p}}}{\psi(s)^{\frac{(n-1)(p-2)}{p}}} \, ds\right)^{\frac{1}{2}} \\ &\leq \left(\frac{e^{-\frac{c(n-1)}{1-\alpha}}r^{1-\alpha}}{k^{n-1} \, r^{\alpha}} \int_{0}^{r_{0}} \psi(s)^{n-1} \, ds + \frac{1}{c(n-1)}\right)^{\frac{1}{p}} \times \\ &\left(\int_{r}^{\infty} \frac{s^{\frac{2\alpha}{p}} e^{-\frac{c(n-1)(p-2)}{(1-\alpha)p} s^{1-\alpha}}}{k^{\frac{(n-1)(p-2)}{p}}} \, ds\right)^{\frac{1}{2}} \end{split}$$

for all  $r > r_0$ , whence  $\lim_{r\to\infty} Q(r) = 0$ . As a consequence, because Q(r) is smooth and positive in  $(0,\infty)$ , it admits a maximum at some  $\overline{r} > 0$ , which is a critical point. Since

$$\begin{aligned} Q'(r) &= \frac{\psi(r)^{n-1}}{p} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{1}{p}-1} \left( \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \\ &- \frac{1}{2 \, \psi(r)^{n-1}} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}-1}, \end{aligned}$$

at  $r = \overline{r}$  we find the identity

$$\int_{\overline{r}}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds = \frac{p}{2} \, \frac{1}{\psi(\overline{r})^{2n-2}} \, \int_{0}^{\overline{r}} \psi(s)^{n-1} \, ds \,, \tag{4.34}$$

so that

$$Q(\overline{r}) = \left(\frac{p}{2}\right)^{\frac{1}{2}} \frac{1}{\psi(\overline{r})^{n-1}} \left(\int_0^{\overline{r}} \psi(s)^{n-1} \, ds\right)^{\frac{p+2}{2p}}.$$
(4.35)

In particular,

$$\sup_{r \in (0,\infty)} Q(r) \le \left(\frac{p}{2}\right)^{\frac{1}{2}} \sup_{r \in (0,\infty)} \frac{1}{\psi(r)^{n-1}} \left(\int_0^r \psi(s)^{n-1} \, ds\right)^{\frac{p+2}{2p}}.$$
(4.36)

Therefore, in order to establish (4.30), it is enough to prove an analogous upper bound on the r.h.s. of (4.36). To this aim, first of all note that the first two inequalities of (4.29) entail

$$\frac{1}{\psi(r)^{n-1}} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{p+2}{2p}} \le \frac{r^{\frac{p+2}{2p}}}{\psi(r)^{\frac{(n-1)(p-2)}{2p}}} \le r^{\frac{(n-2)(2^*-p)}{2p}} \,,$$

that is, upon taking the supremum over  $(0, r_0)$ ,

$$\sup_{r \in (0,r_0)} \frac{1}{\psi(r)^{n-1}} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{p+2}{2p}} \le r_0^{\frac{(n-2)(2^*-p)}{2p}}, \tag{4.37}$$

where in the case n = 2 we mean  $(n - 2)(2^* - p) = 4$ . On the other hand, by exploiting (4.32), (4.33), (4.37) and the fact that  $\psi$  is nondecreasing, we obtain:

$$\begin{split} \sup_{r \in (r_0,\infty)} \frac{1}{\psi(r)^{n-1}} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{p+2}{2p}} \\ &\leq \sup_{r \in (r_0,\infty)} \left( \frac{1}{\psi(r_0)^{\frac{2p(n-1)}{p+2}}} \int_0^{r_0} \psi(s)^{n-1} \, ds + \frac{1}{\psi(r)^{\frac{2p(n-1)}{p+2}}} \int_{r_0}^r \psi(s)^{n-1} \, ds \right)^{\frac{p+2}{2p}} \\ &\leq \sup_{r \in (r_0,\infty)} \left( r_0^{\frac{(n-2)(2^*-p)}{p+2}} + \frac{r^{\alpha}}{c(n-1)\,\psi(r)^{\frac{(n-1)(p-2)}{p+2}}} \right)^{\frac{p+2}{2p}} \\ &\leq \sup_{r \in (r_0,\infty)} \left( r_0^{\frac{(n-2)(2^*-p)}{p+2}} + \frac{r^{\alpha}}{c(n-1)\,k^{\frac{(n-1)(p-2)}{p+2}} e^{\frac{c(n-1)(p-2)}{(p+2)(1-\alpha)}r^{1-\alpha}}} \right)^{\frac{p+2}{2p}} \\ &\leq \left( r_0^{\frac{(n-2)(2^*-p)}{p+2}} + \frac{[\alpha(p+2)]^{\frac{\alpha}{1-\alpha}}}{[c(n-1)]^{\frac{1-\alpha}{1-\alpha}}} r_0^{-\frac{(n-1)(p-2)}{p+2}} e^{\frac{c(n-1)(p-2)}{(p+2)(1-\alpha)}r_0^{1-\alpha}} - \frac{\alpha}{1-\alpha}} \right)^{\frac{p+2}{2p}} \end{split}$$

$$(4.38)$$

where we have computed explicitly the last supremum in (4.38) (over the whole  $\mathbb{R}^+$  actually) recalling the definition of k given in (4.32). Hence, by combining (4.36), (4.37) and (4.38), we end up with

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \le \left(\frac{p}{2}\right)^{\frac{1}{2}} \times \left( r_0^{\frac{(n-2)(2^*-p)}{p+2}} + \frac{[\alpha(p+2)]^{\frac{\alpha}{1-\alpha}}}{[c(n-1)]^{\frac{1}{1-\alpha}} (p-2)^{\frac{\alpha}{1-\alpha}}} r_0^{-\frac{(n-1)(p-2)}{p+2}} e^{\frac{c(n-1)(p-2)}{(1-\alpha)(p+2)} r_0^{1-\alpha} - \frac{\alpha}{1-\alpha}} \right)^{\frac{p+2}{2p}} + \frac{[\alpha(p+2)]^{\frac{\alpha}{1-\alpha}}}{[c(n-1)]^{\frac{1}{1-\alpha}} (p-2)^{\frac{\alpha}{1-\alpha}}} r_0^{-\frac{(n-1)(p-2)}{p+2}} e^{\frac{c(n-1)(p-2)}{(1-\alpha)(p+2)} r_0^{1-\alpha} - \frac{\alpha}{1-\alpha}} \right)^{\frac{p+2}{2p}}$$

from which (4.30) easily follows just by letting  $p \downarrow 2$  and (in the case n = 2 only)  $p \to \infty$ .

We are now in position to prove Theorem 4.1.

Proof of Theorem 4.1. We consider only the case  $p < 2^*$  since, as recalled in the Introduction, it is well known that the Euclidean Sobolev inequality holds on any Cartan-Hadamard manifold.

Let us first establish the validity of (4.9) and then show optimality according to (4.10). To our purposes, we introduce the following function:

$$\psi_{\star}(r) := \left(\frac{\operatorname{meas}\left(S_{r}\right)}{\omega_{n-1}}\right)^{\frac{1}{n-1}} \qquad \forall r \ge 0, \qquad (4.39)$$

where  $\omega_{n-1}$  is the Hausdorff measure of the Euclidean unit sphere  $\mathbb{S}^{n-1}$  and  $S_r$  is the sphere of radius r in  $\mathbb{M}^n$  centered at the pole o. It is an elementary fact that  $\psi_{\star} \in \mathcal{A}$ . Indeed, recalling (4.16), it is apparent that  $\psi_{\star} \in C^{\infty}((0,\infty))$  thanks to the regularity of  $A(r,\theta)$  outside the pole (recall the corresponding discussion in Subsection 4.2.1). On the other hand, by applying (4.25) with  $\varepsilon = r$ , we easily deduce that  $\psi_{\star} \in C^1([0,\infty)), \psi_{\star}(0) = 0$  and  $\psi'_{\star}(0) = 1$ . Now we aim at showing that  $\psi_{\star}$  fulfills the hypotheses (4.29) of Lemma 4.2 for some positive  $r_0 = r_0(\beta, C_0, R_0), c = c(n, \beta, C_0, R_0)$  and  $\alpha = \beta/2$ . Indeed, thanks to (4.18), (4.21) and Lemma 4.1, the following inequalities hold:

$$\frac{\frac{\partial}{\partial r}A(r,\theta)}{A(r,\theta)} \ge \frac{n-1}{r} \qquad \forall (r,\theta) \in (0,\infty) \times \mathbb{S}^{n-1} \setminus \mathcal{P}$$
(4.40)

 $\operatorname{and}$ 

$$\frac{\frac{\partial}{\partial r}A(r,\theta)}{A(r,\theta)} \ge c r^{-\frac{\beta}{2}} \qquad \forall (r,\theta) \in [r_0,\infty) \times \mathbb{S}^{n-1} \setminus \mathcal{P},$$
(4.41)

for suitable constants  $r_0, c > 0$  as above. By integrating (4.40) from  $\varepsilon > 0$  to  $r > \varepsilon$ , we obtain:

$$A(r,\theta) \ge \frac{A(\varepsilon,\theta)}{\varepsilon^{n-1}} r^{n-1} \qquad \forall (r,\theta) \in (\varepsilon,\infty) \times \mathbb{S}^{n-1} \setminus \mathcal{P}$$

whence, upon integrating over  $\mathbb{S}^{n-1}$ ,

$$\operatorname{meas}\left(S_{r}\right) \geq \omega_{n-1} r^{n-1} \frac{\operatorname{meas}\left(S_{\varepsilon}\right)}{\operatorname{meas}\left(S_{\varepsilon}\right)} = \omega_{n-1} r^{n-1} \left(1 + O(\varepsilon^{2})\right), \qquad (4.42)$$

where we have used (4.16) and again (4.25). If we let  $\varepsilon \downarrow 0$  in (4.42), we therefore end up with

$$\psi_{\star}(r) \ge r \qquad \forall r \ge 0.$$

The fact that  $\psi'_{\star}(r) \geq 0$  everywhere is a trivial consequence of (4.40), so that we are left with establishing the last inequality of (4.29). To this aim, note that the integration of (4.41) over  $\mathbb{S}^{n-1}$  yields

$$\frac{d}{dr} \operatorname{meas} \left( S_r \right) = \int_{\mathbb{S}^{n-1}} \frac{\partial}{\partial r} A(r, \theta) \, d\theta$$
$$\geq c \, r^{-\frac{\beta}{2}} \int_{\mathbb{S}^{n-1}} A(r, \theta) \, d\theta$$
$$= c \, r^{-\frac{\beta}{2}} \operatorname{meas} \left( S_r \right) \qquad \forall r \geq r_0 \,,$$

which readily entails

$$\frac{\psi'_{\star}(r)}{\psi_{\star}(r)} \ge \frac{c}{n-1} r^{-\frac{\beta}{2}} \qquad \forall r \ge r_0 \,,$$

namely the last inequality of (4.29) with  $\alpha = \beta/2$ , upon relabeling c. Hence, we have proved that the function  $\psi_{\star}$  defined in (4.39) satisfies all of the assumptions of Lemma 4.2, with  $\alpha = \beta/2$ . Thus, as a consequence of (4.30), we deduce that

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi_\star(s)^{n-1} \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{\psi_\star(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \le \frac{C \, p^{\frac{2+\beta}{2(2-\beta)}}}{(p-2)^{\frac{\beta}{2-\beta}}} \tag{4.43}$$

for a suitable  $C = C(n, \beta, C_0, R_0) > 0$ . Thanks to (4.43), we can apply Proposition 4.1 with  $w(r) = \psi_{\star}(r)^{n-1}$ , which ensures the validity of the Sobolev-type inequalities

$$\left(\int_{0}^{\infty} |g(r)|^{p} \psi_{\star}(r)^{N-1} dr\right)^{\frac{1}{p}} \leq \left(1 + \frac{p}{2}\right)^{\frac{1}{p}} \left(1 + \frac{2}{p}\right)^{\frac{1}{2}} \frac{C p^{\frac{2+\beta}{2(2-\beta)}}}{(p-2)^{\frac{\beta}{2-\beta}}} \left(\int_{0}^{\infty} |g'(r)|^{2} \psi_{\star}(r)^{N-1} dr\right)^{\frac{1}{2}} \qquad (4.44)$$
$$\forall g \in C_{c}^{1}([0,\infty)), \qquad \forall p \in (2,2^{*}).$$

Finally, we need to show how to pass from (4.44) to (4.9). To this purpose, it is enough to observe that  $f \in C^1_{c:rad}(\mathbb{M}^n)$  implies  $r \mapsto f(r) \in C^1_c([0,\infty))$ , which, along with Fubini's Theorem, allows us to apply (4.44) in the following way:

$$\begin{split} \|f\|_{L^{p}(\mathbb{M}^{n})} &= \left(\int_{0}^{\infty} \int_{\mathbb{S}^{n-1}} |f(r)|^{p} A(r,\theta) \, d\theta \, dr\right)^{\frac{1}{p}} \\ &= \left(\omega_{n-1} \int_{0}^{\infty} |f(r)|^{p} \, \psi_{\star}(r)^{n-1} \, dr\right)^{\frac{1}{p}} \\ &\leq \omega_{n-1}^{\frac{1}{p}} \left(1 + \frac{p}{2}\right)^{\frac{1}{p}} \left(1 + \frac{2}{p}\right)^{\frac{1}{2}} \frac{C p^{\frac{2+\beta}{2(2-\beta)}}}{(p-2)^{\frac{\beta}{2-\beta}}} \left(\int_{0}^{\infty} |f'(r)|^{2} \, \psi_{\star}(r)^{n-1} \, dr\right)^{\frac{1}{2}} \\ &= \omega_{n-1}^{\frac{1}{p}-\frac{1}{2}} \left(1 + \frac{p}{2}\right)^{\frac{1}{p}} \left(1 + \frac{2}{p}\right)^{\frac{1}{2}} \frac{C p^{\frac{2+\beta}{2(2-\beta)}}}{(p-2)^{\frac{\beta}{2-\beta}}} \times \\ &\qquad \left(\int_{0}^{\infty} \int_{\mathbb{S}^{n-1}} |f'(r)|^{2} A(r,\theta) \, d\theta \, dr\right)^{\frac{1}{2}} \\ &= \omega_{n-1}^{\frac{1}{p}-\frac{1}{2}} \left(1 + \frac{p}{2}\right)^{\frac{1}{p}} \left(1 + \frac{2}{p}\right)^{\frac{1}{2}} \frac{C p^{\frac{2+\beta}{2(2-\beta)}}}{(p-2)^{\frac{2-\beta}{2-\beta}}} \, \|\nabla f\|_{L^{2}(\mathbb{M}^{n})} \,, \end{split}$$

$$\tag{4.45}$$

namely (4.9) upon relabelling C.

Let us now deal with optimality. It is enough to consider any function  $\psi \in \mathcal{A}$  such that

$$\psi''(r) \ge 0 \quad \forall r > 0, \qquad \frac{\psi''(r)}{\psi(r)} = C_0 r^{-\beta} \quad \forall r \ge R_0,$$
 (4.46)

which ensures that the associated model manifold  $\mathbb{M}_{\psi}^{M}$  is Cartan-Hadamard and complies with (4.8). Indeed, following e.g. [120, Lemma 4.1], it is not difficult to prove that (4.46) implies

$$\frac{\psi'(r)}{\psi(r)} \sim \sqrt{C_0} r^{-\frac{\beta}{2}} \qquad \text{as } r \to \infty \,, \tag{4.47}$$

where by  $a(r) \sim b(r)$  we mean that the ratio a(r)/b(r) tends to 1. In particular, (4.47) entails

$$\frac{\sqrt{C_0}}{2} r^{-\frac{\beta}{2}} \le \frac{\psi'(r)}{\psi(r)} \le 2\sqrt{C_0} r^{-\frac{\beta}{2}} \quad \forall r \ge r_0 \implies \psi(r) \ge c_1 e^{c_2 r^{\frac{2-\beta}{2}}} \quad \forall r \ge r_0,$$
(4.48)

 $r_0, c_1, c_2 > 0$  being suitable constants that depend on  $\psi$  but not on  $p \in (2, 2^*] \cap (2, \infty)$ , whose exact values are not relevant to our purposes. Hence, (4.48) plus a simple integration by parts in the same spirit as above, yield

$$\int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds \ge \frac{r^{\frac{\beta}{2}}}{2\sqrt{C_0}(n-1)\,\psi(r)^{n-1}} \qquad \forall r \ge r_0 \,. \tag{4.49}$$

Similarly,

$$\int_{r_0}^r \psi(s)^{n-1} ds \ge \frac{1}{2\sqrt{C_0}(n-1)} \left[ r^{\frac{\beta}{2}} \psi(r)^{n-1} - r_0^{\frac{\beta}{2}} \psi(r_0)^{n-1} - \frac{\beta}{2} \int_{r_0}^r \frac{\psi(s)^{n-1}}{s^{\frac{2-\beta}{2}}} ds \right] \qquad \forall r \ge r_0 \,,$$

which implies, upon picking  $r_0$  so large that

$$\frac{\beta}{4\sqrt{C_0}(n-1)\,r_0^{\frac{2-\beta}{2}}} \le 1\,,$$

the validity of the estimate

$$\int_{r_0}^r \psi(s)^{n-1} \, ds \ge \frac{1}{4\sqrt{C_0}(n-1)} \left[ r^{\frac{\beta}{2}} \, \psi(r)^{n-1} - r_0^{\frac{\beta}{2}} \, \psi(r_0)^{n-1} \right] \qquad \forall r \ge r_0 \,.$$

$$(4.50)$$

From (4.50), it is then apparent that one can select another  $\hat{r}_0 > r_0$  in such a way that

$$\int_{r_0}^r \psi(s)^{n-1} \, ds \ge \frac{r^{\frac{\beta}{2}} \, \psi(r)^{n-1}}{8\sqrt{C_0}(n-1)} \qquad \forall r \ge \hat{r}_0 \,. \tag{4.51}$$

Hence, by combining (4.48), (4.49) and (4.51), we deduce that

$$\left(\int_{0}^{r} \psi(s)^{n-1} ds\right)^{\frac{1}{p}} \left(\int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} ds\right)^{\frac{1}{2}} \ge C \frac{r^{\frac{\beta(p+2)}{4p}}}{\psi(r)^{\frac{(n-1)(p-2)}{2p}}} \\ \ge C \frac{r^{\frac{\beta(p+2)}{4p}}}{e^{c_{2} \frac{(n-1)(p-2)}{2p}} r^{\frac{2-\beta}{2}}} \qquad \forall r \ge \hat{r}_{0} ,$$

$$(4.52)$$

where from here on C denotes a general positive constant that can be taken independent of  $p \in (2, 2^*] \cap (2, \infty)$ , that we shall not relabel. A straightforward calculation shows that the maximum over  $r \in (0, \infty)$  of the last term in (4.52) is attained at

$$\overline{r} = \left[\frac{\beta(p+2)}{c_2(n-1)(2-\beta)(p-2)}\right]^{\frac{2}{2-\beta}},$$

which ensures that

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \ge C \left( \frac{p+2}{p-2} \right)^{\frac{\beta}{2-\beta}} \tag{4.53}$$

provided

$$\left[\frac{\beta(p+2)}{c_2(n-1)(2-\beta)(p-2)}\right]^{\frac{2}{2-\beta}} \ge \hat{r}_0.$$
(4.54)

It is therefore clear that (4.53)-(4.54), together with (4.28), yield (4.10) at least in the case  $n \ge 3$ , where  $2^* < \infty$ . What is left in the case n = 2 is just the correct estimate on the behaviour of the supremum in the l.h.s. of (4.53) as  $p \to \infty$ . To this aim, it is enough to observe that, as a simple consequence of the fact that  $\psi(r) \sim r$  as  $r \to 0$ , there holds

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi(s) \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{\psi(s)} \, ds \right)^{\frac{1}{2}} \ge C \sup_{r \in \left(0, \frac{1}{\sqrt{e}}\right)} r^{\frac{2}{p}} \left( -\log r \right)^{\frac{1}{2}} \ge C p^{\frac{1}{2}} \,,$$

which, upon exploiting again (4.28), ensures the validity of (4.10) also as  $p \rightarrow \infty$ .

#### 4.3.2 The quasi-Euclidean case

Similarly to Subsection 4.3.1, we start by a crucial result showing that, for appropriate weights  $w(r) \equiv \psi(r)^{n-1}$  ( $\psi \in \mathcal{A}$ ) associated with quasi-Euclidean model manifolds (still according to the terminology of Subsection 4.1.2), one can bound quantitatively the supremum appearing in the statement of Theorem 4.1.

**Lemma 4.3.** Let  $\psi \in A$  satisfy the following assumptions:

$$\psi(r) \ge r \quad \forall r > 0 \,, \quad \psi'(r) \ge 0 \quad \forall r > 0 \,, \quad \frac{\psi'(r)}{\psi(r)} \ge \frac{c}{r} - \frac{c'}{r^q} \quad \forall r \ge r_0 \,, \quad (4.55)$$

for some c > 1, c' > 0, q > 1 and  $r_0 > 0$ . Let  $n \in \mathbb{N}$  be larger than or equal to 2. Then

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \le C \sqrt{p} \qquad \forall p \in \left[ \frac{2\tilde{n}}{\tilde{n}-2}, 2^* \right),$$
(4.56)

where  $\tilde{n} := \tilde{n}(n,c) := c(n-1) + 1$  and C is a positive constant depending only on  $n, c, c', q, r_0$ .

*Proof.* The strategy relies on arguments close to the ones used in the proof Lemma 4.2. Indeed, let Q(r) be defined by (4.31). By integrating the last differential inequality in (4.55) from  $r_0$  to  $r > r_0$  (and taking advantage of the first one as well), we obtain:

$$\psi(r) \ge k r^c \quad \forall r \ge r_0, \qquad \text{where } k = k(c', q, r_0) > 0, \qquad (4.57)$$

which ensures that Q(r) is a smooth function of r > 0, given the finiteness of the integrals involved. Moreover, because  $\psi(r) \sim r$  as  $r \to 0$  and  $p < 2^*$ , it is immediate to check that  $\lim_{r\to 0} Q(r) = 0$ . In order to deal with the limit at infinity, we need again some integral bounds. Clearly, the last inequality of (4.55) yields

$$\frac{\psi'(r)}{\psi(r)} \ge \frac{c+1}{2r} \quad \forall r \ge \tilde{r}_0, \qquad \text{for some } \tilde{r}_0 = \tilde{r}_0(c', q, r_0) > r_0; \tag{4.58}$$

note that inequality (4.58) can be rewritten as

$$\psi(r)^{n-1} \le \frac{2r}{(c+1)(n-1)} \frac{d}{dr} \left(\psi^{n-1}\right)(r) \qquad \forall r \ge \tilde{r}_0.$$
(4.59)

Hence, if we integrate (by parts) (4.59) between  $\tilde{r}_0$  and  $r > \tilde{r}_0$ , we deduce that

$$\int_{\tilde{r}_{0}}^{r} \psi(s)^{n-1} ds \leq \frac{2}{(c+1)(n-1)} \int_{\tilde{r}_{0}}^{r} s \frac{d}{ds} (\psi^{n-1})(s) ds$$

$$= \frac{2}{(c+1)(n-1)} \left[ r \psi(r)^{n-1} - \tilde{r}_{0} \psi(\tilde{r}_{0})^{n-1} - \int_{\tilde{r}_{0}}^{r} \psi(s)^{n-1} ds \right]$$

$$\leq \frac{2r \psi(r)^{n-1}}{(c+1)(n-1)} .$$
(4.60)

Let us now rewrite (4.58) as

$$\frac{1}{\psi(r)^{n-1}} \le -\frac{2r}{(c+1)(n-1)} \frac{d}{dr} \left(\frac{1}{\psi(r)^{n-1}}\right) \qquad \forall r \ge \tilde{r}_0 \,. \tag{4.61}$$

The integration (by parts) of (4.61) between  $r>\tilde{r}_0$  and  $\infty$  (along with (4.57)) ensures that

$$\begin{split} \int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds &\leq -\int_{r}^{\infty} \frac{2s}{(c+1)(n-1)} \frac{d}{ds} \left( \frac{1}{\psi(s)^{n-1}} \right) \, ds \\ &= \frac{2}{(c+1)(n-1)} \frac{r}{\psi(r)^{n-1}} \\ &\quad + \frac{2}{(c+1)(n-1)} \int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds \,, \end{split}$$

whence

$$\int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds \le \frac{2}{(c-1)(n-1) + 2(n-2)} \, \frac{r}{\psi(r)^{n-1}} \qquad \forall r \ge \tilde{r}_0 \,. \tag{4.62}$$

By plugging estimate (4.60) into (4.31), exploiting (4.57), the fact that  $\psi(r)$  is nondecreasing and (4.62), we obtain:

$$\begin{split} Q(r) &= \left( \int_{0}^{\tilde{r}_{0}} \psi(s)^{n-1} \, ds + \int_{\tilde{r}_{0}}^{r} \psi(s)^{n-1} \, ds \right)^{\frac{1}{p}} \left( \int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \\ &\leq \left( \int_{0}^{\tilde{r}_{0}} \psi(s)^{n-1} \, ds + \frac{2r \, \psi(r)^{n-1}}{(c+1)(n-1)} \right)^{\frac{1}{p}} \left( \int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \\ &\leq \left( \frac{1}{r \, \psi(r)^{n-1}} \int_{0}^{\tilde{r}_{0}} \psi(s)^{n-1} \, ds + \frac{2}{(c+1)(n-1)} \right)^{\frac{1}{p}} \times \\ & \left[ \left( r \, \psi(r)^{n-1} \right)^{\frac{2}{p}} \int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds \right]^{\frac{1}{2}} \\ &\leq \left( \frac{1}{r \, \psi(r)^{n-1}} \int_{0}^{\tilde{r}_{0}} \psi(s)^{n-1} \, ds + \frac{2}{(c+1)(n-1)} \right)^{\frac{1}{p}} \left( \frac{2r^{\frac{p+2}{p}} \psi(r)^{-\frac{(n-1)(p-2)}{p}}}{(c-1)(n-1) + 2(n-2)} \right)^{\frac{1}{2}} \\ &\leq C \, \frac{r^{\frac{p+2}{2p}}}{\psi(r)^{\frac{(n-1)(p-2)}{2p}}} \\ &\leq C \, r^{\frac{p+2}{2p} - \frac{c(n-1)(p-2)}{2p}}, \end{split}$$

for all  $r > \tilde{r}_0$ , where from here on C stands for a general suitable positive constant depending only on  $n, c, c', q, r_0$  (that we shall not relabel). In particular,

$$\limsup_{r \to \infty} Q(r) \le C$$

since

$$p \ge \frac{2\tilde{n}}{\tilde{n}-2} \qquad \Longrightarrow \qquad \frac{p+2}{2p} - \frac{c(n-1)(p-2)}{2p} \le 0. \tag{4.63}$$

There are two possibilities: either Q(r) does not admit an internal maximum, in which case

$$Q(r) < C \qquad \forall r > 0, \qquad (4.64)$$

or Q(r) does admit an internal maximum at some  $\overline{r} > 0$ , which is a critical point. Note that, in this case, by carrying out the same computations as in the proof of Lemma 4.2, equations (4.34), (4.36) and (4.37) (with  $r_0$  replaced by  $\tilde{r}_0$ ) are still true. On the other hand, by exploiting (4.37) (with  $r_0$  replaced by  $\tilde{r}_0$ ), (4.57), (4.60) and the fact that  $\psi(r)$  is nondecreasing, we end up with

. . .

$$\begin{split} \sup_{r \in (\tilde{r}_{0},\infty)} \frac{1}{\psi(r)^{n-1}} \left( \int_{0}^{r} \psi(s)^{n-1} ds \right)^{\frac{p+2}{2p}} \\ &\leq \sup_{r \in (\tilde{r}_{0},\infty)} \left( \frac{1}{\psi(\tilde{r}_{0})^{\frac{2p(n-1)}{p+2}}} \int_{0}^{\tilde{r}_{0}} \psi(s)^{n-1} ds + \frac{1}{\psi(r)^{\frac{2p(n-1)}{p+2}}} \int_{\tilde{r}_{0}}^{r} \psi(s)^{n-1} ds \right)^{\frac{p+2}{2p}} \\ &\leq \sup_{r \in (\tilde{r}_{0},\infty)} \left( \tilde{r}_{0}^{\frac{(n-2)(2^{*}-p)}{p+2}} + \frac{1}{\psi(r)^{\frac{2p(n-1)}{p+2}}} \frac{2r \,\psi(r)^{n-1}}{(c+1)(n-1)} \right)^{\frac{p+2}{2p}} \\ &\leq \sup_{r \in (\tilde{r}_{0},\infty)} \left( \tilde{r}_{0}^{\frac{(n-2)(2^{*}-p)}{p+2}} + \frac{2}{(c+1)(n-1)} \frac{r}{\psi(r)^{\frac{(n-1)(p-2)}{p+2}}} \right)^{\frac{p+2}{2p}} \\ &\leq \sup_{r \in (\tilde{r}_{0},\infty)} \left( \tilde{r}_{0}^{\frac{(n-2)(2^{*}-p)}{p+2}} + C \,r^{1-\frac{c(n-1)(p-2)}{p+2}} \right)^{\frac{p+2}{2p}} \\ &= \left( \tilde{r}_{0}^{\frac{(n-2)(2^{*}-p)}{p+2}} + C \,\tilde{r}_{0}^{1-\frac{c(n-1)(p-2)}{p+2}} \right)^{\frac{p+2}{2p}}, \end{split}$$

where in the last line we have taken advantage of (4.63). Hence, by collecting (4.36), (4.37) (with  $r_0$  replaced by  $\tilde{r}_0$ ) and (4.64), we finally obtain

$$\begin{split} \sup_{r \in (0,\infty)} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \\ \leq \left( \frac{p}{2} \right)^{\frac{1}{2}} \left( \tilde{r}_0^{\frac{(n-2)(2^*-p)}{p+2}} + C \, \tilde{r}_0^{1-\frac{c(n-1)(p-2)}{p+2}} \right)^{\frac{p+2}{2p}} \lor C \,, \end{split}$$

which establishes (4.56) up to relabelling C.

We can now prove Theorem 4.2.

Proof of Theorem 4.2. We argue similarly to the proof of Theorem 4.1: our aim is to show that the function  $\psi_{\star}$  defined by (4.39) satisfies the hypotheses of

Lemma 4.3. In order to establish the validity of the first two inequalities of (4.55), as well as the fact that  $\psi_{\star} \in \mathcal{A}$ , one can reason exactly in the same way. As concerns the third one, some adaptations have to be performed: we shall mainly refer to [120, Subsection 8.1]. First of all, note that the general solution of the differential equation

$$\phi''(r) = C_1 r^{-2} \phi(r) \qquad \forall r \in \mathbb{R}^+$$

is explicit, i.e.

$$\phi(r) = a_1 r^{q_1} + a_2 r^{q_2} \qquad \forall r \in \mathbb{R}^+$$

for arbitrary real constants  $a_1$  and  $a_2$ , where  $q_{1,2} = (1 \pm \sqrt{1 + 4C_1})/2$ . It is not difficult to show (just by following the same ideas as in [120, Subsection 8.1]) that one can construct a function  $\psi \in \mathcal{A}$  such that

$$\psi''(r) = C_1 r^{-2} \psi(r) \quad \forall r \ge 2R_0, \qquad \psi''(r) \le C_1 r^{-2} \psi(r) \quad \forall r \in (R_0, 2R_0), \psi''(r) = 0 \quad \forall r \in [0, R_0],$$

which therefore complies with (4.3.2) for every  $r \ge 2R_0 =: r_0$  and constants  $a_1 > 0, a_2 \in \mathbb{R}$  depending only on  $C_1, R_0$ . In view of (4.11), we are in position to apply the Laplacian-comparison results of Subsection 4.2.2 (specifically (4.20)), guaranteeing that

$$\mathsf{m}(r,\theta) \ge (n-1)\frac{\psi'(r)}{\psi(r)} \ge (n-1)\left(\frac{q_1}{r} - \frac{h}{r^{1+\sqrt{1+4C_1}}}\right)$$
(4.65)

for all  $r \ge r_0$ , where h is a suitable positive constant depending on  $a_1, a_2, q_1, q_2, r_0$ . Thanks to (4.65), we can now proceed as in the proof of Theorem 4.1, observing that

$$\begin{split} \psi'_{\star}(r) &= \frac{1}{\omega_{n-1}(n-1)} \left( \frac{\operatorname{meas}\left(S_{r}\right)}{\omega_{n-1}} \right)^{\frac{1}{n-1}-1} \int_{\mathbb{S}^{n-1}} \mathsf{m}(r,\theta) \, A(r,\theta) \, d\theta \\ &\geq \frac{1}{\omega_{n-1}(n-1)} \left( \frac{\operatorname{meas}\left(S_{r}\right)}{\omega_{n-1}} \right)^{\frac{1}{n-1}-1} \times \\ & \left(n-1\right) \left( \frac{q_{1}}{r} - \frac{h}{r^{1+\sqrt{1+4C_{1}}}} \right) \int_{\mathbb{S}^{n-1}} A(r,\theta) \, d\theta \\ &= \left( \frac{q_{1}}{r} - \frac{h}{r^{1+\sqrt{1+4C_{1}}}} \right) \psi_{\star}(r) \end{split}$$

for all  $r \ge r_0$ . Hence, the function  $\psi_{\star}$  satisfies the hypotheses of Lemma 4.3 with  $c = q_1$ , c' = h and  $q = 1 + \sqrt{1 + 4C_1}$ . As a consequence, we deduce that

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi_\star(s)^{n-1} \, ds \right)^{\frac{1}{p}} \left( \int_r^\infty \frac{1}{\psi_\star(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \le C \sqrt{p} \qquad p \in \left[ \tilde{2}, 2^* \right),$$
(4.66)

where  $\tilde{2}$  is defined in (4.13) and *C* is a positive constant as in the statement. Once (4.66) has been established, the conclusion follows as in the proof of Theorem 4.1, i.e. by applying Proposition 4.1 and carrying out the same computations that led to (4.45). Let us finally deal with optimality. To this aim, it is enough to consider any function  $\psi \in \mathcal{A}$  such that

 $\psi^{\prime\prime}(r)\geq 0 \quad \forall r>0 \qquad \text{and} \qquad \psi(r)\asymp r^{\frac{\bar{n}-1}{n-1}} \quad \text{as} \ r\to\infty\,,$ 

where by  $a(r) \simeq b(r)$  we mean that the ratios a(r)/b(r), a'(r)/b'(r), a''(r)/b''(r)tend to some positive numbers and  $\tilde{n}$  is related to  $C_1$  by (4.13). This ensures that the associated model manifold  $\mathbb{M}_{\psi}^n$  is Cartan-Hadamard and complies with (4.11). Recalling that  $\tilde{n} - 2 > 0$ , we therefore obtain

$$\left(\int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds\right)^{\frac{1}{2}} \asymp r^{-\frac{\bar{n}-2}{2}} \quad \text{as } r \to \infty \tag{4.67}$$

and

$$\left(\int_0^r \psi(s)^{n-1} \, ds\right)^{\frac{1}{p}} \asymp r^{\frac{\tilde{n}}{p}} \qquad \text{as } r \to \infty \,, \tag{4.68}$$

for all  $p \ge 2$ . The combination of (4.67)–(4.68) then yields

$$\left(\int_{0}^{r} \psi(s)^{n-1} ds\right)^{\frac{1}{p}} \left(\int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} ds\right)^{\frac{1}{2}} \asymp r^{\frac{2\tilde{n}-p\tilde{n}+2p}{2p}} \quad \text{as } r \to \infty. \quad (4.69)$$

Clearly, the r.h.s. of (4.69) stays bounded as  $r \to \infty$  if and only if  $2\tilde{n} - p\tilde{n} + 2p \leq 0$ , namely  $p \geq \tilde{2}$ . Hence, thanks to Proposition 4.1, we can conclude that in this case (4.12) fails for all  $p \in [2, \tilde{2})$ . As for the behaviour of the optimal constant as  $p \to \infty$  (for n = 2), one reasons exactly as in the end of the proof of Theorem 4.1.

Remark 4.4 (On the Cartan-Hadamard assumption). It is worth pointing out that, in Theorems 4.1 and 4.2, in general it is not possible to drop the assumption that M is a Cartan-Hadamard manifold, i.e. it is not enough to require that the sectional curvatures satisfy only (4.8) or (4.11). Indeed, consider a model manifold  $\mathbb{M}^n_{\psi}$  with  $\psi \in \mathcal{A}$  such that  $\psi(r) \simeq e^{-r^{\alpha}}$  as  $r \to \infty$ , where  $\alpha = \beta/2 \in$ (0,1). It is straightforward to check that such a manifold complies with (4.8); however, it is apparent that all of the inequalities (4.9) fail, since the supremum appearing in (4.27) in Proposition 4.1 is identically  $\infty$  because of the second integral. Similarly, with regards to Theorem 4.2, one can consider a model manifold  $\mathbb{M}^N_{\psi}$  with  $\psi(r) = r^{q_2}$  for large r, where  $q_2 < 0$  is the power appearing in (4.3.2). It is plain that all such manifolds are not Cartan-Hadamard, since a local change of sign of the second derivative of  $\psi$  necessarily occurs.

Remark 4.5 (The case  $p \in [1,2)$ ). Throughout the whole paper we have assumed that  $p \geq 2$ . In fact there is a simple reason for such a restriction: it was proved in [118, Theorem 4.6] that on any Cartan-Hadamard manifold the inequality  $||f||_p \leq C ||\nabla f||_2$  always fails as soon as p is strictly smaller than 2. Moreover, since the argument used in the corresponding proof relies only on radial functions, the inequality is false even if restricted to  $C_{crad}^1(\mathbb{M}^n)$ .

### 4.4 The Poincaré inequality: McKean's Theorem and related issues

As discussed in the Introduction, one of the main motivations for this work was a celebrated paper by H.P. McKean [167], which is fully devoted to the proof of

the following result.

**Theorem 4.4** (McKean 1970, original statement). Consider a smooth, ndimensional, simply-connected Riemannian manifold M with negative sectional curvatures Sect bounded away from 0: specifically, suppose Sect  $\leq -k$  for some constant k > 0. Then the spectrum of the corresponding Laplace-Beltrami operator  $\Delta$  acting in  $L^2(M)$  is also bounded from 0: specifically, the top of the spectrum lies to the left of

$$-\frac{k\left(n-1\right)^{2}}{4}\,,$$

and this bound is sharp.

We point out that Theorem 4.4 can be rephrased equivalently by asserting that on any Cartan-Hadamard manifold  $\mathbb{M}^n$  with sectional curvatures bounded from above by -k < 0, the following *Poincaré inequality* 

$$\|f\|_{L^{2}(\mathbb{M}^{n})} \leq \frac{2}{\sqrt{k}(n-1)} \|\nabla f\|_{L^{2}(\mathbb{M}^{n})} \qquad \forall f \in C^{1}_{c}(\mathbb{M}^{n})$$

holds. This is the form of the statement that we shall refer to below.

The original proof of McKean is far from trivial. He had already understood that it all amounted to establishing the inequality for radial functions, since the extension from radial to nonradial in the pure Poincaré case (p = 2)is straightforward, see the proof of Theorem 4.5 below. However, in order to prove that the weight associated with the volume measure on  $\mathbb{M}^n$  (recall Subsection 4.2.1) satisfies a differential inequality of the type of (4.70) (actually of second order) w.r.t. the variable r, which is at the core of the problem, he employs several technical tools that involve the second fundamental form. Jacobi fields and the so-called index form of Morse theory. Here we shall only use elementary arguments related to weighted one-dimensional inequalities, in the spirit of Section 4.3. Of course the main nontrivial result behind our methods lies in the Laplacian-comparison Theorem recalled in Subsection 4.2.2, which allows one to pass from model manifolds to general manifolds with very little effort. Furthermore, through these techniques, we are able to slightly generalize McKean's Theorem, by requiring that only the *radial* sectional curvatures are negative away from zero.

In order to carry out our alternative proof, we need a preliminary lemma.

**Lemma 4.4.** Let  $\psi \in C^1([0,\infty)) \cap C^\infty((0,\infty))$  be a positive function on  $(0,\infty)$  such that  $\psi(0) = 0$ . Let  $n \in \mathbb{N}$  be larger than or equal to 2. If

$$\frac{\psi'(r)}{\psi(r)} \ge \sqrt{k} \qquad \forall r > 0, \quad for \ some \ k > 0, \qquad (4.70)$$

then

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{1}{2}} \left( \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \le \frac{1}{\sqrt{k} \, (n-1)} \, . \tag{4.71}$$

*Proof.* For convenience, let us assume that k = 1: the general case can be obtained by a simple scaling argument, as we shall see in the end of the proof. So, given any  $\varepsilon > 0$ , upon integrating (4.70) from  $\varepsilon$  to r we infer that

$$\psi(r) \ge \psi(\varepsilon) e^{r-\varepsilon} \quad \forall r \ge \varepsilon.$$
 (4.72)

On the other hand, inequality (4.70) can be rewritten as

$$\psi(r)^{n-1} \le \frac{1}{n-1} \frac{d}{dr} \left(\psi^{n-1}\right)(r) \qquad \forall r > 0$$

so that an integration between 0 and r yields (recall that  $\psi(0) = 0$ )

$$\int_0^r \psi(s)^{n-1} \, ds \le \frac{1}{n-1} \, \psi(r)^{n-1} \qquad \forall r > 0 \,. \tag{4.73}$$

Similarly, another way of rewriting (4.70) is

$$\frac{1}{\psi(r)^{n-1}} \le -\frac{1}{n-1} \frac{d}{dr} \left(\psi^{-n+1}\right)(r) \qquad \forall r > 0;$$
(4.74)

by integrating (4.74) from r to  $\infty$  we obtain

$$\int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds \le \frac{1}{n-1} \, \frac{1}{\psi(r)^{n-1}} \qquad \forall r > 0 \,, \tag{4.75}$$

where we have exploited the fact that  $\lim_{r\to\infty} \psi(r) = \infty$ , trivial consequence of (4.72). By combining (4.73) and (4.75), we finally deduce that

$$\int_0^r \psi(s)^{n-1} \, ds \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \le \frac{1}{(n-1)^2} \qquad \forall r > 0 \,,$$

namely (4.71) for k = 1. In order to deal with the general case, it is enough to apply the just proved result to  $r \mapsto \sqrt{k} \psi(r/\sqrt{k})$ .

We are now ready to give an elementary proof of McKean's Theorem, by taking advantage of Lemma 4.4 along with the basic facts in Riemannian geometry recalled in Subsections 4.2.1–4.2.3.

**Theorem 4.5** (McKean's Theorem revisited). Let  $\mathbb{M}^n$  be a Cartan-Hadamard manifold satisfying

$$\operatorname{Sect}_{\omega}(x) \le -k \qquad \forall x \in \mathbb{M}^n ,$$
 (4.76)

where k > 0 and  $\text{Sect}_{\omega}(x)$  denotes the sectional curvature w.r.t. any 2-dimensional tangent subspace  $\omega$  at x containing a radial direction. Then

$$\|f\|_{L^{2}(\mathbb{M}^{n})} \leq \frac{2}{\sqrt{k}(n-1)} \|\nabla f\|_{L^{2}(\mathbb{M}^{n})} \qquad \forall f \in C^{1}_{c}(\mathbb{M}^{n}).$$
(4.77)

*Proof.* Thanks to (4.76), we can apply the Laplacian-comparison result recalled in Subsection 4.2.2 with the explicit model function  $\psi(r) = \sinh(\sqrt{k}r)$ , which corresponds to the hyperbolic space of curvature -k and trivially satisfies

$$\frac{\psi''(r)}{\psi(r)} = k \qquad \forall r > 0 \,.$$

Hence, upon recalling identity (4.18), we deduce that

$$\begin{aligned} \frac{\frac{\partial}{\partial r}A(r,\theta)}{A(r,\theta)} = &\mathsf{m}(r,\theta) \\ \geq &(n-1)\frac{\psi'(r)}{\psi(r)} \\ = &(n-1)\sqrt{k} \coth\left(\sqrt{k}r\right) \\ \geq &(n-1)\sqrt{k} \quad \forall r > 0, \quad \forall \theta \in \mathbb{S}^{n-1} \setminus \mathcal{P}, \end{aligned}$$

namely

$$\frac{\frac{\partial}{\partial r}\psi_A(r,\theta)}{\psi_A(r,\theta)} \ge \sqrt{k} \qquad \forall r > 0 \,, \quad \forall \theta \in \mathbb{S}^{n-1} \setminus \mathcal{P} \,, \tag{4.78}$$

where  $\psi_A(r,\theta) := A(r,\theta)^{\frac{1}{n-1}}$ .

Thanks to the basic properties of the function  $(r, \theta) \mapsto A(r, \theta)$  described in Subsection 4.2.1, in view of (4.78) we can apply Lemma 4.4 to  $\psi \equiv \psi_A(\cdot, \theta)$ , for every fixed  $\theta \in \mathbb{S}^{n-1} \setminus \mathcal{P}$ , which ensures that

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi_A(s,\theta)^{n-1} \, ds \right)^{\frac{1}{2}} \left( \int_r^\infty \frac{1}{\psi_A(s,\theta)^{n-1}} \, ds \right)^{\frac{1}{2}} \le \frac{1}{\sqrt{k} (n-1)}$$
$$\forall \theta \in \mathbb{S}^{n-1} \setminus \mathcal{P} \,. \tag{4.79}$$

As a consequence, from Proposition 4.1 with p = 2 and  $w(r) \equiv \psi_A(r, \theta)$  we deduce that

$$\int_{0}^{\infty} g(r)^{2} A(s,\theta) dr \leq \frac{4}{k \left(N-1\right)^{2}} \int_{0}^{\infty} g'(r)^{2} A(r,\theta) dr$$

$$\forall g \in C_{c}^{1}([0,\infty)), \quad \forall \theta \in \mathbb{S}^{n-1} \setminus \mathcal{P}.$$
(4.80)

On the other hand, if  $f \in C_c^1(\mathbb{M}^n)$  then  $r \mapsto f(r,\theta) \in C_c^1([0,\infty))$  for every  $\theta \in \mathbb{S}^{N-1}$ , so that by exploiting (4.80) with  $g(r) \equiv f(r,\theta)$ , integrating over  $\mathbb{S}^{n-1}$  and using Fubini's Theorem, we end up with

$$\begin{split} \int_{\mathbb{S}^{n-1}} \int_0^\infty f(r,\theta)^2 A(r,\theta) \, dr \, d\theta &\leq \frac{4}{k \left(n-1\right)^2} \, \int_{\mathbb{S}^{n-1}} \int_0^\infty \left| \frac{\partial}{\partial r} f(r,\theta) \right|^2 A(r,\theta) \, dr \, d\theta \\ &\leq \frac{4}{k \left(n-1\right)^2} \, \int_{\mathbb{S}^{n-1}} \int_0^\infty \left| \nabla f(r,\theta) \right|^2 A(r,\theta) \, dr \, d\theta \,, \end{split}$$

namely (4.77), recalling (4.17).

e had alreadv b

As concerns the sharpness of the constant, which of course had already been established by McKean, note that it is easily verified e.g. by observing that the r.h.s. of (4.79) is attained on hyperbolic space, i.e. when  $\psi_A(r,\theta) \equiv \sinh(\sqrt{kr})$ .

#### 4.4.1 Negative curvatures outside a ball

In fact the previous techniques allow us to obtain a McKean-type result under the weaker assumption that curvatures are bounded above by a negative constant only in the complement of a ball. To the best of our knowledge, this result is new even if, in some sense, expectable.

We first establish the following lemma.

**Lemma 4.5.** Let  $\psi \in C^1([0,\infty)) \cap C^\infty((0,\infty))$  be a positive function on  $(0,\infty)$  such that  $\psi(0) = 0$  and  $\psi'(0) > 0$ . Let  $n \in \mathbb{N}$  be larger than or equal to 2. If

$$\psi'(r) \ge 0 \quad \forall r \ge 0 \qquad and \qquad \frac{\psi'(r)}{\psi(r)} \ge c \quad \forall r \ge r_0$$

$$(4.81)$$

for some  $c, r_0 > 0$ , then there exists a positive constant C, depending only on  $N, c, r_0$ , such that

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi(s)^{n-1} \, ds \right)^{\frac{1}{2}} \left( \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \right)^{\frac{1}{2}} \le C \,. \tag{4.82}$$

*Proof.* First of all we observe that the integration of the last inequality in (4.81) from  $r_0$  to  $r > r_0$ , along with the fact that  $\psi$  is positive, yields the bound from below

$$\psi(r) \ge K e^{cr} \quad \forall r \ge r_0 , \qquad \text{where } K := \psi(r_0) e^{-cr_0} .$$
 (4.83)

We proceed similarly to the proof of Lemma 4.4. Upon rewriting the last inequality in (4.81) as

$$\psi(r)^{n-1} \le \frac{1}{c(N-1)} \frac{d}{dr} \left(\psi^{n-1}\right)(r) \qquad \forall r \ge r_0$$

and integrating between  $r_0$  and  $r > r_0$ , we obtain:

$$\int_{r_0}^r \psi(s)^{n-1} ds \leq \frac{1}{c(n-1)} \int_{r_0}^r \frac{d}{ds} \left(\psi^{n-1}\right)(s) ds$$
  
=  $\frac{1}{c(n-1)} \left[\psi(r)^{n-1} - \psi(r_0)^{n-1}\right]$  (4.84)  
 $\leq \frac{1}{c(n-1)} \psi(r)^{n-1}.$ 

Another way of rewriting such an inequality is

$$\frac{1}{\psi(r)^{n-1}} \le -\frac{1}{c(n-1)} \frac{d}{dr} \left(\psi^{-n+1}\right)(r) \qquad \forall r \ge r_0, \qquad (4.85)$$

whence by integrating (4.85) from r to  $\infty$  we infer that

$$\int_{r}^{\infty} \frac{1}{\psi(s)^{n-1}} \, ds \le \frac{1}{c(n-1)} \, \frac{1}{\psi(r)^{n-1}} \qquad \forall r \ge r_0 \,, \tag{4.86}$$

where we have used the property  $\lim_{r\to\infty} \psi(r) = \infty$ , consequence of (4.83). The combination of (4.84), (4.86) and the first inequality of (4.81) yields

$$\begin{split} \hat{Q}(r) &:= \int_0^r \psi(s)^{n-1} \, ds \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \\ &= \left( \int_0^{r_0} \psi(s)^{n-1} \, ds + \int_{r_0}^r \psi(s)^{n-1} \, ds \right) \int_r^\infty \frac{1}{\psi(s)^{n-1}} \, ds \\ &\leq \frac{1}{c(n-1)} \frac{1}{\psi(r)^{n-1}} \int_0^{r_0} \psi(s)^{n-1} \, ds + \frac{1}{c^2(n-1)^2} \\ &\leq \frac{r_0}{c(n-1)} + \frac{1}{c^2(n-1)^2} \end{split}$$

for all  $r \geq r_0$ , namely

$$\sup_{r \in [r_0,\infty)} \hat{Q}(r) \le \frac{r_0}{c(n-1)} + \frac{1}{c^2(n-1)^2} =: C^2.$$
(4.87)

We are left with bounding the analogous supremum for r ranging between 0 and  $r_0$ . There are two possibilities: either  $\sup_{r \in (0,r_0)} \hat{Q}(r) \leq C^2$ , in which case (4.82) trivially follows, or  $\sup_{r \in (0,r_0)} \hat{Q}(r) > C^2$ , in which case there necessarily exists  $\bar{r} \in (0, r_0)$  such that  $\hat{Q}(\bar{r}) = \sup_{r \in (0,\infty)} \hat{Q}(r)$  (note that  $\lim_{r \to 0} \hat{Q}(r) = 0$ ). In particular, we can reason exactly as in the proof of Lemma 4.2 to deduce the analogue of (4.35) with p = 2:

$$\sqrt{\hat{Q}(\overline{r})} = \frac{1}{\psi(\overline{r})^{n-1}} \int_0^{\overline{r}} \psi(s)^{n-1} \, ds \le \overline{r} \le r_0 \,, \tag{4.88}$$

where we have taken advantage again of the elementary fact that  $\psi(r)$  is nondecreasing. Hence, as a consequence of (4.87)–(4.88),

$$\sup_{r\in(0,\infty)}\sqrt{\hat{Q}(r)} \le C \lor r_0\,,$$

namely (4.82) up to relabelling C.

We are now ready to prove the following version of McKean's Theorem outside a ball.

**Theorem 4.6** (McKean's Theorem outside a ball). Let  $\mathbb{M}^n$  be a Cartan-Hadamard manifold such that

$$\operatorname{Sect}_{\omega}(x) \leq -k \qquad \forall x \in \mathbb{M}^n \setminus B_{R_0}$$

for some  $k, R_0 > 0$ , where  $\text{Sect}_{\omega}(x)$  denotes the sectional curvature w.r.t. any 2-dimensional tangent subspace  $\omega$  at x containing a radial direction. Then there exists a positive constant  $C_P$ , depending only on k and  $R_0$ , such that

$$\|f\|_{L^{2}(\mathbb{M}^{n})} \leq C_{P} \|\nabla f\|_{L^{2}(\mathbb{M}^{n})} \qquad \forall f \in C_{c}^{1}(\mathbb{M}^{n}).$$
(4.89)

*Proof.* By applying Lemma 4.1 with  $\beta = 0$ , we infer that

$$\mathsf{m}(r,\theta) \ge c \qquad \forall (r,\theta) \in [r_0,\infty) \times \mathbb{S}^{n-1}$$

for suitable positive constants  $r_0 = r_0(k, R_0)$  and  $c = c(n, k, R_0)$ , that is

$$\frac{\frac{\partial}{\partial r}\psi_A(r,\theta)}{\psi_A(r,\theta)} \ge c \qquad \forall r > r_0 \,, \quad \forall \theta \in \mathbb{S}^{n-1} \backslash \mathcal{P} \,, \qquad \text{where } \psi_A(r,\theta) := A(r,\theta)^{\frac{1}{n-1}} \,.$$

$$(4.90)$$

Thanks to (4.22) and (4.23), in view of (4.90) we can apply Lemma 4.5 to  $\psi \equiv \psi_A(\cdot, \theta)$  at each fixed  $\theta \in \mathbb{S}^{n-1} \setminus \mathcal{P}$ , which ensures that

$$\sup_{r \in (0,\infty)} \left( \int_0^r \psi_A(s,\theta)^{n-1} \, ds \right)^{\frac{1}{2}} \left( \int_r^\infty \frac{1}{\psi_A(s,\theta)^{n-1}} \, ds \right)^{\frac{1}{2}} \le C \qquad \forall \theta \in \mathbb{S}^{n-1} \setminus \mathcal{P}$$

where C is the same constant as in (4.82). Hence, by Proposition 4.1 with p = 2and  $w(r) \equiv \psi_A(r, \theta)$ , we end up with

$$\int_{0}^{\infty} g(r)^{2} A(s,\theta) dr \leq 4C^{2} \int_{0}^{\infty} g'(r)^{2} A(r,\theta) dr$$

$$\forall g \in C_{c}^{1}([0,\infty)), \quad \forall \theta \in \mathbb{S}^{n-1} \setminus \mathcal{P}.$$
(4.91)

Once (4.91) has been established, inequality (4.89) follows with  $C_P = 2C$  just by reasoning as in the final part of the proof of Theorem 4.5.

# 4.5 Failure of the inequalities in the nonradial framework

In this section we prove Theorem 4.3 by constructing an explicit sequence of nonradial functions that make the Rayleigh quotient associated with inequality (4.15) blow up.

Proof of Theorem 4.3. For later convenience, we set  $\lambda := (2-\beta)/2 \in (0,1)$ . We want to prove that the Sobolev-type inequality

$$\left(\int_{\mathbb{M}^n} \left|f\right|^p\right)^{\frac{1}{p}} \le C\left(\int_{\mathbb{M}^n} \left|\nabla f\right|^2\right)^{\frac{1}{2}},\tag{4.92}$$

supposed to be valid for all  $f \in C_c^1(\mathbb{M}^n)$ , actually *fails* as soon as  $p < 2^*$  under the running assumptions on  $\mathbb{M}^n$ . To this purpose, we provide the following family of functions that we denote by  $f_R$ , for which the Rayleigh quotient of (4.92) blows up as  $R \to +\infty$ :

$$f_R(x) := \left(1 - \frac{\mathrm{d}(x, o_R)}{R^{1-\lambda}}\right)_+ \qquad \forall x \in \mathbb{M}^n \,, \tag{4.93}$$

where  $o_R \in S_R$ , namely  $R = d(o_R, o)$ . Note that each  $f_R$  is in fact only Lipschitz regular, but this is not an issue (one can always regularize it in order to obtain a  $C_c^1$  function close enough to  $f_R$ , see also Remark 4.3). In view of (4.93), we have:

$$|\nabla f_R(x)| = \frac{1}{R^{1-\lambda}} \chi_{B_{R^{1-\lambda}}(o_R)}(x) \qquad \forall x \in \mathbb{M}^n , \qquad (4.94)$$

$$|f_R(x)| \ge \frac{1}{2} \chi_{B_{\frac{R^{1-\lambda}}{2}}(o_R)}(x) \qquad \forall x \in \mathbb{M}^n,$$
(4.95)

 $B_r(o_R)$  being the Riemannian ball of radius r > 0 centered at  $o_R$ . Thanks to (4.95), the  $L^p$  norm of  $f_R$  is readily estimated from below:

$$\int_{\mathbb{M}^n} |f_R|^p \ge \frac{1}{2^p} \nu \left( B_{\frac{R^{1-\lambda}}{2}}(o_R) \right) \ge \frac{\omega_{n-1}}{2^{p+n} n} R^{(1-\lambda)n}, \qquad (4.96)$$

where in the last inequality we have used the simple fact that, because  $\mathbb{M}^n$  is Cartan-Hadamard, the volume of balls (w.r.t. any pole) grows at least with Euclidean rate (recall that  $\omega_{n-1}$  is the Hausdorff measure of the Euclidean unit sphere of dimension n-1). This is just a consequence of Laplacian comparison, see Section 4.2 (in particular Subsections 4.2.1–4.2.3). Let us now deal with the  $L^2$  norm of the gradient. By (4.94), we have:

$$\int_{\mathbb{M}^N} |\nabla f_R|^2 \, d\nu = \frac{1}{R^{2(1-\lambda)}} \, \nu(B_{R^{1-\lambda}}(o_R)) \,; \tag{4.97}$$

in order to estimate the volume in the r.h.s. of (4.97), we need to exploit (4.14).

First of all, note that in  $B_{R^{1-\lambda}}(o_R)$  condition (4.14) can be rewritten as follows:

$$\operatorname{Ric}(x) \geq -\frac{C_2}{\operatorname{d}(x, o)^{2-2\lambda}}$$

$$\geq -\frac{C_2}{\left[\operatorname{d}(o, o_R) - \operatorname{d}(x, o_R)\right]^{2-2\lambda}}$$

$$\geq -\frac{C_2}{\left(R - R^{1-\lambda}\right)^{2-2\lambda}}$$

$$\geq -\frac{2C_2}{R^{2-2\lambda}} \quad \forall x \in B_{R^{1-\lambda}}(o_R),$$
(4.98)

provided R is so large that  $(1 - R^{-\lambda})^{2-2\lambda} \ge 1/2$  and  $R - R^{1-\lambda} \ge R_0$ . Thanks to (4.98), we can then apply comparison with the surface measure of the balls in the hyperbolic space with sectional curvature

$$-\frac{2C_2}{(N-1)R^{2-2\lambda}}:=-\frac{\hat{C}^2}{R^{2-2\lambda}}\,,$$

which corresponds to the model function

$$\psi(r) = \frac{R^{1-\lambda}}{\hat{C}} \sinh\left(\frac{\hat{C}}{R^{1-\lambda}}r\right)$$

(recall again Subsections 4.2.1–4.2.3, here the reference "pole" is  $o_R$ ). To this aim we take advantage, in particular, of the validity of (4.98) along radial directions emanating from  $o_R$ , which yields

$$\nu(B_{R^{1-\lambda}}(o_R)) \le \omega_{n-1} \frac{R^{(1-\lambda)(n-1)}}{\hat{C}^{n-1}} \int_0^{R^{1-\lambda}} \left[\sinh\left(\frac{\hat{C}}{R^{1-\lambda}}r\right)\right]^{n-1} dr$$
  
=  $\omega_{n-1} \frac{R^{(1-\lambda)n}}{\hat{C}^n} \int_0^{\hat{C}} \sinh(s)^{n-1} ds$ . (4.99)

Hence, by virtue of (4.97) and (4.99), we obtain:

$$\int_{\mathbb{M}^n} |\nabla f_R|^2 \, d\nu \le \frac{\omega_{n-1} \int_0^{\hat{C}} \sinh(s)^{n-1} \, ds}{\hat{C}^n} \, R^{(1-\lambda)(n-2)} \,. \tag{4.100}$$

So, if inequality (4.15) was true, in view of (4.96) and (4.100) we would end up with

$$\left(\frac{\omega_{n-1} R^{(1-\lambda)n}}{2^{p+n} n}\right)^{\frac{1}{p}} \le C \left(\frac{\omega_{n-1} \int_0^{\hat{C}} \sinh(s)^{n-1} ds}{\hat{C}^n} R^{(1-\lambda)(N-2)}\right)^{\frac{1}{2}},$$

namely

$$R^{\frac{1}{p}-\frac{1}{2^*}} \le \overline{C}(n,\lambda,C,p) \,,$$

and the contradiction is achieved upon letting  $R \to \infty$ , since  $p < 2^*$ .

## 4.6 The porous medium equation on Cartan-Hadamard manifolds

Theorem 4.1 has some interesting consequences concerning smoothing effects for the porous medium equation (4.6), at least when the initial datum belongs to  $L^1(\mathbb{M}^n)$  and is radially symmetric with respect to the pole o. We shall denote by  $L^1_{rad}(\mathbb{M}^n)$  the space constituted by all such functions.

**Theorem 4.7.** Let  $\mathbb{M}^n$  be a Cartan-Hadamard manifold such that

$$\operatorname{Sect}_{\omega}(x) \le -C_0 r^{-\beta} \qquad \forall x \in \mathbb{M}^n \setminus B_{R_0}, \qquad (4.101)$$

for some  $\beta \in (0,2)$  and  $C_0, R_0 > 0$ . Then there exists a positive constant K > 0, depending only on  $m, n, \beta, C_0, R_0$ , such that for any initial datum  $u_0 \in L^1_{rad}(\mathbb{M}^n)$  the solution u of (4.6) satisfies the smoothing estimate

$$\|u(t)\|_{L^{\infty}(\mathbb{M}^{n})} \leq K \left[ \log \left( t \, \|u_{0}\|_{L^{1}(\mathbb{M}^{n})}^{m-1} + e \right) \right]^{\frac{2+\beta}{(m-1)(2-\beta)}} t^{-\frac{1}{m-1}} \qquad \forall t > 0.$$
(4.102)

Moreover, the result is optimal w.r.t. long-time dependence, in the sense that if (4.101) holds with reverse inequality and  $\operatorname{Sect}_{\omega}(x)$  replaced by  $\operatorname{Ric}_o(x)$ , then there exist initial data  $u_0 \in L^1_{\operatorname{rad}}(\mathbb{M}^n)$  for which the analogue of (4.102) holds with reverse inequality for large t.

The above result is a consequence of arguments that follow the lines of [118, Theorem 3.1] and [120, Theorem 3.2]. For the reader's convenience here we write down a concise proof (mostly borrowed from the proof of [118, Theorem 3.1]), which should allow one to realize how the Sobolev-type inequalities (4.9) come into play.

Proof. Let q > 0 and  $\sigma = p/2 \in (1, 2^*/2]$ , both being for the moment free parameters. We can suppose with no loss of generality that  $u_0 \in L^1(\mathbb{M}^n) \cap$  $L^{\infty}(\mathbb{M}^n)$ . In order to make rigorous the computations we shall carry out, one needs some approximation procedures, which we skip because they are out of the scope of this section: see [118] and references therein for more details. To improve readability, throughout the proof we mean  $\|\cdot\|_q \equiv \|\cdot\|_{L^q(\mathbb{M}^n)}$ .

So, by multiplying the differential equation in (4.6) by  $u^q$ , integrating by parts and using (4.9), we obtain:

$$\frac{d}{dt} \|u(t)\|_{q+1}^{q+1} = -\frac{4q(q+1)m}{(m+q)^2} \left\|\nabla\left(u^{\frac{q+m}{2}}\right)(t)\right\|_2^2 \\
\leq -\frac{4q(q+1)m}{(m+q)^2 C_{\sigma}^2} \|u(t)\|_{\sigma(q+m)}^{q+m},$$
(4.103)

where

$$C_{\sigma} := \frac{C p^{\frac{2+\beta}{2(2-\beta)}}}{(p-2)^{\frac{\beta}{2-\beta}}}$$
(4.104)

and C is the same constant as in (4.9). Taking advantage of standard interpolation and the well-known fact that the  $L^1$  norm does not increase along the evolution, we infer that

$$\|u(t)\|_{q+1} \le \|u(t)\|_{\sigma(q+m)}^{\frac{\sigma(q+m)q}{[\sigma(q+m)-1](q+1)}} \|u_0\|_{1}^{\frac{\sigma(q+m)-(q+1)}{[\sigma(q+m)-1](q+1)}} \quad \forall t > 0.$$
(4.105)

For notational convenience, let us assume that  $||u_0||_1 = 1$  (the case of a general  $L^1$  norm can be handled by a routine time-scaling argument). As a consequence, from (4.103) and (4.105) there follows

$$\frac{d}{dt}\|u(t)\|_{q+1}^{q+1} \le -\frac{4q(q+1)m}{(m+q)^2 C_{\sigma}^2} \|u(t)\|_{q+1}^{(q+1)\frac{\sigma(q+m)-1}{\sigma q}}.$$
(4.106)

The integration of (4.106) yields

$$y(t)^{\frac{\sigma m-1}{\sigma q}} \le \frac{1}{\frac{1}{y(0)^{\frac{\sigma m-1}{\sigma q}} + \frac{4m(q+1)(\sigma m-1)}{\sigma(q+m)^2 C_{\sigma}^2}t}} \quad \forall t > 0, \qquad y(t) := \|u(t)\|_{q+1}^{q+1},$$

whence

$$\|u(t)\|_{q+1} \le \left[\frac{\sigma(q+m)^2 C_{\sigma}^2}{4m(q+1)(\sigma m-1)}\right]^{\frac{\sigma q}{(q+1)(\sigma m-1)}} t^{-\frac{\sigma q}{(q+1)(\sigma m-1)}} \qquad \forall t > 0.$$
(4.107)

By previous results (see e.g. [117, Corollary 5.6] or [34, Theorem 1.5]), the validity of (4.1) for a fixed  $p/2 = \sigma = \sigma_0 \in (1, 2^*/2)$  entails the smoothing estimate

$$\|u(t)\|_{\infty} \le K t^{-\frac{\sigma_0}{(\sigma_0-1)(q+1)+\sigma_0(m-1)}} \|u_0\|_{q+1}^{\frac{(\sigma_0-1)(q+1)}{(\sigma_0-1)(q+1)+\sigma_0(m-1)}} \qquad \forall t > 0, \quad (4.108)$$

where from here on by K we shall denote a general positive constant that depends only on  $m, n, \beta, C_0, R_0$  (which will not be relabelled). Therefore, the combination of (4.107) (evaluated at time t/2) and (4.108) (with the time origin shifted from 0 to t/2) yields

$$\|u(t)\|_{\infty} \leq K \left[ \frac{\sigma(q+m)^2 C_{\sigma}^2}{4m(q+1)(\sigma m-1)} \right]^{\frac{\sigma q(\sigma_0-1)}{(\sigma m-1)[(\sigma_0-1)(q+1)+\sigma_0(m-1)]}} \times t^{-\frac{\sigma_0(\sigma m-1)+\sigma q(\sigma_0-1)}{(\sigma m-1)[(\sigma_0-1)(q+1)+\sigma_0(m-1)]}}$$
(4.109)

for all t > 0. Because q > 0 is a free parameter and (4.109) holds at any time for any such q, we can let  $q = \log(t + e)$  in (4.109) to obtain

for all t > 0. If  $\sigma \in (1, \sigma_0)$ , it is apparent that the first two factors in the r.h.s. of (4.110) can be bounded from above by another general positive constant K, so that (4.110) reads

$$\|u(t)\|_{\infty} \leq K C_{\sigma}^{-\frac{2\sigma(\sigma_{0}m-1)}{(\sigma_{0}-1)[\log(t+e)+1]+\sigma_{0}(m-1)\}}} \times \left[\log(t+e) C_{\sigma}^{2}\right]^{\frac{\sigma}{\sigma m-1}} t^{-\frac{\sigma}{\sigma m-1}}, \quad (4.111)$$

for all t > 0, which implies, upon recalling (4.104),

$$\|u(t)\|_{\infty} \leq K \left(\sigma - 1\right)^{\frac{2\beta\sigma(\sigma_{0}m-1)}{(2-\beta)(\sigma m-1)\{(\sigma_{0}-1)|\log(t+e)+1\}+\sigma_{0}(m-1)\}}} \times \left[\log(t+e)\left(\sigma - 1\right)^{-\frac{2\beta}{2-\beta}}\right]^{\frac{\sigma}{\sigma m-1}} t^{-\frac{\sigma}{\sigma m-1}}$$
(4.112)

for all t > 0. We can now set

$$\sigma = 1 + \frac{\sigma_0 - 1}{\log(t + e)} \,,$$

so that from (4.112) (using the fact that the first factor stays bounded) one deduces that

$$\begin{aligned} \left\| u(t) \right\|_{\infty} &\leq K \left[ \log(t+e) \right]^{\frac{2+\beta}{(m-1)(2-\beta)} - \frac{(2+\beta)(e_0-1)}{[(m-1)\log(t+e)+m(\sigma_0-1)](m-1)(2-\beta)}} \\ &\times t^{-\frac{1}{m-1} + \frac{\sigma_0-1}{[(m-1)\log(t+e)+m(\sigma_0-1)](m-1)}} \end{aligned}$$

for all t > 0, i.e.

$$\|u(t)\|_{\infty} \le K \left[\log(t+e)\right]^{\frac{2+\beta}{(m-1)(2-\beta)}} t^{-\frac{1}{m-1}} \qquad \forall t > 0$$

which is equivalent to (4.102) in the case  $||u_0||_1 = 1$ .

As concerns optimality, it is enough to recall that [120, Theorem 3.2] ensures, provided the curvature assumption (4.101) holds with reverse inequality and  $\operatorname{Sect}_{\omega}(x)$  is replaced by  $\operatorname{Ric}_{o}(x)$ , that any (nontrivial) bounded, compactly supported and positive initial datum  $u_0$  gives rise to a solution of (4.6) satisfying (in particular) the lower bound

$$\|u(t)\|_{\infty}^{m-1} \ge \hat{C} \,\frac{(\log t)^{\frac{2+\beta}{2-\beta}}}{t} \qquad \text{for large } t \,, \tag{4.113}$$

where  $\hat{C}$  is a suitable positive constant depending on  $\mathbb{M}^n, m, u_0$ . It is plain that (4.113) matches (4.102) (with respect to time behaviour) from below.

Remark 4.6 (The case  $\beta = 0$ ). It is worth pointing out that Theorem 4.7 actually holds for  $\beta = 0$  as well: in fact in such case the result is true for all  $L^1(\mathbb{M}^n)$ initial data, not only the radial ones. This is a direct consequence of Theorem 4.6 and [118, Theorem 2.1], whereas optimality follows from the sharp estimates of [221].

Finally, in the quasi-Euclidean case, thanks to Theorem 4.2 we can obtain the analogue of Theorem 4.7. The proof follows by combining [117, Corollary 5.6] (the fact that it is stated on Euclidean domains is inessential) and [120, Theorem 6.2], so we shall omit it since the argument is just a simplified version of the one used in the proof of Theorem 4.7.

**Theorem 4.8.** Let  $\mathbb{M}^n$  be a Cartan-Hadamard manifold such that

$$\operatorname{Sect}_{\omega}(x) \leq -C_1 r^{-2} \qquad \forall x \in \mathbb{M}^n \setminus B_{R_0}$$

$$(4.114)$$

for some  $C_1$  and  $R_0 > 0$ . Then there exists a positive constant K > 0, depending only on  $N, C_1, R_0$ , such that for any initial datum  $u_0 \in L^1_{rad}(\mathbb{M}^n)$  the solution u of (4.6) satisfies the smoothing estimate

$$\|u(t)\|_{L^{\infty}(\mathbb{M}^{n})} \leq K t^{-\frac{\tilde{n}}{2+\tilde{n}(m-1)}} \|u_{0}\|_{L^{1}(\mathbb{M}^{n})}^{\frac{2}{2+\tilde{n}(m-1)}} \qquad \forall t > 0, \qquad (4.115)$$

where  $\tilde{n}$  is defined in (4.13).

Moreover, the result is optimal w.r.t. long-time dependence, in the sense that if (4.114) holds with reverse inequality and  $\operatorname{Sect}_{\omega}(x)$  replaced by  $\operatorname{Ric}_o(x)/(n-1)$ , then there exist initial data  $u_0 \in L^1_{\operatorname{rad}}(\mathbb{M}^n)$  for which the analogue of (4.115) holds with reverse inequality for large t.

# Part III Appendixes

### Appendix A

### A general result on Riemannian manifolds with boundary

In this appendix we prove a general result on Riemannian manifolds with boundary, in particular let (M, g) be a *n*-dimensional orientable compact Riemannian  $C^2$ -manifold with boundary  $\partial M$ . For  $\delta \in \mathbb{R}_+$  we denote

$$M_{\delta} := \{ p \in M : d_M(p, \partial M) > \delta \},\$$

and for  $r \in \mathbb{R}_+$  and  $z \in M$  we denote by  $\mathcal{B}_r(z)$  the geodesic ball centred at z of radius r, i.e.

$$\mathcal{B}_r(z) = \{ p \in M : d_M(z, p) < r \}$$

where  $d_M$  is the geodesic distance on M induced by g. Moreover we denote by  $|\cdot|_g$  the Riemannian volume with respect to g.

**Proposition A.1.** Assume that there exist positive constants c and  $\delta_0$  such that

$$\mathcal{B}_r(z)|_g \ge cr^n \,, \tag{A.1}$$

and  $B_r(z)$  belongs to the image of the exponential map, for every  $z \in M_{\delta}$  and  $0 < r \le \delta < \delta_0$ .

Fix p and q in a connected component of  $M_{\delta}$ . Then there exists a piecewise geodesic path  $\gamma : [0,1] \to M_{\delta/2}$  connecting p and q of length bounded by  $\delta N_{\delta}$ , where

$$N_{\delta} := \max\left(4, \frac{2^n |M|_g}{c\delta^n}\right) \,. \tag{A.2}$$

*Proof.* We proceed in three steps:

- 1. We observe that we can join p and q by a continuous path  $\tilde{\gamma} : [0, 1] \to M_{\delta}$  such that  $\tilde{\gamma}(0) = p$  and  $\tilde{\gamma}(1) = q$ .
- 2. We construct a chain of pairwise disjoint geodesic balls  $\{\mathcal{B}_0, \ldots, \mathcal{B}_{N+1}\}$  of radius  $\delta/2$  such that:

 $- \mathcal{B}_0$  is centered at p;

- $\mathcal{B}_i$  is centered at  $\tilde{\gamma}(t_i)$ , where  $\{t_i\}$  is an increasing sequence in [0, 1]; -  $\mathcal{B}_{N+1}$  contains q;
- $\mathcal{B}_i$  is tangent to  $\mathcal{B}_{i+1}$  for any  $i = 0, \ldots, N$ .

In order to construct this chain we consider the increasing sequence  $\{t_0, t_1, \ldots, t_N\}$ in [0, 1] recursively defined as follows:

$$t_0=0\,,$$

 $\operatorname{and}$ 

$$t_{i+1} := \inf \left\{ t \in [0,1] : \mathcal{B}_{\delta/2}(\tilde{\gamma}(s)) \cap \bigcup_{j=0}^{i} \mathcal{B}_{\delta/2}(\tilde{\gamma}(t_j)) = \emptyset, \, \forall s \in [t,1] \right\}, \quad (A.3)$$

if the set in brackets is non-empty, and  $t_{i+1} = t_N$  otherwise. Therefore, by construction,  $\{t_0, t_1, \ldots, t_N\}$  is an increasing sequence in [0, 1] satisfying

$$\mathcal{B}_{\delta/2}(\tilde{\gamma}(t_i)) \cap \mathcal{B}_{\delta/2}(\tilde{\gamma}(t_j)) = \emptyset \quad \text{for } i \neq j, \, i, j = 0, \dots, N \,, \tag{A.4}$$

 $\operatorname{and}$ 

$$\mathcal{B}_{\delta/2}(\tilde{\gamma}(t_i)) \subset M_{\delta/2} \quad i = 0, \dots, N$$

We complete the sequence by adding  $t_{N+1} = 1$  as the last term. Since

$$\left| \bigcup_{j=0}^{N} \mathcal{B}_{\delta/2}(\tilde{\gamma}(t_j)) \right|_{g} \le |M|_{g} \,,$$

from (A.1) and (A.4) we get

$$N+1 \le N_{\delta} \,. \tag{A.5}$$

From (A.3) it is clear that

$$\overline{\mathcal{B}}_{\delta/2}(\tilde{\gamma}(t_i)) \cap \bigcup_{j=0}^{i-1} \overline{\mathcal{B}}_{\delta/2}(\tilde{\gamma}(t_j)) \neq \emptyset, \quad \text{for every } i = 1, \dots, N.$$

3. We construct the piecewise geodesic path  $\gamma$ . The idea is the following: for every  $i = 0, \ldots, N$  we choose a tangency point  $p_i$  between  $\mathcal{B}_i$  and  $\mathcal{B}_{i+1}$ . The piecewise geodesic path  $\gamma$  is constructed by connecting  $\tilde{\gamma}(t_i)$  with  $p_i$  and  $p_i$  with  $\tilde{\gamma}(t_{i+1})$  by using geodesic radii, for  $i = 0, \ldots, N-1$ , and connecting  $\tilde{\gamma}(t_N)$  with q by using a geodesic path contained in  $\mathcal{B}_{N+1}$ . Hence

$$\operatorname{length}(\gamma) \le N\delta \le \delta N_{\delta} \,,$$

as required.

In order to construct the path  $\gamma$  we set

$$\sigma(i) = \max\{j > i : \overline{\mathcal{B}}_{\delta/2}(\tilde{\gamma}(t_i)) \cap \overline{\mathcal{B}}_{\delta/2}(\tilde{\gamma}(t_j)) \neq \emptyset\}.$$

Then we set  $\sigma^2(i) = \sigma(\sigma(i))$ ,  $\sigma^3(i) = \sigma(\sigma(\sigma(i)))$  and so on and fix  $\tau \in \mathbb{N}$  such that  $\sigma^{\tau}(0) = N$ . We define  $\gamma_1$  as a minimal geodesic joining p and  $\tilde{\gamma}(t_{\sigma(0)})$  and such that

$$\gamma_1 \subset \overline{\mathcal{B}}_{\delta/2}(p) \cup \overline{\mathcal{B}}_{\delta/2}(\tilde{\gamma}(t_{\sigma(0)}));$$

for  $i = 2, ..., \tau$ , we let  $\gamma_i$  be a minimal geodesic joining  $\tilde{\gamma}(t_{\sigma^i(0)})$  and  $\tilde{\gamma}(t_{\sigma^{i+1}(0)})$ and such that

$$\gamma_i \subset \overline{\mathcal{B}}_{\delta/2}(\tilde{\gamma}(t_{\sigma^i(0)})) \cup \overline{\mathcal{B}}_{\delta/2}(\tilde{\gamma}(t_{\sigma^{i+1}(0)})).$$

Moreover, we let  $\gamma_{\tau+1}$  be a minimal geodesic joining  $\tilde{\gamma}(t_N)$  and q and such that

$$\gamma_{\tau+1} \subset \overline{\mathcal{B}}_{\delta/2}(\tilde{\gamma}(t_{\sigma^{\tau+1}(0)})) \cup \overline{\mathcal{B}}_{\delta/2}(q)$$

Let  $\gamma$  be the piecewise geodesic obtained as the union of  $\gamma_1, \ldots, \gamma_{\tau+1}$ . It is clear that each  $\gamma_i$  has length  $\delta$  for  $i = 1, \ldots, \tau$ , and  $\leq \delta$  for  $i = \tau + 1$ . Since  $\tau \leq N$ , from (A.5) we obtain

$$\operatorname{length}(\gamma) \le (\tau + 1)\delta \le \delta N_\delta \,,$$

as required.

The second result of this appendix is the following Proposition in which we give an upper bound of the diameter of M when  $\partial M = \emptyset$ . The proof is analogue to the one of Proposition A.1 and it is omitted.

**Proposition A.2.** Assume  $\partial M = \emptyset$  and that there exist positive constants c and  $\delta$  such that

$$|\mathcal{B}_r(z)|_g \ge cr^n \,, \tag{A.6}$$

for every  $z \in M$  and  $0 < r \leq \delta$ . Fix p and q in M, then there exists a piecewise geodesic path  $\gamma : [0,1] \to M$  connecting p to q of length bounded by  $\delta N_{\delta}$  where  $N_{\delta}$  is given by (A.2).

In particular the diameter of M is bounded by  $\delta N_{\delta}$ .

### Appendix B

### Sharp anisotropic Sobolev inequalities with weight in convex cones

In this appendix we prove a sharp version of the anisotropic Sobolev inequality in cones by suitably adapting the optimal transportation proof of the Sobolev inequality in [72, Theorem 2]. As we shall see, the proof not only applies to the case of arbitrary norms, but it also allows us to cover a large class of weights. In particular, our result extends the weighted isoperimetric inequalities from [46, Theorem 1.3] to the full Sobolev range  $p \in (1, n)$  (note that the case p = 1 can be recovered letting  $p \to 1^+$ ).

**Theorem B.1.** Let  $p \in (1, n)$ . Let  $\Sigma$  be a convex cone and H a norm in  $\mathbb{R}^n$ . Let  $w \in C^0(\overline{\Sigma})$  be positive in  $\Sigma$ , homogeneous of degree  $a \ge 0$ , and such that  $w^{1/a}$  is concave in case a > 0. Then for any  $f \in \mathcal{D}^{1,p}(\Sigma)$  we have

$$\left(\int_{\Sigma} |f(x)|^{\beta} w(x) \, dx\right)^{p/\beta} \le C_{\Sigma}(n, p, a, H, w) \int_{\Sigma} H^{p}(\nabla f(x)) \, w(x) \, dx \qquad (B.1)$$

where

$$\beta = \frac{p(n+a)}{n+a-p} \,. \tag{B.2}$$

Moreover, inequality (B.1) is sharp and the equality is attained if and only if  $f = \mathcal{U}_{\lambda,x_0}^{H,a}$ , where

$$\mathcal{U}_{\lambda,x_0}^{H,a}(x) := \left(\frac{\lambda^{\frac{1}{p-1}}c(n,p,a,H,w)}{\lambda^{\frac{p}{p-1}} + \tilde{H}_0(x-x_0)^{\frac{p}{p-1}}}\right)^{\frac{n+a-p}{p}}$$
(B.3)

with  $\lambda > 0$  where

$$\widetilde{H}_0(\zeta) := H_0(-\zeta) \qquad \forall \zeta \in \mathbb{R}^n ,$$
(B.4)

and  $H_0$  denotes the dual norm associated to H, namely

$$H_0(\zeta) := \sup_{H(\xi)=1} \zeta \cdot \xi \qquad \forall \zeta \in \mathbb{R}^n$$

Furthermore, writing  $\Sigma = \mathbb{R}^k \times \mathcal{C}$  with  $k \in \{0, \ldots, n\}$  and with  $\mathcal{C} \subset \mathbb{R}^{n-k}$  a convex cone that does not contain a line, then:

- (i) if k = n then  $\Sigma = \mathbb{R}^n$  and  $x_0$  may be a generic point in  $\mathbb{R}^n$ ;
- (*ii*) if  $k \in \{1, \ldots, n-1\}$  then  $x_0 \in \mathbb{R}^k \times \{\mathcal{O}\}$ ;
- (iii) if k = 0 then  $x_0 = \mathcal{O}$ .

*Proof.* We aim at proving that for any nonnegative  $f, g \in L^{\beta}(\Sigma)$  with  $||f||_{L^{\beta}(\Sigma)} = ||g||_{L^{\beta}(\Sigma)}$  and such that  $\nabla f \in L^{p}(\Sigma)$ , we have that

$$\int_{\Sigma} g^{\gamma} w \, dx \le \frac{\gamma}{n+a} \left( \int_{\Sigma} H^p(\nabla f) \, w \, dx \right)^{1/p} \left( \int_{\Sigma} H_0^{p'} g^{\beta} w \, dx \right)^{1/p'}, \qquad (B.5)$$

with equality if  $f = g = \mathcal{U}_{\lambda, x_0}^{H, a}$ . The value of  $\gamma$  will be specified later. As shown in [72], inequality (B.5) implies the Sobolev inequality (B.1).

Let F and G be probability densities on  $\Sigma$  and let  $T : \Sigma \to \Sigma$  be the optimal transport map (see e.g. [225]).<sup>1</sup> It is well known that, by the transport condition  $T_{\#}F = G$ , one has

$$|\det(DT)| = \frac{F}{G \circ T}$$

(see for instance [?, Section 3]). Then, if we choose

$$F = f^{\beta}w$$
 and  $G = g^{\beta}w$ ,

the Jacobian equation for T becomes

$$|\det(DT)| \frac{w \circ T}{w} = \frac{f^{\beta}}{g^{\beta} \circ T}.$$

We observe that, since

$$T_{\#}(f^{\beta}w) = g^{\beta}w \,,$$

then for any  $0 < \gamma < \beta$  we have

$$\int_{\Sigma} g^{\gamma} w \, dx = \int_{\Sigma} (g^{\gamma-\beta} \circ T) f^{\beta} w \, dx = \int_{\Sigma} \left[ |\det(DT)| \, \frac{w \circ T}{w} \right]^{\frac{\beta-\gamma}{\beta}} f^{\gamma} w \, dx \,. \tag{B.6}$$

We choose  $\gamma$  such that

$$\frac{\beta - \gamma}{\beta} = \frac{1}{n+a}$$
 i.e.  $\gamma = \frac{p(n+a-1)}{n+a-p}$ .

<sup>&</sup>lt;sup>1</sup> As explained in [97] (see also [98]), the argument that follows can be made rigorous using the fine properties of BV functions (we note that T belongs to BV, being the gradient of a convex function). However, to emphasize the main ideas, we shall write the whole argument when  $T: \Sigma \to \Sigma$  is a  $C^1$  diffeomorphism, and we invite the interested reader to look at the proof of [97, Theorem 2.2] to understand how to adapt the argument using only that  $T \in BV_{\text{loc}}(\Sigma; \Sigma)$ .

Alternatively, arguing by approximation, one can assume that w is strictly positive in  $\overline{\Sigma} \setminus \{0\}$ , and that f and g are both strictly positive and smooth inside  $\overline{\Sigma}$ . Then, if  $T: \Sigma \to \Sigma$  denotes the optimal transport map from  $f^{\beta}w$  to  $g^{\beta}w$ , [?, Theorem 1 and Remark 4] ensure that  $T: \Sigma \to \Sigma$  is a diffeomorphism. This allows one to perform the proof of (B.5) avoiding the use of the fine properties of BV functions.

Since  $T = \nabla \varphi$  for some convex function  $\varphi$ , then DT is symmetric and nonnegative definite. In particular det $(DT) \ge 0$ , and it follows from Young and the arithmetic-geometric inequalities that

$$\left[ |\det(DT)| \frac{w \circ T}{w} \right]^{\frac{1}{n+a}} \leq \frac{n}{n+a} \det(DT)^{1/n} + \frac{a}{n+a} \left( \frac{w \circ T}{w} \right)^{1/a}$$
$$\leq \frac{1}{n+a} \left[ \operatorname{div}(T) + a \left( \frac{w \circ T}{w} \right)^{1/a} \right].$$

Also, from the concavity of  $w^{1/a}$  we have that

$$a\left(\frac{w\circ T}{w}\right)^{1/a} \le \frac{\nabla w\cdot T}{w}$$

(see [46, Lemma 5.1]), hence

$$\left[\left|\det(DT)\right|\frac{w\circ T}{w}\right]^{\frac{1}{n+a}} \le \frac{1}{n+a}\left(\operatorname{div}\left(T\right) + \frac{\nabla w\cdot T}{w}\right).$$
(B.7)

(If a = 0 then w is just constant and (B.7) corresponds to the arithmetic-geometric inequality.) Noticing that

$$\operatorname{div}(T) + \frac{\nabla w \cdot T}{w} = \frac{1}{w} \operatorname{div}(Tw)$$

combining (B.6) and (B.7) we have

$$\int_{\Sigma} g^{\gamma} w \, dx \leq \frac{1}{n+a} \int_{\Sigma} \operatorname{div} (Tw) f^{\gamma} \, dx$$
$$= -\frac{\gamma}{n+a} \int_{\Sigma} w f^{\gamma-1} T \cdot \nabla f \, dx + \frac{1}{n+a} \int_{\partial \Sigma} w f^{\gamma} T \cdot \nu \, d\sigma \, .$$

Here we notice that, since  $T(x) \in \overline{\Sigma}$  for any  $x \in \overline{\Sigma}$ , the convexity of  $\Sigma$  implies that  $T \cdot \nu \leq 0$  on  $\partial \Sigma$ . Thus we obtain

$$\int_{\Sigma} g^{\gamma} w \, dx \leq -\frac{\gamma}{n+a} \int_{\Sigma} f^{\gamma-1} T \cdot \nabla f \, w \, dx \leq \frac{\gamma}{n+a} \int_{\Sigma} f^{\gamma-1} \tilde{H}_0(T) H(\nabla f) \, w \, dx \,,$$

where the last inequality follows from the definition of the dual norm  $H_0$  of H and from the definition of  $\tilde{H}_0$  (B.4). Finally, setting  $p' = \frac{p}{p-1}$ , it follows by Holder's inequality that

$$\begin{split} \int_{\Sigma} f^{\gamma-1} \tilde{H}_0(T) H(\nabla f) \, w \, dx &\leq \left( \int_{\Sigma} f^{p(\gamma-1) - \frac{p\beta}{p'}} H^p(\nabla f) \, w \, dx \right)^{1/p} \times \\ & \left( \int_{\Sigma} \tilde{H}_0^{p'}(T) \, f^{\beta} w \, dx \right)^{1/p'} \\ &= \left( \int_{\Sigma} H^p(\nabla f) \, w \, dx \right)^{1/p} \left( \int_{\Sigma} \tilde{H}_0^{p'} g^{\beta} w \, dx \right)^{1/p'} \,, \end{split}$$

where we used the transport condition  $T_{\#}(f^{\beta}w) = g^{\beta}w$  and the identity

$$\gamma - 1 - \frac{\beta}{p'} = 0 \,.$$

Hence, by this chain of inequalities we get (B.5).

In order to prove the sharpness of our Sobolev inequality we choose  $f = g = \mathcal{U}_{1,\mathcal{O}}^{H,a}$ . In this particular case the transport map reduces to the identity map  $T(x) = \nabla \varphi(x) = x$  and  $\det(DT) = 1$ . Also the homogeneity of w implies that  $\nabla w \cdot x = a w$ . This implies that all the inequalities in the previous computations become equalities and we obtain (B.1).

Finally, to prove the characterization of the minimizers one can argue as in [98, Appendix A] and [72, Section 4]. More precisely, choose  $g = \mathcal{U}_{1,\mathcal{O}}^{H,a}$  and let f be a minimizer. As noticed in the proof of [72, Theorem 5], one can assume that  $f \geq 0$ .

First one shows that the support of f is indecomposable (this is a measuretheoretic notion of the concept that  $\{f > 0\}$  is connected, see [98, Appendix A] for a definition and more details). Indeed, otherwise one could write  $f = f_1 + f_2$ with

$$\int_{\Sigma} H^{p}(\nabla f)w(x)dx = \int_{\Sigma} H^{p}(\nabla f_{1})w(x)dx + \int_{\Sigma} H^{p}(\nabla f_{2})w(x)dx$$

and then by applying (B.1) and the fact that f is a minimizer, we would get

$$\left(\int_{\Sigma} f^{\beta} w(x) dx\right)^{p/\beta} \ge \left(\int_{\Sigma} f_1^{\beta} w(x) dx\right)^{p/\beta} + \left(\int_{\Sigma} f_2^{\beta} w(x) dx\right)^{p/\beta}$$

Since

$$\int_{\Sigma} f^{\beta} w(x) dx = \int_{\Sigma} f_1^{\beta} w(x) dx + \int_{\Sigma} f_2^{\beta} w(x) dx$$

(because  $f_1$  and  $f_2$  have disjoint support), by concavity of the function  $t \mapsto t^{p/\beta}$  we conclude that either  $f_1$  or  $f_2$  vanishes.

Once this is proved, one can then argue as in the proof of [72, Proposition 6] to deduce (from the fact that all the inequalities in the proof given above much be equalities) that T must be of the form  $T(x) = \lambda(x - x_0)$  for some  $\lambda > 0$  and  $x_0 \in \Sigma$ , from which the result follows easily. Finally, properties (i) - (ii) - (iii) on the location of  $x_0$  follow for instance from the fact that T has to map  $\Sigma$  onto  $\Sigma$ .

### Appendix C

# Symmetry results for critical p-Laplace equation in $\mathbb{R}^n$

The goal of this section is to revisit the proof of Theorem 3.1 when the cone  $\Sigma$  is  $\mathbb{R}^n$  and the norm  $H(\cdot)$  is the Euclidean norm  $|\cdot|$ , which is the simplest possible case. In this case, a proof of Theorem 3.1 is already available in literature [48, 203, 224] and asymptotic estimates on u and  $\nabla u$  are already known (see Lemma C.1 below). The knowledge of those asymptotic estimates allows us to give a more readable version of the proof of Theorem 3.1 and to emphasize the main ideas without entering in technical details.

The theorem that we are going to prove in this Appendix is the following:

**Theorem C.1.** Let  $n \ge 2$  and 1 . Let u be a solution to

$$\begin{cases} \Delta_p u + u^{p^* - 1} = 0 & \text{ in } \mathbb{R}^n \\ u > 0 & \text{ in } \mathbb{R}^n \\ u \in \mathcal{D}^{1, p}(\mathbb{R}^n) \end{cases}$$
(C.1)

then  $u(x) = \mathcal{U}_{\lambda,x_0}(x)$ , for some  $\lambda > 0$  and  $x_0 \in \mathbb{R}^n$ , where

$$\mathcal{U}_{\lambda,x_0}(x) := \left(\frac{\lambda^{\frac{1}{p-1}} \left(n^{\frac{1}{p}} \left(\frac{n-p}{p-1}\right)^{\frac{p-1}{p}}\right)}{\lambda^{\frac{p}{p-1}} + |x-x_0|^{\frac{p}{p-1}}}\right)^{\frac{n-p}{p}}.$$
 (C.2)

We will need the following three preliminary results which are collected in the following three lemmas: in the first lemma we prove explicit growth conditions on u,  $\nabla u$  and  $\nabla^2 u$ , in the second lemma we recall the Newton's inequality and in the third lemma we recall a general differential identity which holds in a very general setting.

**Lemma C.1.** Let  $n \ge 2$ , 1 and <math>u be a solution to (C.1). Then

(i) there exist two constants  $C_1$  and  $C_2$  depending only on n, p and u such that

$$\frac{C_1}{1+|x|^{\frac{n-p}{p-1}}} \le u(x) \le \frac{C_2}{1+|x|^{\frac{n-p}{p-1}}} \quad and \quad |\nabla u(x)| \le \frac{C_2}{1+|x|^{\frac{n-1}{p-1}}}, \quad (C.3)$$

for every  $x \in \mathbb{R}^n$ .

(ii) There exist a radius  $R_0$  and a constant  $C_3$  depending only on n, p and u such that

$$|\nabla u(x)| \ge \frac{C_3}{|x|^{\frac{n-1}{p-1}}},$$
 (C.4)

for every  $x \in \mathbb{R}^n \setminus B_{R_0}$ .

(iii) There exists a costant  $C_4$  depending only on n, p and u such that such that

$$|\nabla^2 u(x)| \le \frac{C_4}{|x|^{\frac{n-1}{p-1}+1}},\tag{C.5}$$

for every  $x \in \mathbb{R}^n \setminus B_{R_0}$ .

*Proof.* (i) and (ii) are shown in [224, Theorem 1.1] and in [203, Theorem 2.2], respectively. In order to prove (iii) we argue as in [29, Theorem 3.3]; let  $\rho > 4R_0$  be fixed. For  $y \in E = \overline{B_4} \setminus B_{1/4}$ , we define

$$\tilde{u}(y) = \rho^{\frac{n-p}{p-1}} u(\rho y) \,. \tag{C.6}$$

We notice that, from (C.1),

$$-\Delta_p \tilde{u} = \frac{1}{\rho^{\frac{p}{p-1}}} \tilde{u}^{p^*-1} \quad \text{in} \quad E.$$
 (C.7)

Hence  $\tilde{u}$  satisfies an elliptic equation (thanks to (C.4)) of the form

$$\sum_{i,j=1}^{n} a_{ij} \partial_{ij}^2 \tilde{u} = -\frac{1}{\rho^{\frac{p}{p-1}}} \tilde{u}^{p^*-1} \,,$$

where the coefficients  $a_{ij}$  are given by

$$a_{ij}(y) = |\nabla \tilde{u}(y)|^{p-2} \left\{ (p-2) \frac{\partial_i \tilde{u}(y) \tilde{\partial}_j u(y)}{|\nabla \tilde{u}(y)|^2} + \partial_{ij}^2 \tilde{u}(y) \right\} \,.$$

From (C.3) and (C.4) we have that there exists  $\gamma$  depending only on n and p such that

$$\frac{1}{\gamma}|\xi|^2 \le a_{ij}(y)\xi_i\xi_j \le \gamma|\xi|^2$$

for every  $y \in E$  and  $\xi \in \mathbb{R}^n$ . Notice that interior Schauder's estimates (see e.g. [112]) apply to  $\tilde{u}(y)$ . This entails  $|\nabla^2 \tilde{u}| \leq C_4$  for some positive constant  $C_4$ , that is

$$|\nabla^2 u(\rho y)| \le C_2 \rho^{\frac{p-n}{p-1}-2}$$

for  $y \in B_2 \setminus \overline{B_{1/2}}$  and *(iii)* follows.

**Lemma C.2** ([55], Lemma 3.2). Let B and C be symmetric matrices in  $\mathbb{R}^{n \times n}$ , and let B be positive semidefinite. Set A = BC. Then the following inequality holds:

$$S_2(A) \le \frac{n-1}{2n} \operatorname{Tr}(A)^2.$$
(C.8)

Moreover, if  $Tr(A) \neq 0$  and equality holds in (C.8), then

$$A = \frac{\operatorname{Tr}\left(A\right)}{n} \operatorname{Id},$$

and B is, in fact, positive definite.

**Lemma C.3** ([29], Lemma 4.1). Let v be a positive function of class  $C^3$  and let  $V : \mathbb{R}^n \to \mathbb{R}^+$  be of class  $C^3(\mathbb{R}^n)$  and such that  $V(\nabla v) \text{div}(\nabla V(\nabla v))$  can be continuously extended to zero at  $\nabla v = 0$ . Let

$$W = \nabla [\nabla_{\xi} V(\nabla v)] = V_{\xi_i \xi_j} (\nabla v) \partial_{ij}^2 v.$$
 (C.9)

Then, for any  $\gamma \in \mathbb{R}$  we have

$$2v^{\gamma}S^{2}(W) = \operatorname{div}\left(v^{\gamma}S^{2}_{ij}(W)V_{\xi_{i}}(\nabla v)\right) - \gamma v^{\gamma-1}S^{2}_{ij}(W)V_{\xi_{i}}(\nabla v)\partial_{j}v \qquad (C.10)$$

and

$$\begin{aligned} \operatorname{div} \left( v^{\gamma} S_{ij}^{2}(W) V_{\xi_{i}}(\nabla v) + \gamma(p-1) v^{\gamma-1} V(\nabla v) V_{\xi_{j}}(\nabla v) \right) \\ &= 2v^{\gamma} S^{2}(W) + \gamma(\gamma-1)(p-1) v^{\gamma-2} V(\nabla v) V_{\xi_{i}}(\nabla v) \partial_{i} v \\ &+ \gamma v^{\gamma-1} \left( (p-1) V(\nabla v) + V_{\xi_{i}}(\nabla v) \partial_{i} v \right) \operatorname{Tr} (W) \\ &+ \gamma v^{\gamma-1} \left( (p-1) V_{\xi_{i}}(\nabla v) V_{\xi_{j}}(\nabla v) \partial_{ij}^{2} v + V_{\xi_{j}\xi_{l}}(\nabla v) \partial_{li}^{2} v V_{\xi_{i}}(\nabla v) \partial_{j} v \right) . \end{aligned}$$

$$(C.11)$$

In particular, if

$$V(\xi) = \frac{|\xi|^p}{p} \quad \text{for } p > 1 \text{ and } \xi \in \mathbb{R}^n , \qquad (C.12)$$

then

$$2v^{\gamma}S^{2}(W) = \operatorname{div}\left(v^{\gamma}S_{ij}^{2}(W)V_{\xi_{i}}(\nabla v) + \gamma(p-1)v^{\gamma-1}V(\nabla v)\nabla_{\xi}V(\nabla v)\right)$$
  
-  $\gamma(\gamma-1)p(p-1)v^{\gamma-2}V^{2}(\nabla v)$   
-  $\gamma(2p-1)v^{\gamma-1}V(\nabla v)\Delta_{p}v$ . (C.13)

Observe that, in this particular case,

$$W(x) := \nabla [|\nabla v(x)|^{p-2} \nabla v(x)].$$

We are now ready to give the proof of Theorem C.1.

Proof of Theorem C.1. We consider the following auxiliary function:

$$v = u^{-\frac{p}{n-p}} . \tag{C.14}$$

An easy computation shows that v is a positive solution to

$$-\Delta_p v + \left(\frac{p}{n-p}\right)^{p-1} \frac{1}{v} + \frac{n(p-1)}{p} \frac{|\nabla v|^p}{v} = 0.$$
 (C.15)

By using Newton's inequality (C.8) we get

$$2v^{\gamma}S_2(W) \le \frac{n-1}{n}v^{\gamma}(\Delta_p v)^2, \qquad (C.16)$$

for any  $\gamma \in \mathbb{R}$ , where W is as in Lemma C.3. From (C.16) and from formula (C.13) (here we can perform an approximation argument as in Subsection 3.2.1 which is simpler in this case because, thanks to Lemma C.1, we know that  $u \in C^{2,\alpha}(\mathbb{R}^n \setminus B_{R_0})$ ) we obtain

$$\frac{n-1}{n}v^{\gamma}(\Delta_{p}v)^{2} \geq \operatorname{div}\left(v^{\gamma}S_{ij}^{2}(W)V_{\xi_{i}}(\nabla v) + \gamma(p-1)v^{\gamma-1}V(\nabla v)\nabla_{\xi}V(\nabla v)\right) - \gamma(\gamma-1)p(p-1)v^{\gamma-2}V^{2}(\nabla v) - \gamma(2p-1)v^{\gamma-1}V(\nabla v)\Delta_{p}v.$$
(C.17)

Observe that choosing  $\gamma = 1 - n$  from Lemma C.1 we get that

$$v^{\gamma} S_{ij}^{2}(W) V_{\xi_{i}}(\nabla v) + \gamma(p-1) v^{\gamma-1} V(\nabla v) \nabla_{\xi} V(\nabla v) = o(|x|^{n-1}), \qquad (C.18)$$

as  $|x| \to \infty$ . Hence, by integrating (C.17) in a ball of radius R, by using the divergence Theorem and by sending R to infinity (using (C.18) and recalling that  $|\nabla u|^{p-2}\nabla u \in W^{1,2}_{loc}(\mathbb{R}^n)$ , see Proposition 3.2) we obtain (here we can perform an approximation argument as in Subsection 3.1.3)

$$\frac{n-1}{n} \int_{\mathbb{R}^N} v^{\gamma} (\Delta_p v)^2 \, dx \ge -\gamma(\gamma-1)(p-1)p \int_{\mathbb{R}^N} v^{\gamma-2} V^2(\nabla v) \, dx -\gamma(2p-1) \int_{\mathbb{R}^N} v^{\gamma-1} V(\nabla v) \Delta_p v \, dx \,,$$
(C.19)

and by using (C.15) (recall that  $\gamma = 1 - n$ ) we get

$$-(n-1)\int_{\mathbb{R}^n} v^{-n-1}V(\nabla v) \, dx + \frac{n-1}{n} \left(\frac{p}{n-p}\right)^{p-1} \int_{\mathbb{R}^n} v^{-n-1} \, dx \ge 0 \,.$$
(C.20)

Now we show that the equality holds in (C.20), indeed by multiplying (C.15) by  $v^{-n}$  and integrating by parts we obtain

$$-(n-1)\int_{\mathbb{R}^n} v^{-n-1}V(\nabla v) \, dx + \frac{n-1}{n} \left(\frac{p}{n-p}\right)^{p-1} \int_{\mathbb{R}^n} v^{-n-1} \, dx = 0 \, .$$

This implies that all the previous inequalities are equalities and then the equality holds in Newton's inequality (C.16), that is (from Lemma C.2)

$$W(x) = \lambda(x) \operatorname{Id}$$
 for a.e.  $x \in \mathbb{R}^n$ , (C.21)

for some function  $\lambda : \mathbb{R}^n \to \mathbb{R}$ . To conclude the proof, we show that the function  $\lambda$  is constant. Since

$$\begin{aligned} \lambda(x) &= \frac{1}{n} \operatorname{Tr} \left( W(x) \right) \\ &= \frac{1}{n} \Delta_p v(x) \\ &= \frac{1}{n} \left( \frac{p}{n-p} \right)^{p-1} \frac{1}{v(x)} + \frac{p-1}{p} \frac{|\nabla v(x)|^p}{v(x)} \,, \end{aligned}$$

and since  $v \in C_{loc}^{1,\alpha}(\mathbb{R}^n)$ , we get that  $\lambda \in C_{loc}^{0,\alpha}(\mathbb{R}^n)$ . Moreover, elliptic regularity theory yields that  $v \in C_{loc}^{2,\alpha}(\mathbb{R}^n \setminus B_{R_0})$ , which implies that  $\lambda \in C_{loc}^{1,\alpha}(\mathbb{R}^n \setminus B_{R_0})$ . From (C.21) we have that

$$\partial_j(|\nabla v(x)|^{p-2}\partial_i v(x)) = \lambda(x)\delta_{ij} \quad \text{for } i, j \in \{1, \dots, n\},$$
(C.22)

which implies that  $|\nabla v|^{p-2}\nabla v \in C^{2,\theta}_{loc}(\mathbb{R}^n \setminus B_{R_0})$ . Then, given  $i \in \{1, \ldots, n\}$ , choosing  $j \neq i$  and using (C.22) we obtain

$$\partial_i \lambda(x) = \partial_i (\partial_j (|\nabla v(x)|^{p-2} \partial_j v(x))) = \partial_j (\partial_i (|\nabla v(x)|^{p-2} \partial_j v(x))) = 0,$$

for any  $x \in \mathbb{R}^n \setminus B_{R_0}$ , which implies that  $\lambda$  is constant on  $\mathbb{R}^n \setminus B_{R_0}$ . In order to deduce that  $\lambda$  is constant in the whole  $\mathbb{R}^n$ , we can argue as in Subsection 3.2.2. For this reason we omit the details and conclude that  $\lambda$  is constant. In particular, recalling (C.21)

$$\nabla[|\nabla v|^{p-2}\nabla v] = W = \lambda \mathrm{Id}$$
 in  $\mathbb{R}^n$ .

Hence  $|\nabla v(x)|^{p-2}\nabla v(x) = \lambda(x-x_0)$  for some  $x_0 \in \mathbb{R}^n$ ; this implies that

$$v(x) = c_1 + c_2 |x - x_0|^{\frac{p}{p-1}},$$

or equivalently (recalling (C.14))  $u(x) = \mathcal{U}_{\lambda,x_0}(x)$ . This completes the proof of Theorem C.1.
### Appendix D

## Serrin vs Alexandrov

In this appendix we show that the celebrated theorems by Alexandrov and Serrin (Theorem A and Theorem B in the Introduction) are in some sense equivalent. In particular, one can use Serrin's Theorem to prove Alexandrov's one and viceversa. The Appendix is divided in two sections: in the first one we show how to deduce Serrin's Theorem from Alexandrov's Theorem, while in the second one we show how to deduce Alexandrov's Theorem from Serrin's Theorem.

We recall that Serrin's result is the following:

let  $\Omega \subset \mathbb{R}^n$  be a bounded domain with boundary of class  $C^2$ . Then there exists a solution  $u \in C^2(\Omega) \cap C^1(\overline{\Omega})$  to

$$\begin{cases} \Delta u = -1 & \text{ in } \Omega \\ u = 0 & \text{ on } \partial \Omega \\ \partial_{\nu} u = c & \text{ on } \partial \Omega \end{cases}$$

if and only if  $\Omega$  is a ball.

While Alexandrov's result is the following:

the sphere is the only  $C^2$ -regular, connected, closed hypersurface embedded in the Euclidean space with constan mean curvature.

### D.1 Serrin implies Alexandrov

Let S be a connected,  $C^2$ -regular and closed hypersurface embedded in  $\mathbb{R}^n$  with constant mean curvature. Thanks to the embeddeness we may assume that  $S = \partial \Omega$ , where  $\Omega \subset \mathbb{R}^n$  is a bounded domain. We want to prove that  $\partial \Omega$  is a sphere. In order to do this, we consider the unique solution to the following Dirichlet problem

$$\begin{cases} \Delta u = 1 & \text{in } \Omega \\ u = 0 & \text{on } \partial \Omega . \end{cases}$$
(D.1)

Our goal is to assume that H, the mean curvature of S, is constant and to show that u satisfies

$$\partial_{\nu} u = c \quad \text{on } \partial\Omega.$$
 (D.2)

Then according to Serrin's result, applied to -u,  $\Omega$  must be a ball and we prove Alexandrov's Theorem.

To prove (D.2), we follow the approach in [194, Theorem 5] (see also [30, Appendix B]). Thanks to (I.15) and thanks to Newton's inequality (I.13) we get, for all  $x \in \Omega$ ,

$$\frac{n-1}{2n} = \frac{n-1}{2n} (\Delta u)^2 \ge \frac{1}{2} \operatorname{div}(S_{ij}^2(\nabla^2 u)\partial_i u) \,. \tag{D.3}$$

By integrating over  $\Omega$ , using the divergence theorem on the right hand side and using that  $\nu = \nabla u / |\nabla u|$  we obtain

$$\begin{aligned} \frac{n-1}{2n} |\Omega| &\geq \frac{1}{2} \int_{\partial\Omega} S_{ij}^2(\nabla^2 u) \partial_i u \nu_j \, d\sigma \\ &= \frac{1}{2} \int_{\partial\Omega} S_{ij}^2(\nabla^2 u) \partial_i u \frac{\partial_j u}{|\nabla u|} \, d\sigma \\ &= \frac{n-1}{2} H \int_{\partial\Omega} |\nabla u|^2 \, d\sigma \,, \end{aligned}$$

where we used the following formula (see e.g. [30, Formula 61])

$$S_{ij}^2(\nabla^2 u)\partial_i u\partial_j u = (n-1)H|\nabla u|^3,$$

and the fact the mean curvature H of  $\partial \Omega$  is constant. Hence we get

$$\int_{\partial\Omega} |\nabla u|^2 \, d\sigma \le \frac{|\Omega|}{nH} \,. \tag{D.4}$$

On the other hand, from Holder's inequality, we have

$$\left(\int_{\partial\Omega} |\nabla u| \, d\sigma\right)^2 \le |\partial\Omega| \int_{\partial\Omega} |\nabla u|^2 \, d\sigma \,. \tag{D.5}$$

Moreover, from the divergence theorem,

$$\Omega| = \int_{\Omega} \Delta u \, dx = \int_{\partial \Omega} |\nabla u| \, d\sigma \,. \tag{D.6}$$

Hence, from (D.5) and (D.6) we get

$$|\Omega|^2 \le |\partial \Omega| \int_{\partial \Omega} |\nabla u|^2 \, d\sigma$$
.

Recalling (D.4) we have proved that

$$|\Omega|^2 \le |\partial\Omega| \frac{|\Omega|}{nH}.$$
 (D.7)

Now, thanks to the *Minkowski's identity* (see e.g. [195])

$$\int_{\partial\Omega} Hx \cdot \nu \, d\sigma = \left| \partial\Omega \right|,\,$$

and the fact that, by assumption, the mean curvature H of  $\partial\Omega$  is constant we obtain that

$$H = \frac{|\partial \Omega|}{n|\Omega|} \,.$$

and hence the equality in (D.7) holds. This entails that equality holds in both Newton and Holder inequalities and hence  $|\nabla u|$  must be constant on  $\partial\Omega$  and from this fact we immediately obtain that (D.2) holds true and we conclude.

### D.2 Alexandrov implies Serrin

Let  $\Omega \subset \mathbb{R}^n$  be a bounded domain with boundary of class  $C^2$  and suppose that there exists a solution  $u \in C^2(\Omega) \cap C^1(\overline{\Omega})$  to

$$\begin{cases} \Delta u = -1 & \text{in } \Omega \\ u = 0 & \text{on } \partial \Omega \\ \partial_{\nu} u = c & \text{on } \partial \Omega . \end{cases}$$
(D.8)

If we show that the mean curvature H of  $\partial\Omega$  is constant, then according to Alexandrov's Theorem,  $\partial\Omega$  must be a sphere.

To prove that the mean curvature of  $\partial\Omega$  is constant, we argue as in [92, 215]. We use the already cited *P*-function introduced by Weinberger

$$P(x) = |\nabla u(x)|^2 + \frac{2}{n}u(x).$$
 (D.9)

We notice that P is subharmonic. Indeed

$$\Delta P = 2|\nabla^2 u|^2 + \frac{2}{n}\Delta u = 2\left(n|\nabla^2 u|^2 - \frac{1}{n}(\Delta u)^2\right),\,$$

and this is non-negative according to Cauchy-Schwarz inequality. Moreover, P is constant on  $\partial\Omega$ , hence from the strong maximum principle we obtain that either

$$P \equiv c^2 \quad \text{in } \Omega. \tag{D.10}$$

or

$$P < c^2 \quad \text{in } \Omega. \tag{D.11}$$

Arguing as in Weinberger's proof of Serrin's theorem (see Introduction to Part I) one can prove that (D.10) holds true. From (D.10) and (D.9) we get

$$|\nabla u(x)| = \sqrt{c^2 - 2u(x)} := g(u(x)),$$
 (D.12)

for all  $x \in \Omega$ . Obviously g is a function of class  $C^1$  in  $(0, \max u)$ . Moreover, since the function  $t \to \sqrt{t}$  is strictly monotone we have that  $\nabla u \equiv 0$  only where u attains its maximum on  $\Omega$ ; then the vector field  $\nu = -\nabla u/|\nabla u|$  is well-defined on the set  $\mathcal{U} := \{x \in \Omega : u(x) \in (0, \max u)\}.$ 

We observe that  $\partial_{\nu} u = -|\nabla u| = -g(u)$  and that

$$\Delta u = \partial_{\nu\nu}^2 u + (n-1)H\partial_{\nu}u \tag{D.13}$$

where H is the mean curvature of the level set of u. From (D.13) and (D.8) we get that H depends only on u, actually on g(u) and on  $\partial^2_{\nu\nu} u$ ; moreover we observe that, on the one hand

$$\partial_{\nu}(|\nabla u|^2) = 2\partial_{\nu}u\partial_{\nu\nu}^2 u\,,$$

and on the other hand, from (D.12),

$$\partial_{\nu}(|\nabla u|^2) = 2g(u)g'(u)\partial_{\nu}u\,,$$

 $\mathbf{SO}$ 

$$\partial_{\nu\nu}^2 u = g(u)g'(u). \tag{D.14}$$

Hence, from (D.13) and (D.14), we get

$$H = \frac{1 + g(u)g'(u)}{(n-1)g(u)} \quad \text{in } \mathcal{U},$$

and this identity says that the mean curvature of each level set of u at height between 0 and max u is constant. By Alexandrov's Theorem we deduce that the connected components of each level set must be spheres. Since  $\partial\Omega$  is connected this implies that  $\Omega$  must be simply connected (otherwise a particular level set would contain two nested spheres of equal radius, which is a contradiction). Therefore each level set consists of exactly one sphere, and because of (D.12) these spheres are concentric and  $\Omega$  is a ball.

# Part IV Bibliography

## Bibliography

- U. ABRESCH, H. ROSENBERG. A Hopf differential for constant mean curvature surfaces in S<sup>2</sup> × ℝ and H<sup>2</sup> × ℝ. Acta Math. 193 no. 2 (2004), 141–174.
- [2] A. AFTALION, J. BUSCA, W. REICHEL. Approximate radial symmetry for overdetermined boundary value problems. Adv. Diff. Eq. 4 no. 6 (1999), 907-932.
- [3] V. AGOSTINIANI, R. MAGNANINI. Symmetries in an overdetermined problem for the Green's function. Discrete Contin. Dyn. Syst. Ser. S 4 no. 4 (2011), 791–800.
- [4] G. ALESSANDRINI, R. MAGNANINI. Symmetry and non-symmetry for the ovedetermined Stekoff problem II. Nonlinear Problems in Applied Mathematics, 1-9, SIAM, Philadelphia, PA, 1996.
- [5] A. D. ALEXANDROV. Uniqueness theorems for surfaces in the large II. Vestnik Leningrad Univ. 12 no. 7 (1957), 15-44 (in Russian); English transl.: Amer. Math. Soc. Transl. 21 no. 2 (1962), 354-388.
- [6] A. D. ALEXANDROV. Uniqueness theorems for surfaces in the large V. Vestnik Leningrad Univ. 13 no. 19 (1958), 5–8. English translation: Amer. Math. Soc. Transl. 21 no. 2 (1962), 412–415.
- [7] A. D. ALEXANDROV. A characteristic property of spheres. Ann. Mat. Pura Appl. 58 (1962), 303–315.
- [8] F. J. ARNOLD. On the Alexandrov-Fenchel inequality and the stability of the sphere. Monatsh. Math. 155 (1993), 1–11.
- [9] D. G. ARONSON, L. A. CAFFARELLI. The initial trace of a solution of the porous medium equation. Trans. Amer. Math. Soc. 280 no. 1 (1983), 351-366.
- [10] T. AUBIN. Équations différentielles non linéaires et problème de Yamabe concernant la courbure scalaire. J. Math. Pures Appl. 55 (1976), 269–296.
- T. AUBIN. Espaces de Sobolev sur les variétés Riemanniennes. Bull. Sci. Math. 100 (1976), 149–173.
- [12] T. AUBIN. Problèmes isopérimétriques et espaces de Sobolev. J. Differential Geom. 11 no. 4 (1976), 573–598.

- T. AUBIN, Y. Y. LI. On the best Sobolev inequality. J. Math. Pures Appl. 78 (1999), 353–387.
- [14] B. AVELIN, T. KUUSI, G. MINGIONE. Nonlinear Calderón-Zygmund theory in the limiting case. Arch. Rational Mech. Anal. (2018), 227–663.
- [15] E. BAER, A. FIGALLI. Characterization of isoperimetric sets inside almost-convex cones. Discrete Contin. Dyn. Syst. 37 no. 1 (2017), 1–14.
- [16] D. BAKRY, T. COULHON, M. LEDOUX, L. SALOFF-COSTE. Sobolev inequalities in disguise. Indiana Univ. Math. J. 44 (1995), 1033–1074.
- [17] D. BAKRY, I. GENTIL, M. LEDOUX. "Analysis and Geometry of Markov Diffusion Operators." Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], 348. Springer, Cham, 2014.
- [18] D. BAKRY, M. LEDOUX, F. Y. WANG. Perturbations of functional inequalities using growth conditions. J. Math. Pures Appl. 87 (2007), 394– 407.
- [19] J. L. BARBOSA, M. DO CARMO. Stability of Hypersurfaces of constant mean curvature. Math. Zeit. 185 no. 3 (1984), 339–353.
- [20] J. L. BARBOSA, M. DO CARMO, M. ESCHENBURG. Stability of hypersurfaces of constant mean curvature in Riemannian manifolds. Math. Zeit. 197 no. 1 (1988), 123–138.
- [21] E. BAER, A. FIGALLI. Characterization of isoperimetric sets inside almost-convex cones. Discrete Contin. Dyn. Syst. 37 (2017), 1–14.
- [22] A. BENNETT. Symmetry in an Overdetermined Fourth Order Elliptic Boundary Value Problem. SIAM J. Math. Anal. 17 no. 6 (1986), 1354– 1358.
- [23] C. BENNETT, R. GULLIVER. Aleksandrov reflection and nonlinear evolution equations I. The n-sphere and n-ball. Calc. Var. Partial Differential Equations 4 no. 3 (1996), 249–264.
- [24] C. BENNETT, R. GULLIVER. Aleksandrov reflection and geometric evolution of hypersurfaces. Comm. Anal. Geom. 9 no.2 (2001), 261–280.
- [25] E. BERCHIO, D. GANGULY, G. GRILLO. Sharp Poincaré-Hardy and Poincaré-Rellich inequalities on the hyperbolic space. J. Funct. Anal. 272 (2017), 1661–1703.
- [26] H. BERESTYCKI, L. A. CAFFARELLI, L. NIRENBERG. Inequalities for second-order elliptic equations with applications to unbounded domains I. Duke Math. J. 81 no. 2 (1996), 467–494.
- [27] H. BERESTYCKI, L. NIRENBERG. On the method of moving planes and the sliding method. Bol. Soc. Bras. Mat. Vol. 22 No. 1 (1991), 1–37.
- [28] A. L. BESSE. "Einstein Manifolds." Reprint of the 1987 edition. Classics in Mathematics. Springer-Verlag, Berlin, 2008.

- [29] C. BIANCHINI, G. CIRAOLO. Wulff shape characterizations in overdetermined anisotropic elliptic problems. Comm. Partial Differential Equations Vol. 43 (2018), 790–820.
- [30] C. BIANCHINI, G. CIRAOLO, P. SALANI. An overdetermined problem for the anisotropic capacity. Calc. Var. Partial Differential Equations 55 no. 4 (2016), Art. 84, 24 pp.
- [31] C. BIANCHINI, A. HENROT, P. SALANI. An overdetermined problem with non constant boundary condition. Interfaces Free Bound. 16 no. 2 (2014), 215-241.
- [32] M. F. BIDAUT-VERON, L. VERON. Nonlinear elliptic equations on compact Riemannian manifolds and asymptotics of Emden equations. Invent. math. 106 (1991), 489–539.
- [33] I. BIRINDELLI, F. DEMENGEL. Overdetermined problems for some fully non linear operators. Comm. Partial Differential Equations 38 no. 4 (2013), 608-628.
- [34] M. BONFORTE, G. GRILLO. Asymptotics of the porous media equation via Sobolev inequalities. J. Funct. Anal. 225 (2005), 33-62.
- [35] M. BONFORTE, G. GRILLO, J. L. VÁZQUEZ. Fast diffusion flow on manifolds of nonpositive curvature. J. Evol. Equ. 8 (2008), 99–128.
- [36] M. BONFORTE, R. G. IAGAR, J. L. VÁZQUEZ. Local smoothing effects, positivity, and Harnack inequalities for the fast p-Laplacian equation. Adv. Math. 224 no. 5 (2010), 2151–2215.
- [37] M. BONFORTE, J. L. VÁZQUEZ. Global positivity estimates and Harnack inequalities for the fast diffusion equation. J. Funct. Anal. 240 no. 2 (2006), 399-428.
- [38] M. BONFORTE, J. L. VÁZQUEZ. Positivity, local smoothing, and Harnack inequalities for very fast diffusion equations. Adv. Math. 223 no. 2 (2010), 529-578.
- [39] B. BRANDOLINI, C. NITSCH, P. SALANI, C. TROMBETTI. On the stability of the Serrin problem. J. Differential Equations 245 no. 6 (2008), 1566– 1583.
- [40] B. BRANDOLINI, C. NITSCH, P. SALANI, C. TROMBETTI. Serrin-type overdetermined problems: an alternative proof. Arch. Ration. Mech. Anal. 190 no. 2 (2008), 267–280.
- [41] S. BRENDLE. Constant mean curvature surfaces in warped product manifolds. Publ. Math. Inst. Hautes Études Sci. 117 (2013), 247–269.
- [42] F. BROCK, A. HENROT. A symmetry result for an overdetermined elliptic problem using continuous rearrangement and domain derivative. Rend. Circ. Mat. Palermo (2) 51 no. 3 (2002), 375–390.
- [43] V.I. BURENKOV. "Sobolev Spaces on Domains." Teubner-texte zur Mathematik, Stuttgart (1998).

- [44] G. BUTTAZZO, B. KAWOHL. Overdetermined boundary value problems for the ∞-Laplacian. Int. Math. Res. Not. IMRN no. 2 (2011), 237–247.
- [45] X. CABRÉ, M. FALL, J. SOLA-MORALES, T. WETH. Curves and surfaces with constant nonlocal mean curvature: meeting Alexandrov and Delaunay. To appear in J. Reine Angew. Math. (Crelle's Journal). arXiv:1503.00469.
- [46] X. CABRÉ, X. ROS-OTON, J. SERRA. Sharp isoperimetric inequalities via the ABP method. J. Eur. Math. Soc. (JEMS) 18 (2016) no. 12, 2971–2998.
- [47] L. A. CAFFARELLI, N. GAROFALO, AND F. SEGÀLA. A gradient bound for entire solutions of quasi-linear equations and its consequences. Comm. Pure Appl. Math. 47 (1994), 1457–1473.
- [48] L. A. CAFFARELLI, B. GIDAS, J. SPRUCK. Asymptotic symmetry and local behavior of semilinear elliptic equations with critical Sobolev growth. Comm. Pure Appl. Math. 42 no. 3 (1989), 271–297.
- [49] H. D. CAO, G. CATINO, Q. CHEN, C. MANTEGAZZA, L. MAZZIERI. Bach-flat gradient steady Ricci solitons. Calc. Var. Partial Differential Equations 49 (2014), 125–138.
- [50] G. CARRON. Inégalités de Hardy sur les variétés riemanniennes noncompactes. J. Math. Pures Appl. 76 (1997), 883–891.
- [51] G. CARRON. Inégalité de Sobolev et volume asymptotique. Ann. Fac. Sci. Toulouse Math. 21 (2012), 151–172.
- [52] S. CHENG, S. YAU. Hypersurfaces with constant scalar curvature. Math. Ann. 225 no. 3 (1977), 195-204.
- [53] A. CIANCHI, V. MAZ'YA. Global lipschitz regularity for a class of quasilinear elliptic equations. Communications in Partial Differential Equations. 36 no. 1 (2010), 100–133.
- [54] A. CIANCHI, V. MAZ'YA. Second-Order Two-Sided Estimates in Nonlinear Elliptic Problems. Arch. Rational Mech. Anal. 229 (2018), 569–599.
- [55] A. CIANCHI, P. SALANI. Overdetermined anisotropic elliptic problems. Math. Ann. 345 no. 4 (2009), 859–881.
- [56] G. CIRAOLO, A. FIGALLI, F. MAGGI, M. NOVAGA. Rigidity and sharp stability estimates for hypersurfaces with constant and almost-constant nonlocal mean curvature. J. Reine Angew. Math. (Crelle's Journal) 741 (2018), 275-294.
- [57] G. CIRAOLO, A. FIGALLI, A. RONCORONI. Symmetry results for critical anisotropic p-Laplacian equations in convex cones. Preprint. arXiv:1906.00622
- [58] G. CIRAOLO, F. MAGGI. On the shape of compact hypersurfaces with almost constant mean curvature. Comm. Pure Appl. Math. **70** (2017), 665-716.

- [59] G. CIRAOLO, R. MAGNANINI, S. SAKAGUCHI. Symmetry of minimizers with a level surface parallel to the boundary. J. Eur. Math. Soc. 17 (2015), 2789–2804.
- [60] G. CIRAOLO, R. MAGNANINI, S. SAKAGUCHI. Solutions of elliptic equations with a level surface parallel to the boundary: Stability of the radial configuration. J. Anal. Math. 128 (2016), 337–353.
- [61] G. CIRAOLO, R. MAGNANINI, V. VESPRI. Hölder stability for Serrin's overdetermined problem. Ann. Mat. Pura Appl. 195 (2016), 1333–1345.
- [62] G. CIRAOLO, R. MAGNANINI, V. VESPRI. Symmetry and linear stability in Serrin's overdetermined problem via the stability of the parallel surface problem. Preprint. arXiv:1410.7791.
- [63] G. CIRAOLO, A. RONCORONI. Serrin's type overdetermined problems in convex cones. To appear in Calc. Var. Partial Differential Equations. arXiv:1806.08553.
- [64] G. CIRAOLO, A. RONCORONI. The method of moving planes: a quantitative approach. Bruno Pini Mathematical Analysis Seminar (2018), 41–77.
- [65] G. CIRAOLO, A. RONCORONI, L. VEZZONI. Quantitative stability for hypersurfaces with almost constant curvature in space forms. Preprint. arXiv:1812.00775.
- [66] G. CIRAOLO, L. VEZZONI. A rigidity problem on the round sphere. Commun. Cont. Math. 19 (2016), 1750001.
- [67] G. CIRAOLO, L. VEZZONI. A sharp quantitative version of Alexandrov's theorem via the method of moving planes. J. Eur. Math. Soc. (JEMS) 20 no. 2 (2018), 261–299.
- [68] G. CIRAOLO, L. VEZZONI. On Serrin's overdetermined problem in space forms. Manuscripta Math. 159 (2019), 445–452.
- [69] G. CIRAOLO, L. VEZZONI. Quantitative stability for hypersurfaces with almost constant mean curvature in the hyperbolic space. To appear in Indiana Univ. Math. J. arXiv:1611.02095.
- [70] M. CHOULLI, A. HENROT. Use of the domain derivative to prove symmetry results in partial differential equations. Math. Nachr. 192 (1998), 91-103.
- [71] M. CLAPP, F. PACELLA. Existence of nonradial positive and nodal solutions to a critical Neumann problem in a cone. Preprint. arXiv:1906.09301.
- [72] D. CORDERO-ERAUSQUIN, B. NAZARET, C. VILLANI. A masstransportation approach to sharp Sobolev and Gagliardo-Nirenberg inequalities. Adv. Math. 182 no. 2 (2004), 307–332.
- [73] G. CRASTA. Existence, uniqueness and qualitative properties of minima to radilly symmetric non-coercive non-convex variational problem. Math. Z. 235 (2000) 569–589.

- [74] G. CRASTA, I. FRAGALÀ. A symmetry problems for the infinity Laplacian. Int. Math. Res. Not. IMRN no. 18 (2015), 8411–8436.
- [75] F. DA LIO, B. SIRAKOV. Symmetry results for viscosity solutions of fully nonlinear uniformly elliptic equations. J. Eur. Math. Soc. (JEMS) 9 no. 2 (2007), 317–330.
- [76] L. DAMASCELLI, S. MERCHÁN, L. MONTORO, B. SCIUNZI. Radial symmetry and applications for a problem involving the  $-\Delta_p(\cdot)$  operator and critical nonlinearity in  $\mathbb{R}^n$ . Adv. Math. **265** no. 10 (2014), 313–335.
- [77] E.- B. DAVIES. "Heat Kernels and Spectral Theory." Cambridge Tracts in Mathematics, 92. Cambridge University Press, Cambridge, 1990.
- [78] E. DE GIORGI. Sulla proprietà isoperimetrica dell'ipersfera, nella classe degli insiemi aventi frontiera orientata di misura finita. Atti Accad. Naz. Lincei. Mem. Cl. Sci. Fis. Mat. Nat. Sez. I 5 no. 8 (1958), 33–44.
- [79] M. DEL PINO, M. MUSSO, F. PACARD, A. PISTOIA. Large energy entire solutions for the Yamabe equation. Journal of Differential Equations 251 (2011), 2568–2597.
- [80] M. DEL PINO, M. MUSSO, F. PACARD, A. PISTOIA. Torus action on S<sup>n</sup> and sign-changing solutions for conformally invariant equations. Annali della Scuola Normale Superiore di Pisa - Classe di Scienze 12 (2013), 209–237.
- [81] C. DELAUNAY. Sur la surface de révolution dont la courbure moyenne est constante. J. Math. Pures. Appl. 6 (1841), 309–320.
- [82] M. DELGADINO, F. MAGGI. Alexandrov's Theorem revisited. Preprint. arXiv:1711.07690v2.
- [83] M. DELGADINO, F. MAGGI, C. MIHAILA, R. NEUMAYER. Bubbling with L<sup>2</sup>-almost constant mean curvature and an Alexandrov-type theorem for crystals. Arch. Rat. Mech. Anal. 230 no. 3 (2018), 1131–1177.
- [84] W. DING. On a conformally invariant elliptic equation on  $\mathbb{R}^n$ . Comm. Math. Phys. **107** (1986), pp. 331-335.
- [85] E. DIBENEDETTO.  $C^{1+\alpha}$  local regularity of weak solutions of degenerate elliptic equations. Nonlinear Anal. 7 no. 8 (1983), 827–850.
- [86] J. DOLBEAULT, M. J. ESTEBAN, M. LOSS. Rigidity versus symmetry breaking via nonlinear flows on cylinders and Euclidean spaces. Invent. Math. 206 (2016), 397-440.
- [87] J. DOLBEAULT, M. J. ESTEBAN, M. LOSS, M. MURATORI. Symmetry for extremal functions in subcritical Caffarelli-Kohn-Nirenberg inequalities. C. R. Math. Acad. Sci. Paris 355 (2017), 133–154.
- [88] K. ECKER. "Regularity Theory for Mean Curvature Flow." Progress in Nonlinear Differential Equations and Their Applications, Birkhäuser Basel, 2014.

- [89] L. ESPOSITO, G. MINGIONE. Some remarks on the regularity of weak solutions of degenerate elliptic systems. Rev. Mat. Complutense 11 (1998), 203-219.
- [90] M. M. FALL, S. JAROHS. Overdetermined problems with fractional Laplacian. ESAIM Control Optim. Calc. Var. 21 no. 4 (2015), 924–938.
- [91] M. M. FALL, I. MINLEND, T. WETH. Serrin's overdetermined problem on the sphere. Calc. Var. Partial Differential Equations 57 no. 1 Art. 3 (2018), 24 pp.
- [92] A. FARINA, B. KAWOHL. Remarks on an overdetermined boundary value problem. Calc. Var. Partial Differential Equations 31 no. 3 (2008), 351– 357.
- [93] A. FARINA, E. VALDINOCI. A pointwise gradient estimate in possibly unbounded domains with nonnegative mean curvature. Adv. Math. 225 no. 5 (2010), 2808–2827.
- [94] A. FARINA, E. VALDINOCI. Partially and globally overdetermined problems of elliptic type. Adv. Nonlinear Anal. 1 no. 1 (2012), 27–45.
- [95] A. FARINA, E. VALDINOCI. On partially and globally overdetermined problems of elliptic type. Amer. J. Math. 135 (2013), 1699–1726.
- [96] W. M. FELDMAN. Stability of Serrin's problem and dynamic stability of a model for contact angle motion. SIAM J. Math. Anal. 50 no. 3 (2018), 3303–3326.
- [97] A. FIGALLI, E. INDREI. A sharp stability result for the relative isoperimetric inequality inside convex cones. J. Geom. Anal. 23 no. 2 (2013), 938-969.
- [98] A. FIGALLI, F. MAGGI, A. PRATELLI. A mass transportation approach to quantitative isoperimetric inequalities. Invent. Math. 182 no. 1 (2010), 167-211.
- [99] I. FONSECA, N. FUSCO. Regularity results for anisotropic image segmentation models. Ann. Scuola Norm. Sup. Pisa Cl. Sci. (IV) 24 (1997), 463– 499.
- [100] R. L. FOSDICK, J. SERRIN. Rectilinear steady flow of simple fluids. Proc. Roy. Soc. (London) Ser. A 332 (1973), 311–333.
- [101] A. R. FOTACHE, M. MURATORI. Smoothing effects for the filtration equation with different powers. J. Differential Equations 263 (2017), 3291– 3326.
- [102] I. FRAGALÀ, F. GAZZOLA. Partially overdetermined elliptic boundary value problems. J. Differential Equations 245 no. 5 (2008), 1299–1322.
- [103] I. FRAGALÀ, F. GAZZOLA, B. KAWOHL. Overdetermined problems with possibly degenerate ellipticity, a geometric approach. Math. Z. 254 no. 1 (2006) 117–132.

- [104] N. FUSCO, F. MAGGI, A. PRATELLI. The sharp quantitative isoperimetric inequality. Ann. of Math. (2) 168 no. 3 (2008), 941–980.
- [105] E. GAGLIARDO. Proprtietà di alcune classi di funzioni di più variabili. Ricerche Mat. 7 (1958), 102–137.
- [106] N. GAROFALO, J. L. LEWIS. A symmetry result related to some overdetermined boundary value problems. Amer. J. Math. 111 no. 1 (1989), 9–33.
- [107] N. GAROFALO, E. SARTORI. Symmetry in exterior boundary value problems for quasilinear elliptic equations via blow-up and a priori estimates. Adv. Differential Equations 4 (1999), 137–161.
- [108] B. GIDAS. Symmetry properties and isolated singularities of positive solutions of nonlinear elliptic equations. Nonlinear partial differential equations in engineering and applied science (Proc. Conf., Univ. Rhode Island, Kingston, R.I., 1979), pp. 255–273, Lecture Notes in Pure and Appl. Math., 54, Dekker, New York, 1980.
- [109] B. GIDAS, W. M. NI, L. NIRENBERG. Symmetry and related properties via the maximum principle. Comm. Math. Phys. 68, no. 3 (1979), 209-243.
- [110] B. GIDAS, W. M. NI, L. NIRENBERG. Symmetry of positive solutions of nonlinear elliptic equations in R<sup>n</sup>. Mathematical analysis and applications, Part A, 369–402, Adv. in Math. Suppl. Stud. 7a, Academic Press, New York-London, 1981.
- [111] B. GIDAS J. SPRUCK. Global and local behavior of positive solutions of nonlinear elliptic equations. Comm. Pure Appl. Math. 34 no. 4 (1981), 525-598.
- [112] D. GILBARG, N. S. TRUDINGER. "Elliptic partial differential equations of second order." Springer-Verlag, Berlin- New York, 1977.
- [113] L. GRAFAKOS. "Classical Fourier Analysis, 2nd ed." Graduate Texts in Mathematics, vol. 249, Springer, New York, 2008.
- [114] R. E. GREENE, H. WU. "Function Theory on Manifolds which Possess a Pole." Lecture Notes in Mathematics, 699, Springer, Berlin, 1979.
- [115] A. GRIGOR'YAN. Analytic and geometric background of recurrence and non-explosion of the Brownian motion on Riemannian manifolds, Bull. Amer. Math. Soc. 36 (1999), 135–249.
- [116] A. GRIGOR'YAN, L. SALOFF-COSTE. Surgery of the Faber-Krahn inequality and applications to heat kernel bounds. Nonlinear Anal. 131 (2016), 243-272.
- [117] G. GRILLO, M. MURATORI. Sharp short and long time L<sup>∞</sup> bounds for solutions to porous media equations with homogeneous Neumann boundary conditions. J. Differential Equations 254 (2013), 2261–2288.
- [118] G. GRILLO, M. MURATORI. Smoothing effects for the porous medium equation on Cartan-Hadamard manifolds. Nonlinear Anal. 131 (2016), 346-362.

- [119] G. GRILLO, M. MURATORI, M. M. PORZIO. Porous media equations with two weights: smoothing and decay properties of energy solutions via Poincaré inequalities. Discrete Contin. Dyn. Syst. 33 (2013), 3599-3640.
- [120] G. GRILLO, M. MURATORI, J. L. VÁZQUEZ. The porous medium equation on Riemannian manifolds with negative curvature. The large-time behaviour. Adv. Math. 314 (2017), 328–377.
- [121] M. GROMOV. Stability and pinching. In: Geometry Seminars. Sessions on Topology and Geometry of Manifolds (Bologna, 1990), Univ. Stud. Bologna, Bologna, (1992) 55–97.
- [122] M. GROSSI, F. PACELLA. Positive solutions of nonlinear elliptic equations with critical Sobolev exponent and mixed boundary conditions. Proc. Roy. Soc. Edinburgh Sect. A 116 (1990), 23–43.
- [123] M. GUEDDA, L. VÉRON. Local and global properties of solutions of quasilinear elliptic equations. J. Differential Equations 76 no. 1 (1988), 159–189.
- [124] G. H. HARDY, J. E. LITTLEWOOD, G. PÓLYA. "Inequalities." Cambridge Mathematical Library, Cambridge University Press, Cambridge, 1988. Reprint of the 1952 edition.
- [125] P. HARTMAN. On complete hypersurfaces of non negative sectional curvatures and constant m'th mean curvature. Transactions of the American Mathematical Society 245 (1978), 363–374.
- [126] L. HAUSWIRTH, F. HÉLEIN, F. PACARD. On an overdetermined elliptic problem. Pacific J. Math. 250 no. 2 (2011), 319–334.
- [127] Y. J. HE, H. Z. LI. Integral formula of Minkowski type and new characterization of the Wulff shape. Acta Math. Sin. 24 no. 4 (2008), 697–704.
- [128] Y. HE, H. LI, H. MA, J. GE. Compact embedded hypersurfaces with constant higher order anisotropic mean curvatures. Indiana Univ. Math. J. 58 no. 2 (2009), 853–868.
- [129] E. HEBEY. "Nonlinear Analysis on Manifolds: Sobolev Spaces and Inequalities." Courant Lecture Notes in Mathematics, 5. New York University, Courant Institute of Mathematical Sciences, New York; American Mathematical Society, Providence, RI, 1999.
- [130] E. HEBEY. Optimal Sobolev inequalities on complete Riemannian manifolds with Ricci curvature bounded below and positive injectivity radius. Amer. J. Math. 118 (1996), 291–300.
- [131] E. HEBEY. "Sobolev Spaces on Riemannian Manifolds." Lecture Notes in Mathematics, 1635. Springer-Verlag, Berlin, 1996.
- [132] E. HEBEY, M. VAUGON. The best constant problem in the Sobolev embedding theorem for complete Riemannian manifolds. Duke Math. J. 79 (1995), 235-279.
- [133] A. HENROT, G. PHILIPPIN. On a class of overdetermined eigenvalue problems. Math. Methods Appl. Sci. 20 no. 11 (1997), 905–914.

- [134] A. HENROT, G. PHILIPPIN. Some overdetermined boundary value problems with elliptical free boundaries. SIAM J. Math. Anal. 29 no. 2 (1998), 309-320.
- [135] A. HENROT, G. PHILIPPIN, H. PRÉBET. Overdetermined problems on ring shaped domains. Adv. Math. Sci. Appl. 9 no. 2 (1999), 737-747.
- [136] H. HOPF. "Differential Geometry in the Large." Lecture Notes in Mathematics 1000 (1989).
- [137] H. HOPF. Zum Clifford-Kleinschen Raumproblem. Math. Ann. 95 (1926), 313-339.
- [138] W. Y. HSIANG. On generalization of theorems of A. D. Alexandrov and C. Delaunay on hypersurfaces of constant mean curvature. Duke Math. J. 49 no. 3 (1982), 485-496.
- [139] W. Y. HSIANG, Z.-H. TENG, W. C. YU. New examples of constant mean curvature immersions of (2k - 1)-spheres into Euclidean 2k-space. Ann. of Math. (2) 117 no. 3 (1983), 609–625.
- [140] W. Y. HSIANG, W. YU. A generalization of a theorem of Delaunay. J. Differential Geom. 16 (1981), 161–177.
- [141] C. C. HSIUNG. Some integral formulas for closed hypersurfaces. Math. Scand. 2 (1954), 286–294.
- [142] H. KARCHER. Riemannian center of mass and mollifier smoothing. Comm. Pure Appl. Math. 30 (1977), 509–541.
- [143] W. KILLING. Ueber die Clifford-Klein'schen Raumformen. Math. Ann. 39 (1891), 257–278.
- [144] P. KOHLMANN. Curvature measures and stability. J. Geom. 68 (2000), 142–154.
- [145] N. J. KOREVAAR. Sphere theorems via Alexandrov for constant Weingarten curvature hypersurfaces - Appendix to a note of A. Ros. J. Diff. Geom. 27 (1988), 221–223.
- [146] N. J. KOREVAAR, R. KUSNER, B. SOLOMON. The structure of complete embedded surfaces with constant mean curvature. J. Differential Geom. 30 no. 2 (1989), 465–503.
- [147] D. KOUTROUFIOTIS. Ovaloids which are almost spheres. Comm. Pure Appl. Math. 24 (1971), 289–300.
- [148] B. KRUMMEL, F. MAGGI. Isoperimetry with upper mean curvature bounds and sharp stability estimates. Calc. Var. Partial Differential Equations 56 no. 2 Art. 53 (2017), 43 pp.
- [149] A. KUFNER, B. OPIC. "Hardy-type Inequalities." Pitman Research Notes in Mathematics Series, 219. Longman Scientific & Technical, Harlow, 1990.
- [150] S. KUMARESAN, J. PRAJAPAT. Serrin's result for hyperbolic space and sphere. Duke Math. J. 91 no. 1 (1998), 17–28.

- [151] U. LANG. Diameter bounds for convex surfaces with pinched mean curvature. Manuscripta Math. 86 (1995), 15-22.
- [152] M. LEDOUX. On manifolds with non-negative Ricci curvature and Sobolev inequalities. Comm. Anal. Geom. 7 (1999), 347–353.
- [153] M. LEDOUX. Spectral gap, logarithmic Sobolev constant, and geometric bounds. Surv. Differ. Geom. 9, 219–240, Int. Press, Somerville, MA, 2004.
- [154] J. M. LEE, T. H. PARKER. The Yamabe problem. Bull. Amer. Math. Soc. 17 (1987), 37–91.
- [155] Y. Y. LI. Harnack type inequality: the method of moving planes. Comm. Math. Phys. 200 no. 2 (1999), 421–444.
- [156] J. LI, C. XIA. An integral formula and its applications on sub-static manifolds. J. Differ. Geom. (2018).
- [157] G. M. LIEBERMAN. Boundary regularity for solutions of degenerate elliptic equations. Nonlinear Anal. 12 no. 11 (1988), 1203–1219.
- [158] H. LIEBMANN. Eine neue Eigenschaft der Kugel. Nachr. Kgl. Ges. Wiss. Göttingen, Math-Phys. Klasse (1899), 44–55.
- [159] P. L. LIONS, F. PACELLA. Isoperimetric inequalities for convex cones. Proc. Amer. Math. Soc. 109 no. 2 (1990), 477–485.
- [160] P. L. LIONS, F. PACELLA, M. TRICARICO. Best constants in Sobolev inequalities for functions vanishing on some part of the boundary and related questions. Indiana Univ. Math. J. 37 no. 2 (1988), 301–324.
- [161] R. LÒPEZ. "Constant Mean Curvature Surfaces with Boundary." Springer-Verlag, Berlin, 2013.
- [162] Z. LU, D. ZHOU. On the essential spectrum of complete non-compact manifolds. J. Funct. Anal. 260 (2011), 3283–3298.
- [163] F. MAGGI. "Sets of Finite Perimeter and Geometric Variational Problems: an introduction to Geometric Measure Theory." Cambridge University Press, Cambridge, 2012.
- [164] R. MAGNANINI. Alexandrov, Serrin, Weinberger, Reilly: symmetry and stability by integral identities. Bruno Pini Mathematical Seminar (2017), 121-141.
- [165] R. MAGNANINI, G. POGGESI. On the stability for Alexandrov's Soap Bubble Theorem. To appear in Jour. Anal. Math. arXiv:1610.07036.
- [166] R. MAGNANINI, G. POGGESI. Serrin's problem and Alexandrov's Soap Bubble Theorem: stability via integral identities. To appear in Indiana Univ. Math. Jour. arXiv:1708.07392.
- [167] H. P. MCKEAN. An upper bound to the spectrum of  $\Delta$  on a manifold of negative curvature. J. Differential Geometry 4 (1970), 359–366.

- [168] W. MEEKS III. The topology and geometry of embedded surfaces of constant mean curvature. J. Differential Geom. 27 no. 3 (1988), 539–552.
- [169] W. H. MEEKS III, P. MIRA, J. PÉREZ, A. ROS. Constant mean curvature spheres in homogeneous three-manifolds. Preprint. arXiv:1706.09394.
- [170] W. H. MEEKS III, P. MIRA, J. PÉREZ, A. ROS. Constant mean curvature spheres in homogeneous three-spheres. Preprint. arXiv:1308.2612.
- [171] R. MOLZON. Symmetry and overdetermined boundary value problems. Forum Math. 3 no. 2 (1991), 143–156.
- [172] S. MONTIEL, A. ROS. Compact hypersurfaces: the Alexandrov theorem for higher order mean curvatures. Pitman Monographs and Surveys in Pure and Applied Mathematics 52 (1991), 279–296.
- [173] J. D. MOORE. Almost spherical convex hypersurfaces. Trans. Amer. Math. Soc. 180 (1973), 347–358.
- [174] M. MURATORI, A. RONCORONI. Sobolev-type inequalities on Cartan-Hadamard manifolds under curvature bounds. Preprint. arXiv:1805.02726.
- [175] V. H. NGUYEN. The sharp Poincaré-Sobolev type inequalities in the hyperbolic space H<sup>n</sup>. J. Math. Anal. Appl. 462 (2018), 1570–1584.
- [176] C. NITSCH, C. TROMBETTI. The classical overdetermined Serrin problem. Complex Var. Elliptic Equ. 63 no. 7-8 (2018), 1107–1122.
- [177] L. NIRENBERG. On elliptic partial differential equations. Ann. Scuola Normale Sup. Pisa 13 (1959), 116–162.
- [178] M. OBATA. The conjecture on conformal transformations of Riemannian manifolds.. J. Diff. Geom. 6 (1971), 247–258.
- [179] R. OSSERMAN. The isoperimetric inequality. Bull. Amer. Math. Soc. 84 (1978), no. 6, 1182–1238.
- [180] F. PACELLA, G. TRALLI. Overdetermined problems and constant mean curvature surfaces in cones. Preprint. arXiv:1802.03197.
- [181] L. E. PAYNE. Some remarks on maximum principles. J. Anal. Math. 30 (1976), 421–433.
- [182] L. E. PAYNE, P. SCHAEFER. Duality theorems in some overdetermined boundary value problems. Math. Methods Appl. Sci. 11 no. 6 (1989), 805– 819.
- [183] I. PERAL. "Multiplicity of solutions for the *p*-Laplacian." Lecture Notes at the Second School on Nonlinear Functional Analysis and Applications to Differential Equations, ICTP, Trieste, 1997.
- [184] P. PETERSEN. "Riemannian Geometry." Grad. Texts in Math. 71. Springer-Verlag, New York, 1998.

- [185] S. PIGOLA, G. VERONELLI. Lower volume estimates and Sobolev inequalities. Proc. Amer. Math. Soc. 138 (2010), 4479–4486.
- [186] S. I. POHŎZAEV. On the eigenfunctions of the equation  $\Delta u + \lambda f(u) = 0$ . Dokl. Akad. Nauk SSSR **165** (1965), 36–39.
- [187] P. POLÁCIK, P. QUITTNER, P. SOUPLET. Singularity and decay estimates in superlinear problems via Liouville-type theorems, I: elliptic equations and systems. Duke Math. J. 139 no. 3 (2007), 555–579.
- [188] J. PRAJAPAT. Serrin's result for domains with a corner or cusp. Duke Math. J. 91 no. 2 (1998), 29–31.
- [189] G. QIU, C. XIA. A generalization of Reilly's formula and its applications to a new Heintze-Karcher type inequality. Int. Math. Res. Not. IMRN no. 17 (2015), 7608–7619.
- [190] G. QIU, C. XIA. Overdetermined boundary value problems in S<sup>n</sup>. J. Math. Study 50 no. 2 (2017), 165–173.
- [191] A. G. RAMM. Symmetry problem. Proc. Amer. Math. Soc. 141 no. 2 (2013), 515-521.
- [192] W. REICHEL. Radial symmetry for elliptic boundary-value problems on exterior domains. Arch. Rational Mech. Anal. 137 no. 4 (1997), 381–394.
- [193] W. REICHEL. Radial symmetry for an electrostatic, a capillarity and some fully nonlinear overdetermined problems on exterior domains. Z. Anal. Anwendungen 15 no. 3 (1996), 619–635.
- [194] R. REILLY. Applications of the Hessian operator in a Riemannian manifold. Indiana Univ. Math. J. 26 (1977), 459–472.
- [195] R. REILLY. Mean curvature, the Laplacian, and soap bubbles. Amer. Math. Monthly. 89 (1982), 180–188.
- [196] A. RONCORONI. A Serrin-type symmetry result on model manifolds: an extension of the Weinberger argument. C. R. Math. Acad. Sci. Paris 356 no. 6 (2018), 648-656.
- [197] A. Ros. Compact hypersurfaces with constant higher order mean curvatures. Rev. Math. Iber. 3 (1987), 447–453.
- [198] A. Ros. Compact hypersurfaces with constant scalar curvature and a congruence theorem. J. Diff. Geom. 27 (1988), 215–220.
- [199] H. ROSENBERG. Hypersurfaces of constant curvature in space forms. Bull. Sci. Math. 117 no. 2 (1993), 211–239.
- [200] R. SCHENIDER. A stability estimate for the ALexandrov-Fenchel inequality, with an application to mean curvature. Manuscripta Math. 69 (1990), 291–300.
- [201] R. SCHOEN. Conformal deformation of a Riemannian metric to constant scalar curvature. J. Differential Geom. 20 (1984) no. 2, 479–495.

- [202] R. SCHOEN. Uniqueness, symmetry, and embeddedness of minimal surfaces. J. Differential Geom. 18 no. 4 (1983), 791–809.
- [203] B. SCIUNZI. Classification of positive  $D^{1,p}(\mathbb{R}^n)$ -solutions to the critical p-Laplace equation in  $\mathbb{R}^n$ . Advances in Mathematics **291** (2016), 12–23.
- [204] J. SERRIN. A symmetry problem in potential theory. Arch. Rational Mech. Anal. 43 (1971), 304–318.
- [205] J. SERRIN. Local behaviour of solutions of quasilinear equations. Acta Math. 113 (1965), 219-240.
- [206] J. SERRIN, H. ZOU. Cauchy-Liouville and universal boundedness theorems for quasilinear elliptic equations and inequalities. Acta Math. 189 no. 1 (2002), 79–142.
- [207] L. SILVESTRE, B. SIRAKOV. Overdetermined problems for fully nonlinear elliptic equations. Calc. Var. Partial Differential Equation 54 no. 1 (2015), 989–1007.
- [208] L. SIMON. "Lectures on geometric measure theory." Proceedings of the Centre for Mathematical Analysis, Australian National University, 3. Australian National University, Centre for Mathematical Analysis, Canberra, 1983. See also https://web.stanford.edu/class/math285/ts-gmt.pdf.
- [209] B. SIRAKOV. Symmetry for exterior elliptic problems and two conjectures in potential theory. Ann. Inst. H. Poincaré Anal. Non Linéaire 18 no. 2 (2001), 135–156.
- [210] S. L. SOBOLEV. Sur un théorème d'analyse fonctionelle. Mat. Sb. (N.S.) 47 (1938), 471–496.
- [211] R. P. SPERB. "Maximum Principles and Their Applications." Mathematics in Science and Engineering, vol. 157, Academic Press Inc. [Harcourt Brace Jovanovich Publishers], New York, 1981.
- [212] W. SÜSS Uber Kennzeichnungen der Kugeln und Affinsphären durch Herrn K.-P. Grotemeyer. Arch. Math. (Basel) 3 (1952), 311–313.
- [213] G. WANG, C. XIA. A characterization of the Wulff shape by an overdetermined anisotropic PDE. Arch. Ratio. Mech. Anal. 199 no. 1 (2011), 99-115.
- [214] J. P. WANG. The spectrum of the Laplacian on a manifold of nonnegative Ricci curvature. Math. Res. Lett. 4 (1997), 473–479.
- [215] H. WEINBERGER. Remark on the preceeding paper of Serrin. Arch. Ration. Mech. Anal. 43 no. 4 (1971), 319–320.
- [216] H. C. WENTE. Counterexample to a conjecture of H. Hopf. Pacific J. Math. 121 (1986), 193-243.
- [217] G. TALENTI. Best constant in Sobolev inequality. Ann. di Mat. Pura ed Appl. 110 (1976), 353–372.

- [218] P. TOLKSDORF. Regularity for a more general class of quasilinear elliptic equations. J. Differential Equations 51 no. 1 (1984), 126–150.
- [219] N. S. TRUDINGER. On Harnack type inequalities and their application to quasilinear elliptic equations. Comm. Pure Appl. Math. 20 (1967) 721-747.
- [220] N. S. TRUDINGER. Remarks concerning the conformal deformation of Riemannian structures on compact manifolds. Ann. Scuola Norm. Sup. Pisa Cl. Sci. 3 (1968), 265–274.
- [221] J. L. VÁZQUEZ. Fundamental solution and long time behavior of the porous medium equation in hyperbolic space. J. Math. Pures Appl. 104 (2015), 454–484.
- [222] J. L. VÁZQUEZ. "Smoothing and Decay Estimates for Nonlinear Diffusion Equations. Equations of Porous Medium Type." Oxford University Press, Oxford, 2006.
- [223] J. L. VÁZQUEZ. "The Porous Medium Equation. Mathematical Theory." The Clarendon Press, Oxford University Press, Oxford, 2007.
- [224] J. VÉTOIS. A priori estimates and application to the symmetry of solutions for critical p-Laplace equations. J. Differential Equations 260 (2016), 149– 161.
- [225] C. VILLANI. "Topics in Optimal Transportation." Graduate Studies in Mathematics, vol. 58. American Mathematical Society, Providence (2003), xvi+370 pp.
- [226] H. YAMABE. On a deformation of Riemannian structures on compact manifolds. Osaka Math. J. 12 (1960), 21–37.
- [227] S. T. YAU. Problem section. Seminar on Differential Geometry, pp. 669-706, Ann. of Math. Stud., no. 102, Princeton Univ. Press, Princeton, N.J., 1982.
- [228] Y. L. XIN. "Geometry of Harmonic Maps." Progress in Nonlinear Differential Equations and their Applications, 23. Birkhäuser Boston, Inc., Boston, MA, 1996.

### Grazie!

Grazie ai miei relatori prof. Giulio Ciraolo e prof. Luigi Vezzoni. Grazie per avermi introdotto al mondo della ricerca, grazie per avermi coinvolto in numerosi progetti, grazie per gli stimoli e gli incoraggiamenti costanti che mi avete fornito e, non da ultima, grazie per la pazienza che avete avuto. Un grazie anche a Skype che ha reso Torino, Pavia, Palermo (e Firenze) molto vicine.

Grazie a prof. Alessio Figalli e a prof. Alberto Farina. Grazie della vostra disponibilità e dell'ospitalità che mi avete riservato. I periodi trascorsi con voi sono stati molto proficui sia dal punto di vista matematico sia da quello umano.

Grazie a prof. Matteo Muratori. Grazie per avermi coinvolto in un progetto di ricerca molto interessante sin dai primi mesi del mio dottorato.

Grazie a tutti i professori, ricercatori, assegnisti e dottorandi che si sono interessati al mio lavoro e con i quali ho avuto occasione di chiacchierare di matematica e non solo...

Grazie a tutti i dottorandi pavesi dei cicli XXIX, XXX, XXXI, XXXI XXXIII e XXXIV. Grazie per aver contribuito a rendere questa avventura di 3 anni indimenticabile! In modo particolare grazie ai miei fantastici compagni d'avventura: Barbara, Maria Gioia, Monica e Nicolò. I tre anni di dottorato sono volati ed è stato un piacere condividerli in toto con voi.

Grazie al Dipartimento di Matematica dell'Università degli Studi di Pavia per l'ambiente armonioso che ha fatto da sfondo a questi tre anni e grazie anche a prof. Giulio Schimperna e ai membri del collegio docenti del dottorato di Pavia-Milano Bicocca per avermi concesso la libertà di avere relatori esterni.

Grazie ai miei genitori. Grazie per aver caldeggiato le mie scelte incoraggiandomi e standomi sempre vicino.

Grazie agli amici (comaschi e non) di sempre per il sostegno e per la costante amicizia, anche a distanza.

Grazie a Irene. Con te al mio fianco è stato tutto più facile, hai riempito di momenti bellissimi e indimenticabili questi ultimi mesi di dottorato. Spero di riuscire a fare lo stesso per te!

#### A voi tutti: GRAZIE!!